Understanding the Conceptual Interplay Between Learners' Motivation and Patterns of Personal Meaning in the Mathematics Classroom: Results From Germany and Finland

Neruja Suriakumaran



Neruja Suriakumaran University of Bremen

A doctoral dissertation submitted to Faculty 3—Mathematics and computer science, University of Bremen in fulfilment of the requirements for the degree of Dr. rer. nat.

Defence of the dissertation: 21st January 2022 First assessor: Prof. Dr Maike Vollstedt (University of Bremen) Second assessor: Prof. Dr Markku Sakari Hannula (University of Helsinki)

Abstract

The orientation of motivation plays an important role in the quality of students' functioning in mathematics learning. In particular, autonomous (i.e., self-determined) motivation has multiple positive effects on their learning processes, for instance as it relates to proactive coping with failure, taking responsibility for one's own learning, volitional actions, and most importantly, well-being in the classroom. Intrinsic motivation represents the hallmark of self-determination. However, not all students are intrinsically motivated; learners seek to fulfil their need for meaning in learning situations by referring to their concept of the personal relevance of studying mathematics. Recent research in the field of motivation has emphasized that teachers' external efforts can affect the learner's autonomous motivation by highlighting the student's sense of the personal relevance of mathematics. The conceptual relationship between the personal relevance attach to studying and their quality of motivation is not clarified yet. While studies have identified motivational profiles so as to identify configurations of behavioural regulations, profiles of personal relevance have not received substantial attention yet. That is why the present investigation examines what patterns of personal relevance (termed as individual relevance systems) move learners to study mathematics and how the learner's individually constructed personal relevance is interconnected with their motivation in the mathematics classroom.

To this effect, two theoretical lenses, self-determination theory (SDT) and social constructivism, are adopted to jointly study why individuals show a certain motivational orientation in line with what kind of individual relevance system they construct in mathematics as a means of orienting themselves towards the social environment in the classroom. In this connection, the motivational construct of personal meaning reflects the personal relevance learners attach to studying mathematics in an academic context. Theoretically, personal meaning encompasses a biological and a social ontological nature, so this construct can be conceptualized as straddling the boundary of both theoretical lenses, SDT and social constructivism. To understand the interaction between both perspectives, I studied the interplay between data from German and Finnish ninth graders (aged 14–16) to contrast the results and refine the theoretical relationships between both affective constructs.

Research findings were gathered within a sequential-dependent research design: The initial empirical work that was carried out using quantitative statistical analyses only captured personal meaning as an individual phenomenon, as it focused on a classical survey method (self-report data). To take into account the social ontological nature of personal meaning, I subsequently considered the approach of networking theories and applied the strategy of coordinating.

I condense the main findings of this present dissertation into two central messages: Firstly, the individual relevance systems of learners found in Germany (GER) and Finland (FIN) can be differentiated into four kinds of mathematics-learning orientation in view of emotional-social integration—learners' starting internalization (either balanced (GER) or autonomous motivation (FIN)); enjoyment in mathematics learning (learners experience wellbeing and integration—autonomous motivation); self-improvement (learners feel freedom of action—autonomous motivation), and external pressure (learners follow obligations—either balanced (FIN) or less controlled motivation (GER)).

The empirical evidence suggests that basic psychological needs can be stimulated by the construction (or fail to be constructed in the event of non-construction) of the three specific personal meanings EXPERIENCE OF AUTONOMY, COMPETENCE, and SOCIAL RELATEDNESS through the learner's reaction to their social environment in the mathematics classroom (known as the inner feedback mechanism). The satisfaction of the respective basic psychological needs for autonomy, competence, and relatedness seems to retroact to the learner's social space in class. So, the basic psychological needs are the factors in personal meaning that have a strengthening or weakening effect on biological (the link between SDT and personal meaning) and social regulatory mechanism (the link between social constructivism and personal meaning). Through this inner feedback mechanism, the biological regulatory mechanism regulates the learner's self-determination and well-being in the classroom (autonomous or controlled motivation), and, in the classroom setting, the social regulatory mechanism regulates the direction of the learner's type of individual relevance system that is produced in line with the biological regulation (geared towards or away from the relevance in individual and mathematics-related personal meanings). The mediating roles of teacher and cultural amplifiers reshape the learner's social regulatory mechanism, and this in turn explain why similar patterns of personal meaning result in different biological regulations across countries.

Secondly, the biological and social regulatory mechanisms are coordinated within the boundary object of personal meaning, and the indicators found within the coordination of the biological and social regulatory mechanisms can be modelled as a hypothetical feedback mechanism to explain the genesis of motivation in the mathematics classroom. The social regulatory mechanism seems to trigger the biological regulatory mechanism, and that in turn seems to feedforward to their social regulatory mechanism. As a result of this linkage, it is not possible to separate the social and biological regulatory mechanisms or view them as independent of each other. I conclude that the boundary object of personal meaning has major significance for the genesis of motivation in the mathematics classroom.

The dissertation concludes with an in-depth discussion of how its findings relate to those of earlier studies, and of the potential for its future application in fields other than the specific area of study, i.e., mathematics-related motivation.

Acknowledgements

First of all, I would like to thank my supervisors, Prof. Dr Maike Vollstedt and Prof. Dr Markku Sakari Hannula. My journey would never have started if Maike hadn't invited me to join the working group "AG Didaktik der Mathematik" in Bremen. I would like to thank you Maike, for putting your faith in me. Thank you also for encouraging me to take part in (inter)national conferences at a very early stage. I would also like to thank you for helping me make my research stay at Helsinki University possible.

There are people who, by their very nature, enrich the lives of others in countless ways without perhaps being aware of it—Markku, you are one of them. First of all, I would like to thank you for welcoming me into your working group during my research stay. You took every opportunity to support me both in my work as a researcher in mathematics education and as a teacher specializing in didactics at the school and university. I particularly enjoyed the working atmosphere in your group, which was characterized by mutual appreciation and respectful interaction. Thank you also for offering to supervise my dissertation and encouraging me to conduct a comparative study based on my work. That idea set me on the road to realizing this project. Thanks to your honest and constructive support, I was able to draw on your knowledge and experiences with regard to theory, methodology, and method. Not only that, but you were always ready to listen and offer advice, and, with your warm and approachable manner, you gave me the opportunity to get to know my own strengths, to grow even further, and to develop my passion for working in research. Thank you for the academic and emotional support and for being the wonderful supervisor—and person—that you are.

I would like to say a big thank you to Prof. Dr Angelika Bikner-Ahsbahs. As your name suggests, Angelika, you have been a constant source of guidance and help throughout this journey, especially when it came to the design of the theory network. Your constructive suggestions and questions provided a helpful challenge to me and my work; I've learned a lot from you, especially about what my values are when it comes to (good) research and how to tread my own path in academia. Our discussions have always been extremely enriching, and I am very grateful to you for taking that time for me. I would especially like to thank you for your encouraging words, especially in the final writing phase; they meant a great deal to me.

I am very happy and proud to have been part of the working group "AG Didaktik der Mathematik" and to have had the opportunity to draw on the varied knowledge and experiences of that group's members. I would also like to thank the working group for the countless times they offered me constructive feedback. Thanks go in particular to Prof. Dr David A. Reid—David, thank you for your sympathetic ear and for sharing your knowledge with me, whether at digital meetings or during fascinating discussions over coffee. Your astute observations and wonderful sense of humour were the perfect antidote to stress—for which I am also eternally grateful. Dr Ingolf Schäfer, I would like to thank you for the plentiful helpful feedback you provided, both theoretical and methodological, and especially for your inspiration as to the selection of my theoretical approaches. Dr Christoph Duchhardt, thank you very much for your methodical feedback, which was very valuable, especially at the beginning of the doctoral process. Special thanks also go to Prof. Dr Christine Knipping and Prof. Dr David A. Reid for

organizing the writing group, which provided great support, particularly when I was in the final phase of writing my dissertation. Kerstin and Roland, I am grateful for your constant technical support. Many thanks also to my wonderful fellow doctoral students Anna, Daniela, Jona, and Erik—how lovely that we could go on this journey together and share our joys, sorrows, and biscuits. Thanks to you, too, my dear Angelie, not only for being the most supportive student assistant anyone could imagine, but also for your friendship, your inspiring and motivating ideas, and, last but not least, our uniquely enriching conversations about research and culture.

The research stay in Helsinki will always be a fond memory for me, not least because of my wonderful colleagues on campus. Jessica F. A. Salminen-Saari, thank you for so generously offering to translate the questionnaire, and for doing it so quickly—the result couldn't have been better. My dear Dr Enrique Garcia Moreno-Esteva, thank you for your mental support and for your countless unforgettable anecdotes about Finland, its culture and people. To my friend Visku Salonen—thank you for answering all my methodical questions. Your encouraging words always came at the right time. And finally, my dear friend Dr Eeva Haataja: my time in Finland, would not have been the same without you. Thank you for your active support, especially in translating the questionnaire, supporting the data collection, and conducting the cognitive lab; for sharing the highs and lows of the doctorate experience with me; and always being ready to listen, no matter what. I am so grateful for our wonderful friendship.

Some of my colleagues provided me with invaluable help with the data collection. Many thanks to Dr Benjamin Thiede and Uwe Schallmaier. Dr Nikola Leufer, thank you for also sharing your knowledge and experience of the academic world with me—our conversations were always enriching in so many ways. To all German and Finnish students and teachers—I am deeply grateful for your time and support!

I have been lucky to receive guidance from numerous consultations with colleagues, who provided me with a lot of support in regard to the selection of methodology and methods in my dissertation; without that support it would have been difficult to progress. I would like to thank the following for their constructive and interesting contributions as well as for their time: Prof. Dr Ralf Bohnsack, Prof. Dr Jan Retelsdorf, Prof. Dr Nils Buchholtz, Dr Martin Scharpenberg, Prof. Dr Christopher Helm, and Dr Felix Knappertbusch. Our conversations were goal-oriented, inspiring, and at the same time motivating.

To my dear friend Katrin Klieme: I am grateful to you for so many things. For answering my methodical questions during the most challenging part of my dissertation to the formatting of my tables and figures (without which it would have been almost impossible for me to make my deadline), you were always at my side, not only as a highly valued scientific colleague but also as my friend. Thank you for all the encouragement, chocolate bunnies, cake breaks, and especially your friendship. I appreciate you not only for your beautiful mind but even more for your unique, wonderful soul.

Many thanks also to Prof. Dr Stefanie Rach, Patrick Fesser, Dr Sebastian Geisler, Dr Kolja Pustelnik, and Karyna Umgelter for your constructive feedback, which was extremely helpful when I was preparing to defend my dissertation.

My entry into the field of mathematics education research would not have been possible without the support of some exceptional people. Tobias Westermann, thanks to your support as my mathematics teacher, I learned to have more faith in my abilities and to develop love for mathematics, which finally made me study the subject. Prof. Dr Martin Stein, thank you for allowing me to enter the world of academia and to discover my passion for research in mathematics education.

Dilan and Shiyami, thank you for being there for me and supporting me in so many ways on this journey.

Dr Thomas Janßen, ever since you showed me round on my first day of work at the University of Bremen you have been there for me, not only as a helpful colleague with your constructive contributions, but also as a dear friend; especially in the final writing phase, you were always ready to support me. Thank you for listening, proofreading, and for giving your feedback.

Prof. Dr Christian Lohmann, thank you very much for sharing your passion for research with me and for offering me your help with all my questions and thoughts without hesitation (even before I started my doctorate). You have taught me a lot about academic life.

Many thanks also to my dear friend Frances Tye and also to Philip Tye for their great support with the language editing. I couldn't have imagined anyone doing a better job. Thank you, Frances, for the lovely, distracting bike trips, which will probably remain unforgettable memories for both of us.

My wonderful friends Katrin B., Tanja, and Chrissa: I'm so happy to have you in my life. Dear Katrin B., thank you for your constant support and for being so generous with your help. My dear Tani, thank you for encouraging me, cheering me up, and making me laugh on this journey. Chrissa, I can't imagine what my doctoral period in Bremen would have been like without you. How wonderful that we found each other and were able to accompany each other on our dissertation journeys from beginning to end. Thank you all for your help and the time we spent together—I will never forget our adventures.

And last of all, my family. My aunt Manjula, my uncle Tirukkaran and my cousins Ashmika, Dhilushan, and Laka—thank you for your support but especially for helping me to keep going during the difficult times. My brother Senthuran and my adorable nephew Suriya Zayn—thank you for your love and the encouragement you gave me just by being there. My beloved grandma Ponmani, thank you for always believing in me—I wish you were still here. The most important thanks go to my mum and best friend Janahi—thank you for giving me the support that no one else could have given me, for staying up late with me, for cheering me on, and laughing with me. Thank you for letting me be your daughter—அம்மா, என் வாழ்வின் வெளிச்சம் நீ.

> I Dedicate this Thesis to all Learners who are on the Way to Becoming Self-Determined and to the Teachers Supporting Them on Their Individual Journeys.

Contents

	Abstract		3		
	Acknowledgements				
	Contents				
	List of figures				
	List of tab	les	13		
1	Introducti	on	15		
2	Theoretica	al framework	21		
	2.1 Self-determination theory (SDT) for understanding motivation to learn				
	 2.1.1 SDT's elements: types of motivation and basic psychological needs				
	2.2 Learne	ers' construction of personal meaning in the mathematics classroom			
	2.2.1 The social constructivist perspective2.2.2 The motivational construct of personal meaning2.2.3 Critical reflection on the concept of personal meaning				
	2.3 Contrasting juxtaposition of SDT and social constructivism				
	 2.3.1 SDT's perspective on personal meaning – theoretical assumptions				
	2.3.3	assumptions Consideration of personal meaning as a boundary object	56 58		
	2.4 Cross-cultural settings for validation and refinement				
	2.4.1 German and Finnish mathematics curricula2.4.2 Answerable questions				
	2.5 Résumé				
	2.6 Research questions				
3	Empirical	work	69		
	3.1 Adopting a sequential-dependent approach to studying personal meaning's multimodal nature				
	3.2 Methodological considerations for the quantitative analyses				
	 3.2.1 Selection of the classical survey instruments				
	3.2.3 3.2.4	Quality criterion and quality assurance Generalizability and representativeness			

	3.3 Empirical methods		
	3.3.1	Data collection	83
	3.3.2	Survey instrument	
	5.5.5	Quantitative statistical analyses	
	3.4 Résun	né	96
4	Theory-di	riven analyses	97
	4.1 Metho	odological considerations for the coordination analysis	97
	4.1.1 4.1.2	Theoretical triangulation Approach of networking theories to studying personal meaning's socia	97 Il nature 99
	4.2 Coord	lination analysis method	100
	4.2.1 4.2.2 4.2.3	Terminologies for a theory-driven interpretation of quantitative data Coordination analysis procedure Critical reflection on applying terms from qualitative research tradition	102 103
	43 Résur	né	109
5	Empirical	results	110
5	5 1 Factor	r structure of personal meaning and regulatory style across countries	
5.1 Factor structure of personal meaning and regulatory style across countries			
	5.2 Drofil	ing classes of personal magning and their outcomes in mativation	131
	5.4 Summ	and then outcomes in motivation	1/0
6	Discussio	n of the empirical results	149
0	6 1 Psych	ometric properties of personal meaning and regulatory style across coun	130
	0.1 1 Sych	oneure properties of personal meaning and regulatory style across cour	150
	6.2 Natur	e of latent mean differences in personal meaning and regulatory style	153
	6.3 Identi	fication of profiles of personal meaning and their motivational outcomes	s154
7	Results ac	ccording to the coordination analysis method	156
	7.1 Mathe	ematics-learning orientations in Germany	158
	7.1.1	Profile 1: Emotional-social integration with a focus on seeking accepta	ince
		from the teacher	158
	7.1.2	Profile 2: Enjoyment in mathematics learning	161
	7.1.3	Profile 4: External pressure focused on adaption to the teacher's perfor	mance
		requirements	167
	7.1.5	Comparison of German learners' mathematics-learning orientations	170
	7.2 Mathe	ematics-learning orientations in Finland	171
	7.2.1	Profile 1: Emotional-social integration focused on own professional	. –
	7 7 7	qualification and seeking acceptance from the teacher	171 174
	1.4.4	rome 2. Enjoyment in matiematics rearning	

	7.2.3 7.2.4 7.2.5	Profile 3: External pressure focusing on adaption to the educational syste Profile 4: Self-improvement with a focus on the individual Comparison of Finnish learners' mathematics-learning orientations	em 177 180 183
	7.3 Summ	nary of the results according to coordination analysis	184
	7.4 Theori	izing the results of the coordination	185
	7.4.1	Interpreting the interaction between biological and social regulatory mechanisms	186
	7.4.2	Comparing the individual relevance systems in Germany and Finland	189
	7.4.3	Conclusions on the effects found in German and Finnish mathematics-le orientations	arning 194
	7.4.4	Model of the feedback mechanism: the genesis of motivation in the mathematics classroom	196
8	Discussion	n	200
	8.1 Main f	findings	200
	8.2 Comp	lementing references to the field of mathematics-related affect	202
	8.2.1 8.2.2 8.2.3	Networking theories to counter the terminological ambiguity References to meta-theoretical knowledge The body as an essential element of mathematical thinking	202 203 206
	8.3 Comp	lementing references to SDT's research landscape	206
	8.4 Retros	spection	208
	8.4.1 8.4.2	Limitations of the theoretical approach Limitations of the methodological approach	208 210
	8.5 Prospe	ects	213
	8.5.1 8.5.2	Implications for future research Implications for teaching and educational policy	213
	8.6 Conclu	usion	215
	Reference	°S	217
	Appendix		237

List of Figures

Figure 1 Self-determination theory's taxonomy of regulatory styles (adapted from Ryan & Deci, 2020)
Figure 2 Model of sign appropriation and use (adapted from Ernest, 2010)
Figure 3 Theoretical concept of personal meaning (adapted from Vorhölter, 2009; see also Vollstedt, 2011b)
Figure 4 Hypothesized link between basic psychological needs and the three specific
personal meanings
Figure 5 Hypothesized links between regulatory styles and personal meanings
Figure 6 Social regulatory mechanism of the individual relevance system (cycle adapted from Ernest, 2010)
Figure 7 Personal meaning characterized as located at the boundary of social constructivism and self-determination theory
Figure 8 Overview of general educational standards in mathematics to be reached after grade nine which are acquired through content-related mathematical competencies and central mathematical ideas (adapted from KMK, 2005, translated by the author)63
Figure 9 Key content areas related to the objectives of mathematics in grades 7–9 and transversal competencies in basic education (adapted from Finnish National Board of Education, 2016; see also Halinen, 2018)
Figure 10 The present investigation's sequential-dependent research design
Figure 11 Overview of the present investigation's empirical work
Figure 12 Networking strategies for connecting theories in mathematics education (adapted from Prediger & Bikner-Ahsbahs, 2014)
Figure 13 Overview of the present investigation's coordination analysis procedure
Figure 14 Four profiles of personal meaning in German students
Figure 15 Characteristics of the latent classes in the outcome variables regulatory styles in German students
Figure 16 Four profiles of personal meaning in Finnish students141
Figure 17 Characteristics of the latent classes in the outcome variables regulatory styles in Finnish students
Figure 18 Profile 1 in Germany: Emotional-social integration with a focus on seeking acceptance from the teacher
Figure 19 Profile 2 in Germany: Enjoyment in mathematics learning
Figure 20 Profile 3 in Germany: Self-improvement with a focus on mastery165
Figure 21 Profile 4 in Germany: External pressure focused on adaption to the teacher's performance requirements

Figure 22 Profile 1 in Finland: Emotional-social integration focused on own professional qualification and seeking acceptance from the teacher	172
Figure 23 Profile 2 in Finland: Enjoyment in Mathematics Learning	175
Figure 24 Profile 3 in Germany: External pressure focused on adaption to the educational system	178
Figure 25 Profile 4 in Germany: Self-improvement with a focus on the individual 1	81
Figure 26 Revised overview: Learner's social regulatory mechanism of the individual relevance system—affected by the mediating role of the teacher and cultural amplifiers	193
Figure 27 Inner feedback mechanism between basic psychological needs and specific personal meanings	195
Figure 28 Hypothetical model of the feedback mechanism between social and biological regulatory mechanisms to explain the genesis of motivation in the mathematics classroom	197

List of Tables

Table 1 Descriptive statistics of sample, means, and standard deviations for all items of personal meaning and regulatory style in German and Finnish students
Table 2 Scale reliabilities for regulatory style and personal meaning scales in German and Finnish students
Table 3 Fit statistics for confirmatory factor analysis of dimension of personal meaning models separately and all together in German students 113
Table 4 Fit statistics for confirmatory factor analysis of dimension of personal meaning models separately and all together in Finnish students 114
Table 5 Fit statistics for confirmatory factor analysis of regulatory style models in German and Finnish students 116
Table 6 Second-order confirmatory factor analysis: factor loadings of the first-order personal meanings and intercorrelations between dimensions of personal meaning in German Students 118
Table 7 Second-order confirmatory factor analysis: factor loadings of the first-order personal meanings and intercorrelations between dimensions of personal meaning in Finnish students
Table 8 Intercorrelations between regulatory styles and personal meanings in German students 122
Table 9 Intercorrelations between regulatory styles and personal meanings in Finnish students 123
Table 10 Intercorrelations of second-order multifaceted dimensions and regulatory styles in German students 124
Table 11 Intercorrelations of second-order multifaceted dimensions and regulatory styles in Finnish students 124
Table 12 Fit statistics for each measurement invariance model of the nine higher-order dimensions of personal meaning and regulatory style 128
Table 13 Latent mean differences in personal meanings and regulatory styles between German and Finnish students 132
Table 14 Fit Indices of latent class analysis based on the 17 personal meanings in the German sample. 134
Table 15 Relationship between LCA membership and outcome variables in German students.
Table 16 Fit Indices of latent class analysis based on the 17 personal meanings in the Finnish sample 139
Table 17 Relationship between LCA membership and outcome variables in Finnish students. 142

Table	18 Mann–Whitney U test statistics for mean difference of personal meanings and regulatory styles in profile 1 from Germany and profile 1 from Finland	45
Table	19 Mann–Whitney U test statistics for mean difference of personal meanings and regulatory styles in profile 2 from Germany and profile 2 from Finland	46
Table	20 Mann–Whitney U test statistics for mean difference of personal meanings and regulatory styles in profile 3 from Germany and profile 4 from Finland	47
Table	21 Mann–Whitney U test statistics for mean difference of personal meanings and regulatory styles in profile 4 from Germany and profile 3 from Finland	48

1. Introduction

Education is the kindling of a flame, not the filling of a vessel.

Socrates, 470 BC

If, according to Socrates, education kindles the flame of knowledge, motivation can be seen as the torch that arouses and sustains that flame within students (Liu et al., 2016, p. 1) in (e.g.,) mathematics-learning situations. Understanding each student's motivation to study mathematics is not easy to manage in the classroom context. Like a snowflake, each learner is unique, and thus they¹ bring their individual motivations with them into the mathematics classroom. Although motivating students to study mathematics is a matter of much debate, teachers may find it overwhelming due to their busy workloads (Reeve, 2016). Therefore, it is sometimes necessary for teachers to apply pressure to students to ensure they complete their work-regardless of whether the student is interested in the topic for its own sake, finds no personal importance in it, or is even bored by it (Reeve, 2016; Vansteenkiste et al., 2018). This appears to teachers to be a necessary strategy if they are to meet the requirements of the mathematics curriculum and handle the increasing amount of responsibilities they hold in that classroom, in the face of a lack of resources. Most importantly, teachers use pressure and control to ensure they do not drop out of the profession, as well as for their own personal wellbeing, in order to function sustainably over time, both in and out of the classroom. Hence, teachers do not intentionally seek to establish a controlling environment or to avoid teaching styles that would motivate their students, such as recognizing learners' perspectives while teaching (Reeve & Cheon, 2021). Exploring what personal relevance drives learners to study mathematics not only helps students to experience a positive learning process (Ryan & Deci, 2017) but also helps educational professionals to make their daily routine smoother and to foster students' motivation to learn mathematics in the classroom.

First of all, what does it mean to be motivated? And why is it important, for a variety of reasons, for learners to be autonomously motivated in the classroom instead of feeling controlled? Learners who are motivated in the mathematics classroom are *moved* to direct their behaviour towards a certain action (Hannula, 2004; Ryan & Deci, 2000a). The extent of motivation can vary, both in terms of quantity and in terms of the learner's quality of functioning. Quantitative motivational processes, or *achievement-related behaviour*, concern *how much* a student is motivated; however, *qualitative motivation* refers to the *why* of orientations towards the *what*, or content, of goal pursuits (Deci & Ryan, 2000; Ryan et al., 1996; Ryan & Deci, 2000a). To illustrate this, a student may be extremely motivated to study mathematical contents out of passion and interest or a desire to use the learned mathematical skills in their future education. Alternatively, a learner may also be motivated to solve difficult mathematical tasks to impress important figures in their life (e.g., teachers, peers, or parents), or they may value the skills for other advantages they bring with them (Ryan & Deci, 2000a).

¹ I use "they" as a generic third-person singular pronoun to ensure bias-free language within this study's context and to refer to all individuals as human beings (American Psychological Association, 2020).

Both motivational processes, quantity and quality, matter. However, when examining these motivations at the level of quantity, there is no substantial difference: that is, all actions can appear to be highly motivated regardless of the nature of that motivation (Ryan & Deci, 2000a). In contrast, the quality aspect of motivation differs in terms of its composition (i.e., autonomous or controlled) and its underlying focus of orientation (e.g., goals, Ryan et al., 1996; Ryan & Deci, 2000a), and this in turn affects the quality of mathematics learning.

For this reason, this study focuses on learners' *quality of motivation* when learning mathematics. Thinking of the energizing nutrients of quality of motivation within educational settings, several scholars (to name but a few: Boekaerts, 1999; Hannula, 2006; Nuttin, 1984; Ryan & Deci, 2020) emphasize the functional significance of *basic psychological needs for autonomy, competence*, and *relatedness* in terms of students' learning processes and optimizing their motivation. These basic psychological needs are interpreted as the roots (Hannula, 2004) and the driving force of autonomous-oriented motivation in learning and its sustainability (Krapp, 2005). Students engage more energetically in learning processes only if they subjectively experience an overall balance of basic-psychological-needs satisfaction (Krapp, 2005).

The implementation of learning processes can be qualitatively improved if learners engage with (e.g.,) mathematics for its own sake; namely, with autonomous motivation. Autonomously motivated learning situations are connected to various robust positive outcomes such as more self-efficacy (Wang et al., 2016), investing more effort in learning mathematics (León et al., 2015; Ryan & Deci, 2020), perseverance, positive emotions, and high performance (Deci & Ryan, 1991; Ryan & Deci, 2017, 2020; Vansteenkiste et al., 2018). Moreover, learning activities that are built on autonomous motivation can more easily be absorbed into those core values which make up the students' "identity as scientists" (Skinner et al., 2017, p. 2436). Focusing on STEM courses, Skinner et al. (2017) found that if learners' classroom experiences fulfilled their basic psychological needs (i.e., were associated with autonomous motivation), students were more engaged and performed to a higher level; they also saw themselves as scientists and positively identified with the field of STEM science. In contrast, controlled motivation is associated with the frustration of basic psychological needs (Haerens et al., 2015), making the need for self-esteem more salient as a result (Ryan & Deci, 2017), and has a detrimental effect on intrinsically motivated behaviour (Ryan & Deci, 2017), as well as leading to ill-being (Stenling et al., 2017). Furthermore, students' learning is suboptimal, particularly when learning activities are complex or demand a conceptual and creative sequence of operations (Benware & Deci, 1984), which is often the case in mathematics. In particular, controlled motivation is associated with a lack of maintenance of learning when the student is not being observed; this unsustainable approach makes dropout more likely (Deci & Ryan, 2000; Pelletier et al., 2001; Ryan & Deci, 2017; Vallerand & Bissonnette, 1992).

Particularly during adolescence, dropping out of mathematics education has an impact on students' cognitive and brain development. In fact, scholars at the University of Oxford discovered that dropping out of mathematics education, specifically at the age of 16, was detrimental for British learners' neurological and cognitive development (Zacharopoulos et al., 2021). In some countries, students can drop out of mathematics voluntarily. The study showed that learners who stopped studying the subject had a lower concentration of GABA (gamma-Aminobutyric acid), a neurochemical that is necessary for brain plasticity within an area of the brain that is important for cognitive functioning, for example logic, mathematics ability, memory, problem-solving, and mathematical reasoning (Zacharopoulos et al., 2021). Previously, when they were still engaged in mathematics, no neurochemical changes had been observed (Zacharopoulos et al., 2021). Zacharopolous et al. did not merely stress the high relevance of mathematics education for students' brain development and cognitive functioning; they also underlined the reciprocal interplay between biology and mathematics education that takes place during adolescence (Zacharopoulos et al., 2021, p. 8).

These results show the importance of autonomously motivated learning, particularly of mathematics, for students' learning processes and personal development. As schematically pointed out by Hannula (2012), this empirical evidence underlined the interplay between (autonomous) motivation, (positive) emotion, and (development and positive impact on) cognition. Vygotsky made this point succinctly when he stated that affect is never external to the learner's intellect (Vygotsky, 1962, p. 8; 1987, p. 50). In terms of this issue, I agree with Vygotsky (1987), who argued that

The inevitable consequence of the isolation of these functions [intellect and affect] has been the transformation of thinking into an autonomous stream. Thinking itself became the thinker of thoughts. Thinking was divorced from the full vitality of life, from the motives, interests, and inclinations of the thinking individual. (...) By isolating thinking from affect at the outset, we effectively cut ourselves off from any potential for a causal explanation of thinking. A deterministic analysis of thinking presupposes that we identify its motive force, that we identify the needs, interests, incentives and tendencies that direct the movement of thought in one direction or another. (p. 50)

On the basis of Vygotsky's argument, Hickey (1997) stated that motivation as an affective construct should not be distinguished from cognitive processes. That is the case because motivation as an individual construct is not separable from cognitive activities that take place while learning. The importance of mathematics learning foregrounds the question: How can educators foster learners' autonomously motivated learning in mathematics? Specifically, it is necessary to find solutions with which to engage students when access to mathematics education in school is limited (e.g., due to the COVID-19 pandemic crisis of 2020/2021), in order to balance the impacts of this on adolescents' (and children's) neurological and cognitive development.

In the case of mathematics education, there is relatively little existing research focusing on subject-related motivation compared to the amount of research in general education. For instance, Middleton and Spanias (1999) emphasized the necessity of clarifying why learners are motivated to study mathematics and of exploring the interrelations of motivational factors that play a determining role in learners' "social and cognitive worlds" (pp. 83–84) if researchers are to get a full picture of the phenomenon, that is, of motivation in mathematics education. In this relation, Hannula (2006) suggested that enhancing the theoretical knowledge of the interplay between what motivates students to study mathematics and how their motivation to learn is regulated—that is, whether it is autonomous or controlled—may help scholars to

understand learners' behaviour in the mathematics classroom. However, students typically have multiple motives within the educational context (Ryan & Deci, 2017, 2020). Recently, researchers have investigated the patterns of various types of motivation (motivational profiles) in mathematics education (e.g., Wang et al., 2017).

Although these results concerning profiling provide an insight into students' patterns of various types of motivation as regards mathematics learning, merely concentrating on why learners show a certain behaviour in the classroom is not enough to foster their motivation in mathematics. Mathematics learning is not interesting for all individuals. Specifically, due to the decrease of interest in mathematics with increasing age (e.g., Frenzel et al., 2010, 2012) it is nearly impossible for teachers to stimulate the intrinsic motivation in mathematics that comes with interest, curiosity, and enjoyment. However, students can also positively engage with mathematical activities by means of a learning environment that is supportive of their basic psychological needs and that highlights the personal relevance to those students (it tends to be more valued on an individual than on a social basis, Ryan & Deci, 2017, 2020; Vansteenkiste et al., 2018). In this connection, Bikner-Ahsbahs (2005) emphasized that mathematics learning can also become more interesting to students when they experience it as personally relevant. For this reason, it is necessary to focus on what kinds of personal relevance students construct when studying mathematics.

Concerning the relevance of mathematics, Niss stated that "although the social significance of mathematics seems to be ever increasing in scope and density, the place, rôle [sic], and function of mathematics are largely invisible to – and unrecognized by – the general public" (Niss, 1994, p. 371). Niss pointed out that despite the commonly accepted relevance of mathematics, mathematics remains subjectively irrelevant for the majority, which he calls a "relevance paradox" (Niss, 1994, p. 371). To highlight learners' personal relevance, researchers therefore need to explore what personal relevance students subjectively construct in mathematics-learning situations. Therefore, they need to understand the learners' individual "mental compasses": that is, the set of personal relevance that provides them with the orientations to study mathematics. The motivational construct personal meaning concerns the personal relevance students attach to mathematics (Vollstedt, 2011a; Vorhölter, 2009). In this connection, Vollstedt (2011a, 2011b) reconstructed 17 personal meanings that learners (aged 15-17) attached to mathematics learning within an educational context. Vollstedt and Vorhölter (2008) made the theoretical claim that each learner constructs a pattern of personal relevance (Vollstedt, 2011b; Vollstedt & Vorhölter, 2008; Vorhölter, 2009) that they assign to mathematics learning. If learners truly construct various patterns of personal meaning, the empirical evidence of those patterns of personal meaning, here termed *individual relevance* systems,² can help to clarify exactly what it is that provides them with a learning orientation in the mathematics classroom.

The cultural context and the mathematics curriculum as its outcome are indivisible from learners' mathematics-classroom environments and the teaching styles that provide the backdrop for motivational factors (Hickey, 1997; Ryan & Deci, 2020). Since each country's educational policy is unique, its principles will also vary in terms of how to support students' mathematics education (Zacharopoulos et al., 2021). Learners' participation in mathematics

² A clear distinction should be made here between this definition and that of Alfred Schutz (1899–1959).

education is determined by the various opportunities provided within the mathematics curricula. Accordingly, an investigation into different cultural contexts that have different foci within mathematics curricula would be fruitful, in order to see which individual relevance systems students construct. For this reason, this dissertation considers data from both German and Finnish ninth graders (aged 14–16; OECD, 2020a, 2020b). Nevertheless, an isolated study of students' individual relevance systems does not provide enough context either, and will need to be supplemented by further research.

To the best of my knowledge, so far, no well-defined conceptual framework has been developed that discusses the theoretical connection between learners' personal relevance in an academic context and the quality of motivation they experience in (mathematics) education. According to this *problematique*, this dissertation's main purpose is to link the theoretical perspectives on "why" students show a certain motivation with those concerning "what" personal relevance they construct in mathematics in order to orient themselves in their social settings of the mathematics classroom. Exploring learners' individual relevance systems in connection with their motivational behaviour has the potential to increase our theoretical knowledge of the interrelation and interplay between the "why" and the "what" of the development of motivation within mathematics education (Hannula, 2006; Middleton & Spanias, 1999). Not all types of personal relevance are supportive for mathematics learning or are close to the learner's sense of self. An exploration of the conceptual relation between personal relevance and motivation also helps to identify those individual relevance systems that promote students' autonomous motivation and those relevance systems which are less supportive. Accordingly, the overall objectives of this thesis are to explore the following questions:

- what kinds of individual relevance system do learners from Germany and Finland construct when learning mathematics?
- how are the individual relevance systems from Germany and Finland associated with motivation in mathematics?

Moreover, the comparison of German and Finnish data could help to contrast and refine our understanding of the relation between learners' individual relevance systems and the quality of their motivation. Based on this conceptual understanding, the interplay between personal relevance and motivation may provide a theoretical explanation for the genesis of motivation in the mathematics classroom. Therefore, the two questions above lead us to the main question with which this study is concerned, namely:

- how does learners' motivation in mathematics classroom emerge from the interplay between learners' individual relevance systems and motivation?

An insight into this interplay may help teachers to address the specific motivational factors that increase the personal relevance of mathematics for learners and that are also associated with autonomous motivation or positive learning processes. Furthermore, such an insight might also help educators to avoid teaching styles that have a negative impact on learners' autonomous motivation and the satisfaction of their basic psychological needs. Accordingly, this study seeks to contribute to an in-depth understanding of students' motivation in mathematics education by focusing on learners' individual voices and on the need to provide them with conditions in mathematics lessons that support the learning process as well as their personal development.

Based on these aims, the present dissertation follows a sequential-dependent research design (Schoonenboom & Johnson, 2017) in order to understand the conceptual interplay between German and Finnish ninth graders' levels of personal relevance and quality of motivation, and the genesis of motivation in mathematics. The following chapter, Chapter 2, introduces the theoretical framework of this study. In relation to this study's research interest, two different theoretical perspectives are adopted – self-determination theory (SDT) and social constructivism—through which to discuss learners' motivation in an academic context in connection with the development of students' individual relevance systems. Therefore, each theoretical lens is introduced separately to highlight both perspectives' strengths and limitations as regards the aims of this study. Based on existing theoretical as well as empirical knowledge, an interconnection between both theories is assumed within the boundary object (Akkerman & Bakker, 2011) of personal meaning as it seems to be grounded on a theoretical conceptualization that synergizes both learners' individual (basic-psychological-needs experiences) and social experiences (the influence of interpersonal "conversations" in the classroom) into their personal relevance construction in mathematics.

Chapters 3 and 4 encompass the methodological considerations of this thesis. Here, I explore and discuss how to methodologically examine personal meaning within the context of SDT and social constructivism. In Chapter 3 I discuss the philosophical background of the quantitative research methods that are used to identify the learners' individual relevance systems and the corresponding motivational propensities in two different cultural contexts. Since the applied quantitative methods show certain limitations in terms of capturing personal meaning in the context of social constructivism, Chapter 4 points out ways in which the approach of networking theories can help to balance the weakness of the quantitative work.

Chapter 5 reports the empirical results collected in Germany and Finland using quantitative research methods, and Chapter 6 provides a discussion of this empirical work. Chapter 7 refers to the results according to the coordination analysis method based on the empirical data. Subsequently, I reflect theoretically on the coordination of the results in order to propose a theoretical model that might explain the genesis of motivation in the mathematics classroom.

In Chapter 8, I discuss the main findings of this dissertation, and I then examine the evidence from both a retrospective and a prospective viewpoint. Besides a review of my study's limitations, I look at its implications for future research and make some suggestions for the optimization of both mathematics teaching in the classroom and educational policy.

2. Theoretical Framework

As a decisive element of research, theory provides the language to articulate all observations that count to the nature of the research object under investigation (Mason & Waywood, 1996; Schoenfeld, 2002). In this spirit, this present study's theoretical framework aims to provide a robust basis on which to elucidate in a cross-cultural context *why* individual learners show a certain motivational orientation in line with *what* individual relevance systems they construct in mathematics to orient themselves in relation to the social environment in the classroom.

For this purpose, I initially elaborate on the theoretical perspective of self-determination theory (SDT; see section 2.1). SDT is a general theory that can (also) help to understand students' motivation to learn (e.g.,) mathematics.³ In relation to the present research interest, this theory's strength is in explaining learners' intrinsic and extrinsic motivational orientations in mathematics-learning settings as an effect of students' individual perceptions of basicpsychological-needs satisfaction. Within SDT, scholars have articulated that autonomously motivated learning in the classroom is more likely when students perceive a personal relevance in their learning content. That requires us to answer the following questions: What personal relevance do learners attach to learning mathematics? Moreover, how do learners construct their individual relevance systems in the context of classroom interactions that orient them, as their mental compass, to engage (or not) in mathematics learning? To find answers to these questions, I present the social constructivist perspective (see section 2.2) and the motivational construct *personal meaning*, reflecting learners' perception of personal relevance when engaging with mathematics in an educational setting. Then, by critically reflecting on both theories' foci and limits (with reference to Radford's system of principles (P), see section 2.3), I propose that personal meaning can be characterized as a boundary object between SDT and social constructivism to connect both theoretical perspectives. Following the present research interest, linking these two theories would help investigate the theoretical connections between learners' personal relevance (individual relevance systems) and their motivational tendency towards the genesis of motivation in the mathematics classroom.

Since both theoretical perspectives emphasize the role of culture, I investigate the hypothesized theoretical connections using data from Germany and Finland to understand how learners' motivation arises in the mathematics classroom with respect to their culture-specific learning contexts. To this effect, I briefly introduce both countries' mathematics curricula and discuss these two educational settings tentatively (see section 2.4). Finally, based on the adopted theoretical perspectives and their ontological basis, I reflect on the remaining challenges and draw implications (see section 2.5) for this study's methodology that will need to be considered in order to examine the research questions of this dissertation (see section 2.6).

³ In the following, "mathematics" is bracketed when the content of the statement is not restricted to mathematics in an academic context. Furthermore, SDT is not limited to application in academic contexts (Ryan & Deci, 2017).

2.1 Self-Determination Theory (SDT) for Understanding Motivation to Learn

As a grand psychological theory (Assude et al., 2008), *self-determination theory* (SDT) is an *organismic dialectical* approach postulating that all individuals as active and growth-oriented organisms have inherent propensities to develop an ever more coherent sense of self (autonomy) and that they are naturally inclined to interconnect with their social worlds. The dialectic between the organism and its social world represents the nutrient medium for the individual's development (Deci & Ryan, 2000; Ryan & Deci, 2002, p. 5, 2017). Addressing this *organismic dialectical metatheory*, SDT states that every learner is proactively striving towards "psychological growth and integration" in order to experience "learning, mastery and connection with others" (Ryan & Deci, 2020, p. 2). Consequently, education should support this innate tendency to learn and to evolve greater autonomy (Deci & Ryan, 2000, p. 242) in order to facilitate students' natural development, that is, learners' individual processes of assimilation (Piaget, 1971, pp. 172–173), and to support them on their individual journeys towards evolving into "self-directed and lifelong learners" (Liu et al., 2016, p. 2).

Based on this meta-theoretical foundation, SDT represents a robust theoretical framework for understanding students' intrinsic and varied extrinsic quality of behaviours in a mathematics learning context by considering social-contextual factors (e.g., indicators of classroom environment like teacher behaviours) that essentially empower or impede autonomous forms of motivation, internalization (increasing ownership of external sources), and mental well-being (Deci & Ryan, 2000; Ryan & Deci, 2020). To explain learners' quality of motivation, SDT delineates various learning outcomes associated with forms of motivation ranging from self-determined to controlled (Deci & Ryan, 2002; Ryan & Deci, 2017). The most self-determined regulatory style is intrinsic motivation, followed by autonomous extrinsic regulatory styles (integrated and identified) and controlled extrinsic forms (introjected and external). Amotivation represents non-regulation. In this connection, SDT conceptualizes three basic psychological needs, the needs for autonomy, competence, and relatedness, as fundamental building blocks for learners' motivational propensity in learning situations. Following the research in the SDT literature, scholars have postulated that the social setting in the (mathematics) classroom can foster autonomously motivated learning outcomes and learners' internalization processes by promoting the satisfaction of students' basic psychological needs (Ryan & Deci, 2017; Vansteenkiste et al., 2020). The degree to which learners experience individual satisfaction of these nutrients is associated with certain dynamics within regulatory styles (autonomous versus controlled orientation in the classroom) and the corresponding content of motivation (or what; Deci & Ryan, 2000). Accordingly, to the extent they feel these three psychological basic needs are being fulfilled, learners can attain greater engagement and learning and can identify with external sources: together, these factors explain the differences between individual students' quality of motivation in (the mathematics) classroom (Deci & Ryan, 2000).

How should we address these basic psychological needs and support autonomously motivated learning in the mathematics classroom? To this effect, as indicated earlier (cf. 1), an emphasis on learners' personal relevance in the mathematics classroom represents a promising step to engage students in mathematics-learning activities. To understand the critical role of learners' personal relevance from the perspective of SDT, I first provide the classic definitions

and the historical roots of the elements of SDT, *regulatory styles* and *basic psychological needs*, that underlie its metatheory and constitute the foundation for the upcoming discussions on *why* the quality of engagement differs between learners in academic learning contexts (see section 2.1.1).

Secondly, I discuss the SDT-framed research (see section 2.1.2) as follows: The first part of the research review provides a general overview of the regulatory styles and basicpsychological-needs experience in terms of students' quality of behaviour in the classroom. On the one hand, this review intends to point out the importance of autonomous learning to promote students' mathematics-learning process and their personal growth. On the other hand, this elaboration aims to emphasize the existing empirical connections between autonomously motivated learning and the concept of personal relevance. Secondly, based on this general overview, I then highlight the SDT-framed research to date that specifically addresses the role of students' personal relevance and autonomy-supportive teaching behaviours that help learners to experience relevance in the classroom. To conclude this subchapter, I discuss SDT's potential usefulness in examining this study's main purpose (see section 2.1.3). To this effect, besides its strengths, I point out issues that are missing from SDT's ontological principles and that need to be compensated for in order to enhance the epistemological discourse on personal relevance and quality of motivation as intended in this dissertation.

2.1.1 SDT's Elements: Types of Motivation and Basic Psychological Needs

Self-determination theory embraces six mini-theories (retrieved from experimental and field studies), and each of them considers one component of motivation or psychological functioning (Ryan & Deci, 2017). In relation to the present research interest, the relevant elements are empirical knowledge about the consequences of intrinsically and extrinsically motivated behaviours and basic-psychological-needs satisfaction. For this reason, two of these six mini-theories are of crucial importance as they provide the basis for understanding learners' motivational tendencies when learning mathematics in the classroom: namely organismic integration theory (OIT; for understanding intrinsic and extrinsic motivation) and basic-psychological-needs theory (BPNT; for considering experiences of basic psychological needs). These two sub-theories aim to clarify the interconnections between learners' motivated behaviours (the experienced level of self-determination—OIT) in the classroom and their "health and well-being" (BPNT; Ryan & Deci, 2002, p. 10).

Intrinsic and Extrinsic Motivation. Retrospectively, experimental research on intrinsic motivation⁴ (Deci, 1971) and the classical differentiation between intrinsically and extrinsically motivated behaviours provided the foundation for Deci and Ryan's research on SDT (Ryan & Deci, 2017, 2020). Early research on intrinsic motivation started in the 1970s when Deci (1971) conducted laboratory and field experiments to examine the effects of external rewards (being rewarded or not) on intrinsic motivation (decrease or enhancement of intrinsic motivation).

Intrinsic motivation. Learners have an internal locus of causality; that is, they feel free to act in a way that advances their interest when involved in intrinsically motivated behaviours

⁴ SDT's first mini-theory was cognitive evaluation theory (CET; Deci, 1975; Ryan & Deci, 2017), which particularly addressed social conditions (e.g., rewards, feedback) that facilitate or diminish intrinsic motivation.

(deCharms, 1968; Ryan et al., 1996). They engage in (mathematics) activities *for their own sake* (Deci & Ryan, 2000; Ryan & Deci, 2020); that is, out of interest, enjoyment, and curiosity, as well as inherent satisfaction (Deci & Ryan, 2000; Ryan & Deci, 2020; Vansteenkiste et al., 2018). In line with Ryan and Deci (2020), intrinsic motivation represents the prototype of self-determination (Deci, 1992, p. 44) and "the energizing basis for natural organismic activity" (Deci & Ryan, 1991, p. 244).

Extrinsic motivation. In contrast to intrinsic motivation, extrinsically motivated behaviours are performed by learners to pursue social-contextual requirements which they personally value (Taylor et al., 2014). As such, those requirements are linked to the learning activity, and learners consider them to be decisive and instrumental in enabling them to be interconnected with the social regulations (Deci & Ryan, 2000). The learner's action is used to achieve a certain outcome, that is, to achieve a particular goal, in which the task is not selected solely on account of the subject matter concerned (e.g., mathematics) (Ryan and Deci, 2017; Vansteenkiste et al., 2018). In the context of extrinsic motivation, learners' intentional actions in the classroom can be distinguished according to their felt locus of causality, which can either be internal or external (deCharms, 1968, pp. 328-329; Deci, 1975; Ryan & Connell, 1989). When learners frame an activity from their socializing environment in the context of the individual values they identify with, an activity which is performed with a feeling of choice and commitment, these autonomous actions refer to an internal perceived locus of causality (similar to intrinsic motivation). In contrast, learners' controlled behaviour (Ryan & Deci, 2017) in class has an external perceived locus of causality – it is regulated by guilt (internal pressure) and threats (external pressure) without learners' having a true choice (Deci et al., 1991; Ryan & Deci, 2020; Taylor et al., 2014).

Empirical work in the 1980s focused on extrinsically motivated actions regarding the perceived locus of causality. This research provided the foundation for investigating different levels of extrinsic motivation, which generated the second pillar of the organismic dialectical tendency, internalization (Ryan & Connell, 1989; Vansteenkiste et al., 2010). Ryan and Deci (2017) defined internalization as the student's internal psychological mechanism for answering to an "externally observable interpersonal and cultural process of socialization" (p. 180). As a non-uniform conception, extrinsic motivation is specified around internalization, the active process in which learners "attempt to transform socially sanctioned mores or requests into personally endorsed values and self-regulations (...). It is the means through which individuals assimilate and reconstitute formerly external regulations so the individuals can be selfdetermined while enacting them" (Deci & Ryan, 2000, pp. 235-236). Based on interview data (retrieved from teachers and students from late elementary school), Rvan and Connell (1989) designed the first survey questionnaires to assess students' self-reported reasons for their academic-related behaviour (Ryan & Deci, 2009). Their analyses revealed types of extrinsic motivation associated with different degrees of internalization. That is, unlike other specifications of motivation (dichotomous view: intrinsic versus extrinsic, Harter, 1981; attribution theory, Lepper, 1983), SDT does not conceptualize extrinsically oriented behaviour as a drive from outside the person that forms a contrasting opposite pole to intrinsically motivated behaviour (Ryan & Deci, 2017; Taylor et al., 2014; Vansteenkiste et al., 2018). Instead, in a more sophisticated way, SDT differentiates these regulatory styles so that they fall along a continuum of increasing ownership (internalization). To support learners'

internalization in the classroom (from a less to a more fully internalized degree), students must initially experience the valence and importance of (mathematics) learning. A successful internalization process helps them get more integration at the intrapsychic and social level (Deci & Ryan, 2000, p. 236), along the lines of their natural trajectories as proposed by SDT's organismic dialectical metatheory.

OIT illustrates the various behavioural outcomes that are associated with either selfdetermined or controlled types of motivation to learn. In this connection, learners' instrumental behaviour in terms of different activity engagement in class is associated with four subtypes of extrinsic motivation arranged along the continuum, as a function of internalization (Ryan & Deci, 2017, pp. 191–192). Figure 1 illustrates schematically the varied regulatory styles and their specific sources. As indicated earlier, intrinsic regulation (the far-right end of the motivational continuum) is the hallmark of self-determination. The central section of Figure 1 depicts the different forms of extrinsic regulation ranging from *external* to *integrated regulation*.

Figure 1

Self-Determination Theory`s Taxonomy of Motivation						
Motivation	AMOTIVATION	EXTRINSIC MOTIVATION				INTRINSIC MOTIVATION
Regulatory Style		External Regulation	Introjection	Identification	Integration	
			In	ternalization		
Attributes	 Lack of perceived competence Lack of value, or Nonrelevance 	 External rewards or punishments Compliance Reactance 	 Ego involvement Focus on approval from self and others 	 Personal importance Conscious valuing of activity Self- endorsement of goals 	 Congruence Synthesis and consistency of identifications 	 Interest Enjoyment Inherent satisfaction
Perceived Locus of Causality	Impersonal	External	Somewhat External	Somewhat Internal	Internal	Internal

Self-Determination Theory's Taxonomy of Regulatory Styles (adapted from Ryan & Deci, 2020).

Within OIT, *external regulation* has an external perceived locus of causality that frames the first type of controlled behaviour (Ryan & Deci, 2020; Vansteenkiste et al., 2018). To avoid negative consequences, students learn to be cooperative in the classroom and work in accordance with the socioculturally required values (Vansteenkiste et al., 2018). They follow the rules they are expected to follow, pay attention in class, and do their schoolwork so as to be rewarded, for example with praise from their teacher (Vansteenkiste et al., 2018). Externally motivated action does not emanate from personal value. Accordingly, no innate attraction to processing the knowledge exists, and thus the learner's behavioural regulation is not internalized (Ryan & Deci, 2020; Vansteenkiste et al., 2018, p. 32).

Rather than responding to pressure from outside (external regulation), *introjected regulated* learners undergo pressure from within themselves, as they force themselves to study (e.g., mathematics) in order to prove their competencies or to feel like a good student (Vansteenkiste et al., 2018). Thereby, their behaviour is a projection of expected characteristics or actions from significant others (Ryan & Deci, 2017). However, the necessity of proving their own value and impressing others is not fully accepted by the learner, and thus it remains external to their sense of self (Deci et al., 1991; Deci & Ryan, 1995, p. 39). This kind of academic motivation is caused by internal coercion that does not reflect the learner's true choice and is thus only partially internalized (Assor et al., 2009). Since this regulation is driven by an internal regulation over time but is still a fragile type of motivational behaviour (Deci & Ryan, 2000). Accordingly, this second type of extrinsic motivation constitutes the second form of controlled motivation.⁵

Identified regulation comprises the first autonomous form of extrinsic motivation, in which students recognize the significance of learning activities in the classroom out of personally meaningful reasons that relate to the subject (mathematics) itself. Identified regulation has a more internal perceived locus of causality (Ryan & Deci, 2017), and thus in this stage, the learner has started to internalize the valence of (mathematics) learning (Deci & Ryan, 1995). This type is more volitional than external or introjected regulated behaviour, as learners identify the personal relevance or importance of (mathematics) learning activity that refers to a desired outcome at the present time or in future (Vansteenkiste et al., 2018), for example, doing practice exercises willingly to improve and succeed at the subject (Taylor et al., 2014) or gain good grades for a professional career. Thus, the necessary schoolwork forms a greater part of the integrated cognitions and affects that shape the learner's individuality (Deci & Ryan, 2000). Nonetheless, due to the usefulness of the action, this type remains extrinsic but reflects a full form of internalization (Vansteenkiste et al., 2018) and thus, these learners' actions are highly volitional; that is, similar to those resulting from intrinsic motivation (Ryan & Deci, 2017, 2020).

The fullest form of internalization is *integrated motivation*,⁶ which is defined as the most autonomous type of extrinsic motivation (Ryan & Deci, 2020, p. 3). This self-determined extrinsic behaviour appears when learners integrate their personally meaningful values into their coherent sense of self (Deci et al., 1991, p. 330); that is, when learners mutually assimilate the identified values (e.g., of mathematics learning) into harmony with comprehensive deeprooted aspects of their self, such as their values, (psychological) needs, and identity (e.g., a personal dream of becoming a psychologist, Deci et al., 1991; Deci & Ryan, 2000; Müller & Palekčić, 2005; Ryan & Deci, 2017). Similar to previous empirical studies (to name but a few: Müller & Palekčić, 2005; Ryan & Connell, 1989; Vallerand et al., 1992), I did not include integrated regulation in this present research to avoid issues that appeared regarding distinguishing between integrated regulation and identified type (Vallerand et al., 1992).

⁵ In empirical studies, the combination of external and introjected regulation reflects the degree of learners' controlled motivation (Farmer et al., 2020; Koestner et al., 2008).

⁶ The sum of self-determined types of motivation, intrinsic, identified regulation (Farmer et al., 2020), and integrated regulation (Koestner et al., 2008) constitutes autonomous motivation in empirical research.

To fully study human behaviours and experiences, Deci and Ryan (1985, p. 150) included a further type of motivation, namely *amotivation* (the far-left category outside of the motivational continuum in Figure 1). A learning situation becomes amotivating when students interpret themselves as unable to master their learning situation because they perceive themselves as incompetent. This category has an impersonal causality (Ryan & Deci, 2017), and thus it is defined neither by personal values nor interest; that is, the learner is neither extrinsically nor intrinsically regulated.

Dozens of studies using survey questionnaires, assessed across numerous cultures, have confirmed the power of these regulatory styles, specifically the knowledge behind the sequential arrangement along the internalization continuum (Deci et al., 1991; Deci & Ryan, 1995; Ryan & Deci, 2020). Thereby, the most researched domain constitutes the academic field (Vallerand & Bissonnette, 1992). More often than not, learners' behaviour in the classroom can become complex, particularly when they are multiply motivated so that different forms cooccur, for example, intrinsic motivation and extrinsic styles; also, different extrinsic regulations can emerge simultaneously for some content (Ryan & Deci, 2017; Ryan & Deci, 2020; Vansteenkiste et al., 2018). That is, the content of schoolwork may simultaneously appear to the learner as interesting (intrinsic) as well as relevant (identified), and sometimes also enjoyable (intrinsic) as it is beneficial for their own self-worth (introjected regulation) in which the personal value to the content is lacking, and so on (Ryan & Deci, 2020; Vansteenkiste et al., 2018). In this connection, the particular content or reason for engaging in a learning activity largely defines which behavioural regulation will most likely become effective (Vansteenkiste et al., 2018, p. 33). For this reason, in addition to examining the specificity of each behavioural regulation, scholars use scores within SDT research that measure, for instance, the composite index, namely the self-determination index (SDI; Levesque et al., 2004, p. 73; Ryan & Deci, 2020). The SDI, computed as SDI = (2 x Intrinsic) + (Identified) - (Introjected) - (2 x External), reflects learners' perceived relative autonomy and is highly predictive of "behavioural persistence" (Ryan & Deci, 2017, p. 195; also see Ryan & Deci, 2020).

In addition, scholars investigated homogenous profiles (Wang et al., 2017) of motivation or of basic psychological needs (Warburton et al., 2020) to examine the different patterns of learners' motivational propensity and establish whether the findings in subgroups result in a theory-consistent explanation of the perceived locus of causality (e.g., the co-occurrence of external and intrinsic regulation towards a goal is theoretically conflicting; Ryan & Deci, 2017, p. 196). As an overall focus, these additional measures help us, in a wide array of ways, to learn more about the predictive power of these regulatory styles (for example, on affective and cognitive aspects, see section 2.1.2; Ryan & Deci, 2020).

This regulation taxonomy is not in itself a developmental continuum, and it is not necessary for learners to proceed through each level of internalization step by step. In fact, learners can optionally internalize a new form of regulation at any level if the social climate in class offers the relevant experiences to satisfy their inter-individual basic psychological needs (Deci & Ryan, 2002).

Basic Psychological Needs. Ryan and Deci (2000b) specified the notion of the basic need concept as follows: "By our definition, a basic need, whether it be a physiological need (...) or a psychological need, is an energizing state that, if satisfied, conduces toward health and well-being but, if not satisfied, contributes to pathology and ill-being" (p. 74). Learners'

behaviour and their full functioning during (mathematics) lessons depend accordingly on the fulfilment of both basic physiological (Hull, 1943) and basic psychological needs (Vansteenkiste et al., 2020). Whereas the satisfaction of primary physical needs for, for example, hunger and thirst, leads to a direct reduction of these needs, basic psychological needs for autonomy, competence, and relatedness are permanent and are capable of extending or intensifying under certain favourable cultural and social conditions (Krapp, 2005, p. 631; Ryan & Deci, 2017).

According to basic-psychological-needs theory (BPNT), basic psychological needs are universal; that is, they are valid across time, age, gender, situation, culture, and social context (Ryan & Deci, 2017; Vansteenkiste et al., 2020). Vansteenkiste et al. (2020) explained further: "The effects associated with need-based experiences should be reflected in myriad cognitive, affective, and behavioral outcomes, while also surfacing at different levels, from the psychological to the neurological/biological" (p. 4). Embedded within the growth-oriented nature of the organismic dialectical metatheory of SDT, basic psychological needs serve as the essential nutrients for studying (mathematics) and for energizing the learner's internalization process and integrative functioning, as well as their active and affirmative involvement in (mathematics) learning (Deci & Ryan, 2016, pp. 15–16). In this connection, relatedness reflects the learner's strong sense of security, trust, and bonding in the (mathematics) classroom, for instance with their teacher or peers. Competence represents feeling efficacious and confident about engaging in (mathematical) tasks to reach the desired outcome as well as feeling mastery. Moreover, autonomy refers to the perception of self-initiating one's own (mathematics) activities, accompanied by a sense of psychological freedom (Deci & Ryan, 2000; Ryan & Deci, 2017, 2020).

Historically, White's (1959) criticism of empirical drive theory (Hull, 1943) describes the functional property of an organism to "interact effectively with its environment" (p. 297) as a part of intrinsically motivated behaviour. However, although the need for competence could clarify various motivational forces of intrinsic motivation, it does not entirely explain why people do not engage in activities for which they have the required competence. In this connection, deCharms (1968) emphasized that in addition to the need for competence it is also important for the learner to feel autonomy, arguing that if

a major factor in the intrinsic dimension is the desire for personal causation, then intrinsically motivating tasks are those in which the person feels that he [i.e., the individual] is in control, that he [the individual] originated the behaviour (as an Origin) with the concomitant feelings of free choice and commitment. (p. 329)

The early studies of Deci (1971) confirmed the importance of these dual basic needs for competence and autonomy to intrinsic motivation and its maintenance over time (Ryan & Deci, 2017, p. 120). Next to the inspiring work of White and deCharms, research with (young) children has stressed that they exhibit intrinsically oriented exploration and childlike curiosity when they are encircled by the feeling of relatedness; that is, warmth, caring, security, and bonding. It is fulfilled through integration and experiencing themselves as significant to others

(e.g., their parents or similar supportive carers, Deci & Ryan, 2000; Ryan & Deci, 2017; Vansteenkiste et al., 2020).

Guided by the definitions of the most important elements of SDT in relation to this present research, I now review SDT-framed research on the educational outcomes that correlate with the regulatory styles and the satisfaction/frustration of the basic psychological needs.

2.1.2 SDT's Research to Date on Learners' Classroom Motivation

General Overview: Why Do Learners Experience a Certain Quality of Motivation in the Classroom? Over the last decades, a multitude of studies within SDT have observed very interesting findings about learning outcomes (e.g., behavioural, affective, and cognitive), outcomes that follow the expected pattern regarding the experienced regulatory styles and basic psychological needs in academic contexts. I illustrate a few results to point out why it is important that learning is autonomously motivated and what difference it makes in this connection to perceive a personal relevance to learning.

According to numerous studies of beneficial outcomes, intrinsic motivation can be understood as a desirable form of motivation that fosters students' (mathematics) learning process in the classroom. Emanating from the self, a student's passion for learning (mathematics) is accompanied by spontaneous cognitions and affects about learning that lead to high-quality performance (Ryan & Deci, 2017; Taylor et al., 2014), a high level of concentration (Deci, 1992) and, especially, perseverance when meeting obstacles (Vansteenkiste et al., 2018). The concept of interest (Krapp, 2000, 2002) is closely aligned with intrinsic motivation (Bikner-Ahsbahs, 2005; Deci, 1992; Krapp, 2000, 2002), whereas the flow experience (Csikszentmihalyi, 1975, p. 36) may reflect the purest version of intrinsic motivation, where the learner joyfully gravitates towards an autotelic learning activity and the experience they associate with the object (mathematics) thereby represents both stimulus and reward simultaneously (Deci, 1992). As indicated above, the satisfaction of autonomy and competence nurtures intrinsic motivation (Deci & Ryan, 2000). However, beneficial outcomes also result from autonomous extrinsic forms; namely from integrated and identified regulation. Vansteenkiste et al. (2018) argued in this relation that the notion of personal relevance is critical for both forms, integrated as well as identified regulation.

As indicated above, integrated regulation reflects the advanced form of autonomous extrinsic motivation that leads to a completion of the learner's internalization process. However, integrated motivation requires self-reflection, self-awareness, and great effort on the part of students in terms of expressing who they are and what they value as individuals (Deci et al., 1991; Vansteenkiste et al., 2018). Accordingly, this motivational behaviour requires a certain maturity and occurs mainly at advanced levels of development; that is, the adult stage (Deci et al., 1991). According to Sheldon and Kasser (2001, p. 492), individuals who feel integrated strive willingly to attain a content that satisfies their psychological needs. Younger students, in particular, are not mature enough to associate a sense of integrated value with academic content. Based on these characteristics, students cannot easily attain integrated regulation within the everyday routine (Vansteenkiste et al., 2018) in the (mathematics) classroom and thus it is not easy for teachers to foster that regulation.

In the case of identified regulation, this behaviour co-exists with personal care, better performance (Barkoukis et al., 2014; Ryan & Deci, 2000a), and perseverance, as well as sustained energy (Ryan & Deci, 2017) as the learner links the important values they personally hold (Vansteenkiste et al., 2018) with their motives to study. Here, learners feel ready to overcome various obstacles that the subject (mathematics) or its tasks represent in order to achieve their personally significant outcomes (Vansteenkiste et al., 2018, p. 35). Further, students cope proactively with disappointments in learning situations (Deci & Ryan, 2000). Theoretically, all three basic psychological needs are satisfied in the presence of identified regulation (Deci & Ryan, 2000; Vansteenkiste et al., 2018). The experiences of competence and relatedness are critical to students' starting to own externally offered values (i.e., beginning of internalization; Deci & Ryan, 2000). The need for competence emerges as a need-substitute (Ryan & Deci, 2017) when, for instance, learners want to compensate for their lacking sense of autonomy by displaying power solely because they want to overcome a sense of being confined (Martela et al., 2019, p. 3; Vansteenkiste et al., 2020). The need for competence is also characteristic of ego concerns in particular (Haerens et al., 2015, p. 33). Along similar lines, striving towards a sense of secure relatedness is desired (as a compensatory strategy, Ryan & Deci, 2017) particularly when situations bring feelings of need frustration, a feeling of being controlled, indifference, or having no true choice, as well as being overburdened (Vansteenkiste et al., 2020). Accordingly, to awaken a sense of personal relevance, the social context (e.g., of mathematics) should specifically satisfy a feeling of autonomy, not only of competence and relatedness (Deci & Ryan, 2000; Vansteenkiste et al., 2018).

Altogether, this specific instrumental behaviour—identified regulation—helps learners to become self-determined (Deci & Ryan, 1995), and as such the more internalized the motivation, the more the mathematics activities' personal relevance becomes part of a learner's identity. Thus, referring to this study's context, the properties of this regulatory style, make it *optimal* for motivating students and thus feasible for teachers to use in terms of fostering students' process of internalization in a mathematics-learning environment.

Speaking of the support of internalization, the needs of competence and relatedness suffice to meet introjected regulation. Although the learner has partially accepted the reason for learning (no external prompts are required; Vansteenkiste et al., 2018), the activities remain conflictual to the self. That is, reasons to study, for example being a smart, hard-working student in order to feel competent in mathematics, have not been identified as self-chosen personal values, namely as being a part of the individual learner (Deci et al., 1991, p. 329). In this connection, the reasons for the learners' behaviour, such as anxiety (Ryan & Deci, 2000b) guilt, shame (Bartholomew et al., 2018), or striving towards self-worth to avoid failure, are within the learners themselves (Ryan & Deci, 2017). Ryan and Deci (2017, p. 186) postulated that this kind of behaviour is especially present in achievement-related fields in which competition and inter-individual comparisons exist. This type of motivation leads to inconsistent self-esteem (Deci & Ryan, 1995), as learners are constantly dependent on others' confirmation of their worth, for example in the domain of achievement, where self-worth functions as a guiding strategy for engaging in the (mathematics) classroom (Ryan & Deci, 2017). Accordingly, it is assumed that this kind of introjected self-perfection is detrimental for learners (Assor et al., 2009). In particular, the confusion of self-worth with achievement that frames the search for self-acceptance in a highly competitive environment is counterproductive

for students when they feel unable to keep up with the class, but also feel pressure to sustain competency and avoid failure (Covington, 1998, pp. 78–79, 2009). In summary, in relation to this investigation, not only is introjected regulation (Assor et al., 2009) suboptimal for students' mathematics-learning processes and their personal development, but so is external regulation, as will now be discussed.

Initiated by an external source, learners perform a behaviour that is driven by fear, pressure, punishments, threats, and rewards (Deci et al., 1991), for example mediated by their mathematics teacher or a parent. As indicated above, the external pressure drives the learner so that they only perceive a low relevance (e.g., a socially valued one) in schoolwork. The externally regulated student complies with expectations as long as the contingency lasts. That is, without the expectations in the classroom that control learners' behaviour, they would not maintain their behaviour in lessons over the long term and would withdraw from the learning activity, to which little attention and effort was devoted anyway (Deci & Ryan, 2000; Ryan & Deci, 2017). Occurring as the inverse of intrinsic motivation (Deci & Ryan, 2000), this type of motivation is predictive of ill-being and poor academic performance, as a consequence of basic-psychological-need frustration (Warburton et al., 2020).

Complete lack of basic-psychological-needs satisfaction leads to an experience of reactance (Brehm, 1966):

if a person's behavioral freedom is reduced with reduction or threatened with reduction, he [the individual] will become motivationally aroused. This arousal would presumably be directed against any further loss of freedom and it would also be directed toward the re-establishment of whatever freedom had already been lost or threatened. Since this hypothetical motivational state is in response to the reduction (or threatened reduction) of one's potential for acting, and conceptually may be considered a counterforce, it will be called "psychological reactance". (p. 2)

Further, the absence of a personal intention induces the students to quit and display amotivation (Bartholomew et al., 2018). In most cases, this takes place after the student has experienced a period of consistently negative feedback and poor performance on their part (Deci & Ryan, 1985). This strongly predicts poor learning outcomes and reduced well-being which culminates in personal helplessness (Abramson et al., 1978, p. 52).

Focusing on the pervasive influences, diverse cultures constitute unique pathways for individuals to satisfy their psychological needs that nurture the process of internalization (Ryan & Deci, 2017). What is more, an individual's motives, orientations and identity are contextualised by cultural norms and values (Ryan & Deci, 2017; Vansteenkiste et al., 2020). As a result, the regulations affecting the individual—both social and personal, such as their motives and aspirations— are also a product of the culture in which that individual lives. Thus, each learner follows a distinct path towards well-being and social integration (Ryan & Deci, 2017). Because different types of content differ on a cross-cultural level, they relate to a greater or lesser extent to basic psychological needs and are correspondingly integrated (Chirkov et al., 2003; Ryan & Deci, 2017). However, many SDT-framed studies across cultures (to name

but a few: Chen et al., 2015; Chirkov et al., 2003) showed that regardless of cultural communities (e.g., individualism versus collectivism, Chirkov et al., 2003), associations between basic psychological needs and wellness exist. It is especially worth noting that autonomy is necessary for well-being whatever the circumstances, and regardless of the place of autonomy in the cultural context. Specifically, even in cultures where autonomy was not valued, the empirical evidence showed it was essential to well-being (Ryan & Deci, 2017). Therefore, SDT is particularly helpful in terms of understanding how individuals internalize content that they have accumulated through cultural and social channels, as well as the ways in which the cultural context encourages autonomy. In relation to this current investigation, it would be interesting to see how far the cultural elements within mathematics curricula contribute to and provide a frame for studying mathematics autonomously and promote increasing internalization (see section 2.4).

To conclude, from the empirical evidence to date, it appears that there is no doubt about the benefits that come with autonomous motivation for students' learning processes and their personal development. Learners need to recognize the personal value of mathematics learning to assimilate increasingly externally offered contingencies and to display autonomous behaviour. SDT postulates that the robustness of a certain motivational direction and learners' internalization process (that of transferring contingencies from external sources into their own) depends on the involvement of learners' inherent basic psychological needs (Deci et al., 1991). In this regard, the experiences of competence and relatedness are indeed vital for the learner's internalization or to start it. However, to experience deep-rooted personal importance and to put one's heart into learning situations in mathematics, the satisfaction of autonomy is essential. Accordingly, learners effectively experience mathematics learning as personally relevant when the learning environment fulfils their three basic psychological needs (Vansteenkiste et al., 2018) or the experience of autonomy in particular. In a nutshell, within mathematics-learning situations, autonomous types of motivation come with the benefit of cultivating each individual student's natural interest and encouraging it to grow. In comparison to the other types of autonomous motivation (intrinsic or integrated regulation), the teacher's external endeavour can address identified regulation by helping students to understand the value and personal relevance of mathematics learning. Thus, to the extent the learner perceives mathematics learning as important, personally relevant, and self-determined, they will identify with the external mathematical content they are learning and actively internalize it through identification (Vallerand et al., 1992, pp. 1006–1007). For this reason, it is a matter of particular interest which mathematics-related personal relevance learners associate with identified regulation.

Importance of Students' Personal Relevance from SDT's Perspective. Scholars within SDT have investigated through varied research settings the process of internalizing by means of the concept personal relevance. Thereby the notion of personal relevance has been addressed using different relevance-related terms, for example rationale, goal, personal meaning, or self-relevance. To give few examples of such a studies, Jang (2008) found that the recognition of rationales provided through an autonomy-supportive climate during uninteresting learning contexts produced positive outcomes for learning, motivation, and engagement. Thereby, teachers explained the value of uninteresting learning activities to their students (college students from educational psychology class) within statistics lessons, so that

the participants experienced increasing internalization over time. These rationales were transferred into self-endorsed motives (identified regulation) and induced learners to make an effort for the still-boring learning materials.

The feeling of being involved in classroom processes and having one's opinions taken seriously was also found to be supportive when Steingut et al. (2017), examined within a metaanalysis (including the academic context in that analysis) the unique properties of the rationales provided to the sample (participants were from various developmental stages-elementary, middle, and high school, college, and adult). Their findings were similar to those of previous research carried out within an educational context with early adolescents (Vansteenkiste et al., 2005) in that the specific type of rationale determines the impact of such rationales, and in this relation prosocial and autonomous rationales (e.g., community- and helpfulness-related goals) foster greater engagement and performance than do controlling rationales (e.g., goals relating to money, marks, and social status). Steingut et al. (2017) claimed that prosocial rationales correlate positively with the satisfaction of the basic psychological needs, which were associated with beneficial learning outcomes. Interestingly, their results pointed out that rationale provision may have a negative effect on the competence need. Referring to Hulleman et al. (2010, p. 881), who conducted their study with college students of mathematics and psychology, Steingut et al. (2017) assumed that a decrease in the learner's perceived competence might be associated with low performance. In that case, students might then feel unable to cope and engage when the teacher emphasizes the great importance of the subject mathematics. These results indicated that not all students profit from rationale provision and that, in mathematics, rationales can be experienced at the same time in different ways. To benefit everyone in the class, teachers need to frame the relevance so as to highlight it in light of the social conditions in the classroom and thus begin to understand learners' own rationales first, before providing or highlighting any other rationales in the mathematics classroom.

Referring to classroom conditions, Assor et al. (2002, p. 272) stressed that certain teacher behaviours in elementary schools-"fostering relevance" as well as "suppressing criticism" and making an effort to comprehend their students' subjective values and learning-can support learners' experiences of autonomy in the classroom. In this line, Reeve (2016) underscored the importance of the teacher-student relationship, and Haerens et al. (2016) emphasized the avoidance of controlling teaching styles (e.g., controlling communicative approaches) in order to foster an autonomy-supportive learning environment in the classroom. Controlling teaching (Haerens et al., 2016; Reeve & Cheon, 2021) can be differentiated into externally controlling ways of controlling learners' behaviour (e.g., using language that pressures the student, such as "You have to!") and internally controlling ways, such as psychological control (e.g., expressing disappointment, triggering shame and a sense of guilt). Haerens et al. (2016) also argued that teachers should refrain from using controlling and pressuring behaviour in cases where, for example, the students are not following instructions or advice, as such behaviour frustrates the learners' basic psychological needs. Such frustration in turn associates with controlled motivation and amotivation (Haerens et al., 2015). In particular, a sense of intimidating obligation takes an emotional toll on the student, affecting their well-being and energy, as well as leading to poor performance (Haerens et al., 2016, p. 60).

To conclude, a focus on the students' relevance, or rationales that are student-centred and support the experience of autonomy, as well as autonomy-supportive teaching styles that foster a positive teacher–student relationship contribute to satisfying learners' basic needs experience. These factors in turn foster their processes of internalization and their striving towards autonomy, which is a constant that is independent of the developmental stage (Assor et al., 2002). In a recent article, based on theoretical and empirical considerations, Vansteenkiste et al. (2018) provided a well-elaborated broad overview on the role of personal meaning and self-relevance for promoting students' learning processes. In this connection, Vansteenkiste et al. (2018) not only placed emphasis on highlighting self-relevance in learning contexts, but also discussed the location of personal meaning and self-relevance in terms of the taxonomy of motivation. As an attempt to conceptualize the relations between self-relevance and the internalization continuum, they argued that "although all types of extrinsically motivated activities may be important to people in a very general sense, the notion of self-relevance applies only to the more internalized forms of extrinsic motivation" (p. 33).

2.1.3 Critical Reflection on SDT

In the following, the elaborated theoretical lens of SDT is discussed in terms of its potential to provide answers for this study's main purpose. According to Radford (2008a), a theoretical lens can be viewed "as a way of producing understandings and ways of action based on" basic principles (P), methodology (M), and research questions (Q) (p. 320). Within this tripartite view (P, M, Q) of a theoretical lens, P embraces the system of principles, M is a set of methodologies (technical procedures of data collection and interpretation) facilitating P in producing relevant data, and Q is a set of research questions. These three elements are closely interwoven with each other and help not only in conceptualizing theories (how is understanding produced?), but also in exploring a theory's *boundary*—that is, the limit of the theory's principles. If such limitations emerge in the context of the intended investigation, "Connecting theories can (...) be accomplished at different levels (principles, methodology, research questions), with different levels of intensity" (Radford, 2008b, p. 14). Next, I will consider SDT's system of principles in order to identify the theory's strength and boundary in terms of this study's research interest; that is, by clarifying the conceptual relation between learners' personal relevance (individual relevance systems) and their motivational tendency in the mathematics classroom.

 P_1 : Briefly, by following an organismic dialectical metatheory, SDT's lens focuses on the understanding of and increasing knowledge about learners' quality of motivation;⁷ that is, the kind of motivation, not the amount. In doing so, SDT distinguishes between the general categories of intrinsic and extrinsic motivation in order to emphasize the motivational dynamics underlying the process of internalization, that is, the function of increasing ownership reflected by the extrinsic regulatory styles: external, introjected, identified, and integrated regulation. P_2 : Thereby, learners' social context can promote intrinsic motivation and internalization by nurturing the fundamental basic psychological needs of autonomy, competence, and relatedness, which are vital for learners' sense of self, which is analogous to

⁷ Again, SDT's theoretical principles are not restricted to the enterprise of schooling (Ryan & Deci, 2017; Vansteenkiste et al., 2020).

SDT's metatheory. These theoretical principles explicitly highlight in this relation the role of social cultural context, which affects the learner's growth and learning in an ongoing dialectic between the learner's basic psychological needs and their social environment (Ryan & Deci, 2000a). Historically, this grand theory emerged over the decades through multiple international research contributions from different contexts. There have been a limited number of qualitative studies that have been important, for instance, to developing the survey instruments (e.g., regulatory styles), but most of the research findings are based on surveys. Thus, SDT's general principles, which have been proven to be robust and predictive, are frequently researched within a quantitative research paradigm (Ryan & Deci, 2020, pp. 17–18). To conclude, the broad framework of SDT seeks to understand the variability in the social cultural conditions that foster individuals' basic-psychological-needs satisfaction in terms of the quality aspect of motivation and differences in personality. In relation to this present study, due to certain limitations, SDT cannot thoroughly explain the interrelations between learners' personal relevance and their motivation.

Based on the well-elaborated knowledge to date, SDT provides an excellent theoretical foundation for understanding learners' classroom behaviour in connection to their individual experiences of basic psychological needs, whereas the central social experiences of classroom motivation are not discussed in the same depth. As indicated above (cf. 2.1), the dialectic between the individual and their social context represents the nutrient medium for an organism's development (i.e., natural growth orientation). Moreover, Vansteenkiste et al. (2018) stated that the notion of personal relevance can only associate with autonomous forms of motivation. Does learners' personal relevance only occur in connection with identified or more internalized motivation? What if learners value the importance of belonging to their mathematics classroom? Would that not be personally meaningful in terms of the organismic integration within SDT? To clarify these questions, it is necessary to find a theoretical lens that explains the development of learners' personal relevance in the mathematics classroom and is compatible with SDT's theoretical conceptualization. To this effect, I include the theoretical approach that is used to understand interpersonal interactions within the classroom, namely social constructivism (see section 2.2.1).

Further, several studies, and in particularly the sophisticated review conducted by Vansteenkiste et al. (2018), have highlighted the importance of personal meaning/importance/self-relevance for the process of internalization and autonomous motivation. They jointly pointed out the significance of understanding exactly what is relevant to learners, instead of providing or clarifying meanings from the teacher's point of view, or the socially valued relevance of learning. Thereby, scholars stated the need for more research (e.g., intervention studies, Vansteenkiste et al., 2018) in future to explore the effect of personal relevance or rationale on motivation (Steingut et al., 2017).

I claim, in this connection, that in order to increase our knowledge of the relationship between the relevance of learning for learners and their motivation, we initially need to clarify the conceptual understanding of learners' personal relevance. To the best of my knowledge, within SDT there exists no theoretical specification of the concept *students' personal relevance* in an academic context. The review of the literature to date (cf. 2.1.2) showed that various scholars have applied this term (relevance) in connection with different notions (e.g., rationale, personal meaning/significance, or self-relevance). Scholars compared their evidence in this

research without clarifying and contrasting their fundamental theoretical conceptualizations of these notions (e.g., Vansteenkiste et al., 2018). Recently, Ryan and Deci (2017, pp. 252-254) discussed the general notion of meaning, in order to point out that meaning cannot be considered as a basic psychological need. In their view, meaning occurs as a consequence of those needs being satisfied. They postulated that from the perspective of SDT, meaning can be considered as an outcome of the individual's tendency to grow (process of intrinsic motivation) and interconnect with their social worlds (organismic integration). However, with regard to learners' relevance, there is still no clear consensus about the conceptual definition within SDT's research landscape. Although the concept of meaning within SDT seems to theoretically relate to learners' personal relevance, previous scholars have not explained how learners construct personal relevance in learning situations and which factors affect this construction. A clear and well-defined conceptual framework is critical to exploring the relations between personal relevance and motivation. For this reason, a concept is necessary which adopts learners' perspective when learning mathematics, in order to explain their construction of personal relevance with reference to their social context in the classroom and their individual experience of basic psychological needs (to explain motivation). The motivational construct "personal meaning" seems to encompass a theoretical conceptualization that will serve this study's purposes (see section 2.2.2).

In the following, I initially provide a general overview of the theoretical perspective of social constructivism and subsequently introduce the theoretical conceptualization of personal meaning.

2.2 Learners' Construction of Personal Meaning in the Mathematics Classroom

In what follows, the general theoretical understanding of the social constructivist approach is delineated.

2.2.1 The Social Constructivist Perspective

Building on the pioneering research of Vygotsky (1962, 1978, 1987), the perspective of social constructivism helps us to understand and enhance the learning processes that take place in classroom interactions (Hickey, 1997, p. 175). The specific focus of this lens is on "the impact of collaboration, social context, and negotiation on thinking and learning" (Hickey, 1997, p. 175). Referring to concepts from anthropology and sociology, the social constructivist philosophy views each student as indivisibly connected to their surrounding social environment (Ernest, 1998, 2010; Hickey, 1997). This perspective follows the guiding idea of "assisted learning" (Hickey, 1997, p. 175) that emerges within the concept of the *zone of proximal development* (ZPD). Vygotsky (1978) stated: "what children can do with the assistance of others might be in some sense even more indicative of their mental development than what they can do alone" (p. 85). ZPD refers to instances in which a learner, who cannot master a task with their current skills, is "scaffolded" by a more skilled person, who helps them to reach their potential or proximal goal (Ernest, 2010; Hickey, 1997; Vygotsky, 1978). In this spirit, *scaffolding* reflects teachers' supporting learners in the process of internalizing socially accessible knowledge and proficiencies in order to grow and become independent (Bakker et
al., 2015; Vygotsky, 1978, p. 130). Without the teacher's guidance the internalization would be too demanding for students to master their proximal goals (Bakker et al., 2015).

Within the Vygotskian theoretical space, the individual's personal development is thereby affected and formed twice, namely through the social and individual internal development processes (Vygotsky, 1978). As a result, we can see the growth and development of the learner's intellect and mind-set as coming from an internalized dialogue with their social world (Ernest, 1998, 2010). Vygotsky's implications (1987) emphasized the vitality of such conversations for the individual's development of knowledge and learning. Guided by Vygotsky's theoretical enterprise, Ernest (2010) proposed a cycle of appropriation, transformation, publication, and conventionalization (see Figure 2) to illustrate the mutual dialogue between the social and the individual that contributes to students' learning; that is, "the development of mind, personal identity, language and knowledge" (Ernest, 2010, p. 44).

Figure 2

Model of Sign A _l	ppropriation a	and Use	(adapted from	Ernest,	2010).
------------------------------	----------------	---------	---------------	---------	--------

		Individual	Collective	
Ownership	Public	Individual's public utilization of sign to express personal meanings	Conventionalization \rightarrow	Conventionalized and socially negotiated sign use (via critical response & acceptance)
		Publication ↑		Appropriation \downarrow
	Private	Individual's development of personal meanings for sign and its use	Transformation \leftarrow	Individual's own unreflective response to and imitative use of new sign utterance

Social Location

According to Ernest (2010, pp. 44–45), this scheme illustrates a cycle on the whole where meanings located within the *individual and private* realms as well as utterances within the *collective and public* realms are reciprocally modelled by means of *internalized* conversation (Ernest, 1998). It demonstrates how individuals come to personally appropriate signs by perceiving their prevalent social use. First and foremost, the initial process is that of *appropriation*, in which a learner reacts without reflecting on common usage, and models themselves and their behaviour on well-worn routines within the relevant practical discourse. On the basis of such experiences, a learner passes through the whole cycle of all four processes of appropriation, transformation, publication, and conventionalization several times, so that the individual perceives the routine of applying and expressing that sign in public multiple times. They learn the appropriate link between that sign and its corresponding implicit rules and the references it implies. Based on the experiences gained from this, the learner constitutes their own meaning for that experienced sign as well as its usage, and through the process of *transformation* the sign becomes an item of personal property which is in private hands. At publication stage, the student is ready to apply the sign within "autonomous conversational"

(Ernest, 2010, p. 45) activities. The meaning of the sign that has been thus produced underlies, then, the *processes of conventionalization*. Here, the expressed sign in the social realm is offered on diverse occasions to receive a response from the public context in the form of renegotiation, critique, or compliance, for example. That is, in relation to ZPD (Vygotsky, 1978), the process of conventionalization "takes place in the visible centre of ZPD" (Ernest, 2010, p. 45) where the individual student is taught through their own experiences and instructed on how to use the sign in the community. Looking at all operations within the cycle, appropriation and publication build the boundary actions that enclose learners' sign-internalization process (receiving and producing signs), which takes place collectively through the dialogue between the individual and the social sphere. Within the privately owned sphere, signs come to be transformed from the collective into the individual sphere by means of the production of personal meanings for the sign and its usage. For this reason, a crucial stage of learning occurs in the private sphere.

To conclude, Ernest argues that one can say that the ZPD (Vygotsky, 1978) includes the overall cycle that explains how an individual learns concept development though signs, that is, how learning and knowledge are produced (Ernest, 2010, p. 45). Signs or texts are a kind of cultural representation (although there are others), and they appear in/are relevant to all four quadrants of the model of sign appropriation and use. Since culture encompasses signs and texts, culture affects learning (Ernest, 2010).

Referring to Radford's system of principles (P), the theoretical stance of social constructivism can be discussed in terms of this present research's main purpose. P_1 : Generally, the social constructivist philosophical ethos explains the conversation between the social context and the individual that shapes knowledge production in the mathematics classroom. P_2 : Thus it places the social sphere in a central position, as individual social phenomena such as conversation are difficult to break down and define individually (Ernest, 1998, pp.135–136). P_3 : On the contrary, the social constructivist approach provides the social experiences of "objective' and 'subjective' knowledge" that are "based on dialectical, socially situated interpersonal 'conversations" (Ernest, 1998, p. 136) within mathematics education.

In his seminal work, Hickey (1997) postulated that we need to consider classroom motivation through a social constructivist approach so as to increase our understanding of motivation in mathematics learning. The social constructive perspective on social interactions helps us to understand how learners' social experiences of autonomy and self-determination are brought about by interpersonal relations between different people in the mathematics classroom. This lens might become the counterpart of learners' individual classroom experiences in that it could help us to comprehensively understand the development of the learners' individual relevance systems in terms of their motivation in mathematics. The motivational construct personal meaning seems to be grounded on a theoretical conceptualization that synergizes both learners' individual and social experiences into their personal relevance construction in mathematics. Before I discuss these properties in more detail (see section 2.3), I introduce the theoretical conceptualization of the motivational construct personal meaning in general, which reflects the personal relevance learners attach to learning mathematics.

2.2.2 The Motivational Construct of Personal Meaning

Generally, all humans *search for meaning* (Frankl, 1978) and desire to have a meaningful life (Bruner, 1990; Frankl, 1978; Ryan & Deci, 2017; Vinner, 2007). The *need for meaning* is fundamental and concerns our longing to understand ideas that were once unknown to us (Combe, 2014). Meaning satisfies this experience of coming to understand something (Combe, 2014; Gebhard, 2014), and Blumenberg expressed this succinctly when he stated that the need for meaning is a demand for a "legibility of the world" (German: *Lesbarkeit der Welt*, Blumenberg, 1981, p. 10, translated by the author). A perceived lack of meaning may have an adverse effect on the individual's well-being, for example by leading to depression (Biller, 1991, p. 105).

The striving towards meaning is not only common within psychology, however. It is also vital in an educational setting, where students need to experience meaning (Vinner, 2007). In this connection, by referring to Postman (1966), Vinner (2007) stressed "that if education does not challenge the students with meaningful goals it will die" (p. 3). Based on the general need for meaning, which must be in place if learning is to be successful, scholars from a Research Training Group (German: Graduiertenkolleg), "Bildungsgangforschung", sponsored by the German Research Foundation (DFG), discussed the issues around the meanings attached by students to their personal biographies and to different subject areas, as well as their feelings about lessons at school. They intended to study which conditions affect the construction of meaning from the learner's perspective, and how they do so. Thereby, they specifically considered students' biographies (that is, life stories and longer-term considerations) and the debate about subjective and objective requirements learners need to face when learning (Koller, 2008). Through this discussion, several researchers from different disciplines (e.g., physical education, physics, linguistics, and mathematics education) discussed the topic of learners' personal meaning in an educational context (see Koller, 2008 for more). Vollstedt and Vorhölter (2008) elaborated the notion of meaning within mathematics education. Next, I will present the motivational construct of personal meaning with regard to mathematics education.

Learning mathematics within a classroom is in its nature a collective and cultural phenomenon (Cobb et al., 1992). However, the teacher's impulses and the mathematics activities engaged in in class are not collectively experienced in exactly the same manner. Within one learning environment, the teacher's instructions are perceived and interpreted differently, as students ask themselves: "What is the personal relevance of learning mathematics for me?" (Kilpatrick et al., 2005; Vollstedt, 2011b). Dependent on their biographical lens, each learner actively constructs a subjective personal relevance for the school subject mathematics that fits their own personal life story and identity (Blumer, 1969; Cobb et al., 1992; Mason & Johnston-Wilder, 2004). The individual world that is constructed in this fashion functions as a *mental compass* providing students with their own meaningful learning orientation in the mathematics classroom. Hence, it does not matter how well prepared mathematics educators and their concepts of teaching styles are as long as they fail to connect to those individual "worlds" that direct students in the classroom (Blumer, 1969). Accordingly, teachers should not just understand how they personally perceive the subject of mathematics and then assign their own perspectives to students; rather, they should understand how mathematics learning appears to those students, so that the stimuli they provide can

successfully match the individual mental compass that each student uses to study mathematics. To stress this notion, Blumer (1969) stated:

The contention that people act on the basis of the meaning of their objects has profound methodological implications. It signifies immediately that if the scholar wishes to understand the action of people it is necessary for him [them] to see them. Failure to see their objects as they see them, or a substitution of his [their] meanings of the objects for their meanings, is the gravest kind of error that the social scientist can commit. It leads to the setting up of a fictitious world. Simply put, people act toward things on the basis of the meaning that these things have for them, not on the basis of the meaning that these things have for them, not on

However, it is not an easy task to capture those different student voices in one classroom when the teacher does not have access to the individual learners' worlds (Blumer, 1969). Referring to this issue in a mathematics-learning context, Vollstedt and Vorhölter (2008) jointly developed the theoretical conceptualization of *personal meaning*, a motivational construct reflecting the personal relevance that learners associated with studying mathematics, and its relation to the activities, content, and people involved (e.g., teacher or peers) in an educational context. Initially, Vollstedt and Vorhölter used existing theoretical as well as empirical evidence to formulate the theoretical concept behind personal meaning (Vorhölter, 2009). Subsequently, two qualitative studies, focusing on the relations between German students' personal meanings and modelling tasks in mathematics (Vorhölter, 2009) and on the different kinds of personal meaning in Germany and Hong Kong (Vollstedt, 2011b), contributed profound and solid knowledge which enriched and refined the conceptualization of personal meaning (Vorhölter, 2008). Both studies were based on Grounded Theory (Strauss & Corbin, 1996).

Briefly, three aspects are essential within the methodology of Grounded Theory (Strauss & Corbin, 1996). Firstly, the type of coding is theoretical. That is, theoretical concepts are constituted that have an explanatory value for the examined phenomena. Secondly, sampling sources for the source data are identified, from which the theory is developed, and further samples are carried out successively and iteratively throughout the entire research process (*theoretical sampling*). Thirdly, as a result of the constant comparisons between the phenomena and research contexts, the theoretical concepts emerge; that is, these sensitizing concepts are empirically based results.

Vorhölter's study considered students from grade 10, whereas Vollstedt studied ninth and tenth graders' kinds of personal meaning. Since this present investigation focuses on the interrelations between the patterns of personal meaning (individual relevance system) students assign to mathematics learning and quality of motivation, I mainly concentrate on the evidence of Vollstedt's study, as elaborated on below.

Figure 3



Theoretical Concept of Personal Meaning (adapted from Vorhölter, 2009; see also Vollstedt, 2011b).

Figure 3 depicts the theoretical conceptualization of personal meaning. The subjectively perceived personal meaning can assume the shape of a personal aim, purpose, goal, value, or a benefit that the individual associates with an object or an action (Vollstedt, 2011b; Vollstedt & Duchhardt, 2019; Vollstedt & Vorhölter, 2008; Vorhölter, 2009). By adopting the perspective of the *individual* learner, this conceptualization describes the construction of personal meaning in the context of the mathematics-learning situation. Within this learning context, students' construction of personal meaning is individually affected by preliminaries, classified as their personal background (stable properties) and personal characteristics (influenceable properties). Learners' personal background refers to attributes (e.g., gender, family, cultural and social contexts, and age) that belong to the individual and as such cannot be influenced by themselves or the outer environment. On the contrary, learners' personal characteristics are variable by nature, referring to concepts from different scientific disciplines, such as educational psychology (interest, Krapp, 2002; basic psychological needs for autonomy, competence, and relatedness, Ryan & Deci, 2017; Vansteenkiste et al., 2020), educational science (developmental tasks; Havighurst, 1972), and particularly with regard to mathematics education (beliefs; Törner, 2002). Based on these prerequisites as well as the learning settings in the classroom, students individually construct a personal meaning that is accordingly associated with the evaluation of the experienced situation and the consequences of their action, which may be either supportive or obstructive to mathematics learning. Learners' need for meaning is permanent, and as such for each (new) item of mathematics-learning content the personal relevance has to be continually interpreted and subjectively constructed anew (Fischer & Malle, 1985). A specific construction cannot persist unaltered when the student's learning situation and the influences on their personal characteristics change. These changes can make their own demands on subsequent situations; for instance, the factors that affect the

construction in a certain way can become personal requirements for the learners and influence their personal development (Vorhölter, 2009). Accordingly, the teacher cannot impose a personal meaning from outside (Meyer, 2008), but can only assist their students to recognize their own personal meaning so that they can find the mental compass that supports them in their personal development.

To conclude, the construction of personal meaning involves an interaction between individual (endogenous personal characteristics) and social (exogenous to the mathematics-learning situation in the classroom) factors (cf. Figure 3). Moreover, personal meaning in its nature can become conscious, albeit not always (Vollstedt, 2011b; Vorhölter, 2009). Vollstedt (2011b) and Vorhölter (2009) postulated that not only may different learners simultaneously assign different meanings to the same mathematics context, but also that one learner can associate various personal meanings to a specific learning situation. However, what kinds of personal meaning do learners construct in mathematics-learning situations?

With regard to this issue, Vollstedt (2011b) identified different kinds of personal meaning students related to the subject mathematics. As mentioned earlier, in a qualitative study, 34 students, ninth and tenth graders (aged 15–17), from Germany and Hong Kong (17 from each country) were interviewed to investigate the influence of culture on the construction of personal meaning, in order to further validate and refine its theoretical conceptualization. The interviews conducted by Vollstedt started with a sequence of stimulated recall (Gass & Mackey, 2000). Thereby the students watched a short video sequence (5–10 min) of their last mathematics lessons. The sequence was intended to provide an example of a situation in which the learners dealt with something new for them. This situation both reaffirm existing kinds of personal meaning and help construct new ones. Then, the scholar asked the students to reflect on the thoughts they had when they were attending the lesson and while watching the sequence. Within subsequent interviews (approx. 35-45 min) several topics inspired by the relational framework of personal meaning were elaborated (cf. Figure 3). Thereby, the scholar discussed questions like: How did you like this mathematics lesson? What was especially interesting? What feelings do you relate to mathematics lessons? Why do you learn mathematics? What can mathematics be used for? (see Vollstedt, 2011a for more). Accordingly, learners' social experiences in the classroom were individually reflected upon within the interviews. As a central result, Vollstedt reconstructed 17 kinds of personal meaning with relation to mathematics learning in an educational context. The 17 kinds of personal meaning varied between different notions ranging from duty (e.g., I mainly deal with mathematics because I have to) to purism of mathematics (e.g., mathematics is beautiful to me as it is unique in its formalism).

In the following, I provide a brief description (see Vollstedt, 2011b, pp. 139–227 for more) of each kind of personal meaning that has been reconstructed on the basis of qualitative empirical data to date (apart from "reference to reality", as explained below).

Active practice of mathematics: When learning/dealing with mathematics, it is personally relevant for the learner to actively deal with the mathematical content themselves. The basic focus lies on their own activity, which gives pleasure to them. Active pursuit can promote the learning process, for instance by being useful in helping the student to prepare for exams or by providing them with a source of enjoyment. The task variety stimulates to deal with them.

*Balance and even-temperedness*⁸: Mathematics learning is personally relevant for the learner when they feel balance. Engagement with mathematical content is freely chosen during free time or in the classroom, in which the learner feels relaxed while they deal with mathematical tasks.

*Classroom management*⁹: This type of personal meaning refers to the classroom structure when learning mathematics. The student learns mathematics within a well-organized learning atmosphere in which the teacher is not only someone they trust but also has authority, cares about the learning atmosphere, and can ensure positive classroom management.

Cognitive challenge: This personal meaning has a strong relation to mathematical content and the learner strives towards solving complex and difficult tasks in order to be and feel challenged for its own sake. When the learner experiences mathematics learning as cognitively challenging, they want to improve their performance or to compete with themselves or their peers.

Duty: The learner deals with mathematics as it is a compulsory school subject and to fulfil external (e.g., from family, people who have promised rewards) and their own expectations. They pile pressure on themselves by competing and are well prepared to meet the requirements in examinations.

Emotional-affective relation to the teacher: The student learns mathematics in the classroom due to the affective bond between them and their mathematics teacher. This bond is strengthened through a positive and friendly learning atmosphere, for example by joking. This respect or appreciation for the teacher might also become the basic centre of the learner's attention.

Examination: Mathematics learning is personally relevant as it is necessary to pass unavoidable exams at school. This personal meaning is about both the effects that passing or failing might have (e.g., for present and future education) and the type of exam preparation (e.g., constant/short-term learning).

Experience of autonomy: Mathematics learning becomes relevant when the student can independently deal with mathematics-related tasks/content and decide which activities to work on as well as complete the learning activities on their own in the classroom. A personal characteristic of the individual is here the focus, namely, the desire for self-determination of one's own actions.

Experience of competence: Learning mathematics is personally relevant when the learner can feel competent and successful. The experience of competence is a consequence of certain prerequisites being fulfilled, either through one's own activities (e.g., presenting one's own solutions in front of the others, achieving high marks) or through interaction with others in the classroom (e.g., supporting other students successfully and independently).

Experience of social relatedness: The learner studies mathematics in the classroom to feel integrated in the group. In this connection, friendship can be intensified through

⁸ Within a previous study (Suriakumaran et al., 2019) that considered the same data base as this present study, balance and even-temperedness did not fit the data; consequently this personal meaning was excluded from this present research.

⁹ As explained below, within subsequent studies a slight change was made concerning the personal meaning "efficiency" from Vollstedt (2011b); it was changed to "classroom management".

mathematics-related or non-mathematics-related activities as well as by solving tasks on the blackboard. Specifically here, the focus is on collective learning.

General education: The student learns mathematics in the classroom to improve their general education in order to participate in society as well as to act as a responsible citizen.

Marks: Learning mathematics is personally relevant to getting good grades that are linked to various goals, such as meeting the learner's own performance standards or feeling competent, evaluating their own qualifications/knowledge, or because further education depends on mathematics performance.

Positive image: Mathematics learning is perceived to be personally relevant in order to gain recognition from significant others (e.g., parents, peers, teachers, or people from the social environment).

Purism of mathematics: The individual has an emotional relation to mathematics itself. Its formal language and logical structure is valued by the learner as it helps students to understand mathematics easily. The applied approach of mathematics in life is perceived as annoying and uninteresting. The student is fascinated by the characteristics of mathematics, such as the following: its answers can be either right or wrong, it is easy to memorize as only formulae are necessary, and it is easy to understand as it is explainable inner-mathematically; i.e., it can be explained using its own terms and frames of reference (and not by taking an applied approach to life).

*Reference to reality*¹⁰: Mathematics learning is personally meaningful as it provides and embraces knowledge that can be found in the student's reality. Its connection to everyday life shows the relevance of studying mathematics.

Self-perfection: Mathematics activities are relevant as they help to improve the learner's personal skills and performance-related factors that bring them to perfection. Lifelong learning is appraised as important. So, the learner strives to learn mathematics as they associate thinking logically and thorough understanding with fun and they enjoy mastering cognitive skills. Mathematics learning helps the learner to improve themselves (including their performance) and their self-confidence.

Support by the teacher: Mathematics learning is perceived to be personally relevant when the learner experiences support from their teacher. The support can be realized through interpersonal activities where the learner experiences the teacher's fair and empathetic personal characteristics or through classroom teaching that is affected by a friendly atmosphere, appreciation, and respect.

Vocational precondition: Learning mathematics is personally relevant for the student when mathematical skills are required or perceived as important prerequisites for their desired profession or further education.

Although various concepts were considered as influencing factors during the coding process (see Vollstedt, 2011b), the three basic psychological needs, beliefs, and interest became relevant concepts (in the sense of Grounded Theory) for the construction of the learner's personal meaning (Vollstedt, 2011b; also in the study by Vorhölter, 2009). Whereas in both empirical studies, those of Vorhölter (2009) and Vollstedt (2011b), the concept of basic

¹⁰ This personal meaning did not emerge in Vollstedt's (2011b) qualitative research. As explained below, it was included on the basis of subsequent studies.

psychological needs proved to be relevant, within Vollstedt's study it was possible to reconstruct three personal meanings that reflected learners' personal meanings in relation to EXPERIENCE(ING) AUTONOMY, COMPETENCE, and SOCIAL RELATEDNESS¹¹ when learning mathematics. Moreover, based on theoretical considerations, Vollstedt (2011b) suggested structuring the 17 personal meanings respectively in terms of the dimensions intensity of relatedness to mathematics and intensity of relatedness to individual (Vollstedt & Duchhardt, 2019, p. 147). This theoretical procedure led to distinguishing the personal meanings into seven superordinate types of personal meaning: The type *fulfilment of societal demands* encompassed duty, examination, positive image, and vocational precondition. Interaction with mathematical contents consists of active practice of mathematics. Efficient and supportive lesson design encompassed classroom management and support by the teacher. The type cognitive selfdevelopment embraced cognitive challenge, EXPERIENCE OF AUTONOMY, purism of mathematics, and self-perfection. Relevance of application consisted of application in life, and well-being due to own achievement compromised EXPERIENCE OF COMPETENCE and marks. The type emotionally effected evolvement contained balance and even-temperedness, emotionalaffective relationship to the teacher, and EXPERIENCE OF RELATEDNESS (see Vollstedt, 2011b for more).

In subsequent studies, a research team at the University of Bremen investigated the assessment and the structure of the various personal meanings by means of a survey questionnaire (Büssing, 2016; Schröder, 2016; cf. Vollstedt & Duchhardt, 2019; Wieferich, 2016). Based on these studies, a reliable questionnaire was developed by following a complex validation procedure (see Vollstedt & Duchhardt, 2019 for more). The results revealed further knowledge about the conceptualization and kinds of personal meaning: Büssing's (2016) study found out that an additional personal meaning needed to be considered in order to depict relevance of application (one of the seven types of personal meaning), so "reference to reality" was included as an additional kind of personal meaning. Moreover, within the personal meaning of efficiency, one of its two facets (learners' efficient ways of working) could not be assessed; consequently, this personal meaning was termed "classroom management" to address the remaining scale, namely efficient classroom management (Vollstedt, 2011b).

However, further results showed that a scale revision was necessary as the scale of classroom management could not fit the data due to poorly fitting items (Vollstedt & Duchhardt, 2019). In addition, the model fit values indicated that the two dimensions suggested by Vollstedt (2011b) to structure her typology, intensity of relatedness to mathematics and intensity of relatedness to individual, were not exact counterparts to the 17 personal meanings (reference to reality was not included). Therefore, Vollstedt and Duchhardt (2019, p. 161) recommended that future research investigate appropriate dimensions or meta-factors by reviewing the relations to relevant concepts, for example motivation, or interest, which theoretically affect the learner's construction of personal meaning.

¹¹ In what follows, the three (specific) personal meanings, EXPERIENCE OF AUTONOMY, COMPETENCE, and SOCIAL RELATEDNESS appear in small capitals in order to differentiate them from the three basic psychological needs.

2.2.3 Critical Reflection on the Concept of Personal Meaning

Referring to Radford's system of principles,¹² the theoretical conceptualization behind this motivational construct addresses the personal relevance learners associate with studying mathematics, which is affected by learners' personal preconditions (personal background and personal characteristics) and their mathematics-learning situation (P_1). The reconstructed personal meaning(s) (which include the later-added reference to reality) cannot be assigned; on the contrary it is constructed by synergizing individual and social experiences within learning situations (P_2). Vorhölter (2009) and Vollstedt (2011b) argued that one learner constructs various personal meaning is quite new, it has proven to be valid within several investigations in qualitative (Vollstedt, 2011b; Vorhölter, 2009) and quantitative paradigms (Suriakumaran et al., 2019; Vollstedt & Duchhardt, 2019). The theoretical conceptualization of personal meaning can be used to study the effects of mathematics-subject-related features (e.g., modelling tasks, Vorhölter, 2009) and the impact of learners' cultural context on the construction of personal meaning (Vollstedt, 2011b).

Even though Vollstedt's personal meanings have been empirically found¹³ in data from Germany (Suriakumaran et al., 2019; Vollstedt, 2011b; Vollstedt & Duchhardt, 2019), Hong Kong (Vollstedt, 2011b), and Finland (Suriakumaran et al., 2019), they cannot (yet) claim a universal existence, as the social cultural context has a huge impact on what relevance learners see in mathematics learning. That is so because the theoretical concept behind personal meaning considers not only the learners' social cultural environment that offers the learning situations, but also the learner as a subject who has a background and long-term perspective shaped by cultural values and goals. Accordingly, within the construction of personal meaning, as explained within the social context in the classroom. For this reason, these personal meanings might not exist in all cultural contexts, particularly when cultures follow distinct values, motives, and so on, and likewise further personal meanings may exist within these different cultural contexts that are not detected yet.

In relation to this current research, the concept of personal meaning has several supportive functions. This motivational construct encompasses a well-elaborated theoretical conceptualization and provides various kinds of personal meaning that learners construct when studying mathematics. Moreover, Vollstedt (2011b) and Vorhölter (2009) stated that learners construct various patterns of personal meaning in relation to learning situations. An insight into those patterns, or (here termed as) learners' *individual relevance systems*, might help us to understand what individual relevance systems students construct as their own mental compasses, and which are preferred most or least.

Nevertheless, certain limitations and questions remain that need to be clarified. If patterns of personal meaning exist, it nonetheless cannot be assumed that this insight into learners'

¹² In general, Radford's triplets (basic principles, methodology, and research questions) are intended to explain the conceptualization of a theory. It should be clarified here that those triplets are used in this context to describe the conceptualization of the motivational construct of personal meaning (see also Suriakumaran et al., 2020).

¹³ A master's student from the University of Bremen investigated Thai secondary students' personal meanings. However, the measurement accuracy was limited as the scales' psychometric properties were not analysed in detail within her research (Mayer, 2017).

individual relevance systems automatically says anything about the development of those systems. Are there certain conditions within the social context or the individual that affect this construction? Furthermore, which motivational propensity (autonomous or controlled) is associated with each constructed individual relevance system in class? If empirical evidence exists for those individual relevance systems, these patterns of personal meaning can be used to examine the theoretical interplay between individual relevance systems and quality of motivation.

Furthermore, Vollstedt (2011b) stated that in both cultural contexts (Germany and Hong Kong), three personal meanings occurred that indicated learners' personal relevance to EXPERIENCING AUTONOMY, COMPETENCE, and SOCIAL RELATEDNESS in mathematics-learning situations. It is notable that not all the affective psychological concepts that are the ingredients of personal meaning (such as interest or mathematical beliefs) were personally relevant for the learner (Vollstedt, 2011b). Although Vollstedt claimed that these three personal meanings turned out to be particularly relevant for students studying mathematics, it is not clear yet how these three personal meanings are either related to the basic psychological needs or differ from the concept of needs. Basic psychological needs are unique, and are based on nine key criteria, as stressed by Vansteenkiste et al. (2020): "psychological, essential, inherent, distinct, universal, pervasive, content-specific, directional, and explanatory" (p. 4). An overall frustration of these three basic psychological needs leads to ill-being. At this stage, it is important to state that basic psychological needs are not the same as goals, motives (Krapp; 2005; Ryan & Deci, 2017) or, as in this case, personal meanings. Since the basic psychological needs are relevant nutrients for learners' construction of personal meaning, it should be clarified how the concept of basic psychological needs is connected with the constructed personal meanings, particularly with EXPERIENCE(ING) AUTONOMY, COMPETENCE, and SOCIAL RELATEDNESS.

A further clarification refers to the two dimensions of personal meaning proposed by Vollstedt (2011b). The dimensions "intensity of relatedness to mathematics" and "intensity of relatedness to individual" were not exact counterparts to the 17 personal meanings. Since this present investigation refers to motivation in mathematics education, the kinds of personal meaning could be structured along the meta-factors "orientation towards the relatedness of mathematics" and "orientation towards the regulatory styles". This approach might be a possible strategy of relating the kinds of personal meaning to motivation in order to increase our understanding of the relationship between personal relevance and motivation.

The following section 2.3 discusses these specific issues by relating the construct of personal meaning to the theoretical lenses of SDT and social constructivism.

2.3 Contrasting Juxtaposition of SDT and Social Constructivism

"There is nothing so practical as a good theory" (Lewin, 1951, p. 169). Lewin, a seminal scholar whose work led to substantial progress in the field of motivation, stressed that a good theory encompasses the potential to accurately and meaningfully verify a prediction in practice. Due to its exactness and practicability on broad contexts, theory focuses on specific questions (and any relevant factors associated with them) about a subject (Schoenfeld, 2002). Accordingly,

one theory cannot provide reliable and predictive responses for all kinds of questions about the researched object of discourse.

On the basis of the theoretical considerations presented above, SDT (2.1) mainly provides a broad framework for understanding the factors behind why learners display certain motivational propensities (intrinsic and types of extrinsic motivation) and why these propensities bring benefits but also an emotional cost while learning. In this relation, the concept of basic psychological needs is outlined as an essential key factor within learners' social context (i.e., of the classroom) that has a robust effect on students' individual well-being and quality of behaviour within educational settings. SDT emphasizes the importance of learners' personal relevance (which is close to the individual's sense of self) within the social context to promote students' internalization, satisfy their basic psychological needs, and foster their learning processes. In a nutshell, SDT is a powerful lens with which to discuss the basic needs mechanisms within the individual that belong to the learner's individual biological nature. Although SDT considers the important role of social context in the classroom, it does not provide a well-founded conceptual base to investigate how the social context affects the construction of personal relevance within the classroom context, and it is not clear what specific personal relevance learners construct within their social context when learning (in this case mathematics).

This can probably be clarified through the lens of social constructivism (2.2.1), which explains students' learning as well as knowledge production as a conversation between the individual and social mechanisms in the mathematics classroom. Thus, the cycle proposed by Ernest (2010) can be applied to understand the development of learners' personal relevance (or individual relevance system) through their experiences in the mathematics classroom. That is, in contrast to SDT, this perspective primarily uses a cyclical model of four processes to describe how learners internalize externally offered sources by reflecting their own experiences. Hence, this lens also concerns the fundamental intercommunication between the social context and the individual learner. But in contrast to SDT, it remains unresolved which key factors guide learners' individual "meaning of sign" development processes in private; that is, the processes of appropriation and transformation (cf. Figure 2).

Accordingly, an isolated view of each of the lenses cannot help us to fulfil the main purpose of this dissertation; that is, to clarify the interplay between learners' individual relevance systems and motivation, as both theories have limitations. Therefore, the use of SDT or social constructivism alone would fail to address the subject of this thesis. However, both lenses can be used to view motivation from different angles by dissecting focus-relevant questions with respect to the role of cultural imprinting. That is, building a bridge between both theoretical pathways, the social and individual context, seems to be a promising step to getting an enriched and more meaningful insight into the subject of mathematics-related motivation. Nevertheless, both views are rooted within different research domains and should be treated accordingly. In this connection, the motivational construct of personal meaning that was introduced earlier (2.2.2) seems to have a specific conceptual nature that is "plastic and solid enough" (Star & Griesemer, 1989, p. 393) to function as a bridge between both theoretical worlds (Akkerman & Bakker, 2011) as well as to be understood from both perspectives.

The connection between theories can take place at the level of principles, methodologies, or of questions, or by combining these elements (Radford, 2008b). In this study, I consider the

connections at the level of principles. In the following, I provide an argument explaining why personal meaning has the potential to be considered as a so-called "boundary object" (Akkerman & Bakker, 2011) between SDT and social constructivism. In this connection, I provide my theoretical assumptions about how personal meaning can be viewed through both theories' viewpoints. Moreover, I discuss how the nature of personal meaning applies to both *ontological* and *epistemological worlds* (Radford, 2008b) when it is considered through the two perspectives of SDT and social constructivism.

2.3.1 SDT's Perspective on Personal Meaning – Theoretical Assumptions

To investigate how the concept of personal meaning can be understood from the perspective of SDT, it is necessary to clarify the conceptual links between SDT's central elements as they relate to learners' quality of motivation (basic psychological needs and regulatory styles) and (kinds of) personal meaning. A theoretical clarification, at the level of theoretical conceptualization, could explain how the nature of personal meaning appears through the lens of SDT.

Basic Psychological Needs and Personal Meaning. As indicated above (2.2.2) the basic needs concept of SDT was found to be a relevant influencing factor among the personal characteristics that made up learners' construction of personal meaning (Vollstedt, 2011b; Vorhölter, 2009). This knowledge was then reinforced further in both of the relevant empirical studies, namely Vorhölter (2009) and Vollstedt (2011b). Whereas both studies confirmed the involvement of the basic psychological needs within the theoretical conceptualization of personal meaning, Vollstedt (2011b) reconstructed three personal meanings that reflect learners' experience of personal relevance when studying mathematics: EXPERIENCE AUTONOMY, COMPETENCE, and SOCIAL RELATEDNESS. As pointed out earlier, the concept of basic psychological needs and the conceptualization of personal meaning are not equal and thus should not be treated as such. To understand their conceptual relationships, theoretical assumptions are made.

According to the conceptualization of personal meaning, for the construction of a personal meaning-no matter what kind-the basic psychological needs are always involved as relevant personal characteristics. It seems that depending on the constructed personal meaning(s), one can follow that the basic psychological needs are experienced/fulfilled to a certain individual level. For instance, the basic psychological needs seem to be experienced differently when a learner constructs purism of mathematics or duty as the personal relevance of studying mathematics that applies to them. Theoretically, I would expect that (at least) the need for autonomy is satisfied when the learner perceives purism of mathematics as their own personal relevance, while a personal relevance of duty indicates frustration of the need for autonomy. Accordingly, I assume that the individual perception of basic psychological needs essentially contributes to learners' construction of personal meaning in the classroom. By referring to the different notions behind the kinds of personal meaning (cf. 2.2.2), among all the various personal meanings, I presume that the three basic psychological needs (autonomy, competence, and relatedness) are closer to the three which we shall call specific personal meanings (to EXPERIENCE AUTONOMY, COMPETENCE, and SOCIAL RELATEDNESS). That is, I assume that through the construction of the three specific personal meanings, the three basic

psychological needs are also addressed, as the specific meanings address within their core the notions of autonomy, competence, and relatedness. Within the specific personal meanings, a fundamental connection can be assumed between the personal meaning EXPERIENCE OF AUTONOMY and the need for autonomy, the EXPERIENCE OF COMPETENCE and the need for competence, and the EXPERIENCE FOR SOCIAL RELATEDNESS and the need for relatedness.

If so, due to the construction of a specific personal meaning in the mathematical context, the corresponding basic psychological need in mathematics lessons can be satisfied. For instance, if the learner constructs the personal meaning EXPERIENCE OF AUTONOMY, they might satisfy the need for autonomy, and so on. And if the three specific meanings are not constructed, the respective basic psychological needs cannot be satisfied in the classroom. That is, if the learner does not construct the personal meaning EXPERIENCE OF COMPETENCE, their need for competence in the mathematics classroom might not be satisfied, for example.

In turn, when their basic psychological needs are satisfied, the learner constructs the specific personal meanings in the classroom. That is, when a student experiences fulfilment of the need for autonomy during mathematics-learning situations, they construct EXPERIENCE OF AUTONOMY, which strengthens their well-being in the classroom. Likewise, if the basic psychological needs are not satisfied, then learners' well-being decreases, as something is missing. To that effect, I assume that the specific personal meanings in the mathematical context correspond to the basic psychological needs. There seems to exist an *inner feedback mechanism* (see Figure 4) between the basic psychological needs and the specific personal meanings in the classroom.

If so, I expect that these three specific personal meanings determine what specific pattern of personal meaning the learner constructs in the classroom (see section 2.3.2). Furthermore, based on this inner feedback mechanism, I would expect to find the same associations between the basic psychological needs and the learners' well-being and internalization continuum as exist between the three specific personal meanings and the learners' well-being and internalization continuum. In a nutshell, if an inner feedback mechanism between the three specific personal meanings and the basic psychological needs exists, then

- 1. the specific personal meanings determine the patterns of personal meaning (individual relevance system).
- 2. the specific personal meanings exert an influence on the learners' self-determination in the classroom (autonomous or controlled motivation).
- 3. the same regulatory mechanisms should also be found between the specific personal meanings and the regulatory styles as are found between the basic psychological needs and the regulatory styles (cf. 2.1.2).

Figure 4

Hypothesized Link Between Basic Psychological Needs and the Three Specific Personal Meanings.



As indicated earlier (cf. 2.1.2), support of autonomy occurs to a varying extent between cultures. Nevertheless, the association between basic psychological needs and well-being and an understanding of the essential role of autonomy for wellness is universal. For this reason, the predicted inner feedback mechanism should exist irrespective of social cultural contexts; that is, both in the data from Germany and the data from Finland.

Regulatory Styles and Personal Meaning. Based on the assumptions above, we shall explore further hypothetical connections between the regulatory styles and kinds of personal meaning. The taxonomic overview of the subtypes of motivation reflects the levels of autonomy leading towards learners' increasing sense of ownership (cf. Figure 1). In this connection, Vansteenkiste et al. (2018, p. 33) stressed that self-relevance is solely associated with autonomous extrinsic regulatory styles (identified, integrated), in which the regulation of society is less involved compared to controlled extrinsic motivation (introjected, external). I completely agree that in this case, autonomous-oriented activities are closer to the individual learner and more connected to their values and personal aspirations, which leads to multiple beneficial outcomes in their learning and promotes their internalization of external content due to the satisfaction of basic psychological needs. However, it might also be personally relevant for the learner to feel that they are a member of their social group, as this aspect, striving towards integration, belongs to their organismic dialectical nature (Deci & Ryan, 2000; Ryan & Deci, 2017). Hence, although controlled forms are alien to the individual's sense of self and thus not as close to the individual as well-internalized (autonomous) extrinsic actions, students occasionally also identify with those actions in order to experience integration within the mathematics classroom. For this reason, I claim, by referring to the theoretical concept of personal meaning, that learning activities in the mathematics classroom can still be personally relevant when those processes are closer to external sources and less a part of the student's

identity (Ryan & Deci, 2020). Those less-internalized extrinsic forms may emerge in connection with substitute personal meanings when learners cannot construct a personal relevance that closely relates to their sense of self and the subject of mathematics (Vorhölter, 2009; Vollstedt, 2011b). For this reason, to fully understand the conceptual relations between motivation and personal meaning and to have a full picture of individual learners' personal relevance in mathematics, it is necessary to consider all the personal meanings that can be constructed within learning situations; that is, regardless of their relation to more-or-less-internalized forms of motivation.

As indicated above (1), I am interested in the patterns of personal meaning (individual relevance systems) that learners construct. Before I empirically relate the individual relevance systems to their motivational propensity, it would be interesting in this regard to investigate the theoretical correlations between the regulatory styles and the 17 personal meanings; that is, to identify which kind of personal meaning refers to which regulatory style.

In the following, hypothetical connections are explored¹⁴on the basis of the existing theoretical knowledge of regulatory styles and personal meanings, and the links between these are subsequently examined empirically. Firstly, this procedure should help us to understand the theoretical connections between the regulatory styles and the various personal meanings. Secondly, if meaningful connections exist, they will be applied in order to understand the constitution of individual relevance systems and their respective types of motivational behaviour in the mathematics classroom. Both subsequent analyses will show to what extent there is evidence that highlights the conceptual relations between regulatory styles and personal meaning, and how personal meaning can be understood through the lens of SDT.

To explore the theoretical links between both concepts, I initially used two theoretically reasonable meta-factors as orientations along which we can assume connections consistent with theory (between the regulatory styles and the kinds of personal meaning). These theoretical correlations apply to an alternative procedure of structuring the personal meanings by examining the relations between personal meaning and motivation (cf. Vollstedt & Duchhardt, 2019). Hence, I used the connectivity to the subject mathematics and the orientation guided by the taxonomy of motivation to understand how the different personal meanings associate with the regulatory styles. As such, the 17 personal meanings can be systematically located in terms of the two meta-factors, *orientation towards the regulatory styles* and *orientation towards the regulatory styles*.

When locating the 17 personal meanings into the matrix, each personal meaning is placed at one specific level of relatedness to mathematics (ranging from very low to complete harmony). Thereby, each personal meaning is associated with several regulatory styles but has a strong relation with one specific regulatory style. As indicated earlier (cf. 2.1.1), theoretically, learners' motives can be multiply motivated (intrinsic and forms of extrinsic or several extrinsic regulated forms). In this connection, Ryan and Deci (2017) stated "the contents of the regulations and values that are culturally transmitted and internalized make an important

¹⁴ In general, scientific instructions (e.g., provided by Kelle & Kluge, 2010, pp. 91–107) help to group empirical cases into multidimensional types in order to investigate theoretical systems. In the present case, I first assume theoretical connections in order to test them subsequently in empirical research. For this reason, no specific instructions are followed, but the procedure for formulating the assumptions follows a similar pathway as the guidelines proposed by Kelle and Kluge (2010).

difference, with some contents more or less fitting with basic psychological needs and therefore more or less easily integrated" (p. 181). In this relation, referring to Ryan and Connell (1989), Vansteenkiste et al. (2018) stated that the correlates of the regulatory styles follow a *simplex pattern*; that is, with correlates "becoming increasingly less negative and more positive as one moves along the continuum from external to intrinsic motivation" (p. 34). Our existing knowledge about regulatory styles' correlational styles (multiply motivated and simplex pattern) were used to assume the connections. Additional knowledge that was used to locate the personal meanings came from the perspective of culture. This present investigation considers data from two different cultural contexts, namely Germany and Finland. Although these countries' educational systems are different (see section 2.4), their cultural norms and values and ways that these influences are transmitted to the educational system, particularly in mathematics curricula, are more autonomy-supportive than authoritarian. Therefore, based on the cultural contexts that are dealt with within this present study (students from Germany and Finland), I assumed the existence of the illustrated links between both personal meanings and regulatory styles (see Figure 5).

In so doing, the 17 personal meanings could be theoretically divided into the following groups: eight are content-specific dimensions¹⁵ of personal meaning; five of them are multifaceted (*application in life, emotional well-being in class, freedom of action, professional qualification,* and *self-development*); and three of them are self-sufficient (duty, positive image, and purism of mathematics). The irregular forms that surround the dimensions reflect the fact that the kinds of/dimensions of personal meaning do not relate only to one specific regulatory style. In the following, I briefly describe the contexts and the characterization of each dimension by referring to the properties of each meta-factor, namely its type of orientations towards the regulatory styles (external–internal locus of causality) and towards the relatedness of mathematics).

¹⁵ The kinds within the multifaceted dimensions of personal meaning are sorted in alphabetical order.

Figure 5

Hypothesized Links Between Regulatory Styles and Personal Meanings.



Application in life: This dimension emerged when I combined the personal meanings of general education and reference to reality. Here I assumed a relatedness of mathematics that is based on a balanced mixture of the social and the individual contexts, as the learner not only constructs these personal meanings in order to participate in the social realm but also values mathematics knowledge as a part of their identity. Hence, I assume that this dimension and its personal meanings, in which learners have a somewhat internal locus of causality, could be *strongly associated with identified regulation*.

Duty: Here the relatedness of mathematics learning is very low. In cases where this meaning is constructed, I expect the learner not to be interested in mathematics, thus the learner participates in class as they intend to avoid sanctions, which in turn are motivated by an intention to regulate society. Accordingly, the learner's actions have an external locus of causality and due to their lack of interest in mathematics and its low relevance for them, they feel controlled in the classroom. Here, I expect a *strong relation to external regulation*. Since this personal meaning focuses uniquely on the external obligations associated with mathematics and the social regulations that preponderate, I did not group this.

Emotional well-being in class: This dimension encompasses personal meanings that relate to emotional-affective factors within the classroom. Specifically, all these personal meanings are largely related to the role of the teacher. Here, I assume a *strong relation to introjected regulation*, as the focus gaining their teacher's approval or making a positive impression on others in the classroom may lead the learner to internally pressure themselves to study mathematics. Accordingly, the learner's actions may have a somewhat external locus of causality and the relatedness of mathematics is perceived as low, as an orientation towards social regulations preponderates.

Freedom of action: This dimension encompasses two out of the three¹⁶ specific personal meanings. Based on the hypothesized links between the three specific personal meanings and basic psychological needs, I expect EXPERIENCE OF AUTONOMY and COMPETENCE to be associated with activities that are in harmony with mathematics as an object. Accordingly, I assume that when both personal meanings merge together as freedom of action, the learner has an internal locus of causality and there might be a *strong experience of intrinsic motivation*.

Positive image: In this case, I expect the learner not to be interested in mathematics, but to value mathematics learning in order to avoid feelings of guilt and shame. Accordingly, the learner's actions have a somewhat external locus of causality, and they see a low relatedness in mathematics as well as feeling controlled internally. Here, I expect a *strong relation to introjected regulation*. Since this personal meaning uniquely focuses on the internal obligations associated with mathematics, I did not group this with other personal meanings that show a similar direction in mathematics learning.

Professional qualification: This dimension encompasses personal meanings that matter for the learner's vocational planning. Since the desired planning for their profession—or for their further education—is personally meaningful, learning mathematics has a high relevance; here, the relatedness of mathematics is high. The perceived locus of causality is somewhat internal, and I expect a *strong association with identified regulation*.

¹⁶ The three specific personal meanings are EXPERIENCE OF AUTONOMY, COMPETENCE, and SOCIAL RELATEDNESS.

Purism of mathematics: This personal meaning represents the hallmark of the subject mathematics. In cases where this meaning is constructed, I expect the learner to be interested in mathematics, thus in mathematics as an object itself. Accordingly, the learner's actions have an internal locus of causality and due to their interest (Krapp, 2002) in mathematics, these actions harmonize with their experiences in the classroom. Since this personal meaning is unique in that it focuses on mathematics as an object, I did not group this with other personal meanings that show a similar direction in mathematics learning. I expect a *strong association with intrinsic regulation*.

Self-development: This dimension encompasses personal meanings that matter for the learner's personal progress. Since the individual values mathematical skills as they guide them towards perfection and improvement, mathematics has a very high relevance in which an orientation towards the relatedness of mathematics preponderates. The perceived locus of causality is internal, and I expect a *strong association with intrinsic regulation*.

Investigating these assumptions—the relation between basic psychological needs and personal meaning as well as the associations between regulatory styles and personal meanings—will show whether personal meaning can be understood from the perspective of SDT. Since both concepts, personal meaning and motivation, emphasize the role of culture, it would be interesting to see when such theoretical correlations exist; whether, for example, they vary across different cultures and whether they follow similar associations (e.g., purism of mathematics relates to intrinsic motivation) when the patterns of personal meaning are related to their corresponding motivational tendency (in this example, autonomous motivation).

2.3.2 Social Constructivist Perspective on Personal Meaning – Theoretical Assumptions

The current investigation is interested in the core subject of motivation in the mathematics classroom. In this connection, as indicated above (cf. 2.2.1), the social constructivist approach helps us in that we can incorporate its view on social interactions in order to explore how students' experiences of autonomy and self-determination are brought about by interpersonal relations with different people in the mathematics classroom. On this specific topic, within his excellent article, Hickey (1997) precisely reviewed and discussed the notion of motivation in the sense of social constructivism. As an important impetus towards classroom motivation, he considered to view classroom motivational issues by means of a social constructivist approach in order to increase awareness of the contradictions in both conceptual understandings. A holistic perspective that surveys each viewpoint using the tools of the others is important as it is an approach that avoids bringing the social constructivist perspective and motivation theory into conflict with each other and that also assists our comprehension and optimization of learning in the classroom context (Hickey, 1997).

In its appreciation of the central role of the social context within its conceptual understanding, SDT seems to be theoretically commensurate with the social constructivist approach. However, it is necessary to look more closely at the way SDT and social constructivism communicate and to identify whether their roots conflict. To achieve this, I propose that personal meaning should also be viewed through the social constructivist perspective. According to the theoretical conceptualization of personal meaning, learners construct their individual relevance systems within the mathematics-learning situation (cf. Figure 3). As a result, they are integrated into the social world of the classroom and take part in interpersonal conversations. Such individual relevance systems, which I want to identify, have to appear in the classroom and the question is how such relevance systems can be created in the classroom at all; I wish to draw attention not only to their existence, but also to how they manifest themselves in a teaching context and to address the issue of how they are created in the first place. It will only be possible to describe these individual relevance systems theoretically after these points have been addressed. I used Ernest's model (2010) create a social regulatory mechanism as a counterpart that applied to everyday lessons through the lens of social constructivism. Effectively, in order to examine how interpersonal dialogue affects learners' constructions, I adapted Ernest's cycle and applied it to the ways in which learners produce patterns of personal meaning in the mathematics classroom; that is, the social regulatory mechanism of the individual relevance system (see Figure 6). The application of this adapted cycle may help us to increase our understanding of the development of learners' individual relevance systems in mathematics.

By respecting the core notions behind each of the processes-appropriation, transformation, publication, and the processes of conventionalization-I theoretically explain the production of the learner's individual relevance system through the following adjustments (see Figure 6): the adapted model schematically explains how the individual relevance system becomes *appropriate* to the learner through their experience of the personal relevance that mathematics learning can have.

Figure 6

Social Regulatory Mechanism of the Individual Relevance System (cycle adapted from Ernest, 2010).

Individual				Collective	
	Public	Individual uses the individual relevance system as mental compass to orientate meaningfully in classroom	Conventionalization \rightarrow	Production of the individual relevance system is promoted by conventionalization experiences	
Ownership		Publication ↑		Appropriation \downarrow	
	Private	Individual's construction of own individual relevance system (via three specific personal meanings)	Transformation ←	Individual receives own impression of what personal relevance mathematics learning can have	

Social Location

As a first instance, the operation of *appropriation* (creating one's own impression) sees the learner receive their experiences within their social context in the mathematics classroom, which provides them with their own impression of what mathematics learning is about. Then, the learner experiences this a number of times, each time completing a cycle in which they process the personal relevance of mathematics learning for them. This cycle consists of four processes: appropriation, transformation, publication, and conventionalization. The learner examines each experience in terms of its relevance to their own meaning construction and aspirations and *transfers* that perception into their own personal meanings for mathematics as a subject.

Within the individual development process, which takes place in private (and which consists of two of the above-mentioned operations, appropriation and transformation), I specifically assume that the three specific personal meanings as key factors guide the learner's construction on the basis of their experience (non-/construction). That is, here the effects of the hypothesized *inner feedback mechanism* (cf. Figure 4) between basic psychological needs and the three specific personal meanings impacts the learner's reflection process in terms of the pattern of personal meaning. For this reason, this process of reflection, which is based on appropriation and transformation, plays a major part in the construction of the individual's relevance system. This is because this reflection process helps prepare the learning orientation for mathematics for the *publication* stage (orientation), by defining the pattern of personal meaning that provides the learner with the mental compass they will use to enter/attend mathematics-classroom interactions. In connection with the others in classroom, through *processes of conventionalization* (reflection), individual relevance systems are individually differentiated as local patterns for the individuals, just as they fit the individual.

It is notable that the model initiated by Ernest (2010) followed a semiotic perspective (development of a concept through signs), whereas the adapted cycle here focuses on the importance of social interactions within classroom situations that affect the learner's development of their individual relevance system through social interactions in the classroom. This theoretical construction is by its nature hypothetical, and will be further examined within the context of this study.

2.3.3 Consideration of Personal Meaning as a Boundary Object

According to the theoretical assumptions presented above, there seem to exist conceptual links between SDT and personal meaning as well as between social constructivism and personal meaning.

On the one hand, SDT's ontological principles (cf. 2.1.3) provide a lens with which to focus on learners' individual experiences. That is, SDT is a powerful lens to discuss the basic psychological needs mechanisms within the individual that belong to learners' individual biological natures. As such, these basic need mechanisms generate motivation. When considering personal meaning from the perspective of SDT's ontological principles, personal meaning appears as an individual psychological phenomenon. That is so because personal meaning encompasses conceptual characteristics: the biological aspect that considers learners as individuals, and which captures, understands, and translates SDT's ontological principles. Based on this structural complexion of personal meaning, conceptual links between SDT's central elements (basic psychological needs and regulatory styles) and personal meaning could be assumed on the level of theory.

When SDT sheds light on personal meaning, within the individual level of personal meaning there seems to exist a *biological regulatory mechanism* that addresses learners' individual experiences of meaning construction. As such, this biological regulatory mechanism may become epistemologically observable in learners' motivation.

On the other hand, the ontological principles of social constructivism (cf. 2.2.1) highlight students' social contexts in the classroom. In general, this lens helps to explore individuals' social experiences, in which they subjectively assimilate the social regulations within the classroom. As such, the conversation—which belongs to learners' social nature—between the social realm and the individual is used to explain the production of learning and knowledge. However, when considering personal meaning by means of the ontological principles of social constructivism, personal meaning appears as a social phenomenon. That is so because personal meaning (also) includes characteristics—or the social aspect that considers learners as members of the social constructivism. Based on this nature of personal meaning, conceptual links between the cycle proposed by Ernest (model of sign appropriation and use) and personal meaning could be combined in the development of the learner's individual relevance system. The social nature of personal meaning helps to transform learners' interpersonal experiences which have a major impact on learners' production of individual relevance systems.

When social constructivism sheds light on personal meaning, within the social level of personal meaning there seems to exist a *social regulatory mechanism*. As such, this social regulatory mechanism may become epistemologically observable in learners' social experiences in the classroom; that is, be reflected within learners' individual relevance systems.

Accordingly, personal meaning seems to have the potential to fulfil "a bridging function" (Akkerman & Bakker, 2011, p. 133) as a boundary object at the boundary of social constructivism and SDT. According to Star and Griesemer (1989), *boundary objects* are analytic concepts that

both inhabit several intersecting social worlds ... and satisfy the informational requirements of each of them [These concepts are] both plastic enough to adapt to local needs and the constraints of the several parties employing them, yet robust enough to maintain a common identity across sites. They are weakly structured in common use, and become strongly structured in individual site use. These objects may be abstract or concrete. They have different meanings in different social worlds but their structure is common enough to more than one world to make them recognizable, a means of translation. (p. 393)

To conclude, if personal meaning can be viewed and understood from two different theoretical angles, SDT and social constructivism, then it seems to encompass two different ontological and epistemological natures. The hypothesized interconnection between both, biological and social regulatory mechanisms, is illustrated in Figure 7. The lens of SDT reflects its biological or individual ontological nature. Epistemologically, this aspect comes to the fore through the learner's motivational behaviour in the classroom. The perspective of social constructivism focuses on the social nature (on the level of ontology) that epistemologically exists in the learner's constructed individual relevance system. Accordingly, personal meaning seems to have a double ontological nature. In this case, it would be interesting to explore how these regulatory mechanisms, biological and social, interact. That is, these different ontological

natures of personal meaning can be used to study the interplay between learners' individual relevance systems and their motivation in mathematics education.

The exploration of the interrelation between learners' individual relevance systems (social regulatory mechanism) and their quality of motivation (biological regulatory mechanism) may clarify how learners' motivation emerges in the mathematics classroom. To elaborate these theoretical links, both ontological natures need to come into a dialogue by means of networking, so that we can study their interplay and how it relates to the genesis of motivation in mathematics.

Figure 7

Personal Meaning Characterized as Located at the Boundary of Social Constructivism and Self-Determination Theory.



2.4 Cross-Cultural Settings for Validation and Refinement

As indicated above, both theoretical viewpoints highlight the central role of teaching in sociocultural classroom contexts. Ryan and Deci (2020) stated that a country's educational reforms and instructed curricula have an impact on learners' motivation (and also on the teacher's behaviour and performance). In particular, an "excessive emphasis on grades, performance goals, and pressures from high-stakes tests" (p. 10) have a negative effect on learners (and also on teachers). In the face of the mandates that the teacher needs to follow in the classroom, certain constraints within the curricula of a subject hamper and challenge autonomy-supportive teaching (Ryan & Deci, 2020). Referring to Hickey's major leap ahead in terms of discussing motivational issues within social constructivism, how are motivational issues (e.g., support of autonomy) targeted within mathematics curricula approaches? Do the mathematics curricula formulate instructions that pay the same attention to learners' intellects

as they do to the individuals' affective states? This is important because "motivation cannot be distinguished from the larger realm of activity, and the individual's activity cannot be distinguished from the larger sociocultural context" (Hickey, 1997, p. 178). Hickey (1997) emphasized that motivation cannot be separated from cognitive activities; therefore, to study one without the other would serve no purpose.

To understand the hypothesized connections between learners' individual relevance systems and their motivation in the mathematics classroom, data from Germany and Finland are considered here. This decision was made for two reasons: Firstly, both theoretical views, SDT and social constructivism, emphasize the pervasive impacts of cultural norms, behaviour, and practices on students' individual relevance systems and motivation. This present study investigates German and Finnish ninth graders.¹⁷ Referring to scholastic performance in mathematics, Finland has earned international renown for its high performance in international comparative studies. In fact, Finnish and German students were both above average in the context of their performance in PISA 2012–2018. Finnish students outperformed their German counterparts by a narrow margin (OECD, 2014a, 2016, 2019). However, a few differences between the two groups were notable in relation to learners' affect in mathematics learning. For instance, in terms of the association between mathematics anxiety and performance, Finnish students felt less anxiety than German students (OECD, 2015, p. 2). Moreover, German students felt more anxious than Finnish students did when their schoolmates performed better than they did (OECD, 2015, p. 3). Further, between 2003 and 2012 the mean index of intrinsic motivation to study mathematics increased in Finland, whereas in Germany the level decreased (OECD, 2013, p. 74). Accordingly, it would be interesting to see which patterns of personal meaning German and Finnish learners construct on the basis of their social regulations and autonomy support in mathematics classroom.

Secondly, analysing results from two cultural contexts may help to contrast and refine the interconnections between learners' individual relevance system (social regulatory mechanism) and motivational behaviour (biological regulatory mechanism) as they relate to the students' development of motivation in the mathematics classroom. To this effect, I provide a brief insight into both countries' mathematics curricula (intended curriculum) in order to explore which aspects of their educational policies and practices support the learning of mathematics in the classroom.

2.4.1 German and Finnish Mathematics Curricula

A country's educational system and curriculum represent an image of its culture and its society's major elements. In terms of the mathematics curriculum, an exploration of its institutional mandates can take place within different arenas. According to Valverde et al. (2002) a country's overall mathematics curriculum can be viewed as a tripartite model, namely the *intended, implemented,* and the *attained curriculum*. As the name suggests, the intended curriculum formulates the official framework of the encompassing guidelines and key approaches like values, goals, competencies, and behaviour that the country's educational authorities usually communicate within official documents, like textbooks, exams, and syllabi

¹⁷ Again, the theoretical conceptualization of personal meaning is based on students' data from grades nine and ten.

for all learners in that society. It also sets out the way in which those general skills are addressed and transmitted in mathematics-classroom practice/interactions and how they should be discussed within the implemented curriculum. This arena particularly focuses on the interpersonal learning activities in the classroom; that is, those that take place between teacher and students and, of course, between students. The curriculum that specifies the attitudes, knowledge, or competencies that teachers should mainly follow in order to guide these social interactions in practice is known as the attained curriculum. The instructions at this level specify the knowledge and skills to be taught in mathematics-learning situations. A detailed comparison of both countries' educational systems at each level would go beyond the scope of this thesis. For this reason, I limit these (tentative) explorations to the German and Finnish intended curricula of the subject mathematics. Notably, the following analyses do not conform to the criteria-guided empirical methods that are generally applied to educational system comparisons. Nevertheless, since the aim here is to acquire an understanding of the philosophy behind the mandated instructions for mathematics learning in the classroom, a tentative analysis might also be helpful in obtaining an impression of the virtues of those instructions, which may affect the construction of ninth graders' patterns of personal meaning and their experiences of autonomy support in the mathematics classroom.

German Mathematics Curriculum. In response to the major educational initiative "Getting ahead through education" (German: *Aufstieg durch Bildung*), German educational policies emphasized an emerging need for skilled and well-educated workers. A list of aims and measures were decided upon by the federal government and all 16 federal states (KMK, 2019, p. 307):

- Education is to have top priority in Germany
- Every child should have the best possible starting conditions
- Everyone should be able to gain school-leaving and vocational qualifications
- Everyone should have the opportunity to get ahead through education
- More young people should take a degree course
- More people should be filled with enthusiasm for scientific and technical vocations
- More people should take advantage of the opportunity for continuing education

What these objectives had in common was their key aims: to make education a higher national priority and to reduce the number of adolescents leaving school without formal qualifications while raising the number of those obtaining them. Compared to the former version (KMK, 2016, p. 289) the updated (in 2019) key objectives particularly emphasized the effort required to address the causes of the problems the educational system faced; it stated that due to "the challenges posed by digitisation and the ongoing migration of refugees, great efforts must be made to develop the German education system in the years ahead" (KMK, 2019, p. 307).

In Germany, the school systems in each state are dependent on the federal state. In fact, the variation among the types of schools (mostly focused on different performance levels or vocational orientations) and the time spans over which students visit them represent a "mishmash". The variation within the 16 federal states aims to accommodate the heterogeneous diversity in the country by providing the best education possible for each student, which reflects

their academic level and requirements (KMK, 2019). To provide adequate care and support for each learner, all federal states provide an early selection process (typically after four or six years of school attendance, depending on the federal state in which students live) in which students are streamed according to their academic performances.

In relation to the German intended curriculum in mathematics, inspired by the PISA results in 2003, the standing conference of the ministers of education and cultural affairs (German: Kultusministerkonferenz-KMK) introduced so-called general educational standards (German: Allgemeine mathematische Kompetenzen; Blum, 2006; KMK, 2005, p. 7) for major subjects in school that should be reached after grade four (primary level) and after grade nine (intermediate degree). To facilitate the acquisition of these educational standards, content-related mathematical competencies¹⁸ (German: various Inhaltsbezogene Kompetenzen) are formulated that are in turn assigned to central mathematical ideas (German: Mathematische Leitideen; Biehler & Hartung, 2006; KMK, 2005, p. 8). Each central idea is a theme that is woven like DNA (i.e., spirally) into the mathematical curriculum and helps learners to achieve an understanding of basic mathematical terms, interdisciplinary thinking, and joined-up thinking (Biehler & Hartung, 2006). Figure 8 provides a brief overview of these major components and the relationships between them.

Figure 8

Overview of General Educational Standards in Mathematics to be Reached After Grade Nine Which are Acquired Through Content-Related Mathematical Competencies and Central Mathematical Ideas (adapted from KMK, 2005, translated by the author).



When looking at these mandated standards, what immediately catches the eye is that all of them (including the mathematical competencies and the central ideas) only discuss those

¹⁸ A detailed description of the content-related mathematical competencies can be found in KMK, 2005, pp. 9–11.

competencies that refer to the subject mathematics itself. The review of the core curriculum (KMK, 2005) pointed out that the promotion of learners' individual development is also an important aspect of mathematics learning: "The mission of school education goes beyond the acquisition of subject-specific skills. Together with other subjects, mathematics lessons also aim at personal development and value orientation" (KMK, 2005, p. 6, translated by the author). However, later amendments to the curriculum did not pick up on this fundamental topic at all. This issue was mentioned by Blum (2006) and Heymann (2005), whose criticism was that these standards should be redefined as *performance standards*, as the only detailed definition they provide concerns what students should have learned in mathematics by the end of grade nine (Blum, 2006, p. 15). They neither consider students' everyday learning situations in class (Blum, 2006, p. 15), nor value individuals' social, ethical, and personal development (Heymann, 2005, p. 40).

Finnish Mathematics Curriculum. The educational philosophy of Finland places emphasis on the guideline "Every pupil is unique and has the right to high-quality education and training" (Finnish National Agency for Education, 2017, p. 6). This ethos highlights in particular the importance of equality to students' growth and learning processes, and emphasizes the need for a high quality of learning if students are to be prepared for lifelong learning (Finnish National Agency for Education, 2017, p. 7). Following a single structure, Finland has a comprehensive system for the basic education covering grades one to nine. Accordingly, students of all academic abilities attend comprehensive school from primary to lower secondary level. They are taught in mixed-ability classrooms, and receive individual support according to their academic needs (Finnish National Agency for Education, 2017).

Guided by the education authorities' quality assurance plan, "Steering instead of controlling" (Finnish National Agency for Education, 2017, p. 13), the national core curriculum for basic education defines seven transversal competencies for all subjects. Besides these, objectives of instructions and content areas related to those objectives are central elements of the syllabus. The objectives of instructions¹⁹ can be differentiated into "significance, values, and attitudes", "working skills", and "conceptual objectives and objectives specific to the field of knowledge" (Finnish National Board of Education, 2016, p. 403). Figure 7 briefly illustrates the content areas in mathematics and the transversal competencies (Finnish National Board of Education, 2016).

¹⁹ A detailed description of the objectives of instructions can be found in Finnish National Board of Education (2016, pp. 402–408).

Figure 9

Key Content Areas Related to the Objectives of Mathematics in Grades 7–9 and Transversal Competencies in Basic Education (adapted from Finnish National Board of Education, 2016; see also Halinen, 2018).



Accordingly, the task of mathematics education encompasses not only developing students' "logical, precise, and creative mathematical thinking" (Finnish National Board of Education, 2016, p. 402) in mathematical modelling and problem solving; it also covers transversal competence areas that are cross-curricular and that largely concern students' personal development, such as self-actualization and developing confidence both as a learner and as an individual in society (Finnish National Board of Education, 2016). Therefore, in mathematics lessons students' learning process is assisted through an approach that supports their development of a healthy self-concept and self-esteem. They are encouraged to share responsibility with others and set their own goals, while the teacher provides them with strategies and methods for self-improvement. What is more, the teacher enables the students to develop their proximal skills through self-assessment and to support each other in terms of cooperative learning based on constructive feedback, an experience during which it is ensured that they are not ignored but heard. To conclude, the Finnish curriculum places emphasis on cooperation rather than competition. Teachers provide the "scaffolding" that supports their students and helps them to become active, self-confident lifelong learners.

2.4.2 Answerable Questions

Based on this tentative examination of the German and Finnish mathematics curricula, it appears that both countries aim to improve learners' mathematics performance but focus on them differently. In general, both countries, both of which are located in Western Europe, tend to teach mathematics in a way that is more supportive of autonomy than it is authoritarian. What is more, Finland is one of those countries that has adopted the German concept of *Bildung* as a key pillar in its approach to education (Autio, 2021). Despite this German influence, the

Finnish system is distinct from the German system in that it is equally concerned with learners' mathematical skills and with their individual development, whereas the German system mainly emphasizes learners' performance skills in mathematics. In a nutshell, the tentative examination of the mathematics curricula showed that in Finland, autonomy support seems to be more strongly represented than in Germany. The empirical evidence will show which similarities and differences result from these issues in terms of learners' individual relevance systems and motivation, and the interplay of both affective concepts. Since this study does not include any empirical classroom observations, its conclusions on the mathematics curricula/learning environment can only be based on theoretical knowledge and should be drawn with caution.

2.5 Résumé

This study's main purpose is to explore the interplay between learners' individual relevance systems and motivation in mathematics. To this effect, this chapter set out to clarify their interrelation on the level of theory. In so doing, it pointed out that we need to consider the learners' individual (through the lens of SDT) and social experiences (through the lens of social constructivism) in the classroom to reach a more comprehensive research result.

The theoretical conceptualization of personal meaning seems to encompass a biological and a social ontological nature, so that this construct can be viewed as lying at the boundary of SDT and social constructivism. If personal meaning is considered as a boundary object, from the perspective of SDT there seems to exist a biological regulatory mechanism explaining learners' motivational behaviour. Likewise, from the perspective of social constructivism there seems to be a social regulatory mechanism explaining the construction of the individual relevance system. Accordingly, personal meaning seems to link both regulatory mechanisms if we consider it as being at the boundary of both theoretical lenses. Studying personal meaning's dual ontological nature might help to understand how both regulatory mechanisms are interconnected in terms of learners' motivation in mathematics. Besides capturing both ontological natures, a robust and systematic procedure is necessary to network the theoretical lenses while respecting both theoretical roots. This examination will show how these lenses are connected and complement each other at the level of theoretical principles (P) so that both theoreties' articulation within the context of personal meaning leads to a common research result (R as fourth element, Radford, 2012).

2.6 Research Questions

As indicated earlier, within SDT's research landscape, for instance, Wang et al. (2017) have considered the patterns of regulatory styles (motivational profiles) in mathematics education in order to identify configurations of behavioural regulations. Focusing solely on these behavioural regulations is not enough to promote motivation in the mathematics classroom. In this connection, research that takes an SDT approach has pointed out that highlighting learners' personal relevance seems to be a promising pathway towards fostering their autonomously oriented motivation. In comparison to the other kinds of autonomous motivation (intrinsic or integrated regulation), the teacher's external efforts can address identified regulation by helping students to understand (e.g.,) the personal relevance of mathematics learning. For this

reason, it is a matter of particular interest which mathematics-related personal relevance learners associate with identified regulation. However, to the best of my knowledge, there is no clear theoretical concept behind learners' personal relevance or its conceptual relation to learners' motivational behaviour.

Studies within mathematics education have addressed this issue and proposed a relational framework of the motivational construct personal meaning based on Grounded Theory (Vollstedt, 2011b; Vorhölter, 2009). However, it has not been clarified how individual and social factors affect the development of learners' individual relevance systems and which kinds of individual relevance systems learners subjectively construct in order to engage in mathematics learning. Although the theoretical concept of personal meaning encompasses the basic psychological needs as relevant concepts, the conceptual relation between personal meaning and motivation in mathematics has not been clarified yet.

For this *problematique*, this dissertation now moves to the next level and focuses on the interconnection between the learners' individual relevance system and their behavioural regulations. To understand their relation in mathematics, two different theoretical lenses are adopted, SDT and social constructivism, in order to obtain a comprehensive research result that involves both learners' individual and social experiences when learning mathematics. Concerning the main objectives of this study, understanding the interplay between the learners' individual relevance systems (or profiles of personal meaning) and motivational behaviour from both theoretical perspectives may help to increase our theoretical knowledge and conceptual understanding of both affective constructs. Furthermore, the comparison of German and Finnish data may allow us to contrast the interconnections and to refine the interplay between personal relevance and motivation. Accordingly, this exploration may contribute to clarifying the genesis of motivation by focusing on the role of the personal relevance experienced by the learner in the mathematics classroom.

The following research questions (RQs) are considered within this investigation as a means of tackling the main objectives of this dissertation. Firstly, in order to investigate the interconnection between individual relevance systems and motivation, empirical evidence is required that allows a fair comparison of German and Finnish ninth graders' data. Based on this clarification, the data of German and Finnish learners can be compared with regard to (dimensions/kinds of) personal meaning and motivation. Secondly, the evidence provided by empirical patterns of personal relevance is necessary to theoretically describe the development of these patterns within the context of social interaction and their association with motivation. To this effect, RQ 1–RQ 3 are examined in the first part of the study.

(RQ 1) What are the psychometric properties of the affective constructs personal meaning and regulatory styles across Germany and Finland?

(RQ 2) What cross-cultural differences exist on the level of personal meanings and regulatory styles in Germany and Finland?

(RQ 3) How many distinct profiles of personal meaning can be detected in Germany and Finland?

Thirdly, to understand the interplay between individual relevance systems and motivation, the interconnections found in German and Finnish samples are separately analysed

on a theoretical level. Based on that, the results are compared and related to the respective mathematics curricula. Drawing on these comparisons, the genesis of motivation in mathematics is theoretically explained. Therefore, RQ 4–RQ 6 are studied in the second part of this investigation.

(RQ 4) How do the individual relevance systems and motivation interact in Germany and Finland, when viewed from a perspective coordinating SDT and social constructivism?

(RQ 5) How does the interaction found in German and Finnish data differ in view of learners' mathematics curricula?

(RQ 6) How does the conceptual framework built by coordinating SDT and social constructivism by means of the boundary object of personal meaning explain the development of motivation in mathematics classroom?

Next, I will present the research design used and the methodological considerations that should be taken into account before exploring the research questions of this dissertation (see chapters 3 and 4).

3. Empirical Work

In general, when scientifically investigating an empirical phenomenon, the questions always arise of whether the respective theoretical object is measurable and how it can be assessed in an appropriate way that respects its theoretical nature. To this effect, researchers' assumptions about the ontological and the epistemological natures of the phenomenon under investigation strongly determine the picture of the scientific world that they build up. This in turn affects the methodological approaches that the researcher selects (Buchholtz, 2021; Slevitch, 2011). Within the mathematicsrelated affect research, both qualitative and quantitative approaches are adequate (Goldin, 2014). These two research paradigms facilitate distinct insights into empirical phenomena. However, the suitable research direction should ultimately depend on two decisive factors, namely the theoretical roots (Radford, 2008a) of the research object and the problem statement that one aims to study. Therefore, it is necessary to identify the methodology that is useful for the research question, concepts, and theories to be used, and also the larger goal of relating theories or, as in this case, viewing an object from the perspective of two theory cultures at the same time. Accordingly, before deciding to apply a certain research approach, the researcher should clarify whether they have explicitly considered the theoretical conceptualization of the phenomenon and whether their research methods are appropriate for the intended targets of the research questions (Prediger et al., 2008; Slevitch, 2011).

Based on the theoretical framework, this study's substantive concept is clear: As elaborated earlier (cf. 2.3.3), the motivational construct of personal meaning (Vollstedt, 2011b; Vorhölter, 2009) can be characterized as located at the boundary of two theories; namely, self-determination theory (SDT; Deci & Ryan, 2000; Ryan & Deci, 2017) and social constructivism (Ernest, 2010; Hickey, 1997; Vygotsky, 1978). Accordingly, personal meaning seems to present as having multimodal ontological natures (Radford, 2014, p. 359); that is, both as an individual psychological phenomenon (viewed through the lens of SDT) and as a social one (viewed through the lens of social constructivism). Corresponding to these two ontological worlds, personal meaning can be viewed from the perspective of both theories and also comes epistemologically to the fore in two different ways. The connection between personal meaning and SDT-that is, the biological regulatory mechanism-becomes observable in learners' motivational behaviour. As the counterpart to these individual experiences, the connection between personal meaning and social constructivism, or the social regulatory mechanism, is observable in those social interactions on the part of learners that account for the construction of that learners' individual relevance systems. Although the biological and the social worlds are ontologically different concepts, they seem to be theoretically aligned by means of the theoretical construct of personal meaning. The clarification of the interconnections between social and biological regulatory mechanisms might help to increase our knowledge about learners' quality of motivation in the mathematics classroom.

For this reason, this study's entire methodology is tailored to studying the double ontological nature of personal meaning. Since learners' social regulation is also determined by culture, I investigated the connections between both theories not only in the German, but also in the Finnish mathematics-learning context. Based on the similarities and the differences between both countries' mathematics curricula (cf. 2.4), the results may provide important evidence about the interplay between learners' social and biological regulatory mechanisms. The question that arises now is that of how best to study the boundary object of personal meaning, which, it seems, can be viewed from the perspectives of two different ontological worlds.

3.1 Adopting a Sequential-Dependent Approach to Studying Personal Meaning's Multimodal Nature

This study integrates two different methodological approaches to studying the interconnection between the biological and social regulatory mechanisms within personal meaning in terms of motivation to learn mathematics. In the following, I briefly outline the necessity of and the justification for my choice to take this approach. Subsequently, I set out the exact nature of this study's methodological combination to anchor that approach within the methodological discourse.

First and foremost, it was necessary to gather empirical data. Personal meaning is produced by individuals. For this reason, the initial empirical work, in which I applied a quantitative research approach, drew on the different kinds of personal meaning as well as the regulatory styles from SDT, as the motivational elements are a central component of the hypothesized biological regulatory mechanism of learners. Subsequently, the empirical profiles that were generated in this way were used to clarify the other social regulatory mechanism in the classroom. To illustrate, if we were to look at the empirical profiles (personal meaning as an individual psychological phenomenon) from the perspective of social constructivism (personal meaning as a social phenomenon), then we would be able to explain how these individual relevance systems arise and unfold in the classroom. This is because, if we can finally use this social regulatory mechanism to explain how these empirical profiles have developed, then we would have good reason to accept this specific link, between biological and social regulatory mechanisms, as meaningful. For this reason, it was necessary to develop these individual relevance systems and the motivational elements empirically so that they were independent of the hypothesized adapted model - the social regulatory mechanism of the individual relevance system (cf. Figure 6). This was done in order that the gathered empirical evidence could then challenge/support the adapted model of social regulatory mechanism by providing the explanation/interpretation of how these profiles had developed, on an evidence-based basis. Within the quantitative research approach, there is no research activity which helps to link theories, i.e., of biological and social regulatory mechanisms. For this reason, in order to capture the social regulatory mechanism by linking both theoretical perspectives, I also used the networking of theories approach.

In a nutshell, the initial empirical work, which focused on a classical survey (self-report data), only captured the biological ontological nature of personal meaning by means of quantitative

statistical analyses. As a response to the weakness of the quantitative research approach, I subsequently considered networking theories approach and applied the coordination analysis method to capture the social ontological nature of personal meaning. This research activity, that of coordinating theories, was helpful not only to studying the second ontological nature of the empirical phenomenon, but also in investigating the articulation of the biological (motivational behaviour) and social regulatory mechanisms (construction of the individual relevance system) as regards learners' motivation in mathematics education.

This study's whole methodology thus falls into a research design that follows a sequential procedure (Schoonenboom & Johnson, 2017). That is, by

using the outcomes of the first research component, the researcher decides what to do in the second component. Depending on the outcomes of the first research component, the researcher will do something else in the second component. If this is so, the research activities involved are said to be sequential-dependent, and any component preceded by another component should appropriately build on the previous component. (p. 115)

According to this working definition, the following figure (Figure 10) provides a condensed overview of this study's overall methodical procedure, which frames the conducted empirical and the theory-driven analyses of the empirical data within a sequential-dependent research design. This present study will also discuss to what extent this procedure helped to counterbalance the methodological issues that emerged.

Figure 10





To reflect this study's dual methodological approaches for the data analyses, I present these in two different chapters. In this chapter, Chapter 3, I first describe the empirical work that solely focuses on quantitative research methods. I discuss in detail the strengths of this empirical approach, and then explain why a mono-methodological quantitative research approach (Kelle & Buchholtz, 2015) is insufficient to study personal meaning's double ontological nature. As a response to that weakness, I describe in Chapter 4 why building the networking strategy of coordinating on the basis of the existing empirical work has potential as a means of reaching the research aim (cf. 2.6) of the present study.

In the following section (see section 3.2), I discuss the philosophical background of the conducted statistical analyses, before presenting the quantitative methodical procedure (see section 3.3) I used to study personal meaning as an individual psychological phenomenon.

3.2 Methodological Considerations for the Quantitative Analyses

As indicated above, the existing empirical work aims to capture learners' individual experiences. Therefore, the empirical data used in this study should reflect learners' mathematics-classroom experiences. As an initial step to achieving this end, empirical evidence needs to be developed in two different cultural contexts. I first clarify which concepts I assessed, and subsequently, why the
quantitative research approach was chosen in particular as a means of studying the empirical evidence at the individual level.

Vorhölter (2009) and Vollstedt (2011b) theoretically stated that learners construct individually preferred patterns of personal meaning (individual relevance systems) within their mathematics-learning environment (cf. 2.2.2). For this empirical evidence, the assessment of the 17 personal meanings is essential to see whether those patterns of personal meaning are also empirically measurable. The regulatory styles of SDT (amotivation, extrinsic regulation, introjected regulation, identified regulation, and intrinsic regulation;²⁰ Ryan & Deci, 2017) can be used to investigate the subjective motivation associated with each individual relevance system (cf. 2.3.1). As indicated above (cf. 2.1.1), learners' motivational behaviour can be multiply motivated. To understand which behavioural regulation will most likely be manifested (Vansteenkiste et al., 2018, p. 33), besides examining the co-occurrence of the specificity regulatory styles, I used a composite index, or the SDI (Levesque et al., 2004, p. 73). Consequently, the 17 personal meanings and the five regulatory styles need to be assessed.

As we have seen, in a previous qualitative study Vollstedt (2011b) reconstructed 17 kinds of personal meaning students relate to learning mathematics in an educational context (cf. 2.2.2). The identification of the patterns of personal meaning and the investigation of their relationship to motivation within learners' natural learning environments by means of qualitative research methods is theoretically also possible. However, this procedure presents various hurdles, since in this context an increased effort generates only a low number of cases that are not selected as random samples (Lamnek & Krell, 2016, p. 16; Neuman, 2011). More often than not, self-report questionnaires are used for profiling individual response patterns (Geiser, 2011; Lazarsfeld & Henry, 1968; Rost, 2006). The application of a survey instrument helps us to straightforwardly gather plenty of heterogeneous empirical cases, which in turn allows us to consider numerous random samples (Kelle & Buchholtz, 2015, p. 333; Loewenthal, 2001; Neuman, 2011). This quantitative procedure of pattern (or profile) identification at the inter-individual level (e.g., latent class analysis, Geiser, 2011; Lazarsfeld & Henry, 1968; Rost, 2006) is one good way to identify learners' constructed individual relevance systems in connection with their quality of motivation. The self-report data thus collected can be used to study how personal meaning's biological regulatory mechanism becomes observable in learners' motivational behaviour. Based on the need to consider a large amount of random cases, in this present investigation the conceptualization and operationalization of personal meaning and the regulatory styles are carried out using a deductive linear pathway ("top-down" procedure; Neuman, 2011, pp. 69–70), by means of quantitative research methods. The decision to use a quantitative approach was also based on the need to apply highly structured and standardized statistical procedures, which are elaborated on below (see section 3.2.2).

After deciding to apply a self-report questionnaire, the criteria for this study's survey (or *self-report*; see Borkenau, 2006) were chosen that would be used to measure the 17 personal meanings and five regulatory styles in Germany and Finland, as I describe in the following section.

²⁰ As indicated earlier (cf. 2.1.1), integrated regulation is excluded as this style cannot be empirically distinguished from identified regulation (Vallerand et al., 1992).

EMPIRICAL WORK

3.2.1 Selection of the Classical Survey Instruments

In order to select the most suitable survey instrument(s), it is important to identify the properties of this study's affective constructs. Both affective constructs, personal meaning (Vollstedt, 2011b; Vorhölter, 2009) and regulatory styles (Deci & Ryan, 2000), are so-called latent psychological constructs (Pospeschill, 2010; p. 177), defined as "mental structures and processes that are not directly observable; rather, they are postulated to explain observable behaviour" (Macho, 2015, p. 1, translated by the author). On the basis of this particular nature of personal meaning and regulatory styles, these theoretical constructs can be measured by a range of specific observable (manifest) variables (Backhaus et al., 2015; Brown, 2006; Moosbrugger & Schermelleh-Engel, 2006; Rost, 2006). In order to measure these constructs, the immanent theoretical concepts need to be specified (Bühner, 2011; Pospeschill, 2010). In this connection, the latent theoretical construct is operationalized through several standardized items that systematically cover the whole range of the latent construct's conceptual knowledge (Rost, 2004; Steyer & Eid, 2001).

Generally, the assessment, evaluation, and subsequent selection of adequate survey instrument(s) are critical when exploring scholars' individual research objectives. More often than not, the empirical phenomenon appears too specific, so that established surveys cannot be applied (Loewenthal, 2001). As a solution, some scholars develop their own survey instrument. In many cases, it is possible to use already existing validated scales with good psychometric properties that can either be applied directly (if the research phenomenon is the same) or after modification as an adapted version of the instrument (Döring & Bortz, 2016, p. 410; Moosbrugger & Höfling, 2006). If scholars need to develop an instrument for a new theoretical concept (such as for the assessment of the construct personal meaning) or a very specific target-for instance if there are no established instruments for the phenomenon of interest-then a robust and systematic procedure is necessary for scale construction (Döring & Bortz, 2016; Pospeschill, 2010). If that is the case, self-developed survey questionnaires should be validated in terms of their psychometric properties before they can be applied in a study (Moosbrugger & Höfling, 2006; Pospeschill, 2010; Schermelleh-Engel et al., 2006). The methodical process used to validate the test standards for a new instrument is made up of a large number of intricate statistical procedures, which are used to assess and evaluate the relevant quality benchmarks. Particularly in cross-cultural studies, the survey instrument should also be available in the corresponding national language (Kelle & Buchholtz, 2015). A literal translation alone is not enough. In order to ensure that the survey items are understood in the same way in all countries, the survey instrument should be subjected to a thorough check by a native speaker based on local knowledge (e.g., population's age group and their social desirability; Chen, 2008) of the population (Kelle & Buchholtz, 2015) as well as through robust statistical estimation measures (e.g., test of measurement invariance). Due to the intricate and laborious nature of the instrument-validation process, academics have recommended using established survey instruments where possible (Döring & Bortz, 2016; Loewenthal, 2001).

In this study, the literature review has drawn the following conclusions on the application of survey instruments: Motivational regulatory styles can be measured, for instance, by a survey questionnaire (Ryan & Connell, 1989; Ryan & Deci, 2017; Thomas & Müller, 2011). Various

empirical studies have confirmed their successful application in different contexts (including mathematics education; Ryan & Deci, 2017). In this present study, a standardized survey instrument (Thomas & Müller, 2011) is used to assess each learners' quality of motivation in mathematics lessons across countries. Since this instrument has been developed for the German-speaking countries, translation work and further statistical examinations are necessary to integrate Finnish students' data into this present investigation (see section 3.3.3).

Regarding personal meaning, as already mentioned (cf. 2.2.2), a research team at the University of Bremen (Büssing, 2016; Schröder, 2016; Vollstedt & Duchhardt, 2019; Wieferich, 2016) developed a survey instrument to assess the 17 personal meanings students assign to mathematics learning. Concerning the validation process of this young instrument, the following insights from previous investigations are of particular importance for this present study: Almost all scales of the 17 personal meanings showed good psychometric properties, apart from the scale for "classroom management" (Vollstedt & Duchhardt, 2019). A revision was thus necessary in order to consider classroom management in this present investigation (see section 3.3.2). I used an adapted version of the PISA 2012 items (for disciplinary climate in the classroom; OECD, 2014b, p. 331) to estimate classroom management that exhibits efficient organization and structure of teaching, which is one facet of the personal meaning "efficiency" (Vollstedt, 2011b). Furthermore, Büssing's (2016) study found that an additional personal meaning needed to be considered in order to depict "relevance of application" (one of the seven types of personal meaning, Vollstedt, 2011b), so "reference to reality" was included within this survey as an additional kind of personal meaning. Moreover, a previous study (Suriakumaran et al., 2019) had considered the same database as this present dissertation, in which the scale for the personal meaning "balance and even-temperedness" did not fit the data. Consequently, this personal meaning was excluded from this research.

A further result concerned Vollstedt's typology (2011b). As indicated earlier, from a theoretical perspective, Vollstedt summarized the 17 reconstructed kinds of personal meaning from her study that were related to two dimensions, intensity of relatedness to mathematics and intensity of relatedness to individual, into seven types of personal meaning (cf. 2.2.2). Model-fit values indicated that these two meta-factors could not hold the data (Vollstedt & Duchhardt, 2019). To seek an adequate measurement model, Vollstedt and Duchhardt (2019, p. 161) recommended that future research investigate appropriate dimensions or meta-factors by reviewing the relations to relevant concepts, for example motivation, or interest, which theoretically affect learners' constructions of personal meaning.

Highlighting the theoretical links between personal meaning and SDT, I hypothesized an alternative segmentation (cf. 2.3.1) for this present study that structured the 17 personal meanings into eight dimensions with respect to two meta-factors: the orientation towards the regulatory styles and the orientation towards the relatedness of mathematics. In this investigation, two crucial aspects motivated my exploration of the eight dimensions in separate measurement models.

Firstly, the categorizations into separate dimensions were made by specifying the theoretical links between the 17 personal meanings and five regulatory styles (cf. overview of predicted di-

mensionality in Figure 5). Both meta-factors thus reflected the theoretical conceptualizations behind personal meaning (Vollstedt, 2011b; Vorhölter, 2009) and regulatory styles (Ryan & Deci, 2017). This suggestion focused in particular on the theoretical correlations between them. Hence, through this investigation, I was able not only to examine how far these eight dimensions or eight measurement models relate to the German and Finnish data, but also how they cross-culturally relate to the regulatory styles and appear in students' profiles of personal meaning (as profile specifications with respect to their particular quality of motivation). In this spirit, testing the empirical factor validity of the predicted dimensions would empirically support the hypothesized links between personal meaning and regulatory styles (SDT).

Secondly, the specification of too many factors in one overall measurement model is challenging, especially if a matured conceptual rationale is not provided (Brown, 2006). That is to say, considering a huge amount of variables (in this case, 17 personal meanings) from a relatively new theoretical construct in one complex measurement model is risky. Therefore, I aimed to differentiate the 17 personal meanings in a meaningful fashion (with respect to this research interest), where I expected high correlations between the factors (kinds of personal meaning within one dimension) to occur. These were the two main factors that motivated the splitting of the 17 kinds of personal meaning into eight dimensions. Regarding the suggestion of Vollstedt and Duchhardt (2019), this specification may provide a meaningful way not only of conducting the statistical analyses, but also of increasing the empirical knowledge about the theoretical concept of personal meaning in relation to quality of motivation.

Considering 17 personal meanings and five regulatory styles within one survey instrument is demanding, particularly in a cross-cultural setting. This complex composition (the combination of a new with an advanced instrument) thus required a highly structured and standardized procedure to be adopted that suits the focus of the current research project. The standardization applied in quantitative research provided a secure framework on which to carry out the appropriate statistical procedures. These standards help us to determine the psychometric properties of these affective constructs. In particular, they provide us with methods for dealing with the methodological challenges and biasing effects associated with studying established properties across cultures (Chen, 2008).

However, when quantitative statistical procedures were conducted, profiles of personal meaning and their corresponding quality of motivation are available in the form of numerical data. The examination of the identified empirical profiles provided an insight into German and Finnish students' individual experiences in the classroom. However, when it comes to interpreting the double ontological natures of personal meaning not only in the biological, but also in the social world, the quantitative mono-methodological research design showed technical limitations. To compensate for the limitations of quantitative methods, methodological strategies are required that support the consideration of personal meaning as a boundary object in both worlds. If we are to counterbalance this weakness by means of an additional strategy, firstly the strengths and limitations of the quantitative tradition should be clarified with regard to this study's research interest.

3.2.2 The Quantitative Research Approach to Studying Personal Meaning's Individual Nature

The quantitative research paradigm refers to basic assumptions that formulate a purely rational structure to plan the required methodical procedure in terms of measuring latent constructs in a cross-cultural context. Which specific limitations of quantitative research need to be compensated for to reach this study's research objectives?

Strengths of the Quantitative Research Approach as Regards the Goal of This Study. To start with the strengths, the hypothetico-deductive method (Lawson, 2015; Popper, 1968) associated with the philosophical perspective of the quantitative paradigm, that is, *critical rationalism* (Popper, 1989) formulated maxims of how to conclude theoretical knowledge by emphasizing the philosophical ethos of Popper's idea of falsification (Popper, 1989, 2000; Przyborski & Wohlrab-Sahr, 2014). Following a deductive direction, this type of research predicts hypotheses that are based on existing theoretical knowledge. Next, consequences are deduced and empirically measured in order to test the theory in practice and gather evidence (Lawson, 2015; Neuman, 2011; Popper, 1968). Therefore, precise steps for theory-driven problems are formulated in such a way that they are empirically testable and falsifiable (Döring & Bortz, 2016; Lamnek & Krell, 2016; Przyborski & Wohlrab-Sahr, 2014). For instance, theoretical antecedents (preferably connected to related qualitative work) depict assumptions about the reality. Then, progress in findings is gained in a stepwise fashion through structured processes of statistical operations that examine the empirical observations that have been obtained in order to address unsustainable theories (Döring & Bortz, 2016; Neuman, 2011; Popper, 2000). In the quantitative approach, the use of standardized (numerical) data, gathered from a large number of cases (in contrast to qualitative research; Döring & Bortz, 2016), is very typical (Kelle & Buchholtz, 2015; Neuman, 2011). The purpose of this is to develop object-related theories further (Przyborski & Wohlrab-Sahr, 2014) and to tentatively approve those theoretical findings that have repeatedly been tested but have not been falsified (Döring & Bortz, 2016, p. 37).

An important line of critical rationalism is the critical review of the research quality, as well as the disclosure of the methodical procedures and survey instruments of an investigation (Döring & Bortz, 2016). The criteria used to standardize these quantitative methods include objectivity (observer independency), reliability (consistency of results) and, most importantly, validity (validity of resulted empirical evidence; see section 3.2.3; Döring & Bortz, 2016; Kelle & Buchholtz, 2015). The applied object-related theories are established at the beginning and the end of the quantitative research frame, whereas the development of new theories is not the focus in this research paradigm (Neuman, 2011; Przyborski & Wohlrab-Sahr, 2014).

At the same time, this tradition contains several methodological issues that may arise in particular for this study's research context: To help ensure that this study adds to our store of knowledge of this subject, this research approach adheres to a falsification process by confronting empirical evidence with potential inaccuracy in theory, as well as taking into account methodical effects that may result from any issues and difficulties associated with the survey instruments (Döring & Bortz, 2016, p. 37). This was necessary because, more often than not, contradictory empirical results sometimes occur as a result of measurement or operationalization errors. An example of this is the *problem of correspondence* (Döring & Bortz, 2016), which refers to the problem of verifying whether the manifest variables are a precise measure of the correct theoretical concepts. This type of inconsistency, between the theoretical knowledge and the standardized data, should not only be viewed as a contradiction of the hypothesized theory to be tested, but also as a cause conditioned by the measurement theory to assess the empirical phenomenon, as both are always connected to one another. That is so because each data set has an individual methodological and methodical formation and its biography ultimately become a crucial element of the study's evidence and in turn the implications derived from that evidence (Radford, 2008a; Schoenfeld, 2002).

Additionally, to ensure a balanced scientific approach, it is recommended that when researchers study and interpret the empirical evidence produced, they should critically analyse alternative theoretical explanations as well as their own theoretical ideas (Döring & Bortz, 2016; Fiedler et al., 2012). Therefore, *meta-theoretical knowledge* that confronts the empirical evidence with related or competing theories is essential (here referring to the affective constructs of personal meaning and motivation; Döring & Bortz, 2016; Fiedler et al., 2012; Przyborski & Wohlrab-Sahr, 2014). However, most quantitative studies ignore the meta-theoretical consideration of the objectrelated theory (Przyborski & Wohlrab-Sahr, 2014).

Moreover, there are some further critical aspects that need to be taken into account in order to compensate for any possible misconstruction of evidence from different cultural contexts. Especially in an educational context, culture-specific knowledge about learners' particular living environments plays a central role (Chen, 2008). The relations between the student, teacher, school, subject curricula, and educational system are shaped by the respective cultural factors, such as social desirability (Chen, 2008). In cross-cultural studies, typically the sociocultural background of the scholars who conduct the study and the subjects who participate in the study vary as a result of missing *heuristics of common-sense knowledge* (Kelle & Lüdemann, 2019, p. 121). Furthermore, limited knowledge concerning the sample's living environment may lead to *problems of operationalization and measurement* (Kelle & Buchholtz, 2015). This issue could arise if scholars have not considered the respondents' linguistic usage or the topics that are relevant to them (in this case, German and Finnish ninth graders) so that the respondents do not understand the scholars' intended message in the survey items (Kelle & Buchholtz, 2015).

It is therefore clear that acting by rote (mechanically implementing statistical methods like following a formula) is not sufficient to meet the requirements of scientific research (Döring & Bortz, 2016). This discussion has illustrated that even a purely quantitative research study cannot function without reference to a qualitative interpretation work, that is, theoretical conclusion (Kleemann et al., 2013).

Given the above concerns, the current study aims to address all of the types of knowledge that emerged during the quantitative data evaluation. Furthermore, experts from Germany and Finland evaluated the survey instrument to ensure it took account of respondents' language and sociocultural issues. This helped to counter heuristics of common sense and problems of operationalization and measurement (see section 3.3.2).

The first part of the empirical data analyses studied the psychometric properties of the latent constructs of personal meaning and regulatory styles using multivariate analysis methods (confirmatory factor analysis). These statistical procedures (see section 3.3.3) aimed primarily to transform the theoretical knowledge into numerical data. The examination included studying how these properties are operationalized in different cultural contexts. As several cross-cultural studies have demonstrated, the use of survey instruments in one culture that originated in another may present significant challenges (Byrne, 2000; Bofah & Hannula, 2015), for example, poor construct validity (Chen, 2008). Therefore, robust methods were used that counter biasing effects in survey instrument (test of measurement invariance; Marsh & Köller, 2004). Further, for the identification of profiles (individual relevance systems of personal meaning), I used explorative research methods; that is, latent class analysis with distal outcomes. To compare similar profile outputs across countries, I conducted Mann–Whitney U tests. For a condensed overview of this investigation's empirical work, see Figure 11 in section 3.3.3.

In contrast to the regulatory styles, a new survey instrument measured the 17 personal meanings in two different cultural settings. Both instruments are at different stages of development. To compensate for problems of correspondence, a rigorous investigation of theoretical concepts' psychometric properties is necessary. If indicators of theory inconsistency exist, this false-positive theoretical and statistical empirical evidence (Döring & Bortz, 2016) may reflect the existence of hidden knowledge in terms of the study's theoretical concepts. A critical discussion of these paradoxical indicators may establish opportunities to enhance concepts' descriptive/explanatory power (Schoenfeld, 2002) or to improve the survey instrument of personal meaning for future research.

Weakness of the Quantitative Research Approach as Regards This Study's Goal. The methodological considerations elaborated from the conducted empirical work provided the philosophical background for the application of self-report questionnaires in the present investigation. With regard to this study's aims, these quantitative statistical analyses support the study of personal meaning as an individual psychological phenomenon when considering the concept through the lens of SDT (biological regulatory mechanism). Therefore, the self-reported data can be explored in terms of the predicted biological regulatory mechanism (connection between SDT and personal meaning) that becomes observable in learners' motivational behaviour in the classroom.

However, to examine whether personal meaning also exists in the social world, in which it becomes observable in learners' social interaction, it is necessary that the empirical evidence, on an individual level, can be interpreted through the lens of social constructivism; that is, in relation to the adapted model predicting the *social regulatory mechanism of* the *individual relevance system* (cf. Figure 6). Therefore, the empirical evidence (the detected profiles of personal meaning and their motivation) and the adapted model should come into a theory-driven dialogue.

Within the quantitative research paradigm, there is no clear description of how to interpret empirical evidence in view of networking theoretical perspectives. Without this interpretation work (Bernard & Ryan, 2010), the multimodal nature of personal meaning cannot be thoroughly discussed as intended in this research. As a response to this weakness of the quantitative approach, it is not only necessary but also justified to incorporate an additional research practice in order to

consider the social nature of personal meaning; namely, networking of theories (cf. Prediger & Bikner-Ahsbahs, 2014; Prediger et al., 2008). In other words, a theory-driven interpretation of quantitative data, in terms of systematically coordinating the two theoretical perspectives, follows the quantitative statistical analyses in order to qualify the quantitative data (Schoonenboom & Johnson, 2017). As indicated above, the methodology of the empirical work and the methodolog-ical considerations of the theory-driven analyses differ. For this reason, I describe the methodological considerations and the applied coordinating analyses method in more detail in the next chapter (see Chapter 4).

After presenting the strengths and limitations of this study's quantitative research approach, I highlight the methodological properties of this empirical work.

3.2.3 Quality Criterion and Quality Assurance

Science should by definition always refer to consistent, common standards that adhere to the relevant quality benchmarks (Döring & Bortz, 2016; Neuman, 2011; Rost, 2004). The level of scientific quality produced by these methods is complex to assess and should always be examined individually, taking into account the strengths and weaknesses of the study in question (Döring & Bortz, 2016). Therefore, it is important to communicate any necessary methodological limitations transparently to the scientific community (Döring & Bortz, 2016; Neuman, 2011). In the following section, I briefly refer to the quality criteria of quantitative research that relate specifically to this study's properties.

Meta-Theoretical Reflection. Before conducting an empirical study, the clarification of meta-theoretical preconditions plays a fundamental role. As theoretical precursors of the substantive framework, meta-theoretical matters encompass comprehensive ideas concerning the research phenomenon, the concept on which it is based, its prerequisites, and its significance (Przyborski & Wohlrab-Sahr, 2014). This philosophical foundation is usually omitted in quantitative studies or is referred to only obliquely (Przyborski & Wohlrab-Sahr, 2014). The reflection of a profound research logic behind empiricism is not only essential for the progress of knowledge in a discipline, but also for the coherent and consistent understanding of the results in scientific discourse (Döring & Bortz, 2016; Przyborski & Wohlrab-Sahr, 2014). Thereby, the argumentation of meta-theory as a concrete scientific justification also forms part of these research results in order that this knowledge can be applied or addressed in future research (Döring & Bortz, 2016; Przyborski & Wohlrab-Sahr, 2014).

In this study, the conceptualization of motivation is briefly elaborated while pointing out the differences between the quality and quantity aspects of motivation (cf. 1). As mentioned earlier, both kinds of motivation are relevant within the academic context. However, in contrast to achievement-related behaviour, the quality of motivation explains the differences in learners' types of behaviour and their quality of functioning (Ryan et al., 1996) in the mathematics classroom.

Furthermore, in previous research (Suriakumaran et al., 2020), so as to study learners' motivation systematically, affective constructs were distinguished from theoretically comparable concepts by highlighting their conceptualizations. In this spirit, certain theoretical aspects of quality of motivation (basic psychological needs and regulatory styles) were targeted as they play a crucial role in the educational context (Krapp, 2005; Nuttin, 1984; Ryan & Deci, 2020). On this basis, scholars identified which motivational construct would increase our knowledge of the quality aspect of motivation by considering the fundamental theoretical elements of motivation. A tentative comparison between values (Wigfield & Eccles, 2000) and personal meaning (Vollstedt, 2011b; Vorhölter, 2009) helped to select an appropriate affective construct, namely personal meaning, which represents learners' personal relevance in a mathematics-learning context (see Suriakumaran et al., 2020 for more). The concept specification (Döring & Bortz, 2016) of personal meaning explicitly considers critical components of quality of motivation (basic psychological needs).

Within the frame of this quantitative study, the theoretical elaborations referred to are those meta-theoretical knowledge aspects that could be ascertained (Przyborski & Wohlrab-Sahr, 2014). If relevant indicators emerged in the evidence, the relevant meta-theoretical knowledge was re-trieved and is discussed here in terms of this present study's research interest.

Construct Validity. Construct validity exists if the empirical results represent statements that reflect the estimated theoretical constructs; that is, if the validity of actual measures can be shown (Döring & Bortz, 2016; Schermelleh-Engel et al., 2006). Generally, latent constructs are measured by means of directly observable indicator variables (Backhaus et al., 2015). When examining latent constructs, the respective theoretical concept should be formulated precisely so that the construct can be empirically reflected (Rost, 2004). Thereby, the manifest variables should represent the various content-related facets of the construct (Moosbrugger & Schermelleh-Engel, 2006; Rost, 2004). The reflective measurement model built in this way (Backhaus et al., 2015) reflects by means of the indicator variables the direction of change that belongs to the associated construct (Rost, 2006). At this point, the roles of reliability and objectivity comes into play; these are briefly explained in what follows. Moreover, with respect to construct validity, the problem of correspondence according to critical rationalism should be taken into account. Therefore, critical reflection regarding the agreement between the intended theoretical concepts and the actual empirically measured data is essential (Döring & Bortz, 2016). To assess the construct validity, I considered certain scientific criteria (see confirmatory factor analysis in section 3.3.3) to guarantee methodical rigour (Backhaus et al., 2015; Döring & Bortz, 2016; Moosbrugger & Schermelleh-Engel, 2006).

Since this study examined data from two different countries, it was essential to check constructs' measurement invariance. In particular, for further investigation of data (e.g., comparison of affective constructs' latent means), it is important to estimate the actually existing measurement invariance in order to guarantee the required level of measurement invariance for such comparisons. At a minimum, (partial) scalar invariance is required to conduct latent mean comparisons over groups (see the test of measurement invariance in section 3.3.3). In addition, data cleansing plays a crucial role, as unstable data quality can reduce construct validity and impair the interpretation of results (Döring & Bortz, 2016). To ensure adequate handling of the missing values, I used the FIML (Full Information Maximum Likelihood) algorithm in Mplus (see missing data in section 3.3.3). **Objectivity**. The existence of test objectivity guarantees that another test leader, independent of the scholar, could achieve exactly the same results by following their procedures in research. The criterion of test objectivity can be judged as fulfilled if the implementation, evaluation, and subsequent interpretation of methodical procedures are explicitly stated in the study (Rost, 2006). According to this, the procedures followed in this work, from data collection through statistical analyses to interpretation of the results, are presented in detail in the following chapters.

Regarding the theory-driven analyses (described in Chapter 4), to guarantee that the interpretation work that had been conducted was *intersubjectively* comprehensible and verifiable (Bohnsack, 2014) in terms of the coordinating theories, these analyses were also discussed with German and Finnish experts who were familiar with their respective educational systems in order to avoid the author's making a one-sided interpretation. The intersubjective decisions that were made with a group of experts give an objectifying character to the theory-driven analyses, especially when tentatively discussing the results in relation to mathematics curricula and the educational systems behind them.

Reliability. In the sense of classical test theory (Brown, 2006), the existence of test objectivity is a prerequisite for measurement accuracy; namely, reliability (Döring & Bortz, 2016). In this study, consistency is estimated using composite scale reliabilities ρ (Bentler, 2009; Raykov, 2009; Sijtsma, 2009). The German survey instrument of personal meaning is not based on established valid scales. The use of these new scales in a different country (other than Germany) may lead to weaker scale reliability. The possibility of reduced reliability (in this case in Finland) can negatively impact on the validity of the results as well as decreasing the statistical significance and effect sizes (Raykov, 2012). These aspects should be taken into account, before proceeding to the comparative analyses.

3.2.4 Generalizability and Representativeness

Since the investigated constructs, personal meaning and regulatory styles, are affected by the cultural context, I collected the sample in two different educational cultures in order to enrich the validity of the evidence as well as to contrast and refine the research results. As already mentioned above, the educational system in Germany is diverse, while the Finnish system is largely uniform (cf. 2.4). In Finland, data from schools from one province were considered, while the German data came from four federal states (see section 3.3.1).

Regarding profiling classes, by using quantitative methods *average types* were constituted; that is, profiles based on learners' mean values of the 17 personal meanings. These types are strongly bound in time and space (Weber, 1972). In accordance with the quantitative tradition, this study thus aims for representative data (although it is very limited here due to the relatively small sample size; see section 3.3.1) and not the representation of the social meaning (in case of *ideal types*; Weber, 1972). However, the empirical profiles in Germany and Finland cannot easily claim validity for the whole country, as they are simply "this dissertation's results", which I have collected from a few locations in these countries. Therefore, I consistently refer to "Results or profiles in Germany/Finland" instead of "German/Finnish results or profiles".

Based on the moderate sample size used in this research, a sound generalizability of research results is limited (Kelle & Buchholtz, 2015). However, if the empirical evidence found in Germany and Finland leads to theory-consistent explanations when testing the applied background theories that have been applied, this should increase our confidence in the validity of the evidence collected, at least on a preliminary basis.

3.3 Empirical Methods

This section describes the particular quantitative research methods that I used in the present investigation. At the end of this chapter, I reflect on the planned empirical work with respect to the remaining challenges.

3.3.1 Data Collection

Data was collected in Germany and Finland. Since the theoretical conceptualization of personal meaning (Vollstedt, 2011b; Vorhölter, 2009) was retrieved from ninth and tenth graders' interview data, I considered learners from a similar age cohort, namely ninth graders. The 276 German participants (Q: 46%) from 17 classes were from four different federal states (Bavaria, Baden-Württemberg, Bremen, and North Rhine-Westphalia) and attended different schools according to their academic performance (three *Gymnasien* (for high achievers), one *Realschule* (for middle achievers), three *Oberschulen* (comprehensive schools with streaming of classes according to academic performance), and two *Hauptschulen* (for low achievers). In this study, the focus is on the personal meanings used by all students when dealing with mathematics in an educational context. Therefore, to guarantee a heterogeneous group of participants, there was no separation between high and low achievers. The data in Finland is from 256 ninth graders (Q: 48%) from four comprehensive schools and 13 classes from the region of Uusimaa.

The permission for the survey was granted according to the respective German data protection act and by the German *Landesdatenschutzgesetze* (Verbund FDB, 2020), whereby I obtained separate authorizations for all four of the German federal states. In Finland, I followed the local regulations, according to which an ethics review was not required to conduct the survey. As a result, permission was requested in an information letter for students and parents, while a detailed description of the research plan was submitted to the teachers. For one school (in Vantaa), the author had to first ask the director of basic education for permission; once this was obtained, the director delegated responsibility for overseeing the study to the school's head teacher. In both countries, the author's colleagues and private contacts helped to find schools to participate in the study. The whole survey (assessment of both instruments: Personal meaning and regulatory styles) of the Germany and Finnish ninth graders by means of a 45-min (one lesson across countries) paper-and-pencil test took place from October 2016 to March 2018 during the school year (not at the beginning or at the end of the school year). In Germany, the author carried out the surveys in all classes on her own, whereas in Finland Dr Eeva Haataja, a Finnish-speaking PhD student from the Department of Education at the University of Helsinki, supported the data-collection processes in some schools.

3.3.2 Survey Instrument

As indicated earlier (cf. 2.2.2), a research team at the University of Bremen (Büssing, 2016; Schröder, 2016; Vollstedt & Duchhardt, 2019; Wieferich, 2016) developed the German instrument to assess students' different personal meanings. As we have seen, besides the personal meanings reconstructed by Vollstedt (2011), according to Büssing's (2016) suggestion, "reference to reality" was also included within this survey as an additional kind of personal meaning. Within a previous study (Suriakumaran et al., 2019) that considered the same data base as this present study, "balance and even-temperedness" did not fit the data; for this reason, this personal meaning was excluded from this research. Furthermore, as a consequence of the validation process (Vollstedt & Duchhardt, 2019), in this study a few items are adapted from established scales (OECD, 2014b; see Table 1) to measure "classroom management". I used an adapted version of PISA 2012 items (*for disciplinary climate in classroom*, OECD, 2014b, p. 331) to estimate classroom management representing efficient organization and structure of teaching; that is, one facet of the personal meaning "efficiency" (Vollstedt, 2011b). To conclude, 17 kinds of personal meaning were considered in this research. The scale for the study contained items that were formulated as self-oriented statements (e.g., "*I engage with mathematics in order to …*").

As mentioned earlier, regulatory styles were assessed using established scales (Thomas & Müller, 2011). This survey questionnaire considers the regulatory styles in an educational context, adapted for the subject mathematics. For both instruments, a 4-point Likert scale (0 = strongly disagree to 3 = strongly agree) is used to rate the survey. The following Table 1 illustrates the scales of 17 different kinds of personal meaning and the five regulatory styles that were considered in this dissertation.

To investigate Finnish students, the German instruments of personal meaning and regulatory styles were translated²¹ into Finnish by Jessica Salminen-Saari and Dr Eeva Haataja. At the time that the data was being collected in Finland, both were PhD students at the University of Helsinki. To avoid translation mistakes or misconceptions (heuristics of common sense and problems of operationalization and measurement), the instrument was validated in a cognitive lab (Zucker et al., 2004). In collaboration with Dr Eeva Haataja, we conducted the cognitive lab with two Finnish ninth graders. Based on their feedback, retrieved from the cognitive lab, some items were correspondingly revised in the final version. Additionally, from the original total²² of 131 items of personal meaning and 21 items of regulatory style, those satisfying expectations regarding psychometric properties and factor loadings across both German and Finnish data were retained. Consequently, the statistical analyses conducted in this study considered a total of 68 items of personal meaning in addition to 17 items of regulatory style. Moreover, mathematics performance (in Germany marks range from 1 to 6 (highest to lowest), whereas in Finland they rank from 10 to 4

²¹ Initially, the author translated the German instruments into English.

²² For the original survey instrument, see the additional file in Appendix A.

EMPIRICAL WORK

(highest to lowest)) and gender were additionally integrated as individual-related data. The focus of this study is neither on students' mathematics performance nor on their gender. Nevertheless, the inclusion of this background information may provide an implicit contribution to the meta-theoretical knowledge relating to this study's research interest.

3.3.3 Quantitative Statistical Analyses

In the following section, I describe the quantitative statistical methods used within this empirical work in more detail. The following overview (Figure 11) summarizes the planned empirical work in a condensed format.

Figure 11

Overview of the Present Investigation's Empirical Work.

Rationale	Data	Analysis Method
Investigation of affective constructs' psychometric properties and their relations in Germany and Finland	Personal meanings and regulatory styles	Reliability CFA Correlation analyses Measurement invariance
Identification of profiles of personal meaning in Germany and Finland	Personal meanings and regulatory styles	Latent class analysis with distal outcomes
Differences in profile outputs in Germany and Finland	Personal meanings and regulatory styles	Mann-Whitney U test

Multivariate Analysis Methods. Software. Statistical analyses were carried out using the Mplus statistical package (version 8; Muthén & Muthén, 1998–2017).

Missing Data. In this present investigation the estimated percentage of missing data was low (at a maximum of 1.9% in Finland and a maximum of 1.1% in Germany). Table 1 provides the exact amount of missing data for each item. I used the Mplus FIML function to treat missing data. The FIML process assumes that values are not randomly omitted and takes into account all the information provided in order to estimate the parameters of the model (Enders, 2010). Complex models require a large sample size and thus listwise deletion was ignored (Hoyle & Gottfredson, 2014).

Table 1.

Descriptive Statistics of Sample, Means, and Standard Deviations for all Items of Personal Meaning and Regulatory Style in German and Finnish Students.

Item		Germany Finland				
	N	М	SD	Ν	М	SD
Active practice						
I enjoy doing calculations.	274	1.52	0.96	255	1.73	0.96
I am happy when I can do mathematical tasks.	272	1.07	0.90	253	1.68	0.98
I like to deal actively with mathematics.	273	1.33	0.91	253	1.75	0.91
<i>Classroom management</i> It annoys me if no one listens to what the teacher says.	274	1.88	0.95	255	1.84	0.96
It bothers me if the maths lesson is noisy and chaotic.	276	2.12	0.90	256	2.08	0.93
I don't like it if my teacher has to wait a long time until it becomes quiet in the lesson.	275	1.98	0.90	254	2.17	0.82
It is important to me that we can work well in maths lessons.	276	2.34	0.62	255	2.36	0.65
It annoys me if the students only start working a long time after the lesson has already begun.	276	1.44	0.88	253	1.45	0.93
Cognitive challenge						
If I cannot solve tricky tasks immediately, I think about it until I get it.	275	1.74	0.91	253	1.97	0.89
It is important to me that I feel challenged from mathematics.	272	1.47	0.83	254	1.63	0.91
I am unmotivated when the tasks in the maths lessons are too simple.	273	1.15	0.90	256	1.24	1.00
Duty						
I mainly deal with mathematics because I have to.	276	1.58	0.94	252	1.37	0.99
I deal with mathematics mainly because it is a compulsory subject.	272	1.76	0.94	249	1.57	1.00
Dealing with mathematics is nothing more than an obligation.	274	1.45	0.90	254	1.30	0.92
To deal with mathematics is an obligation to me.	275	1.55	0.97	252	1.54	0.98
Emotional-affective relation to the teacher						
It is important to me that I have a good re- lationship to my teacher in maths les- sons	273	2.20	0.80	248	1.98	0.91
I pay more attention in maths lessons if I like the teacher.	273	1.99	0.92	255	2.22	0.83

276	1.64	0.93	252	1.51	0.98
275	1.98	0.82	247	1.91	0.86
273	1.55	0.94	256	1.69	0.94
273	1.52	0.99	248	1.76	1.04
274	1.97	0.90	256	1.75	0.96
275	1.78	0.83	253	1.74	0.88
275	1.70	0.77	251	1.69	0.83
271	1.34	0.79	251	1.36	0.88
275	1.91	0.81	254	1.98	0.91
272	2.47	0.66	255	2.51	0.73
276	2.33	0.75	253	2.37	0.83
270	2.19	0.76	249	2.00	0.87
273	1.89	0.90	254	1.61	0.96
275	1.89	0.75	254	1.87	0.79
275	1.46	0.92	246	1.70	0.94
274	2.10	0.72	253	2.13	0.80
273	2.24	0.78	249	2.08	0.94
271	1.79	0.77	251	1.96	0.85
	 276 275 273 273 274 275 271 275 271 275 276 270 273 275 275 274 273 274 273 271 	2761.642751.982731.552731.522741.972751.702751.702751.912752.472762.332702.192731.892751.892751.462742.102732.242711.79	2761.640.932751.980.822731.550.942731.520.992741.970.902751.780.832751.700.772711.340.792751.910.812762.470.662762.190.762751.890.902751.890.752751.460.922742.100.722732.240.782711.790.77	2761.640.932522751.980.822472731.550.942562731.520.992482741.970.902562751.780.832532751.700.772512711.340.792512751.910.812542762.330.752532702.190.762492731.890.902542751.460.922462751.460.922462742.100.722532732.240.78249	2761.640.932521.512751.980.822471.912731.550.942561.692731.520.992481.762741.970.902561.752751.780.832531.742751.700.772511.692711.340.792511.362751.910.812541.982722.470.662552.512762.330.752532.372702.190.762492.002731.890.902541.612751.460.922461.702742.100.722532.132751.460.922461.702742.100.722532.132731.290.762492.082741.460.922461.702742.100.722532.132731.460.922461.702742.100.722532.132751.460.782492.082711.790.772511.96

citizens should have mathematical skills.						
I think that learning mathematics is im-						
portant because it is of major im-	273	1.66	0.85	251	1.83	0.89
Mathematics is important to me because						
thinking mathematically has many ad-	275	1.81	0.83	252	1.86	0.93
vantages.						
I deal with mathematics because logical thinking pertains to general education.	272	2.11	0.77	251	2.06	0.88
Marks						
I deal with mathematics in order to be proud of my marks.	274	1.81	0.91	254	2.11	0.82
I make an effort in mathematics in order to be pleased with my good marks.	274	2.08	0.84	255	2.30	0.76
I deal with mathematics as good marks strengthen my self-confidence.	274	1.73	0.93	254	2.01	0.88
Positive image						
It is important to me that I am given credit for my achievements in mathematics from people who are important for me.	270	1.43	0.98	249	1.74	0.88
It is important to me that others get a posi- tive impression of me through my mathematics achievements.	273	1.34	0.90	249	1.51	0.94
I deal with mathematics in order to get appreciation for my achievements.	273	1.22	0.87	249	1.45	0.86
It is important to me that my family can be proud of my mathematics achieve- ments.	272	1.92	0.96	247	1.90	0.87
Purism of mathematics						
In its purity, mathematics is uniquely beautiful to me.	270	1.36	0.95	255	1.38	0.97
The structure of mathematics fascinates me.	273	1.27	0.91	253	1.44	0.97
The beauty of mathematics appears to me in simple results for difficult problems.	270	1.40	0.87	253	1.45	0.94
I am overjoyed by the uniqueness of the mathematical technical language.	269	1.07	0.81	254	1.14	0.84
Mathematics is beautiful to me as it is unique in its formalism.	270	1.10	0.85	250	1.12	0.91
Due to the logical structure of mathemat-	070	1.75	0.01	254	1 50	0.00
ics it is easy for me to understand the content.	273	1.75	0.81	256	1.70	0.92
I am excited about mathematics because mathematical statements are either right or wrong.	274	1.37	0.81	254	1.30	0.90
Reference to reality						
It is important to me that the content of maths lessons refers to everyday life.	273	1.45	0.88	251	1.35	0.86

I understand mathematics more easily if tasks refer to examples from my life.	272	1.68	0.89	250	1.48	0.90
I prefer mathematical tasks which apply to situations from real life.	273	1.77	0.87	251	1.81	0.82
Tasks referring to everyday life make me feel more involved with the mathemat- ical content.	271	1.43	0.86	251	1.30	0.86
Self-perfection						
I work on many mathematical tasks in or- der to become better and faster.	273	1.26	0.82	255	1.68	0.87
I deal with mathematics in order to im- prove my logical thinking.	273	1.60	0.81	255	1.99	0.87
I deal with mathematics in order to in- crease my mathematical self-confi- dence.	271	1.20	0.80	253	1.64	0.91
Mathematics is important to me because I can develop myself in a number of ways.	273	1.41	0.84	254	1.51	0.89
Support by the teacher						
My teacher's support is important for my learning process in mathematics.	272	2.04	0.84	255	1.98	0.89
It is important to me that my teacher com- municates my progress in mathematics to me	271	2.10	0.81	250	1.88	0.87
It is important to me that my teacher is in- terested in my progress in maths les- sons.	270	1.96	0.81	247	1.85	0.81
Vocational precondition						
I deal with mathematics as I need it for my desired profession.	270	1.30	1.06	251	1.85	0.99
I deal with mathematics because I will have a wider selection of professions later on	273	1.75	0.92	251	2.08	0.88
I try to be good at mathematics because thereby I will have better opportunities	273	2 02	0.87	250	2 32	0.80
in a profession later on.	275	2:02	0.07	230	2.52	0.00
I learn mathematical content because I have to apply it during my studies/vo-	271	1.79	0.88	248	1.97	0.83
cational training. I will not need school mathematics for my professional life.	274	1.87	0.95	251	2.26	0.86
Amotivation:						
In maths lessons, I am just physically pre- sent but I don't think.	273	0.69	0.78	252	0.83	0.89
If the teacher does not notice, I do other things.	274	1.12	0.84	249	.1.29	0.90
I do not care about maths lessons.	274	0.64	0.84	254	0.79	0.83

External regulation:

I work and learn in the subject mathemat-						
because otherwise I will get into trouble at home.	273	0.98	0.97	247	1.00	0.97
because otherwise I will get in trouble with my teacher.	273	0.82	0.72	247	0.90	0.82
because my parents require that from me.	273	1.26	1.03	247	1.55	0.92
Introjected regulation:						
I work and learn in the subject mathemat- ics						
because I like to be better than my classmates.	272	1.09	0.95	247	1.13	0.98
because I want the other students to think I am smart.	273	1.16	0.97	247	1.22	0.94
because I like to receive praise.	270	1.43	0.97	245	1.19	0.89
Identified regulation: I work and learn in the subject mathemat- ics						
because thereby I will have more op- portunities to choose a profession.	272	1.92	0.91	246	2.30	0.79
in order to be able to do further educa- tion.	273	1.98	0.95	247	2.40	0.73
because thereby I can get a better job. Intrinsic regulation:	273	2.04	0.89	247	2.26	0.85
<i>I work and learn in the subject mathemat- ics</i>						
because it is fun.	271	1.45	0.95	247	1.49	0.95
because I like to deal with mathemat- ics.	270	1.42	0.94	246	1.42	0.93
because I like to solve mathematical tasks.	273	1.37	0.96	247	1.57	0.98
because I like to think about the topics in the subject mathematics.	273	1.25	0.87	247	1.43	0.91
because I enjoy dealing with mathe- matics.	272	1.16	0.88	246	1.48	0.95

Reliability of Scales.²³ Using latent variable modelling, I assessed the scale reliability ρ (Raykov, 2012). The estimator ρ reflects the composite reliability, an alternative to the more widely used Cronbach's alpha. The decision to choose ρ instead of Cronbach's alpha was underpinned by the need to obtain a precise estimate of the respective latent constructs and avoid inconsistency, such as overestimation or underestimation of the reliability (Bollen, 1989; Geldhof et al., 2014; Raykov, 2009). For all 17 personal meanings and five regulatory styles, I estimated the scale reliability ρ (see Table 2). Regarding Germany, the composite reliabilities for personal meanings and regulatory styles were acceptable to good. The values for personal meanings varied between .61 (EXPERIENCE OF SOCIAL RELATEDNESS) and .87 (purism of mathematics) and for types of motivation from .76 (amotivation) to .90 (intrinsic regulation). Furthermore, for Finland, the values of the measured composite reliabilities were also acceptable to good. They ranged from .66 (EX-PERIENCE OF SOCIAL RELATEDNESS) to .90 (purism of mathematics) for personal meanings, and between .71 (external regulation) and .89 (intrinsic regulation) for regulatory styles. As imported affective constructs, the scales did not result in lower reliability in Finland than in Germany, despite the translation of material from the original language.

²³ Reliabilities are measured in nine separate measurement models reflecting the eight dimensions of personal meaning and one model encompassing the regulatory styles (see section 5.1 for more).

Table 2.

Scale Reliabilities for Regulatory Style and Personal Meaning Scales in German and Finnish Students.

	Amount of -	$\overline{ ho}$			
Scale	items	Germany	Finland		
Regulatory Style					
Amotivation	3	.76	.76		
External regulation	3	.81	.71		
Introjected regulation	3	.82	.84		
Identified regulation	3	.89	.89		
Intrinsic regulation	5	.90	.89		
Personal meaning					
Active practice	3	.82	.77		
Classroom management	5	.78	.80		
Cognitive challenge	3	.67	.67		
Duty	4	.81	.87		
Emotional-affective relation to the teacher	4	.70	.75		
Examination	3	.78	.83		
EXPERIENCE OF AUTONOMY	4	.71	.72		
EXPERIENCE OF COMPETENCE	4	.70	.72		
EXPERIENCE OF RELATEDNESS	4	.61	.66		
General knowledge	4	.78	.84		
Marks	3	.85	.84		
Positive image	4	.72	.79		
Purism of mathematics	7	.87	.90		
Reference to reality	4	.82	.80		
Self-perfection	4	.71	.81		
Support by the teacher	3	.70	.78		
Vocational precondition	5	.81	.80		

Confirmatory Factor Analysis. I recorded the reliability and the validity of the hypothetical constructs by means of confirmatory factor analysis (CFA). As an advanced method of multivariate analysis, CFA is used to operationalize theoretically based latent variables using directly observed indicators on an empirical level (Backhaus et al., 2015). Following a reflective measurement model (Backhaus et al., 2015; Moosbrugger & Schermelleh-Engel, 2006), the indicator variables entirely represent the latent variable. For this reason, the assignment of the observed variables, the determination of the amount of factors, and their meaning in terms of the theoretical consideration should be explicitly considered (Backhaus et al., 2015; Brown, 2006).

Data were analysed with the Mplus robust maximum likelihood estimator (MLR). The estimator MLR is robust to non-normality of the manifest variables. Because of the sample's nested design (interest in modelling the differences at the individual level, not across classes), design-based correction of standard errors, and model-fit statistics, MLR is used with the analysis function "complex" (Muthén & Muthén, 1998–2017; Muthén & Satorra, 1995). In both countries, I considered "student's grade of school" as the clustering level. For the model evaluation, several model-fit indices should be considered. In this study, statistical goodness-of-fit of CFA was measured using the Comparative Fit Index (CFI), the Tucker Lewis Index (TLI), the root mean square error of approximation (RMSEA), and the standardized root mean square residual (SRMR). For chi-square difference testing, the Satorra-Bentler (Satorra & Bentler, 2001) scaled χ^2 is included to estimate chi-square under non-normality. The CFI and TLI fluctuate from 0 to 1; indices > 0.90 show an acceptable model fit to the data, while those ≥ 0.95 are an excellent model fit (Hu & Bentler, 1999). The SRMR are indicated as reasonable if the values are $\leq .08$ (Hu & Bentler, 1999).

Test of Measurement Invariance. The concept of measurement invariance is an essential requirement for the comparison of relationships and latent means of constructs in at least two different grouping variables, as here in Germany and Finland. Hence, students should respond in the same way across groups if they have the same score on the underlying construct. This statistical method tests whether the cross-cultural comparisons are equivalent and the survey instrument is operating as theoretically intended between both countries, and allows the cross-cultural generalizability of the estimated factor models (Steenkamp & Baumgartner, 1998). When testing measurement invariance, the context conditions of an investigation comparing German and Finnish students can be defined. Non-invariance indicates invalid or biased group comparison due to a different contextual understanding of the items (DIF items; French & Holmes Finch, 2016). If non-invariance is ignored, the results provide an unfair, faulty, and meaningless comparison (i.e., comparing apples and oranges; Cheung & Rensvold, 2002). In this study, students may understand DIF items differently due to the influence of their educational systems, inspired by the corresponding cultural context.

The existence of a measurement invariance can be gradually examined to demonstrate the validity of the hypothesized model in different groups. This investigation is carried out on the basis of a sequence of hierarchically nested linear multiple-group CFAs. These nested CFAs become

increasingly restricted. While the least restrictive baseline model has no invariance constraints, the most constrained model requires the parameters in question to be equal across the countries (Bollen, 1989; Brown, 2006; Chen, 2007; Meredith, 1993). Within these nested models, the initial three models of the series have a certain role when comparing the mean structure in Germany and Finland, namely configural, metric, and scalar invariance. The baseline model, so-called configural invariance, indicates with acceptable model fit, identical factor amount, and factor loading patterns between indicators and latent variables. In the subsequent model, the factor loadings are required to be equal across groups by testing metric invariance. This level constitutes weak measurement invariance. Metric invariance is a precondition for all other upcoming logically constituted levels of increasingly stringent restrictiveness. Scalar invariance requires identical factor loadings (metric invariance) and item intercepts of indicator variables and implies strong measurement invariance. Latent mean comparisons across groups are only possible if models show scalar invariance. When scalar invariance is reached, the latent constructs respond to the certain peculiarity of indicator variables. Hence, the presentation of configural, metric, and scalar invariance supports the comparison of latent mean differences for a meaningful interpretation (Cheung & Rensvold, 2002). Lastly, the most restrictive level is strict measurement invariance, which confirms equal item uniqueness. As this level is not compulsory to a comparison of the construct's latent means, this model was not estimated in this case.

However, achieving scalar invariance is demanding, especially when relatively new instruments are implemented in the survey (as in the present study). In this case, as an alternative approach, *partial (scalar) invariance* facilitates the comparison of differences in latent means (Byrne et al., 1989; Steenkamp & Baumgartner, 1998). In other words, if additional model restrictions (from metric to scalar invariance) lead to a significant drop in goodness of model fit, individual indicators can be identified and freely estimated in the groups. The identification of indicators should be based on the theoretical considerations that could follow an explorative approach (Vandenberg & Lance, 2000), for instance by considering modification indices (MIs) and expected parameter changes. Meaningful and valid group comparisons for the measured constructs are possible, if at least two indicators of each factor supports metric and scalar invariance (Byrne et al., 1989; Steenkamp & Baumgartner, 1998).

The chi-square difference test was not conducted because of its sensitivity to sample size (Chen, 2007). Instead, Chen's (2007) and Cheung and Rensvold's (2002) guiding criteria were used when comparing the set of nested parsimonious-to-constraint model change. With regard to Chen (2007), a decrease of the model fit CFI at (Δ CFI) \leq .010 in the more parsimonious model or a drop in RMSEA (Δ RMSEA) < .015 indicates empirical support for the less constrained model and the respective level of invariance.

Profile Identification - Latent Class Analysis. The statistical procedure of latent class analysis (LCA) was conducted to classify German and Finnish students into homogenous profiles based on the 17 personal meanings in mathematics. This exploratory classification was based on students' response patterns regarding the observed items (Lazarsfeld & Henry, 1968). The interindividual differences in the patterns that were observed are attributable to class affiliation (Geiser,

2011; Rost, 2006). These latent classes correspond to class-specific response profiles (Geiser, 2011). Profiles were conducted using Mplus version 8 (Muthén & Muthén, 1998–2017). Students' individual mean scores were used to conduct LCA. Mplus robust maximum likelihood (MLR) estimated the profile solutions from one to five profiles of personal meaning. The Mplus analysis option "mixture" was used for the nested design of the data.

Regarding profiling classes, I used listwise deletion of responses with incomplete data (cases with ≥ 2 missing values of 17 personal meanings). The number of samples to be analysed (after listwise deletion, N = 264 in Germany and N = 243 in Finland) was relatively small, but met the minimum requirements for performing an LCA (Nylund et al., 2007). Further missing values were handled by Mplus FIML (Asparouhov & Muthén, 2010). To avoid local maxima in the bootstrap associated with suboptimal results (Geiser, 2011), 3000 random sets of start values and 100 initial stage iterations were set. Local maxima tended to be more common as the number of classes increased. This issue was considered when setting the LRTSTARTS (Mplus analysis option, Geiser, 2011).

Moreover, to identify the optimal amount of classes several goodness-of-fit indicators were used, the information criteria used were: Akaike's Information Criterion (AIC), the Bayesian Information Criterion (BIC), sample-sized-adjusted BIC (aBIC), and the Vuong–Lo–Mendell–Rubin likelihood ratio test (VLMR), as well as the inference statistical results of the model-comparing bootstrap likelihood ratio test (BLRT). Furthermore, it is preferable that the average latent class assignment probabilities on the main diagonal should be >.80 (Rost, 2006, p. 278) and that accuracy of classification entropy should be near 1 (values close to 0 represent a low level of accuracy). To conclude a meaningful interpretation, LCA models should not have groups with < 5 % of the cases (Marsh et al., 2009).

The final and most important decision as to the "best" amount of profiles was made by examining the class interpretability in terms of the theoretical conclusions (Kleemann et al., 2013). Therefore, it should be taken into account whether the profiles of personal meaning can be well differentiated with relation to the substantial theory behind that concept (Marsh et al., 2009). What is more, the procedure suggested by Geiser (2011) entails a parsimonious approach according to which an LCA model with a small number of classes is preferred. As each LCA is very specific with respect to the scholar's research interest, there is no clear description of how to interpret and then validate the LCA output. To assess the quality of LCA models, I investigated the interpretability initially as follows: Response values should be either high or low (i.e., not medium) for all factors in the profile (Geiser, 2011). The 4-point Likert scale as a whole gives clues, ranging from 0 (strongly disagree) to 3 (strongly agree). In this present study, first I examined whether each profile can be clearly interpreted by considering its mean scores ($\geq 1.8 =$ constructed personal meanings and $\leq 1.2 =$ non-constructed personal meanings). Average values are probably made up of different components and therefore they do not provide any information when clarifying the profiles (Geiser, 2011). Therefore, I ignored the personal meanings with moderate mean values in all LCA models across countries. After detecting the best LCA solution for each country, I added the regulatory styles to the LCA model. As distal outcomes, regulatory styles represent additional profile indicators without affecting the identified profiles of personal meaning (Muthén & Muthén, 1998–2017). Moreover, the replicability of the latent class solution in a different data set, and coherence of profiles with external variables (here regulatory styles as covariates) represent support for a valid class solution (Geiser, 2011). To investigate the validity of the profile solution, further analyses (Mann–Whitney U test) in both countries were carried out to contrast the profile output. Furthermore, within the theory-driven analyses, I analysed and validated each profile with respect to this study's conceptual framework, which was built by coordinating theories (see section 4.2.2).

Mann–Whitney U test. This analysis is mainly used as a non-parametric alternative procedure to the independent *t–test*. The Mann–Whitney U test is used as a guide to test whether the profile output (the learners' profiles of personal meaning and their quality of motivation) varies between countries. For this analysis, IBM SPSS Statistics (version 27; IBM Corp., 2020) was used.

3.4 Résumé

This study's empirical work focuses on a classical survey (self-reported) that was carried out using standard quantitative research methods. As well as a new questionnaire (instrument for 17 personal meanings), I applied a standardized instrument (for five regulatory styles) to assess the latent variables under investigation. It is obvious that this study's properties (two theories behind 22 latent variables, investigated by means of two methodological approaches in two different cultural contexts) represent various methodical challenges. For this reason, an in-depth and critical examination of the measurement uncertainty and data analysis might be fruitful in terms of reaching the overall research goal (cf. 2.6). If we rely on the applied quantitative methods (including the coordination analysis method), it should be noted that this present quantitative study is more explorative than hypothesis-testing in its nature. Based on the methods used here, this study cannot explain causal relationships between learners' personal meanings and their quality of motivation.

4. Theory-Driven Analyses

As discussed above (cf. 3), I set out to study personal meaning's multimodal ontological natures (Radford, 2014, p. 359). This concept seems to exist both on an individual (through the lens of SDT) and a social level (through the lens of social constructivism). The empirical findings, self-report data gathered using quantitative statistical methods, captured only one of these aspects, namely personal meaning as an individual psychological phenomenon. Based on that weakness, an additional method was applied to reconstruct learners' social experiences in order to capture personal meaning as a social phenomenon. In this way, personal meaning could be considered from the viewpoint of both theoretical lenses, SDT and social constructivism, as a means of exploring how personal meaning interconnects the predicted social and biological regulatory mechanisms that seem to affect learners' motivation in mathematics (cf. 2.3.3). For this purpose, a research frame was necessary that networked the multimodal ontological natures of personal meaning, namely the social and biological regulatory mechanisms.

Stressing the philosophical ethos of triangulation, the networking strategy of coordinating represents a research practice that methodologically considers both theoretical approaches by respecting their "diversity as richness" (Prediger et al., 2008, p. 170). Before I describe the coordination analysis method, I introduce the philosophical background behind this research practice.

4.1 Methodological Considerations for the Coordination Analysis

As indicated earlier, the affective construct of personal meaning reflects the learner's perceived personal relevance with respect to the subject of mathematics in an educational setting. Theoretically, this construct may enhance our conceptual understanding of the relation between students' individual relevance systems and their quality of motivation, which could be of use in engaging students in learning. Considering the phenomenon as situated at the boundary of SDT and social constructivism can be useful in order to shed new light on the characteristics that make up personal meaning. Even though each of the two theoretical lenses provides valuable information about subjects' motivation when learning mathematics, a clarification of their interplay may provide sophisticated knowledge in terms of how they jointly interact while affecting the quality of motivation in mathematics lessons. In this spirit, I use the idea of theoretical triangulation to reflect on connecting both *epistemological viewpoints* (Sabena et al., 2014, p. 189).

4.1.1 Theoretical Triangulation

Triangulation is a good way to gain more convincing evidence in research practice. At first, the term referred to a trigonometrical concept applied in land surveying and geodesy (Denzin, 1989; Fielding & Fielding, 1986; Flick, 2006; Kelle & Buchholtz, 2015). In the sense of the term first used by Denzin, triangulation expresses the combination of different perspectives used for an investigation (Denzin, 1989; Flick, 2008).

The knowledge of a theory is, at some point, exhausted in terms of examining an empirical object. Theoretical triangulation (also triangulation of different data sources, survey or analysis methods, and scholars' perspectives) is a research strategy used to deepen knowledge of the research phenomenon and to increase the explanatory power of the empirical evidence (Denzin, 1989; Flick, 2008). According to Denzin, theoretical or perspective triangulation works by

approaching data with multiple perspectives and hypotheses in mind. Data that would refute central hypotheses could be collected, and various theoretical points of view could be placed side by side to assess their utility and power (...). Such strategies would permit sociologists to move away from polemical criticisms of various theoretical perspectives, since pitting alternative theories against the same body of data is a more efficient means of criticism. (Denzin, 1970, p. 303)

This research strategy is particularly supportive when different theories shed light on one empirical phenomenon; in such cases, it can be used to examine the data's theoretical coherence from different points of view (Flick, 2008, pp. 14–15). However, the application of perspective triangulation does not automatically lead to enhanced validity of the empirical analysis (Bikner-Ahsbahs & Prediger, 2014, p. 243). In order to benefit from this approach, and not least to compare the results of multiple theoretical lenses, theories should target common properties of the empirical phenomenon as well as share the same conceptual understanding of the researched phenomenon. Only if both perspectives examine the same research object from complementary angles can the results be compared, and triangulation lead to a dense and coherent understanding of the studied object under investigation (Bikner-Ahsbahs & Prediger, 2008).

Triangulation of theories is a well-known research practice in the field of mathematics education (Bikner-Ahsbahs & Prediger, 2014, p. 243). In qualitative empirical case studies this tradition of linking theoretical lenses is particularly widespread, but it is rarely realized in quantitative empirical investigations. This is not surprising, since qualitative research essentially aims to develop or explore (Bikner-Ahsbahs, 2015; Przyborski & Wohlrab-Sahr, 2014) theories, categories, or concepts (Kelle & Buchholtz, 2015) by following inductive procedures (Neuman, 2011). Nonetheless, connecting theories in quantitative studies would be enriching, not only to solve practical problems in research analysis but also to broaden one's mind with respect to (at least) implicit knowledge behind the applied theoretical approaches, for example by pointing out their methodical benefits and technical obstacles for future research (Prediger et al., 2008, pp. 175–176).

In this study's initial quantitative approach, highly structured statistical methods were used to measure the required affective constructs of personal meaning and regulatory styles. Now, I draw on the strengths of triangulating theories to compensate for the weakness that arose in quantitative mono-methodological research (Kelle & Buchholtz, 2015). That is, the main purpose of including theoretical triangulation is to bypass the methodological gap by systematically networking both theoretical perspectives (SDT and social constructivism); that is, to qualify the quantitative data (Schoonenboom & Johnson, 2017). In terms of triangulating

theories, the research activity of networking theories not only supports the examination of the double ontological natures of personal meaning sequentially; it also helps to bring both (complementary) theoretical lenses into a dialogue concerning their individual and conjoint understandings of learners' multimodal (Radford, 2014) motivational experiences in the German and Finnish mathematics-classroom settings.

4.1.2 Approach of Networking Theories to Studying Personal Meaning's Social Nature

The members of a working group at the Congress of the European Society for Research in Mathematics Education (CERME) were the first scholars to discuss the idea of networking theories as a research practice. They elaborated several ways of scientifically linking multiple theoretical perspectives, and proposed strategies for networking theories in the field of mathematics education (cf. Bikner-Ahsbahs & Prediger, 2014; Prediger et al., 2008). These strategies provide the robustness to handle both lenses' specific conceptual identities without neglecting theoretical contradictions (Bikner-Ahsbahs & Prediger, 2014; Janßen, 2016, p. 80). The following figure (Figure 12) delineates all strategies on a scale that systematically places them in order of their connectivity, which ranges from *ignoring other theories* to *unifying globally*.

Figure 12

Networking Strategies for Connecting Theories in Mathematics Education (adapted from Prediger & Bikner-Ahsbahs, 2014).



With respect to the networking analysis carried out in this dissertation, I used the strategy of *coordinating*.

Following the idea of triangulation, combining and coordinating means looking at the same phenomenon from different theoretical perspectives as a method for deepening insights into the phenomenon. The distinction between combining and coordinating is drawn according to the degree of integration of theory elements with respect to their compatibility. *Combining* theoretical approaches does not necessitate the complete compatibility of the theoretical approaches under consideration. Even theories with conflicting basic assumptions can be combined in order to get a multi-faceted insight into an empirical phenomenon in view. In contrast, we use the word *coordinating* when a conceptual framework (...) is built by fitting together elements from different theories for making sense of an empirical phenomenon. A conceptual framework is not a new theoretical approach but a pragmatic bricolage for the purpose of understanding empirical phenomena. (Prediger & Bikner-Ahsbahs, 2014, pp. 119–120)

Initially, I brought the theories together by *combining* them. However, simply juxtaposing both lenses (Sabena et al., 2014, p. 189), SDT and social constructivism, is insufficient to increase our knowledge about learners' quality of motivation in the mathematics classroom with respect to personal meaning. For my theory-driven analyses, I thus used a lens that emerged from coordinating (Prediger & Bikner-Ahsbahs, 2014) social and biological regulatory mechanisms through the medium of the motivational construct personal meaning.

In the following sections, I describe the applied coordination analysis method that I used to study the theoretical connections in Germany and Finland. To systematically structure the coordination analysis, coordinating the theories on quantitative data, I used three (adjusted) terminologies (essentially taken from the qualitative research paradigm, see section 4.2.1) that guided my interpretation procedure (see section 4.2.2): namely "identify with" (German: *Positiver Gegenhorizont*), "not identify with" (German: *Negativer Gegenhorizont*), and enacting potential (German: *Enaktierungspotential*). The use of such a methodological mixture (Kelle & Buchholtz, 2015, p. 330), using terms from qualitative research to interpret standardized data obtained by quantitative research methods, raises various questions; for example, how to interpret the empirical results (referring to the nature of the qualitative or quantitative paradigm) and where to anchor this mixed-research approach in the methodological research landscape. To avoid methodological risks and misinterpretations, I briefly elaborate the critical aspects associated with this mixture in order to clarify the depth to which these qualitative terminologies can lead by applying them to a body of quantitative data (see section 4.2.3).

4.2 Coordination Analysis Method

According to the working definition of coordinating theories, an application of this strategy can only be realized by considering theoretical lenses with *compatible cores* that *complement* each other (Prediger et al., 2008, p. 172). A theoretical assurance of this maxim can be done by highlighting the theories' commonalities and specificities. That emphasizes whether the different epistemic views (Sabena et al., 2014) are complementary or not. The existence of complementary views (*well-fitting elements*) enables the researcher not only to adopt different perspectives but also to constitute a *conceptual framework* to specifically target the research object under investigation (Prediger et al., 2008, p. 172).

Theoretically, I conceptualized personal meaning as located at the boundary of SDT and social constructivism (cf. 2.3.3). Thereby, the potential emerged to build a conceptual framework by coordinating both epistemic angles, which seem, in their natural state, to be compatible and complementary. On a theoretical level, I carefully elaborated how the theoretical boundaries between SDT and social constructivism can be bridged by considering personal meaning as a boundary object. Based on the level of principles (Radford, 2008a), I compared both theories' informative values in terms of quality of motivation in mathematics (cf. 2.3).

For the record, each theoretical angle shed light on different issues (Sabena et al., 2014, p. 188) surrounding learners' quality of motivation in the mathematics classroom. On the one hand, the solid framework of SDT is qualified to provide an insight into learners' regulatory styles (amotivation, external regulation, introjected regulation, identified regulation, intrinsic regulation) referring to their perceived basic psychological needs (for autonomy, competence, and relatedness). From an epistemic point of view (Sabena et al., 2014, p. 189), SDT stresses the importance of students' personal relevance in academic contexts but does not provide an insight into the "*what*" (kinds of, or patterns in the personal relevance of studying mathematics) that directs learners' behaviour in the mathematics classroom. The insight into learners' "what" may help to understand *why* they react in a certain way in class (controlled or autonomous behaviour). Moreover, although the dialectical conversation between the individual and their social context represents the nutrient medium for an organism's development, learners' social experiences are not likewise elaborated within SDT. At this point, SDT's principles (Radford, 2008a) show a theoretical limit.

Social constructivism, on the other hand, provides an insight into learners' experiences of social interaction in the classroom. Ernest proposed a model providing a micro insight into the "production of learning and of knowledge" (Ernest, 2010, p. 45) in a classroom setting that explains "how signs become appropriated by an individual through experiencing their public use" (Ernest, 2010, p. 44). Here, social constructivism is, from an epistemic point of view, limited to explaining learners' individual behaviour as a counterpart that associates with certain social experiences in the mathematics classroom.

The theoretical conceptualization of personal meaning (Vollstedt, 2011b; Vorhölter, 2009) seems to provide a basis on which to connect SDT and social constructivism's perspectives in order to jointly study learners' quality of motivation in class. Theoretically, when we consider the empirical phenomenon of personal meaning as a "bracket" at the boundary of two theoretical perspectives, the object "personal meaning" seems to link both theories in an interesting manner (cf. 2.3.3). In other words, through this joint view of social constructivism it may be possible to explain how the development of learners' individual relevance systems takes place; that is, explain learners' reactions to the outside world (referring to social regulatory mechanism). Further, SDT could explain how learners' inner worlds (referring to biological regulatory mechanism) react to the socially experienced individual relevance systems in the mathematics classroom. Since the theoretical cores of both perspectives refer to individuals' experiences, but perceived in different worlds, the perspectives of what (individual relevance system) and why (motivational behaviour) might complement and articulate each other in a useful way (Prediger et al., 2008, p. 172). The following coordination analysis of this conceptual framework, which considers both regulatory mechanisms as they relate to personal meaning, would not only provide an insight into the respective world (of social or biological regulation of personal meaning); if the assumed connection exists, the complementary understanding as a common view that has emerged through the conceptual framework would also clarify how learners' motivation develops in the German and Finnish mathematics classrooms.

Next, to interpret and relate these numerical data systematically with respect to the conceptual framework built by means of coordination, I used three (adjusted) terminologies: "identify with", "not identify with", and "enacting potential". Before I describe my

coordination analysis procedure, I briefly introduce these terminologies and exemplify how they are commonly used within the qualitative research paradigm.

4.2.1 Terminologies for a Theory-Driven Interpretation of the Quantitative Data

In the qualitative research field, *group discussion* represents a research practice (amongst others) that is used to investigate and reconstruct ideal-typical (Weber, 1972) orientation frameworks constituted by a pattern of thought and action on the part of an observed group of actors with respect to a specific research context (Kleemann et al., 2013). According to Bohnsack (1989), this method aims to reconstruct the respondents' orientation framework (the so-called orientation pattern²⁴). To confirm and validate the identified orientation frameworks of groups, a systematic comparative analysis is carried out in order to saturate the identified types. Finally, a series of contrasting comparisons of types aims to generate a typology that reflects the range of the multiple reconstructed orientation patterns of the researched participants (Kleemann et al., 2013). The typology illustrates multidimensional types of orientation, representing the generalization of orientation patterns that result in the investigated research context (Bohnsack, 2014; Kleemann et al., 2013).

Individuals rarely refer directly to their orientation pattern in group discussions (Kleemann et al., 2013). Their orientation pattern needs to be catalysed from their descriptions and examples (Kleemann et al., 2013, p. 161). Researchers use three structuring features that facilitate the excavation process of actors' orientation patterns: Positive counter-horizon (German: Positiver Gegenhorizont) and negative counter-horizon (German: Negativer Gegenhorizont), as well as enacting potential (German: Enaktierungspotenzial; Bohnsack, 1989, pp. 26–27; Bohnsack, 2014, p. 138). These three features provide "access and contours to their orientation pattern" (Kleemann et al., 2013, p. 161, translated by the author). A positive counter-horizon refers to the "point of reference (actions, people, attitudes, etc.)" that represents what "belongs to one's own will" (Kleemann et al., 2013, p. 238, translated by the author). It provides orientation regarding a certain context, and the offered area is accepted by the actor. On the contrary, the term negative counter-horizon describes the "reference point with which one distinguishes oneself and thus shows what does not belong to one's own will" (Kleemann et al., 2013, p. 238, translated by the author). It also provides orientation regarding a certain context, which is rejected by the actor. The restructured orientation therefore stands on a continuum between two opposing horizons. (Bohnsack, 1989, p. 27). Another key point of reference in group discussions is the "chance of practical realisation of one's own orientation" (Kleemann et al., p. 234, translated by the author). The chance of applying one's own basic orientation in the real world is known as enacting (Bohnsack, 1989, p. 26; Kleemann et al., 2013, p. 161). The individuals and their orientation patterns within their field of experience (i.e., that which provides, or fails to provide orientation) are situated in varying degrees of relation to these three structural components (Bohnsack, 1989, p. 27; Bohnsack, 199, 2014; Kleemann et al., 2013). Thus, these three components make up the orientation framework around the learner's field of experiences (Bohnsack, 1989, 1999). One example of

²⁴ This definition of action makes sense if it is based on the ideal-typical constructed pattern of orientation. In this context, orientation pattern is understood in the sense of Weber's ideal type (Bikner-Ahsbahs, 2015; Bohnsack, 2014, p. 146; Weber, 1972).

their use from a qualitative research study is provided by Kleemann et al. (2013), who investigated the importance of school and its significance in learners' lives:

The students at the only grammar school (Gymnasium—for high achievers) in a small town developed despite all their differences (lower and upper secondary, girls and boys) a common sense of unity; this is displayed somewhat condescendingly towards students from other types of schools. The orientation framework here consists of an understanding of belonging to an elite. One's own school and the level of education it provides are the positive counter-horizon of self-worth, while peers with less education represent a negative counter-horizon. To be able to let the other students feel this evaluation through their self-presentation, is how the students exercise their enacting potential. (Kleemann et al., 2013, pp. 161–162, translated by the author)

The use of the three structuring terminologies²⁵ helps to capture the group's orientation pattern, and the comparison of these orientation frameworks enables us to generalize a multidimensional constructed typology. The configuration of these types demonstrates the level of difference between the reconstructed orientation frameworks and sheds light on the themes and concerns the group members are focused on (on the example shown above "the significance of attending a particular school", see Kleemann et al., 2013).

In contrast to the original application of these three terms within the qualitative approach, in the present study the empirical analysis helped to identify learners' individual relevance systems (patterns of personal meaning). To capture and compare the different profiles' mathematics-learning orientations, I used the structuring elements, adjusted as "identify with" (positive counter-horizon), and "not identify with" (negative counter-horizon). That is, I reorganized the *computed* profiles of personal meaning as orientation patterns that are constituted by these directions; that is, "identify with", and "not identify with". Hence, I followed the reverse order to the conventional procedure in qualitative research. Further, the adaptation of enacting potential made it possible to investigate the theoretical coherence between the social and biological natures of personal meaning, by connecting both ontological natures of personal meaning with respect to the conceptual framework. In the following section, I describe in more detail the systematic use of this procedure of structuring components to coordinate the analysis.

4.2.2 Coordination Analysis Procedure

To investigate the hypothesized connection between social and biological regulatory mechanisms through personal meaning (conceptual framework built by coordinating theories), adjusted terminologies from qualitative research (identify with, not identify with, and enacting potential) were used to help structure the coordination analysis method systematically. In the previous section, I detailed how the structuring elements provide access to groups' orientation patterns, as extracted from interview data material within a qualitative research approach. Next,

²⁵ Of course, this was done in combination with other methodical steps that are not discussed here.

I used these terminologies in an adjusted manner to interpret the numerical data in line with this investigation's conceptual framework. While describing the coordinating analysis method, I define how I adapted these terms with respect to this study's research target. The following figure, Figure 13, provides an overview of the evaluation steps of the coordination analysis I conducted.

Figure 13





The upper side (greyed out) of Figure 13 has two parts: the processes of the social regulatory mechanism (which refer to social constructivism) and those of the biological regulatory mechanism (which refer to SDT). The analysis sequences started first on the lefthand side, that of social regulations. To reconstruct²⁶ the learner's experiences of social interaction, I aimed at a systematic interpretation of the mean values with respect to the social regulatory mechanism. Again, based on the predicted social regulatory mechanism, I assume that the learner's individual relevance system is affected by all regulations (of appropriation, transformation, publication, and processes of conventionalization²⁷) in the mathematics classroom. To explain *theoretically* how these individual relevance systems arise and unfold in the classroom (but also to examine whether the identified empirical profiles make sense on a theoretical level), I link these social processes (apart from processes of conventionalization) to each identified empirical profile of personal meaning.

Thereby, as indicated earlier (2.3.2), the publication stage provides the relevance system in which the learner produces their reactions to their social experiences in the mathematicslearning environment. As mentioned earlier, this very specific level of publication represents an outcome that eventuates after a series of experiences on the part of the learner, illustrated by the adapted cycle of appropriation, transformation, publication, and processes of conventionalization (the social regulatory mechanism of the individual relevance system, cf. Figure 6). By means of the classical survey instrument (self-report), I recorded what learners published by recounting to me; namely, their individual experiences dating from a certain point in the learner's life and not the numerous experiences that led to this production. Firstly, starting from the learner's constructed relevance system, I initially described the profile by means of the eight dimensions of personal meaning (cf. Figure 5) and called this the publication of the social regulatory mechanism. This procedure was helpful in terms of summarizing the profile outcome (regarding the constructed relevance system, $M \ge 1.8$ cut-off point; for nonconstructed personal meanings, $M \le 1.2$ cut-off point). Secondly, I evaluated the appearance of the eight dimensions of personal meaning in the students' relevance systems; that is, I studied whether the theoretical composition of the eight dimensions stayed conjoint for profile specification or diverged. This step, publication of social regulation, thus provides a summarized overview of each learner's individual relevance system in general.

Based on the general profile summary, I then worked out the profile's orientation pattern in more detail. I therefore used the adjusted terminologies to theoretically reconstruct the individual's social experiences in class as follows: Based on the numerical responses from their self-reporting (ticked boxes on the 4-point Likert scale), the individual's learning orientation in mathematics was identified by one scope that is accepted and another one that is rejected for one's own profile orientation. I interpreted these ticked boxes as *identify with* if personal meanings were constructed and *not identify with* when personal meanings were not constructed. Based on Bohnsack (1989, 1999, 2014), I used the term "identify with" for a scope that is

²⁶ This notion of reconstruction is not comparable with the reconstruction of qualitative data (see section 4.2.3).

²⁷ The operation "processes of conventionalization" triggers individuals' learning orientations in the classroom; that is, it influences their orientation towards "identify with" or "not identify with". In order to reconstruct "processes of conventionalization", the detected profiles' classroom environments need to be observed. Empirical observations are not included in this study. For this reason, no clear statements can be made about this operation. Processes of conventionalization are thus excluded from the coordination analysis. However, when implicit knowledge regarding this process emerges in the evidence, it will be discussed later.

accepted as a means of personal identification, a pole that provides orientation and in the context of which the profile identifies with studying mathematics in the classroom. The term "not identify with" applies to the exact reverse of this situation. This refers to a scope that provides no orientation for the learner and is avoided; thus, the profile is that of an individual who does not identify with that pole. For the record, the profile's poles were reconstructed from the individual's self-reports (biological regulatory mechanism) in terms of which scopes were accepted and which were rejected for their own relevance system as a means of providing orientation towards mathematics learning in a social classroom setting. The consideration of the opposing poles also helps validate the meaningfulness of the statistical profiles; that is, by showing whether they are contrariwise and mutually representing opposite of one another. In this connection, I specifically noted the learner's reaction to the three specific personal meanings (EXPERIENCE OF AUTONOMY, COMPETENCE, and SOCIAL RELATEDNESS). I related the reconstructed profile's orientation to the regulations *transformation* and *appropriation* as part of the social regulatory mechanism.

Then in step two, based on the profile's orientation I formulated a theory-compliant hypothesis about what the learner's motivational tendency in class may look like. In other words, after *theoretically describing the social nature* of personal meaning, I built a bridge between social and biological regulation in order to connect the outer social regulations within the classroom context with the learner's inner biological regulation.

In step three, I analysed to what extent the theoretical explanation of the learner's social regulation could be reflected within the empirically measured quality of motivation (covariates of each profile). This step not only reflects the theoretical coherence (between social regulatory mechanism and biological regulatory mechanism) but also evaluates whether the different perspectives consistently complement each other within the predicted conceptual framework (Prediger et al., 2008). This is because, if we can finally use this social regulatory mechanism to explain how these empirical profiles have developed, then we would have good reasons to accept this specific link—between biological and social regulatory mechanisms—as meaningful. The ability to biologically experience one's own relevance system (orientation pattern) in practice is referred to here as *enacting potential* (Bohnsack, 1989).

After following these three steps, I interpreted the results from the German and Finnish data with respect to the conceptual framework behind them (bottom half of Figure 13). Thus, theorization of the coordination of the results was carried out in order to understand how quality of motivation develops in the German and Finnish mathematics classrooms. Therefore, the coordinating view integrated both theoretical lenses to capture the resulting theoretical interaction. On the basis of the obtained interaction between theories across countries, I interpreted these indicators with respect to learners' educational systems in Germany and Finland. Finally, after critically reflecting the predicted links within the conceptual framework, I explored the extent to which both epistemic views contribute to the elucidation of learners' motivation development in the mathematics classroom.

4.2.3 Critical Reflection on Applying Terms from the Qualitative Research Tradition

This study's empirical work solely focuses on quantitative research methods and the obtained standardized data is subsequently interpreted by means of adapted terminologies ("identify with", "not identify with", and "enacting potential"). Basically, these terminologies stem from the qualitative research tradition. Due to this mixture of approaches, that is, the use of qualitative terms to analyse numerical data, it is necessary to clarify how these terms are understood in the present methodology. For this purpose, I briefly outline the general methodological standards of qualitative and quantitative research that go along with the application of these structuring categories. While highlighting the differences in these standards, I simultaneously point out which relevant methodological properties are applied in this study.

The quantitative and qualitative research traditions use different standards: within the qualitative research paradigm, the group discussion method is applied to examine narrative interview material (Bohnsack, 2014). This material is closer to the actions of each participant and focuses on the respondent's point of view. By applying the structuring terms to non-standardized material and through the flexible process of reconstruction, the investigator aims to capture the respondent's holistic view by treating them as a subject (and not as an object; Lamnek & Krell, 2016).

In contrast, in the quantitative tradition, selected items that trace back to other students' views are prefabricated by scholars. For instance, the items used for profile identification can only consider the theoretical knowledge specified within the classical survey instrument of personal meaning. That is, by reconstructing students' mathematics-learning orientations, I limit respondents' actions to the given scales of personal meaning. Hence, the given items will only record the knowledge that stems from the 17 kinds of personal meaning. An insight into additional facets of the 17 kinds or other kinds that differ from Vollstedt's (2011b) results cannot be captured; for example, if personal meanings or facets exist that relate to Finnish culture specifically. For this reason, this reconstructive work within the qualitative tradition where the investigator absorbs the perspective of the researched actor until theoretical saturation is reached.

Furthermore, the understanding of typology differs between the research traditions with reference to their genesis: a typology that arises from qualitative research can consist of ideal types. In this tradition, types correspond with the ideal-typical understanding in the sense of Max Weber (1972), in which an observed action is ideal-typically constructed²⁸ based on the participant's individual biography (Bohnsack, 2014). In contrast, quantitative methods generate average types (here: profiles based on learners' mean values of the 17 personal meanings) that are strongly bound in time and space (Weber, 1972). In accordance with the quantitative tradition, this study thus aims for representative data (although it is very limited here due to the relatively small sample size), and not the representation of the social meaning (ideal types; Bikner-Ahsbahs, 2015; Weber, 1972).

²⁸ A detailed discussion about the construction of ideal types is provided by Bikner-Ahsbahs (2015).
Despite the use of these terminologies from the qualitative paradigm, the study carried out here belongs to the quantitative field. Consequently, while the ideas of these adapted terms can meaningfully support the coordination analysis process, they cannot, however, reach the same level of analysis as that which they would traditionally produce when applied to qualitative data.

4.3 Résumé

Altogether, this investigation's theory-driven activity tries to counterbalance the limitations of quantitative mono-methodology in two ways. Firstly, this networking strategy helps to capture the social nature of the construct personal meaning (challenge/support the adapted model of social regulatory mechanism with the explanation/interpretation of how these profiles have developed). Secondly, the research activity of coordinating theories shows promise as a means of interpreting and validating the empirical profiles of personal meaning. Following an explorative approach, the whole theory-driven analysis will thus lead to a specification of the evidence-based effects across the biological and social worlds of personal meaning. Based on these evidence-based indicators, found across countries, I will critically review the conceptual framework as regards the conceptual interplay between individual relevance systems and motivation in the mathematics classrooms of Germany and Finland.

5. Empirical Results

In this chapter, I present this study's empirical results; discussion of these results follows in the next chapter, 6. Initially I investigated the psychometric properties of personal meaning and regulatory styles assessment in Germany and Finland to see whether the scales could be used for a mean comparison concerning the latent variables across countries (reliability through composite scale reliability ρ , factor structure through confirmatory factor analysis (CFA), measurement invariance, and latent mean differences). Subsequently, I identified and compared students' profiles of personal meaning and their motivational outcomes in Germany and Finland (latent class analysis and Mann–Whitney U test).

The present investigation aimed at understanding how learners' motivation arises in German and Finnish mathematics classrooms with respect to the construction of students' individual relevance systems. As an initial step to this end, I had three research questions (RQs) that I examined successively when preparing the empirical database by means of classical survey instruments.

Concerning RQ 1, I determined the psychometric properties of personal meaning and regulatory styles assessment to see how these properties compare cross-culturally (see section 5.1). Therefore, I examined the scale reliability and factor structure of both constructs. For personal meaning, I tested for empirical support for each of the predicted eight dimensions of personal meaning in separate measurement models. In the next step, I investigated whether the eight dimensions could also be differentiated within one comprehensive model. Concerning regulatory styles, I tested a five-factorial model including all regulatory styles. Moreover, I studied the intercorrelations between personal meaning and regulatory styles in both groups for the theoretically assumed links (cf. 2.3.1). In addition, I investigated the measurement invariance across countries to delineate the generalizability of the underlying constructs. For personal meaning, I based this test on the resulting best factor model (eight separate models). These analyses helped to avoid biasing effects (i.e., cultural bias) in the outcome and to consider any methodological issues concerning the new (personal meaning) and established (regulatory styles) instruments applied here.

The need to translate instruments can be associated with a reduction in the quality of psychometric properties and inconsistencies with the original settings, even in cases where the translation is of a high standard (Bofah & Hannula, 2015). For this reason, with respect to the survey instruments used, I assumed there would be challenges concerning the use of the survey instruments in Finland, as both scales had been developed for German-speaking countries (Thomas & Müller, 2011; Vollstedt & Duchhardt, 2019), although they resulted in good psychometric properties in the country of origin. Regarding the overall model of personal meaning, I assumed there would be problems of measurement when estimating a too-complex model combined with a new theory upon which it was based. Concerning the intercorrelations between both affective constructs, I expected support for the predicted strong interrelations between the 17 personal meanings and regulatory styles (cf. 2.3.1) and relatively similar correlation patterns for German and Finnish data that relate to the cultural properties of Western European countries (cf. 2.4).

Regarding RQ 2, I inspected potential differences between German and Finnish learners' latent means of personal meanings and regulatory styles (see section 5.2). In line with previous research (based on item response theory (IRT); Suriakumaran et al., 2019), I expected to find cultural differences that reflect the respective mathematics curricula.

In relation to RQ 3, I identified profiles (individual relevance systems) of personal meaning showing distinct mathematics-learning orientations (see section 5.3). Moreover, I added the regulatory styles as covariate outcome variables. With respect to their cultural diversity, I conducted a separate latent class analysis in order to detect the best class solution for each country. Analogously to the second research question, I expected to find different response patterns of personal meaning that refer to the specific mathematics-learning experiences that are inherent in the German and Finnish educational systems.

On the basis of these statistical analyses, I developed empirical data that provided the foundation for the subsequent coordination analysis in the theory-driven work (see Chapter 7). In the following sections, I report the empirical results in more detail.

5.1 Factor Structure of Personal Meaning and Regulatory Style Across Countries

Separate measurement models. As indicated earlier, for each scale of personal meaning and regulatory styles the composite reliabilities ρ (Bollen, 1989; Geldhof et al., 2014; Raykov, 2012) were rated as acceptable to good for both countries (see 'Reliability of Scales' in section 3.3.3 for more). In order to further examine the internal psychometric properties and the factor structure of the 17 personal meanings, I initially computed separate measurement models for each of the dimensions of personal meaning. Thus, I tested to what degree the theoretical differentiation of the dimensionality of personal meaning holds empirically in Germany (GER) and Finland (FIN).

As indicated earlier (cf. 2.3.1), I expected the following theoretical separation in terms of the dimensionality of personal meaning: "purism of mathematics" was hypothesized to be one-dimensional, whereas "self-development" was assumed to comprise three factors (kinds of personal meaning): active practice of mathematics, cognitive challenge, and self-perfection. For "freedom of action", I predicted two separable factors: EXPERIENCE OF AUTONOMY and EXPERIENCE OF COM-PETENCE. The dimension of "professional qualification" was hypothesized to be three-dimensional: examination, marks, and vocational precondition. For "application in life", two different factors were predicted: general knowledge and reference to reality. "Emotional well-being in class" was expected to have four factors: classroom management, emotional-affective relation to the teacher, EXPERIENCE OF SOCIAL RELATEDNESS, and support by the teacher. Moreover, I expected "positive image" and "duty" to be one-dimensional.

To support the predicted separation of the eight dimensions of personal meaning, I compared the fit values of the assumed multifactor latent models with the alternative one-factor models for each multifaceted dimension respectively, following Satorra-Bentler's (Satorra & Bentler, 2001) scaled chi-square difference testing. These comparisons indicated the measured goodness-of-fit

values of the expected multidimensional models. For the alternative one-factor models (Kleinke et al., 2017), the multidimensional models with correlated factors were each combined into one factor (model with no separation). Nonetheless, these one-factor models entail a theoretical understanding of the constructs that is different from that of the postulated models. Apart from the goodnessof-fit indication, the model comparisons empirically support the meaningfulness of each theoretically developed multifactor model. For personal meaning, Table 3 (GER) and Table 4 (FIN) show the estimated fit statistics for the CFA measurement models in Germany and Finland.

Table 3.

Fit Statistics for Confirmatory Factor Analysis of Dimension of Personal Meaning Models Separately and all Together in German Students.

5 5 5	2 3	5	0	1 2 0			
GER Model	χ^2	df	CFI	TLI	RMSEA	SRMR	
Purism of mathematics							
One Factor	23.83	14	.983	.974	.050	.029	
Self-development							
One Factor	97.04	35	.914	.889	.080	.052	
Three Factors	47.23	32	.979	.970	.042	.036	
Freedom of action							
One Factor	87.54	20	.763	.668	.111	.077	
Two Factors	24.82	19	.980	.970	.033	.036	
Emotional well-being in class							
One Factor	392.78	104	.650	.597	.100	.093	
Four Factors	145.56	98	.942	.929	.042	.052	
Professional qualification							
One Factor	528.28	44	.468	.334	.200	.139	
Three Factors	66.61	41	.972	.962	.048	.041	
Application in life							
One Factor	272.35	20	.455	.237	.214	.151	
Two Factors	27.03	19	.983	.974	.039	.035	
Positive image							
One Factor	3.66	2	.989	.967	.055	.022	
Duty							
One Factor	5.18	2	.986	.957	.076	.024	
Overall models							
2 nd order	3546.484	2168	.804	.794	.048	.091	

Note. CFI =comparative fit index; TLI = Tucker–Lewis index; RMSEA = root mean square error of approximation; SRMR = standardized root mean square residual.

Table 4.

Fit Statistics for Confirmatory Factor Analysis of Dimension of Personal Meaning Models Separately and all Together in Finnish Students.

		e e	6			
FIN Model	χ^2	df	CFI	TLI	RMSEA	SRMR
Purism of mathematics						
One Factor	28.35	14	.977	.965	.063	.030
Self-development						
One Factor	80.04	35	.940	.923	.071	.043
Three Factors	58.56	32	.965	.951	.057	.036
Freedom of action						
One Factor	48.52	20	.922	.890	.075	.053
Two Factors	20.45	19	.996	.994	.017	.037
Emotional well-being in class						
One Factor	360.44	104	.766	.729	.098	.085
Four Factors	159.04	98	.944	.932	.049	.052
Professional qualification						
One Factor	388.94	44	.608	.511	.175	.111
Three Factors	69.91	41	.967	.956	.052	.040
Application in life						
One Factor	341.47	20	.459	.243	.252	.169
Two Factors	20.77	19	.997	.996	.019	.031
Positive image						
One Factor	3.05	2	.995	.984	.046	.023
Duty						
One Factor	5.10	2	.989	.966	.078	.020
Overall models						
2 nd order	3554.515	2168	.830	.821	.050	.082

Note. CFI =comparative fit index; TLI = Tucker–Lewis index; RMSEA = root mean square error of approximation; SRMR = standardized root mean square residual.

Overall, in the German sample the hypothesized (multi-)factor models (i.e., eight dimensions of personal meaning) yielded a good model fit (fit statistics ranging from CFI = [.942; .989], TLI = [.929; .967], RMSEA = [.033; .076], SRMR = [.022; .052]) whereas the alternative one-factor models (of the multifaceted five dimensions) all performed poorly where applicable (fit statistics ranging from CFI = [.455; .915], TLI = [.237; .889], RMSEA = [.80; .214], SRMR = [.052; .151]). For more details see Table 3. In Germany, each latent model recorded a good fit for the predicted eight dimensions of personal meaning. All factors had at least three manifest variables that loaded on the factor with loadings > .3.²⁹

The same holds for the Finnish sample, with fit statistics for the hypothesized (multi-)factor models all indicating a good fit (fit statistics ranging from CFI = [.944; .997], TLI = [.932; .996], RMSEA = [017; .078], SRMR = [.020; .052]), whereas the alternative one-factor models (of the multifaceted five dimensions) all performed poorly where applicable (fit statistics ranging from CFI = [.459; .940], TLI = [.243; .923], RMSEA = [.071; .252], SRMR = [043; .169]). For more details see Table 4. All factors had at least three indicators that loaded on the factor with loadings >.3.

In a nutshell, these results showed empirical support for the predicted dimensionality of personal meaning in both Germany and Finland.

According to the underlying self-determination theory (SDT), the latent model of regulatory styles was expected to be five-dimensional: amotivation, external regulation, introjected regulation, identified regulation, and intrinsic regulation. Due to the robustness of SDT, I disclaimed a comparison of the five-factor model with an alternative model. Regarding regulatory styles, Table 5 gives an account of the fit statistics for the CFAs in Germany and Finland.

 $^{^{29}}$ I also estimated the standardized factor loadings (Appendix B) and intercorrelations (Appendix C1–C2) in the 17factor model with all personal meanings together (see the additional files in the Appendix). As the further results will show (cf. second-order model with eight dimensions), this model did not fit the data across countries. However, all factor loadings were higher than the cut-off mark >.3.

Table 5.

Fit Statistics for Confirmatory Factor Analysis of Regulatory Style Models in German and Finnish Students.

Model	χ^2	df	CFI	TLI	RMSEA	SRMR
Regulatory styles						
Five Factors (GER)	185.59	109	.966	.958	.050	.044
Five Factors (FIN)	169.60	109	.968	.960	.047	.053

Note. CFI =comparative fit index; TLI = Tucker–Lewis index; RMSEA = root mean square error of approximation; SRMR = standardized root mean square residual.

In the German sample,³⁰ the theoretical five-factor solution resulted in a good fit of data (χ^2 = 185.59, *df* = 109, CFI = .966, TLI = .958, RMSEA = .050, SRMR = .044). Furthermore, all standardized factor loadings were higher than the cut-off mark >.3. The correlations (see Table 8) between the regulatory styles varied from |.21| (intrinsic regulation and external regulation) to |.54| (intrinsic regulation and amotivation).

Accordingly, in the Finnish sample the theoretical five-factorial measurement model fitted the data and recorded reasonable goodness-of-fit for the robust SDT underlying it ($\chi^2 = 169.60$, df = 109, CFI = .968, TLI = .960, RMSEA = .047, SRMR = .053). Furthermore, all standardized factor loadings were higher than the cut-off mark >.3. The correlations (see Table 9) between the regulatory styles varied from |.23| (introjected regulation and amotivation) to |.73| (intrinsic regulation and amotivation).

In summary, the CFA results in Germany and in Finland supported the theoretical dimensionality of personal meaning for each predicted dimension. Moreover, the results supported the factor structure of the regulatory styles in a five-dimensional model across countries.

Second-order model with eight dimensions of personal meaning. Subsequently, I estimated a second-order model with all the dimensions together in an overall model. This specification examined more closely the predicted dimensionality of personal meaning. Therefore, I tested whether the differentiation of personal meanings into eight dimensions holds empirically in one comprehensive model. Thereby, I intended not only to study the associations between the eight predicted dimensions, but also to find an adequate model for further analysis (test of measurement invariance). In this second-order model, the first-order factors were depicted by kinds of personal meaning that loaded on the respective second-order factor, i.e., on the respective dimension of personal meaning. By means of the second-order model, I estimated the associations between all eight dimensions of meaning: application in life, duty, emotional well-being in class, freedom of action, positive image, professional qualification, purism of mathematics, and self-development. In this present investigation, the second-order model recorded a poor fit both in Germany ($\gamma^2 = 3546.484$. df = 2168, CFI = .804, TLI = .794, RMSEA = .048, SRMR = .091) and in Finland ($\gamma 2 = 3554.515$, *df* = 2168, CFI = .830, TLI = .821, RMSEA = .050, SRMR = .082). Tables 6 (GER) and 7 (FIN) show the results from these second-order CFA models with factor loadings of the first-order factors and the resulting intercorrelations between the second-order dimensions. However, due to the poor overall model fit, these estimated model parameters are to be interpreted with care.

³⁰ In addition, Appendix D1–D4 report the standardized factor loadings with respect to the corresponding country (GER/FIN; see the additional file in the appendix).

Table 6.

Personal								
meaning	Ι	II	III	IV	V	VI	VII	VIII
Purism of mathematics	1.00							
Active practice of mathematics		.91						
Self-perfection		.86						
Cognitive challenge		.86						
Experience of competence			.70					
Experience of autonomy			.78					
Experience of social relatedness				.68				
Emotional-affective relation to				.77				
the teacher								
Classroom management				.51				
Support by the teacher				.88				
Marks					.84			
Examination					.64			
Vocational precondition					.50			
General education						.92		
Reference to reality						(.28)		
Positive image							1.00	
Duty								1.00
Purism of mathematics	-							
Self-development	.93	-						
Freedom of action	.78	.88	-					
Emotional w-b in class	(.19)	(.20)	.78	-				
Professional qualification	.40	.47	.80	.64	-			
Application in life	.78	.91	.86	(.42)	.63	-		
Positive image	.35	.35	.72	.63	.85	(.34)	-	
Duty	66	75	46	(.14)	(11)	52	(07)	-

Second-Order Confirmatory Factor Analysis: Factor Loadings of the First-Order Personal Meanings and Intercorrelations Between Dimensions of Personal Meaning in German Students.

Note. I = Purism of mathematics, II = Self-development, III = Freedom of action, IV = Emotional well-being (w-b) in class, V = Professional qualification, VI = Application in life, VII = Positive image, VIII = Duty.

Correlations in parentheses were not statistically significant; correlations printed in italics were significant at p < .01; all other correlations reported were significant at p < .001.

Table 7.

Second-Order Confirmatory Factor Analysis: Factor Loadings of the First-Order Personal Mean-
ings and Intercorrelations Between Dimensions of Personal Meaning in Finnish Students.

Personal								
meaning	Ι	II	III	IV	V	VI	VII	VIII
Purism of mathematics	1.00							
Active practice of mathematics		.93						
Self-perfection		.94						
Cognitive challenge		.90						
Experience of competence			.93					
Experience of autonomy			.79					
Experience of social relatedness				.85				
Emotional-affective relation to				.95				
the teacher								
Classroom management				.61				
Support by the teacher				.88				
Marks					.72			
Examination					.62			
Vocational precondition					.75			
General education						.92		
Reference to reality						.32		
Positive image							1.00	
Duty								1.00
Purism of mathematics	-							
Self-development	.95	-						
Freedom of action	.69	.91	-					
Emotional w-b in class	.31	.43	.71	-				
Professional qualification	.63	.83	.94	.72	-			
Application in life	.84	.97	.85	.54	.94	-		
Positive image	.45	.57	.72	.61	.83	.60	-	
Duty	62	68	45	(07)	47	60	19	-

Note. I = Purism of mathematics; II = Self-development; III = Freedom of action; IV = Emotional well-being (w-b) in class; V = Professional qualification; VI = Application in life; VII = Positive image; VIII = Duty.

Correlations in parentheses were not statistically significant; correlations printed in italics were significant at p < .01; all other correlations reported were significant at p < .001.

In Germany, apart from reference to reality, all factor loadings of the first-order factors were significant and scored values higher than the cut-off mark >.3. The factor loadings ranged from .50 (vocational precondition) to .91 (active practice of mathematics). The correlations between the eight dimensions were not weak, ranging from .35 (p < .01) (positive image and purism of mathematics, positive image and self-development) to .93 (p < .001) (self-development and purism of mathematics).

Regarding Finland, the first-order factor loadings were all significant and higher than the cut-off mark >.3. The loadings varied from .32 (reference to reality) to .95 (emotional-affective relation to the teacher). Here, most of the variance was well explained. Regarding the relations between the dimensions, the correlations between the higher-order dimensions ranged from .31 (p < .001) (emotional well-being in class and purism of mathematics) to .97 (p < .001) (application in life and self-development).

Altogether, these results indicated that this model structure did not fit the data. In both independent data sets, the fit statistics were behaving in a consistent way: the CFI fit resulted in a poor index, indicating that the variables used were not highly correlated (Brown, 2006; Maydeu-Olivares, 2019; Shi et al., 2019). Although the items were assigned to the respective factor, the drop in TLI and CFI suggested a need to consider modification of the measurement model by revising the theoretical factor structure (Shi et al., 2019). Moreover, although RMSEA (Shi et al., 2020) was supported, the weak SRMR fit reflected that even despite the large amount of variables (and high *df*), certain important relations between a couple of variables were still not measured appropriately (Brown, 2006; Maydeu-Olivares, 2019; Savalei, 2012). In conclusion, these results showed that the estimated model is too parsimonious to record essential connections between some factors (Maydeu-Olivares, 2019), and the model did not record the data well (Brown, 2006; Shi et al., 2019). Thus, next to the huge number of variables, the theoretical relations between the dimensions (Backhaus et al., 2015) decreased the fit statistics.

In a nutshell, it was not possible to test the separation of the dimensions in one complete model.³¹ In the second-order model the complex composition of model structure, but certainly also the theoretical correlations between the factors, induced these poor results. For personal meaning, the eight separate measurement models for each dimension of personal meaning yielded the best

³¹ By revising the factor structure, I also estimated an overall model with all 17 personal meanings together. The computation of a 17-factor model with 68 items is risky (Brown, 2006). As assumed from a literature review, in both groups the 17-factor model did not record a reasonable fit: Germany ($\chi 2 = 3140.704$, df = 2074, CFI = .848, TLI = .834, RMSEA = .043, SRMR = .061), Finland ($\chi 2 = 3243.324$, df = 2074, CFI = .857, TLI = .843, RMSEA = .047, SRMR = .062). Next to the huge number of variables, the theoretical relations among the 17 factors (Backhaus et al., 2015) decreased the fit statistics: In both independent data sets, the CFI fit resulted in a poor index, indicating that the variables used were not highly correlated. Across countries, SRMR values improved in comparison to the second-order model, indicating that the model based on the average of the residual correlations (Hu & Bentler, 1999) captures the data well (Shi et al., 2019). Although the items were assigned to the respective factors, the drop in TLI and CFI suggests a need to consider a modification of the measurement model by an alternative factor structure with respect to the theory (Brown, 2006; Savalei, 2012; Shi et al., 2019, 2020).

fit compared to both comprehensive models tested. As a consequence of these CFA results, I decided to continue the measurement invariance analyses with the eight separate measurement models of personal meaning and the five-dimensional model of motivational regulations separately.

Intercorrelations between personal meaning and regulatory styles. To investigate the intercorrelations between the theoretically assumed eight dimensions of personal meaning and the five regulatory styles, I initially studied the relations between the 17 kinds of personal meaning and regulatory styles. This step was necessary to inspect how the 17 personal meanings associate individually and subsequently as dimensions with the regulatory styles. Table 8 (GER) and Table 9 (FIN) report the intercorrelations between the 17 personal meanings and regulatory styles. In the next step, I examined the relations between dimensions of personal meaning and regulatory styles. Table 10 (GER) and Table 11 (FIN) provide the associations between the dimensions of personal meaning and regulatory styles. In the following section, I refer solely to the significant relations (p < .01 and p < .001).

Table 8.Intercorrelations Between Regulatory Styles and Personal Meanings in German Students.

				-					-												
	Act	Cla	Cog	Dut	Emo	Exa	Au	Co	So	Gen	Mark	Pos	Puri	Rea	Sel	Supp	Voc	1	2	3	4
1 Am	63	54	51	.57	(18)	33	57	36	42	55	41	25	47	(11)	51	(18)	37				
2 Ext	20	(.00)	(10)	.41	(.15)	.19	(12)	(.05)	(04)	(06)	.23	.37	(17)	(.09)	(02)	.31	(02)	.34			
3 Intro	.38	.28	.38	14	.29	.37	.42	.54	(.19)	.42	.57	.86	.41	(.21)	.49	.35	.41	24	.28		
4 Iden	.23	.22	.32	17	(.17)	(.14)	.32	(.12)	(.10)	.51	.30	.26	.24	(.17)	.40	(.09)	.87	30	(.04)	.33	
5 Intr	.94	(.24)	.81	70	(.03)	(.19)	.70	.23	(.23)	.65	(.20)	.23	.79	(.12)	.69	(15)	.46	54	21	.31	.28

Notes. N = 276; Act = Active practice; Cla = Classroom management; Cog = Cognitive challenge; Dut = Duty; Emo = Emotional-affective relation to the teacher; Exa = Examination; Au = EXPERIENCE OF AUTONOMY; Co = EXPERIENCE OF COMPETENCE; So = EXPERIENCE OF SOCIAL RELATEDNESS; Gen = General education; Mark = Marks; Pos = Positive image; Puri = Purism of mathematics; Rea = Reference to reality; Sel = Self-perfection; Supp = Support by the teacher; Voc = Vocational precondition; Am = Amotivation; Ext = External regulation; Intro = Introjected regulation; Iden = Identified regulation; Intr = Intrinsic regulation.

Correlations in parentheses were not statistically significant; correlations printed in italics were significant at p < .01; all other correlations reported were significant at p < .001.

Table 9.Intercorrelations Between Regulatory Styles and Personal Meanings in Finnish Students.

				0					0												
	Act	Cla	Cog	Dut	Emo	Exa	Au	Со	So	Gen	Mark	Pos	Puri	Rea	Sel	Supp	Voc	1	2	3	4
1 Am	82	50	68	.79	24	40	66	60	31	66	40	29	63	(13)	67	(16)	56				
2 Ext	27	(18)	19	.33	(.02)	(05)	(24)	28	(06)	(11)	(02)	(.23)	(15)	(06)	(20)	(.04)	(07)	.41			
3 Intro	.37	.31	.41	(13)	.37	.32	.38	.49	.40	.27	.29	.58	.35	.17	.36	(.25)	.23	23	(.14)		
4 Iden	.44	.23	.47	37	.33	.45	.45	.55	(.30)	.60	.53	.50	.38	(.08)	.49	.23	.84	35	(04)	.24	
5 Intr	.95	.42	.80	67	.19	.34	.68	.56	.32	.67	.33	.36	.81	(.17)	.77	(.06)	.50	73	27	.39	.33

Notes. N = 256; Act = Active practice; Cla = Classroom management; Cog = Cognitive challenge; Dut = Duty; Emo = Emotional-affective relation to the teacher; Exa = Examination; Au = EXPERIENCE OF AUTONOMY; Co = EXPERIENCE OF COMPETENCE; So = EXPERIENCE OF SOCIAL RELATEDNESS; Gen = General education; Mark = Marks; Pos = Positive image; Puri = Purism of mathematics; Rea = Reference to reality; Sel = Self-perfection; Supp = Support by the teacher; Voc = Vocational precondition; Am = Amotivation; Ext = External regulation; Intro = Introjected regulation; Iden = Identified regulation; Intr = Intrinsic regulation.

Correlations in parentheses were not statistically significant; correlations printed in italics were significant at p < .01; all other correlations reported were significant at p < .001.

Table 10.

	Amotivation	External	Introjected	Identified	Intrinsic
Self-development	63	15	.44	.31	.94
Freedom of action	63	(04)	.64	.31	.64
Emotional w-b in class	37	.20	.39	(.18)	(.03)
Professional qualification	53	(.08)	.62	.91	.52
Application in life	58	(05)	.46	.56	.69

Intercorrelations of Second-Order Multifaceted Dimensions and Regulatory Styles in German Students.

Note. Correlations in parentheses were not statistically significant; correlations printed in italics were significant at p < .01; all other correlations reported were significant at p < .001.

Table 11.

Intercorrelations of Second-Order Multifaceted Dimensions and Regulatory Styles in Finnish Students.

	Amotivation	External	Introjected	Identified	Intrinsic
Self-development	78	24	.40	.50	.90
Freedom of action	70	31	.52	.59	.68
Emotional w-b in class	29	(00)	.38	.33	.22
Professional qualification	66	(08)	.36	.92	.57
Application in life	69	(12)	.30	.62	.70

Note. Correlations in parentheses were not statistically significant; correlations printed in italics were significant at p < .01; all other correlations reported were significant at p < .001.

Intercorrelations found in German data. As a first step, I examined which of the 17 personal meanings each of the regulatory styles recorded the strongest positive relation to. To gain a first overview, for each regulatory style I identified the highest correlating kind of personal meaning. In Germany, intrinsic regulation had the highest positive relation with the active practice of mathematics (r = .94, p < .001), whereas identified regulation scored the highest relation with vocational precondition (r = .87, p < .001). For introjected regulation, I found the highest positive association with positive image (r = .86, p < .001). External regulation (r = .41, p < .001) and amotivation (r = .57, p < .001) had the strongest positive reference with duty.

With respect to the multifaceted dimensions, intrinsic regulation scored the strongest positive correlation with self-development (r = .94, p < .001), whereas identified regulation correlated highly positively with professional qualification (r = .91, p < .001). Introjected regulation recorded the strongest positive association with freedom of action (r = .64, p < .001) and external regulation recorded only a moderate association with emotional well-being in class (r = .20, p < .001). For amotivation, I found no positive association regarding the multifaceted dimensions.

Intercorrelations found in Finnish data. In Finland, I also examined which of the 17 personal meanings each of the regulatory styles recorded the highest positive relation to. Intrinsic regulation had the highest positive relation with the active practice of mathematics (r = .95, p < .001), whereas

identified regulation correlated highly positively with vocational precondition (r = .84, p < .001). For the style of introjected regulation, I found the highest positive association with positive image (r = .58, p < .001). External regulation (r = .33, p < .01) and amotivation (r = .79, p < .001) associated positively with duty.

Regarding the positive relations to the multifaceted dimensions of personal meaning, intrinsic regulation scored a strong positive correlation with self-development (r = .90, p < .001) and identified regulation correlated highly positively with professional qualification (r = .92, p < .001). Introjected regulation associated strongly positively with freedom of action (r = .52, p < .001). For external regulation and amotivation, I found no positive association in relation to the multifaceted dimensions.

Altogether, the correlation analyses in both independent data sets came up with interesting insights. In Germany, regarding the predicted strongest positive relations (cf. 2.3.1) between the 17 personal meanings and regulatory styles, five personal meanings (duty, examination, EXPERI-ENCE OF COMPETENCE, general education, and marks) showed a different relation to quality of motivation. Apart from general education, the personal meanings recorded associations with controlled types instead of with the predicted self-determined regulations. Surprisingly, examination (correlating highly positive with introjected regulation and not at all with identified regulation) was closer to controlled motivation than to self-determined motivation. In general, almost all personal meanings showed intercorrelations with regulations from bordering regions, apart from the active practice of mathematics and purism of mathematics (both correlated with intrinsic and introjected regulations, but not with identified regulation). As theoretically assumed, this result suggested that personal meanings do not refer to just one specific type of motivation. On the contrary, as expected, most of the personal meanings had one positive correlational emphasis and additional associations with regulations that were close to the prioritized positive correlation. Moreover, reference to reality did not yield any significant relations, neither positive nor negative, with types of motivation. EXPERIENCE OF SOCIAL RELATEDNESS did not show any positive reference to types of motivation.

On the level of dimensions, the predicted intercorrelations could be partly replicated in the German sample. Surprisingly, freedom of action and application in life showed the strongest correlations with regulations other than those that had been predicted. Here, freedom of action exhibited the same strong relation with both autonomous and controlled types of motivation. Apart from application in life (expected strongest relation with identified regulation) and freedom of action (expected to relate only with intrinsic regulation or self-determined regulation), all dimensions related to the regulations as predicted. For application in life and freedom of action, consideration of the facets of dimensions explains the respective correlation. Concerning application in life, reference to reality resulted in no significant correlation, whereas general education had the strongest relation with intrinsic regulation. With respect to freedom of action, both factors, EXPERIENCE OF AUTONOMY and EXPERIENCE OF COMPETENCE, recorded moderate to strong relations with intrinsic and introjected types of motivation. In summary, in contrast to the dimensions, consultation of the

correlation analysis with the 17 personal meanings provided a more detailed insight into the interrelations between personal meaning and quality of motivation in Germany.

In Finland, almost all personal meanings recorded the predicted association structure with regulatory styles. With respect to the theoretically assumed highest positive relations among the 17 personal meanings and regulatory styles, five personal meanings, classroom management, duty, general education, reference to reality, and support by the teacher, showed a different relation to regulatory styles. Apart from duty and reference to reality, the personal meanings yielded relations with (more) self-determined regulatory styles. Interestingly, in Finland certain personal meanings also resulted, next to the association with introjected regulation, in minimum moderate positive relations with self-determined types of motivation: classroom management, emotional-affective relation to the teacher, examination, EXPERIENCE OF COMPETENCE, EXPERIENCE OF SOCIAL RELAT-EDNESS, positive image. In general, most personal meanings showed intercorrelations with regulatory styles of neighbouring types, apart from classroom management and EXPERIENCE OF SOCIAL RELATEDNESS (both correlated with intrinsic and introjected regulations). Also here, the theoretical assumptions were mirrored in the empirical data. Most of the personal meanings not only referred strongly to one certain type of motivation, but also to regulations that were close to the prioritized positive correlation. For all personal meanings, I found at least a low positive relation to the regulatory styles.

On the level of dimensions, the assumed intercorrelations could mostly be replicated. Regarding the predicted strongest correlations, only application in life resulted in a rather self-determined association, showing the highest correlation with self-determined regulatory styles rather than with identified regulation, as had been expected. For all other dimensions, the expected strongest association with each respective regulation was confirmed. Again, in contrast to the dimensions, consideration of the correlation analysis with the 17 personal meanings provided a more in-depth insight into the interrelations between the constructs of personal meaning and quality of motivation in Finland.

On a correlational level, results from the two countries were similar, although not identical. In order to further compare both countries, analyses of measurement invariance were conducted to determine the validity of such a comparison.

Measurement invariance across countries. To inspect the latent mean structure of the 17 personal meanings and five regulatory styles across countries, I estimated two-group CFAs. As personal meanings could only be measured by eight separate measurement models, eight two-group CFAs recorded to what extent the eight dimensions of application in life, duty, emotional well-being in class, freedom of action, positive image, professional qualification, purism of mathematics, and self-development can be generalized across both countries. For regulatory styles, one two-group CFA was conducted to estimate a certain level of measurement invariance regarding the five styles.

As previously stated, to test for measurement invariance, I made a comparison between the model fit indices for models with dissimilar sets of invariance constraints. As outlined by Meredith

(1993), the constraints ranked from the baseline model (no restrictions for parameters – configural invariance) to strong factorial invariance (constraints on measurement intercepts and factor loadings – scalar invariance) that represent the precondition for latent mean comparisons over groups (Meredith, 1993). For each two-group CFA model, I consistently implemented the appropriate constraints from the configural model (M1) over the metric (M2) to the (partial) scalar measurement model (M3/M3p). Following an exploratory approach, I evaluated the decrease in goodnessof-fit (Chen, 2007; Cheung and Rensvold, 2002) that resulted with the respective change in parameter constraints. Table 12 reports the model fit statistics of the measurement invariance test.

Table 12.

Fit Statistics for Each Measurement Invariance Model of the Nine Higher-Order Dimensions of Personal Meaning and Regulatory Style.

Tested Model	χ^2	df	CFI	$\Delta_{ m CFI}$	TLI	RMSEA	$\Delta_{\rm RMSEA}$	SRMR	Δ_{SRMR}
Purisms of mathematics									
M1: configural invariance	54.32	28	.978		.967	.059		.030	
M2: weak measurement invariance	59.55	34	.978	.000	.973	.053	.006	.035	.005
M3: strong measurement invariance	72.02	40	.973	.005	.972	.055	.002	.039	.004
Positive image									
M1: configural invariance	8.04	4	.988		.963	.062		.022	
M2: weak measurement invariance	12.76	7	.982	.006	.970	.056	.006	.036	.006
M3: strong measurement invariance	28.06	10	.945	.037	.934	.083	.027	.050	.014
M3p: partial strong measurement invariance	14.87	9	.982	.000	.976	.050	.006	.042	006
Duty									
M1: configural invariance	10.29	4	.987		.962	.077		.022	
M2: weak measurement invariance	13.69	7	.987	.000	.977	.060	.017	.029	.007
M2p: partial weak measurement invariance	13.42	6	.985	.002	.970	.068	.009	.029	007
M3p: partial strong measurement invariance	17.46	9	.983	.002	.978	.060	008	.033	.004
Emotional w-b in class									
M1: configural invariance	293.30	196	.948		.936	.043		.052	
M2: weak measurement invariance	317.88	208	.941	.007	.932	.045	002	.059	007
M3: strong measurement invariance	367.44	220	.921	.020	.914	.050	.005	.066	007
M3p: partial strong measurement invariance	341.83	218	.934	.007	.927	.046	.001	.062	.003
Professional qualification									
M1: configural invariance	123.00	82	.976		.968	.043		.040	
M2: weak measurement invariance	133.80	90	.974	.002	.968	.043	.000	.046	.006
M3: strong measurement invariance	175.41	98	.954	020	.949	.054	.011	.052	.006
M3p: partial strong measurement invariance	149.19	97	.969	.005	.965	.045	.002	.048	.002
Self-development									
M1: configural invariance	96.09	64	.978		.969	.043		.036	
M2: weak measurement invariance	103.57	71	.978	.001	.972	.042	002	.042	003
M3: strong measurement invariance	156.88	78	.946	032	.938	.062	.020	.054	.012

M3p: partial strong measurement invariance	111.09	76	.976	.002	.972	.042	.000	.045	.003
Application in life									
M1: configural invariance	47.19	38	.991		.987	.030		.033	
M2: weak measurement invariance	56.54	44	.988	.003	.985	.033	.003	.042	.009
M3: strong measurement invariance	77.61	50	.973	015	.970	.046	.013	.051	.009
M3p: partial strong measurement invariance	68.97	49	.981	.007	.978	.039	.006	.049	.007
Freedom of action									
M1: configural invariance	46.07	38	.988		.982	.028		.036	
M2: weak measurement invariance	53.33	44	.986	.002	.982	.028	.000	.047	.011
M3: strong measurement invariance	74.95	50	.962	024	.958	.043	.015	.048	.001
M3p: partial strong measurement invariance	64.16	49	.977	.009	.974	.034	.006	.046	.001
Regulatory style									
M1: configural invariance	335.92	218	.971		.964	.045		.049	
M2: weak measurement invariance	359.62	230	.968	.003	.962	.046	.001	.052	001
M3: strong measurement invariance	443.48	242	.951	.017	.944	.056	010	.057	005
M3p: partial strong measurement invariance	405.06	240	.960	.008	.954	.051	.005	.055	.003

Notes. CFI = comparative fit index; TLI = Tucker–Lewis index; RMSEA = root mean error of approximation; SRMR = standardized root mean squared residual.

For all two-group CFAs, the initial configural model across countries showed a good fit to the data (M1). This reflected that all models fit the data across groups by estimating the same loading structure of factors.

Next, I compared the model fit from configural invariance (M1) with metric invariance (M2), for which factor loadings are constrained. Here, I found support for metric invariance for all twogroup CFAs apart from duty. In line with Chen (2007), the two-group CFA for model fit for duty did not reveal an acceptable cut-off limit (Δ RMSEA 0.015); whereas CFI and SRMR supported the model fit, metric invariance was not supported for duty. This indicated that the measurement properties of the scale were not manifested equally over countries and duty cannot be generalized to measure the same latent construct in Germany and Finland. According to Chen (2008), non-metric invariance would probably result by importing a scale from one linguistic and cultural setting to another that indicated difference in conceptualization and meaning of construct. Moreover, a failure in loading invariance may result when respondents tend to tick or avoid extremes on scales (Chen, 2008).

To ensure that this scale was manifested in the same way in Germany and Finland, I examined the partial metric model, following the guidelines in the modification indices (Byrne et al., 1989; Cheung & Rensvold, 1998; Rensvold & Cheung, 1998). Here, partial measurement invariance allowed certain factor loadings to be freely estimated and some to be invariant across groups. As a consequence, I allowed one out of four items of the scale "duty" to be non-invariant across countries. As the majority of factor loadings were invariant across groups, the resulting model supported partial metric invariance for factor loadings. Thus, all investigated scales (application in life, duty (partial), emotional well-being in class, freedom of action, positive image, professional qualification, purism of mathematics, self-development, and regulatory styles) have invariant factor loadings and thus showed that in both countries the constructs are manifested similarly (Millsap & Olivera-Aguilar, 2012).

To explain the latent factors on a mean level, scalar invariance requires items' factor loadings (M2; weak measurement invariance) and items' intercepts to be equal across groups. Scalar invariance (M3; strong measurement invariance), thus, shows that potential differences of variable means are a result of the influence of common factors. As the scale for duty only reached partial metric invariance, I proceeded to also test partial scalar invariance for this variable. With regard to scalar invariance (M3), the results of application in life, emotional well-being in class, professional qualification, and regulatory styles revealed a drop in fit that was not within the acceptable cut-off limits of CFI (Δ CFI > 0.010). While RMSEA showed an acceptable drop in fit, CFI did not result in a reasonable fit. Further, the resulting model 3 for freedom of action, positive image, and self-development did not yield an acceptable drop of both CFI (Δ CFI 0.010) and RMSEA (Δ RM-SEA 0.015). The results of model 3 for purism of mathematics and duty (partial) fit the data adequately when comparing scalar invariance to metric invariance. Thus, scalar invariance (partial scalar for duty) was supported only for those two, and not for the other seven models.

As indicated earlier (3.3.3), scalar invariance allows us to compare the same constructs across different groups and to interpret the mean differences as attributable to valid cultural differences, thus avoiding measurement bias (Chen, 2008). Here, various parameters might cause the lack of scalar invariance, such as translation inaccuracy and norms of social desirability (Hui & Triandis, 1985) within learners' responses (Chen, 2008). However, scalar invariance, or rather the invariance of item intercepts, is crucial to drawing reliable conclusions with regard to group differences on a mean level.

For this purpose, I have investigated partial invariance (e.g., Byrne et al. 1989; Steenkamp & Baumgartner, 1998) for the item intercepts in all seven models. Again, partial scalar invariance models allow some item intercepts to be invariant and others to be estimated freely. Following an exploratory approach proposed by Byrne et al. (1989) and Cheung & Rensvold (1998), I examined the possibilities of employing partial invariance models. As recommended (cf. "Test of measurement invariance" in section 3.3.3), for every scale I freely estimated one item of each subfactor and constrained all remaining items to display equal intercepts.

Altogether, all nine models yielded an equivalent estimated factor loading structure and supported metric invariance across groups. Thereby, for eight models I found support for full metric invariance and for duty, I found partial metric invariance by allowing one certain factor loading to be freely estimated. The attainment of (partial) metric invariance indicated that each item of the scale loaded to the specified latent factor in a similar way and with equivalent magnitude over groups. Alongside the inspection of equal factor loadings, I tested whether the item intercepts were invariant in both data sets. At least two items of each factor supported full metric and full scalar invariance (Byrne et al., 1989; Steenkamp & Baumgartner, 1998). This guaranteed a meaningful and valid interpretation of group differences for the measured constructs. However, the conclusions on potential group differences should be carefully interpreted, since 11 indicators (DIF items) out of 85 are biased.

5.2 Latent Mean Differences in Personal Meanings and Regulatory Styles

Latent mean differences in personal meanings and regulatory styles. After reaching (partial) scalar invariance, I examined the latent mean structure for the nine factor models over countries. Here, I chose the German sample as the reference group where the factor means were constrained at zero and the latent means for the Finnish group were freely estimated (Kleinke et al., 2017). A positive value in difference thus reflected a higher latent mean value for the Finnish group and a negative value mirrored a higher mean score for the reference group Germany. As asserted by Hancock (2001, pp. 375–377), these differences that emerged between the latent mean values are comparable to Cohen's *d* effect size values where d = .2 (but not 0; Cohen, 1988) can be interpreted as a "small", d = .5 as a "medium", and d = .8 as a "large" effect size (Cohen, 1988, pp. 24–27). Table 13 illustrates the results for the standardized latent mean differences over countries.

Table 13.

man and 1 million statemes.			
Factor	М	SE	р
Personal meaning			
Active practice	+0.75	0.12	<.001
Classroom management	-0.02	0.10	.814
Cognitive challenge	+0.26	0.10	.013
Duty	-0.20	0.09	.026
Emotional-affective relation to the teacher	-0.22	0.09	.020
Examination	+0.21	0.10	.033
EXPERIENCE OF AUTONOMY	+0.00	0.10	.987
EXPERIENCE OF COMPETENCE	-0.04	0.10	.697
EXPERIENCE OF SOCIAL RELATEDNESS	+0.01	0.10	.899
General education	+0.19	0.10	.053
Marks	+0.41	0.11	<.001
Positive image	+0.39	0.11	<.001
Purism of mathematics	+0.06	0.09	.465
Reference to reality	-0.17	0.10	.095
Self-perfection	+0.66	0.11	<.001
Support by the teacher	-0.21	0.09	.026
Vocational precondition	+0.57	0.11	<.001
Regulatory style			
Amotivation	+0.24	0.10	.017
External regulation	+0.28	0.12	.022
Identified regulation	+0.49	0.11	<.001
Intrinsic regulation	+0.13	0.09	.168
Introjected regulation	+0.07	0.09	.427

Latent Mean Differences in Personal Meanings and Regulatory Styles Between German and Finnish Students.

Notes. M = mean difference German – Finnish students.

The comparison of personal meanings and regulatory styles on the latent mean level produced results in which Finnish students were in a stronger position. Regarding personal meanings, significant mean differences between German and Finnish learners were found for five personal meanings (p < .001). The Finnish group perceived a higher personal relevance in learning mathematics through the active practice of mathematics (M = +0.754, p < .001), marks (M = +0.408, p< .001), positive image (M = +0.386, p < .001), self-perfection (M = +0.663, p < .001), and vocational precondition (M = +0.573, p < .001). For classroom management (M = -0.024, p = .814), EXPERIENCE OF AUTONOMY (M = +0.002, p = .987), EXPERIENCE OF COMPETENCE (M = +0.039, p= .697), EXPERIENCE OF SOCIAL RELATEDNESS (M = +0.013, p = .899), and purism of mathematics (M = +0.063, p = .465) German and Finnish learners ascribe similar relevance. I found no significant (p < .001) mean differences concerning the personal meanings of cognitive challenge (M =+0.256, p = .013), duty (M = -0.198, p = .026), emotional-affective relation to the teacher (M = -0.215, p < .020), examination (M = +0.207, p = .033), general education (M = +0.186, p = .053), reference to reality (M = -0.167, p = .095), and support by the teacher (M = -0.208, p = .026).

The mean comparison for regulatory styles in relation to learning mathematics resulted in a significantly positive score for Finnish students regarding identified regulation (M = +0.489, p < .001). Both groups revealed relatively the same amount for introjected regulation (M = +0.074, p = .427), whereas no significant mean differences were yielded with respect to amotivation (M = +0.243, p = .017), external regulation (M = +0.278, p = .022, and intrinsic regulation (M = +0.125, p = .168).

The reported standardized mean differences among German and Finnish students showed substantial differences, with Finnish learners scoring higher latent mean values for meanings related more strongly to the subject of mathematics than German students. These test results were also reflected in that the Finnish group perceived higher identified regulation than German learners. Both groups showed self-determined motivational behaviour, whereas the overall self-determination index³² (SDI) was somewhat higher in Finland (SDI = 1.79) than in Germany (SDI = 1.37). These preferences and the attitude towards mathematics learning might relate to the curriculum setting of the respective culture. In terms of these differences, a discussion with respect to the subject-specific tasks in Germany and Finland might be fruitful (see Chapter 6).

Although partial scalar invariance allowed a fair comparison of the latent factors on a mean level, the results from previous analyses (cf. correlation analyses and measurement invariance) indicated the existence of cultural differences in learners' responses (Chen, 2008). Similar to those results, the significant differences in latent means again showed considerable differences between both groups. It seems that aspects from both countries' educational systems and mathematics curricula contribute to different preferences and perceptions of mathematics. To understand these differences, I provide tentative explanations for these results by referring to both countries' mathematics curricula (see section 6.2). As a consequence of this empirical evidence, the following latent class analysis considered both groups in a separate analysis in order to respect their cultural diversity. That is, for each country I conducted an individual latent class analysis to see which group solution is the best for the respective country. In the following section, I present the results of the latent class analysis that followed an exploratory approach and yielded surprising results.

³² Self-determination index (SDI; Levesque et al., 2004; Vallerand et al., 1992) is calculated as follows: SDI = $(2 \times \text{intrinsic motivation})$ + identified regulation – introjected regulation – $(2 \times \text{external regulation})$. The SDI can therefore summarize self-determined behaviour (positive scores) or controlled motivation (negative values).

5.3 Profiling Classes of Personal Meaning and Their Outcomes in Motivation

Latent class analysis. Since I was interested in the patterns of personal meaning, I used mean values on each of the German and Finnish students for the 17 personal meanings.³³I considered several important goodness-of-fit indices proposed by Geiser (2011), Marsh et al. (2009), and Nylund et al. (2007), for the assessment of the quality of the latent class analysis (LCA) models. Table 14 presents the goodness-of-fit indexes of the conducted LCA in Germany.

 Table 14.

 Fit Indices of Latent Class Analysis Based on the 17 Personal Meanings in the German Sample.

	J					8		\mathbf{T}
	р	LL	AIC	BIC	aBIC	VLMR	BLRT	Entropy
1-Class	34	-4528.309	9124.619	9246.201	9138.404	-	-	-
2-Class	69	-4118.395	8374.790	8621.531	8402.766	.0550	≤.001	.878
3-Class	104	-3954.775	8117.550	8489.449	8159.716	.1840	≤.001	.873
4-Class	139	-3814.390	7906.779	8403.836	7963.136	.2043	≤.001	.887
5-Class	174	(-3720.146)	7788.291	8410.507	7858.838	.3745	≤.001	.904

Note. N = 264; Scaling = scaling factor associated with MLR log likelihood estimates; AIC = Akaike's Information Criterion; BIC = Bayesian Information Criterion; aBIC = sample-size adjusted BIC; VLMR = Vuong–Lo–Mendell–Rubin; BLRT = Bootstrap Likelihood Ratio Test.

Regarding Germany, the fit statistics of the LCA models AIC, BIC, and aBIC dropped as the class amount increased. From four- to five-group solutions, AIC and aBIC only decreased and BIC increased. Based on their simulation study, Nylund et al. (2007) proposed using the BIC index in particular to determine the best class solution and thus the four-class solution was the best (Rost, 2006). The VLMR results were not significant for all models. However, the simulation study by Nylund et al. (2007) additionally recommended considering the bootstrap likelihood ratio test (BLRT) as this is a superior tracer for statistical model comparisons with the VLMR test. Therefore, I additionally consulted BLRT to statistically compare the LCA models. For all models, BLRT was < .001. In all models, entropy was close to 1 and thus reflects an overall high level of classification immunity (Geiser, 2011). The average class assignment probabilities for the most likely class membership were greater than the cut-off mark .80 (Geiser, 2011) for each class on the main diagonal. Further, all solutions had profiles with more than 5% of the cases (Marsh et al., 2009). Aside from that, examination of log likelihood (LL) values reflected a rapid decrease from the two-class solution to the four-class model. For the five-class solution the best LL could not be replicated and the preliminary report (not trustworthy) of LL increased compared to the four-class model. According to the parsimonious principle (Geiser, 2011), the class solution should preferably represent the "new knowledge" by means of a minimal number of classes. Finally, by examining the interpretability (as the most important criterion; Geiser, 2011) of the class solutions, the

³³ I was interested in students' response patterns based on the 17 personal meanings. Therefore, I did not use students' mean values on the dimension level for the class profiling. However, appearance of the eight dimensions of personal meaning and the corresponding motivational outcome were part of the theory-driven analyses (cf. Chapter 7).

interpretation amount increased from the two-class solution, whereas the additional class in the five-group solution did not increase the interpretation value compared to the four-profile solution. Effectively, a class solution that is easy to interpret should have mainly high (here ≥ 1.8) or low (here ≤ 1.2) values, and none from the average range (Geiser, 2011). Taking all of this into account, I identified the four-group solution as the best LCA model for Germany. According to this criterion, all profiles in the four-class solution were easy to interpret.

Following this, I included the regulatory styles (intrinsic, identified, introjected, external motivation, and amotivation) within the four-LCA model as the distal outcomes. As external validation criteria (Geiser, 2011), the regulatory styles (as covariates) also showed theory-compatible connections (see Chapter 7) to the four-class solution when I compared the differences between the profiles with respect to the five regulatory styles. The external validation criteria, using regulatory styles as covariates, and the reflection of the individual class properties among the profiles, provided usefulness and certainty for the identified four-class model (Geiser, 2011). Figure 14 illustrates the identified four profiles of personal meaning and Table 15 reports the characteristics of the profile membership in the regulatory styles in Germany. Additionally, Figure 15 reflects learners' differences in their biological natures (standardized characteristics of the empirical profiles on the regulatory styles) within the empirical individual relevance systems.

For better traceability, I provide a detailed description and interpretation of the profiles' motivational outcomes (the biological regulatory mechanism) in the following chapter (Chapter 7) in detail in order to check both perspectives, biological and social regulations, for coherence. I provide here only the significant differences³⁴ in numerical format without reporting them comprehensively.

³⁴ The differences were the result of Mplus chi-square statistics (Muthén & Muthén, 1998–2017).

Figure 14

Four Profiles of Personal Meaning in German Students.



1		1			
		Differences between			
Regulatory style	Profile 1	Profile 2	Profile 3	Profile 4	Profiles
Intrinsic regulation	0.93 (.05)	1.95 (.08)	1.81 (.09)	0.23 (.06)	2=3>1>4
Identified regulation	1.84 (.08)	2.52 (.08)	1.87 (.10)	1.47 (.14)	2>1=3>4
Introjected regulation	1.24 (.08)	1.87 (.09)	0.90 (.09)	0.50 (.09)	2>1>3>4
External regulation	1.18 (.07)	1.04 (.09)	0.70 (.10)	1.15 (.13)	1=4=2>3
Amotivation	0.94 (.06)	0.37 (.05)	0.68 (.07)	1.46 (.14)	4>1>3>2

Table 15.Relationship Between LCA Membership and Outcome Variables in German Students.

Figure 15



Characteristics of the Latent Classes in the Outcome Variables Regulatory Styles in German Students.

Note. The results were standardized to help in the interpretation of this histogram.

Table	16 reports	the fit index	es of the co	nducted LCAs	in Finland.
1	1010				

Ta	ble	16.
1	010	T O •

Fit Indices of Latent Class Analysis Based on the 17 Personal Meanings in the Finnish Sample.

	р	LL	AIC	BIC	aBIC	VLMR	BLRT	Entropy
1-Class	34	-4396.477	8860.96	8979.72	8871.94	-	-	-
2-Class	69	-3769.548	7677.10	7918.12	7699.40	≤.001	≤.001	.942
3-Class	104	-3576.896	7361.79	7725.07	7395.41	.0367	≤.001	.931
4-Class	139	-3450.538	7179.08	7664.61	7224.00	.3009	≤.001	.930
5-Class	174	(-3370.602)	7089.20	7697.10	7145.44	.2357	≤.001	.949

Note. N = 243; Scaling = scaling factor associated with MLR log likelihood estimates; AIC = Akaike's Information Criterion; BIC = Bayesian Information Criterion; aBIC = sample-size adjusted BIC; VLMR = Vuong–Lo–Mendell–Rubin; BLRT = Bootstrap Likelihood Ratio Test.

Concerning Finnish learners, the goodness-of-fit indexes of the LCA models AIC, BIC, and aBIC dropped as the class numbers rose. From four- to five-group solutions AIC and aBIC only decreased but BIC increased. The VLMR results were significant for the two-class and marginally significant for the three-class model. For the four- and five-class solutions, VLMR resulted in no significant value. Similar to the LCA on the German data set, I consulted BLRT to statistically compare the LCA models. For all models, BLRT was \leq .001. In all models, entropy was close to 1 and thus reflected an overall high level of classification certainty (Geiser, 2011). For each class solution, the average class assignment probabilities for the most likely class membership were greater than the cut-off mark .80 (Geiser, 2011; Rost, 2006). Moreover, almost all class solutions had profiles with more than 5% of the cases, but one class of five-class model was close to the cutoff mark (n = 15 from N = 243). Aside from that, examination of LL values reflected a rapid decrease from the two-class to the four-class solution. For the five-class solution, the best LL could not be replicated. Finally, by examining the interpretability of the class solutions, the interpretation amount increased from the two-class solution, whereas the additional class in the five-group solution did not increase the interpretation value compared to the four-profile solution. Taking everything into account (including the parsimonious principle; Geiser, 2011), I concluded that in the case of Finland, the four-group solution was the optimal LCA model. All the profiles in the fourclass model were well interpretable. Figure 16 illustrates the four profiles of personal meaning. Having ascertained the best class solution for Finland, I included the regulatory styles (intrinsic, identified, introjected, external motivation, and amotivation) as covariates in the final LCA model. Table 17 reports the characteristics of the profile membership on the regulatory styles in Finland. Additionally, Figure 17 reflects learners' differences in terms of their biological natures (standardized characteristics of the empirical profiles on the regulatory styles) within the empirical individual relevance systems.

Similar to the LCA in Germany, as a support for an external validation (Geiser, 2011), the regulatory styles as covariates also showed theory-compatible connection to the four-class solution as I compared the difference between the profiles with respect to the five regulatory styles (see

Chapter 7 for more). This aspect and the reflection of the individual profile characteristics between the four classes highlighted usefulness and certainty for the identified four-class solution in Finland (Geiser, 2011).

Figure 16

Four Profiles of Personal Meaning in Finnish Students.



1		1				
	M (SE)				Differences Betwe	en
Regulatory style	Profile 1	Profile 2	Profile 3	Profile 4	Profiles	
Intrinsic regulation	1.28 (.07)	2.21 (.06)	0.66 (.07)	2.19 (.13)	2=4>1>3	
Identified regulation	2.33 (.06)	2.85 (.03)	1.82 (.10)	2.23 (.14)	2>1=4>3	
Introjected regulation	1.23 (.07)	1.57 (.11)	0.79 (.07)	1.01 (.16)	2>1=4>3	
External regulation	1.19 (.06)	1.07 (.09)	1.26 (.08)	0.84 (.13)	2=1=3>4	
Amotivation	1.02 (.06)	0.48 (.07)	1.48 (.08)	0.63 (.09)	3>1>2=4	

Table 17.Relationship Between LCA Membership and Outcome Variables in Finnish Students.

Figure 17

Characteristics of the Latent Classes in the Outcome Variables Regulatory Styles in Finnish Students.



Note. The results were standardized to help in the interpretation of this histogram.

Contrary to expectations (existing cultural differences may lead to different class solutions), in both countries a four-class solution resulted as the best. The class solution determined by exploration in Germany could be similarly replicated with a new or independent data set from Finland. In line with Geiser (2011), these results underpin the replicability of the four-class solution (Geiser, 2011). A brief overview of the numerical profile characteristics in Germany and Finland reflected certain similarities and differences in the data. To see whether the class profiling resulted in the exact same results in both countries, I compared the profile outputs that scored similar high and low values. To compare the profile outputs, I conducted Mann–Whitney U tests. More detailed analyses of all profiles confirmed that the German and Finnish classes follow similar learning orientations, but they do encompass unique characteristics with respect to their cultural settings (see Chapter 7). To support the upcoming interpretation about the differences between German and Finnish students' mathematics-learning orientations, I statistically identified the central tendency on which the profiles with similar response patterns differ. Although the profiles showed corresponding results across countries, Mann–Whitney U tests revealed specific significant differences.

Mann–Whitney U test. As indicated earlier, I detected four profiles of personal meaning in Germany and Finland that showed similar response patterns. An overview of all eight profiles (based on high and low values of each of the German and Finnish classes) indicated that Profile 1–GER and Profile 1–FIN (Table 18), Profile 2–GER and Profile 2–FIN (Table 19), Profile 3–GER and Profile 4–FIN (Table 20), and Profile 4–GER and Profile 3–FIN (Table 21) had many similarities, but also certain differences. On the basis of Mann–Whitney U tests, I compared these profiles (pairwise) in order to see to what extent these profiles corresponded to each other and whether they showed any significant differences with respect to their preferences (see Tables 18–21). Again, a detailed description and interpretation of all profiles follow in Chapter 7. Now, I provide only a report of the statistically significant (Bonferroni adjusted *p*-value) differences when a kind of personal meaning/regulatory style was realized (\geq 1.8 mean values) in one group but not at all (\leq 1.2 mean values) in the other, or if one group perceived more relevance for the constructed personal meaning or a rather higher value for the regulatory style than the other group.
Table 18.

	М		U	Ζ	р	r
-	GER	FIN	_			
Active practice of mathematics	0.95	1.58	6387.50	7.36	<.001	.55
Classroom management	1.96	2.05	4306.50	1.15	.249	.08
Cognitive challenge	1.21	1.48	4964.00	3.34	<.001	.25
Duty	1.86	1.69	3338.00	-1.71	.087	12
Emotional-affective relation to	2.12	2.02	3328.00	-1.74	.081	13
the teacher						
Examination	1.68	1.85	4438.00	1.55	.121	.11
EXPERIENCE OF AUTONOMY	1.59	1.62	4112.00	0.58	.559	.04
EXPERIENCE OF COMPETENCE	2.27	2.24	3668.00	-0.74	.459	05
EXPERIENCE OF SOCIAL RE-	2.01	2.14	4457.00	1.61	.106	.12
LATEDNESS						
General education	1.65	1.88	4979.50	3.17	.002	.23
Marks	1.97	2.17	4598.50	2.05	.040	.15
Positive image	1.58	1.71	4410.00	1.46	.143	.11
Purism of mathematics	1.07	1.19	4442.00	1.55	.120	.11
Reference to reality	1.74	1.59	3280.50	-1.88	.060	14
Self-perfection	1.23	1.63	5879.00	5.84	<.001	.43
Support by the teacher	2.27	2.08	3061.50	-2.56	.010	19
Vocational precondition	1.60	2.09	5660.50	5.31	<.001	.40
Amotivation	0.94	1.03	4267.50	1.18	.235	.08
External regulation	1.19	1.19	4011.00	0.28	.778	.02
Introjected regulation	1.24	1.24	4065.00	0.44	.659	.03
Identified regulation	1.85	2.34	5343.00	4.29	<.001	.32
Intrinsic regulation	0.93	1.29	5320.00	4.18	<.001	.31

Mann–Whitney U Test Statistics for Mean Difference of Personal Meanings and Regulatory Styles in Profile 1 From Germany and Profile 1 From Finland.

Note. GER = Germany; FIN = Finland.

Profile 1 from Germany versus Profile 1 from Finland. The comparison between the first two profiles indicated that Finnish learners perceived relevance in vocational precondition (M = 2.09, SD = .48), whereas German students did not (M = 1.60, SD = .59), U = 5660.50, p = <.001, r = .40. Regarding types of motivation, Finnish students had higher experience in identified regulation (M = 2.34, SD = .58) than German learners (M = 1.85, SD = .72), U = 5343.00, p = <.001, r = .32.

Table 19.

	М		U	Ζ	р	r
-	GER	FIN	_			
Active practice of mathematics	1.97	2.45	3198.50	4.35	. <.001	.37
Classroom management	2.34	2.44	2528.00	1.31	.190	.11
Cognitive challenge	1.89	2.07	2741.00	2.28	.022	.19
Duty	1.24	0.94	1609.00	-2.81	.005	24
Emotional-affective relation to	2,18	2.36				
the teacher			2652.50	1.87	.061	.16
Examination	2.19	2.43	2771.50	2.43	.015	.20
EXPERIENCE OF AUTONOMY	2.16	2.21	2338.00	0.46	.645	.03
EXPERIENCE OF COMPETENCE	2.62	2.73	2580.00	1.58	.112	.13
EXPERIENCE OF SOCIAL RE-	2.16	2.42				
LATEDNESS			2906.50	3.02	.002	.26
General education	2.46	2.62	2753.00	2.35	.019	.20
Marks	2.49	2.76	2961.50	3.50	<.001	.30
Positive image	1.98	2.27	2924.00	3.098	.002	.26
Purism of mathematics	1.91	2.05	2496.50	1.31	.187	.11
Reference to reality	1.75	1.81	2297.50	0.27	.784	.02
Self-perfection	1.97	2.47	3665.00	6.44	<.001	.55
Support by the teacher	2.37	2.43	2363.00	0.57	.563	.04
Vocational precondition	2.29	2.67	3095.00	3.88	<.001	.33
Amotivation	0.37	0.49	2494.00	1.19	.230	.10
External regulation	1.05	1.07	2278.00	0.18	.852	.01
Introjected regulation	1.87	1.58	1854.00	-1.71	.086	14
Identified regulation	2.52	2.86	2784.00	2.77	.006	.23
Intrinsic regulation	1.95	2.22	2734.50	2.24	.025	.19

Mann–Whitney U Test Statistics for Mean Difference of Personal Meanings and Regulatory Styles in Profile 2 From Germany and Profile 2 From Finland.

Note. GER = Germany; FIN = Finland.

Profile 2 from Germany versus profile 2 from Finland. The Mann–Whitney U test between the two second profiles of both countries showed a significantly greater mean value for Finnish students (M = 2.45, SD = .37) with respect to the personal meaning active practice of mathematics than German learners (M = 1.97, SD = .68), U = 3198.50, p = <.001, r = .37). Moreover, Finnish learners had a higher mean score in marks (M = 2.76, SD = .43) than German learners (M = 2.49, SD = .58), U = 2961.50, p = <.001, r = .30. Further, Profile 2 in Finland had a higher mean value in self-perfection (M = 2.47, SD = .33) than Profile 2 in Germany (M = 1.97, SD = .42), U =3665.00, p = <.001, r = .55). In addition, the Finnish learners perceived more relevance in vocational precondition (M = 2.67, SD = .35) than German learners (M = 2.29, SD = .60), U = 3095.00, p = <.001, r = .33). Table 20.

	М		U	Ζ	р	r
-	GER	FIN	_			
Active practice of mathematics	1.66	2.29	1559.50	4.74	<.001	.48
Classroom management	1.75	1.85	1026.50	.40	.686	.04
Cognitive challenge	1.79	2.35	1459.00	3.92	<.001	.15
Duty	1.19	0.77	615.00	-2.92	.003	30
Emotional-affective relation to	1.53	1.60				
the teacher			1005.50	0.23	.814	.02
Examination	1.48	1.47	939.00	-0.30	.760	.03
EXPERIENCE OF AUTONOMY	1.75	2.08	1303.00	2.66	.008	.27
EXPERIENCE OF COMPETENCE	1.92	2.09	1142.50	1.35	.175	.13
EXPERIENCE OF SOCIAL RE-	1.67	1.59				
LATEDNESS			871.00	-0.85	.392	08
General education	1.90	2.30	1421.50	3.62	<.001	.37
Marks	1.42	1.98	1410.50	3.53	<.001	.36
Positive image	1.09	1.33	1164.00	1.51	.129	.15
Purism of mathematics	1.56	2.00	1481.50	4.25	<.001	.44
Reference to reality	1.41	1.30	911.00	-0.53	.595	05
Self-perfection	1.37	2.11	1755.50	6.33	<.001	.65
Support by the teacher	1.38	1.35	849.00	-1.04	.295	10
Vocational precondition	1.73	2.17	1369.50	3.17	<.001	.32
Amotivation	0.69	0.64	929.50	-0.38	.700	03
External regulation	0.70	0.85	1083.00	0.87	.384	.08
Introjected regulation	0.90	1.02	1014.00	0.30	.76	.03
Identified regulation	1.88	2.23	1258.00	2.30	.021	.23
Intrinsic regulation	1.82	2.19	1324.00	3.14	.002	.32

Mann–Whitney U Test Statistics for Mean Difference of Personal Meanings and Regulatory Styles in Profile 3 From Germany and Profile 4 From Finland.

Note. GER = Germany; FIN = Finland.

Profile 3 from Germany versus Profile 4 from Finland. The third comparison between Profile 3 from Germany and Profile 4 in Finland indicated higher mean values for Finnish learners for several personal meanings. In Profile 4 – FIN, Finnish students perceived relevance in the active practice of mathematics (M = 2.29, SD = .42), but German learners did not see this as relevance (M = 1.37, SD = .44), U = 1755.50, p = <.001, r = .65. For cognitive challenge, Profile 4 – FIN scored a higher mean value (M = 2.35, SD = .50) than class 3 in Germany (M = 1.79, SD = .464), U = 1459.00, p = <.001, r = .15. Additionally, the profile in Finland perceived more relevance in general education (M = 2.30, SD = .42) than German learners did (M = 1.90, SD = .48), U = 1421.50, p = <.001, r = .37). Regarding marks, Finnish learners had a higher mean score in marks (M = 1.98, SD = .83) than German students (M = 1.42, SD = .68), U = 1410.50, p = <.001, r = .36. In Profile 4 – FIN, the learner constructed purism of mathematics (M = 2.00, SD = .44), whereas German student did not (M = 1.56, SD = .51), U = 1481.50, p = <.001, r = .44. Further, Profile 4

in Finland had a higher mean value in self-perfection (M = 2.11, SD = .44) than Profile 3 in Germany (M = 1.38, SD = .44), U = 1755.50, p = <.001, r = .65). In Profile 4 – FIN, learners saw more relevance in vocational precondition (M = 2.17, SD = .55) than German students (M = 1.73, SD = .61), U = 1369.50, p = <.001, r = .32.

Table 21.

Mann–Whitney U Test Statistics for Mean Difference of Personal Meanings and Regulatory Styles in Profile 4 From Germany and Profile 3 From Finland.

	М		U	Ζ	р	r
-	GER	FIN	_			
Active practice of mathematics	0.39	0.95	1870.00	4.30	<.001	.42
Classroom management	1.56	1.55	1288.50	0.26	.794	.02
Cognitive challenge	0.67	1.02	1563.50	2.96	.003	.29
Duty	2.25	1.97	952.50	-2.05	.040	20
Emotional-affective relation to	1.74	1.48				
the teacher			979.50	-1.86	.063	18
Examination	1.22	1.07	1153.00	-0.67	.500	06
EXPERIENCE OF AUTONOMY	0.91	1.15	1609.00	2.47	.013	.24
EXPERIENCE OF COMPETENCE	1.87	1.46	739.00	-3.51	<.001	34
EXPERIENCE OF SOCIAL RE-	1.64	1.41				
LATEDNESS			931.00	-2.20	.028	21
General education	1.12	1.18	1288.00	0.25	.796	.02
Marks	1.35	1.62	1443.50	1.33	.182	.13
Positive image	0.99	1.15	1461.50	1.45	.146	.14
Purism of mathematics	0.51	0.64	1409.00	1.08	.276	.10
Reference to reality	1.10	1.19	1349.0	0.67	.498	.06
Self-perfection	0.67	0.90	1657.00	2.82	.005	.27
Support by the teacher	1,93	1,44	809.00	-3.03	.002	30
Vocational precondition	1.14	1.54	1675.50	2.91	.004	.28
Amotivation	1.47	1.49	1305.00	0.59	.551	.05
External regulation	1.16	1.27	1338.50	0.60	.544	.06
Introjected regulation	0.51	0.79	1536.00	1.99	.046	.19
Identified regulation	1.48	1.82	1542.00	2.01	.044	.19
Intrinsic regulation	0.23	0.66	1710.50	3.49	<.001	.34

Note. GER = Germany; FIN = Finland.

Profile 4 from Germany versus Profile 3 from Finland. Finally, I compared Profile 4 – GER and Profile 3 – FIN. This test indicated a greater mean value for German students for EXPERIENCE OF COMPETENCE (M = 1.87, SD = .69) than for Finnish students (M = 1.46, SD = .51), U = 739, p = <.001, r = -.34.

EMPIRICAL RESULTS

5.4 Summary of the Empirical Results

The assessment of the psychometric properties of both affective constructs produced similar results across countries. In this present investigation, data from both German and Finnish ninth graders showed acceptable to good reliabilities. Compared to previous studies (Vollstedt & Duchhardt, 2019), it was possible to measure classroom management by using an adapted scale (OECD, 2014b, p. 331). Moreover, CFA exhibited reasonable fit statistics supporting the factor validity of regulatory styles. These results emphasized further the robustness of SDT and the validity of the established instrument for regulatory styles (Thomas & Müller, 2011). Regarding personal meaning, the predicted dimensionality of personal meaning held empirically, though only in eight separate measurement models representing the respective dimensions. The estimation of all personal meanings in one comprehensive model (all dimensions within the second-order model) did not fit the data. In both countries, correlation analyses mostly supported the predicted links between the personal meanings and regulatory styles. Moreover, the intercorrelations between constructs varied in both groups. Generally, the Finnish correlation pattern was dominated by intercorrelations between personal meanings and self-determined types of motivation, whereas in Germany relations between meanings and controlled regulations preponderated. Advanced methods of multivariate statistical analysis ensured a reliable/valid interpretation of the nature of the differences in constructs' mean structure by supporting (partial) scalar invariance. In particular, considerable mean differences emerged for highly mathematics-related personal meanings and for identified regulation, where Finnish students scored significantly higher mean values than German learners. Despite remarkable cultural differences, profiling learners' individual relevance systems yielded four distinct profiles of personal meaning for both countries as the best solution. The four-class solutions had similar response patterns over groups, but also resulted in significant differences. The implications of the empirical work are of crucial importance for further coordination analysis. In the following discussion (see Chapter 6), I therefore reflect on the empirical results in more detail.

6. Discussion of the Empirical Results

The results recorded from my empirical work addressed three major research questions (RQs): RQ 1–What are the psychometric properties of the affective constructs personal meaning and regulatory styles across Germany and Finland? RQ 2–What cross-cultural differences exist on the level of personal meanings and regulatory styles in Germany and Finland? RQ 3–How many distinct profiles of personal meaning can be detected in Germany and Finland?

6.1 Psychometric Properties of Personal Meaning and Regulatory Style Across Countries

Initially, I studied the psychometric properties and evaluated the factor structure of both constructs before I identified the profiles of personal meaning. The estimated composite scale reliabilities ρ resulted in acceptable to good values over countries. In the previous study of Vollstedt and Duchhardt (2019), it was not possible to measure "classroom management" in the German data. In this present investigation, items from a modified scale (OECD, 2014b) helped to estimate classroom management so that all 17 meanings could be considered in this study. Even though Germany and Finland are distinct cultural settings, both imported scales (developed originally in the German language) lead to good construct reliabilities. Despite good translation, translated instruments may lead to poor psychometric properties and might not be the same as in the original settings. Therefore, yielding acceptable (to good) psychometric properties is not a matter of course, as shown by, for example, Bofah and Hannula (2015). The reliable Finnish scales can mostly be attributed to a concise and solid translation³⁵ of the German scales into the Finnish language and the revisions proposed in the cognitive lab by Finnish native speakers to avoid problems of operationalization and measurement (Kelle & Buchholtz, 2015).

With regard to construct validity, the factor structure of regulatory styles could be replicated across countries. This result also supported the robustness of SDT as well as of the established instrument that was applied (Thomas & Müller, 2011) to study the biological nature of personal meaning in the sense of convergent validity evidence. Although the instrument of personal meaning is new, the results of this construct's psychometric properties supported its robustness in measurement (Vollstedt & Duchhardt, 2019) and conceptual rationale (Vollstedt, 2011b). Concerning personal meaning, I assessed its dimensionality in separate measurement models. In both countries, I found empirical support for the predicted dimensionality. This in turn supported the theoretical technique of specifying the dimensionality by means of the two meta-factors that refer to the theoretical link between personal meaning and regulatory styles. Thus, this yielded implicit knowledge regarding the associations between the theoretical conceptualization of personal meaning and regulatory styles (SDT). Moreover, the differentiation of the 17 personal

³⁵ Thank you again, Jessica Salminen-Saari and Dr Eeva Haataja, for your very precise and quick translation work. I also wish to thank the Finnish students who helped work out the translation mistakes in the cognitive lab!

meanings justified the further statistical analysis (test of measurement invariance) in a meaningful way.

Nevertheless, measuring all eight dimensions together in one comprehensive model, within the second-order model, consistently yielded a weak fit across countries. As described in the literature (Brown, 2006), the complex higher-order structure recorded poor model fit indices. A detailed inspection of the fit values reflected a modification of the factor structure in order to reduce the complexity of the model and to revise the model's substantive basis, as the factors used did not exhibit a high correlation with each other. Where the sample is small, it is advised that parameter numbers be kept low so as to ensure parameter estimates are stable and the quality of the model is sufficient (Little et al., 2002). Having a moderate sample size, here especially for the second-order model,³⁶ fit statistics resulted in that crucial relations between some factors could not be estimated as the model did not capture the data well. That is to say, in the second-order model it was not possible to record all the important information among variables as the model was too parsimonious (Brown, 2006; Maydeu-Olivares, 2019; Savalei, 2012). These results also bring implications to the fore that refer to the conceptual rationale of personal meaning. A model is as robust as the theoretical basis of the individual factors and their intercorrelations (Brown, 2006, p. 165-166). These results indicated that the estimated model is too parsimonious to capture very important associations between some factors and therefore a revision is necessary to obtain a "clear rationale" for the model (Brown, 2006, p. 166).

These results possibly reflected the fact that the 17 kinds of personal meaning together did not completely account for all theoretically possible facets of the overall construct personal meaning. The previous study by Vollstedt (2011b) reconstructed the 17 personal meanings from data from Germany and Hong Kong. These different kinds refer to those data sets and thus reflect their cultural settings. The low intercorrelations between the 17 kinds may suggest that there are more kinds of personal meaning, that is, other kinds than Vollstedt's personal meanings, around the world that need to be detected to complete the rationale of this motivational construct. If so, future research will have to think about a robust methodological solution to consider all factors within one model.

Whereas the composite construct of reliabilities and the factor structure of constructs showed similar behaviour, the correlation patterns between personal meaning and regulatory styles were different in both groups. The correlation analyses among both constructs resulted in interesting insights into groups. While the correlation pattern in Germany displayed the most relations of personal meanings with types of controlled motivation, personal meanings in Finland had the most relations with regulations of self-determined motivation. In both data sets, from Germany and from Finland, the predicted relations (cf. 2.3.1) among the 17 personal meanings and regulatory styles fitted the data fairly well. In both countries, the kinds of personal meanings did not only correlate with one specific type of motivation. Instead, most of them displayed one specific strongest positive correlation with a certain regulatory style and at the same time further associations with neighbouring regulatory styles. These insights into groups supported the prediction that personal

³⁶ Similar results were indicated for the alternative 17-factor model of personal meaning.

meanings can be perceived in various ways (Deci & Ryan, 1993) but prioritize the one that is compliant with the respective cultural setting. In Finland, I found at least a low positive relation to regulatory styles for all personal meanings, but in Germany I detected no correlation whatsoever (positive or negative) for reference to reality. In a nutshell, the data of Finnish students recorded more stable (moderate or strong) relations among the 17 personal meanings and self-determined types of motivation than did the German data. One reason could be that these results indicate more complex inter-individual differences (Geiser, 2011) in the German data than in the Finnish data. This complexity in response structure might relate to the use of a German sample from different types of schools and federal states versus the Finnish data that comes from comprehensive schools located together in one Finnish region.

In summary, correlation analyses of data from Germany and Finland (West European countries) indicated empirical support for the expected positive linear connection between the motivational construct of personal meaning and quality of motivation. Concerning the dimensionality of personal meaning, the expected relations between the dimensionality of personal meaning and regulations were supported across countries, apart from a few exceptions. These results reflect that next to similarities, fine differences in cultural settings exist. Generally, the consideration of the single personal meanings provided a more detailed insight into constructs' interrelations than associations on the level of dimensions. This aspect further supported the decision to examine learners' individual relevance systems (profiles of personal meaning) as a pattern of the 17 kinds of personal meaning. Altogether, correlation analyses support an empirical link among *socially* experienced personal meanings and *biologically* perceived quality of motivation.

For both affective constructs, I could evaluate the measurement invariance analyses of groups with eight two-group CFA models for personal meaning and one five-factorial two-group CFA for types of motivation. Following an exploratory approach, I found for all two-group models support for (partial) scalar invariance, reflecting that the nature of latent mean differences in constructs can be interpreted. In order to make a meaningful interpretation of cross-cultural dissimilarities in mean structure, partial scalar invariance is essential (Byrne et al., 1989; Steenkamp & Baumgartner 1998). The detected non-invariant items may be caused by different conceptual understandings, small translation mistakes, or cultural preferences (Chen, 2008). In order to construct a fair and valid group comparison, all scales contained at least two indicators having metric and scalar invariance (Byrne et al., 1989; Steenkamp & Baumgartner, 1998). Moreover, the addressed biased items, compared to the total number of items, represented a relatively small amount to affect the mean comparison within the affective constructs (Chen, 2008; Steenkamp & Baumgartner, 1998). In a nutshell, the results from the measurement invariance test, similar to the distinct correlation structure within countries, support the idea that differences exist within countries.

6.2 Nature of Latent Mean Differences in Personal Meaning and Regulatory Style

Cultural differences arose in the mean factor structure of personal meaning and regulatory styles. In general, the differences that indicated that the Finnish students were in a stronger position. In comparison with German learners, Finnish students had higher mean scores for personal meanings that reflected a strong relation to the individual learner perspective (active practice of mathematics and self-perfection). Interestingly, in Finland, these personal meanings correlated highly with intrinsic motivation. These specific preferences might fit with certain aspects of the Finnish mathematics curriculum. Thereby, it seems that the mathematics class is organized in such a manner that learners receive support to develop a positive attitude regarding the subject of mathematics as well as a positive self-conception when dealing with it (Finnish National Board of Education, 2016, p. 402). This consistent focus on these specific attitudes to the subject during mathematics lessons might support Finnish students' construction of mathematics-related personal relevance. Unlike in the Finnish curriculum, there were hardly any detailed statements in the German mathematics core curriculum (KMK, 2005) on how to support individual learners' personal development (particularly through transversal competencies) in mathematics. Instead, the German educational standards (including content-related mathematical competencies and central mathematical ideas) strongly highlighted various mathematical competences.

Moreover, and significantly, Finnish learners placed more importance on marks, positive image, and vocational precondition than German students. Since in Finland students have to take a decision in the ninth grade about their upper secondary education (OECD, 2020a), choosing between academic or vocational tracks, the importance of mathematics and beneficial marks in mathematics as a major subject seem to matter. In Germany, "good" grades are usually the crucial factor in school reports in the tenth grade when most students transfer to further education tracks (OECD, 2020b). These differences in educational systems might explain the distinct importance in the relevance of mathematics for German and Finnish ninth graders. The latent cross-cultural mean structure further indicated that learners from both countries assign the same priority for classroom management, EXPERIENCE OF AUTONOMY, EXPERIENCE OF COMPETENCE, EXPERIENCE OF SOCIAL RELATEDNESS, and purism of mathematics. Surprisingly, all three specific personal meanings,³⁷ which seem to have a connection with basic psychological needs, have equivalent importance (no differences emerged on a latent mean level) independent of cultural setting.

Regarding personal meaning, the resulting differences showed similarities to the findings of a previous study based on *item response theory* (IRT), partial credit models, and differential item functioning (DIF) analysis (Suriakumaran et al., 2019). However, the resulting differences in the former IRT-based study and the present investigation showed few differences in the mean structure. A statistically significant (according to the Benjamini–Hochberg procedure on a 5% false discovery rate) difference in cognitive challenge could not be found in the present study, while positive image resulted in a difference (p < .001) that did not become apparent in the IRT-based study. Although both methodological approaches, *classical test theory* (CTT) and IRT, result

³⁷ The three specific personal meanings are EXPERIENCE OF AUTONOMY, COMPETENCE, and SOCIAL RELATEDNESS.

in comparable outputs (Brown, 2006; Meade & Lautenschlager, 2004), CTT and IRT models have differences within their estimation framework, to name but two: item parameterization (Macintosh & Hashim, 2003; Muthén et al., 1991) and standard error of measurement (Brown, 2006; De Ayala, 2009). Regarding model estimations, CTT aims to clarify the interrelations between items, whereas IRT seeks to model the relationship between latent trait and item responses (Brown, 2006; De Ayala, 2009; Hambleton et al., 1991). Although the exact same instrument of personal meaning was applied to the same data set, the effects of the measurement theories are visible. Since a detailed comparison of these measurement theories would go beyond the scope of this thesis, I have limited the explanations given thus far.

Concerning the regulatory styles, Finnish students scored a higher mean value for identified regulation than German learners. On the one hand, this fact might also go along with the circumstances (as mentioned above) that Finnish learners face in grade nine; namely the upcoming graduation and the transition to an academic or vocational track. On the other hand, the personal meanings (active practice of mathematics, self-perfection, marks, and vocational precondition) favoured by Finnish students reflected, in a previous analysis (cf. correlation analyses), strong positive correlations with self-determined regulatory styles (intrinsic and identified regulation). In general, these results and the fact that the Finnish learners scored a higher SDI than German students show that Finnish students perceive the motivation in mathematics to be less external. Interestingly, these empirical results underline previous PISA results in terms of mathematics anxiety. The PISA results highlighted that Finnish students had among the lowest anxiety levels, and in terms of the association between mathematics anxiety and performance, Finnish students had less anxiety than German students (OECD, 2015, p. 2). Moreover, the change between 2003 and 2012 in the mean index of intrinsic motivation to study mathematics increased in Finland over those years, whereas in Germany it decreased. The further coordination analyses of this study will show whether these findings remain consistent within countries' sub-groups (i.e., profiles found in Germany and Finland; see Chapter 7).

6.3 Identification of Profiles of Personal Meaning and Their Motivational Outcomes

The previous statistical analyses exhibited cross-cultural differences in correlation patterns, in latent mean structure, and in the test of measurement invariance. With respect to both groups' cultural diversity, I conducted two separate latent class analyses for each group to identify their best class solution. Accordingly, the scales of personal meaning provide a solid foundation to empirically map German and Finnish individuals to profiles of personal meaning by means of latent class analysis. Surprisingly, the comparison, followed by an exploratory procedure, of a one-to five-class solution resulted in a four-class model for both groups as the best.

Within the empirical individual relevance systems, certain factors from learners' biological natures are reflected. We assume this must be the case, because otherwise the empirical profiles would develop in the same way and have the same results and there would only be one profile. Since there are very different profiles, from a variety of classes, we can assume that students in the classes have completely different profiles as well, and this fits the picture of a biological regulatory

mechanism. The four profiles detected in Germany and Finland recorded similar response patterns, and the comparison (Mann–Whitney U tests) revealed fine but significant differences in profiles of personal meaning and their motivational outcome. Most differences in nuance refer to the latent mean structure of the respective country. Potentially, the upcoming coordination analysis method could clarify to what extent these subtle differences have an effect with regard to the interaction of learners' social and biological regulatory mechanisms (see Chapter 7). I will now examine to what extent these profiles are found in the classroom and what this means for the classroom in Germany and Finland. The upcoming analyses are a step towards supporting this present dissertation's theoretical work in that I seek to strengthen the empirical profiles (i.e., by clarifying the social regulatory mechanism).

7. Results According to the Coordination Analysis Method

In this chapter, I present the theory-driven analyses according to coordinating (conceptual framework built by coordinating theories) biological (SDT) and social regulatory mechanisms (social constructivism) through the empirical phenomenon personal meaning. As mentioned earlier, this study's empirical work only captured one ontological essence of personal meaning, its individual nature. To capture its second ontological nature, personal meaning as a social phenomenon, as well as to evaluate the validity of the cyclical model concerning the social regulatory mechanism of the individual relevance system posited earlier (cf. Figure 6), it was necessary to draw hypothetical conclusions in order to examine the theoretical assumptions. Therefore, I used the empirically developed data. The data consist of the empirical profiles of personal meaning (individual relevance systems) from Germany and Finland, their corresponding regulatory styles, and the self-determination index (SDI) predicting learners' relative autonomy.

In sections 7.1 and 7.2, I report the results of the coordination analyses based on the empirical data from Germany and Finland. Subsequently, in section 7.4, I interpret the coordination of the results using the hypothesized conceptual framework (cf. Figure 7); this was constructed by linking SDT and social constructivism by means of the boundary object personal meaning, in order to propose a theoretical model explaining the genesis of motivation in mathematics. As an initial step to this end, I had three (further) research questions (RQs) that I examined successively when theorizing the coordination of the results.

Several issues were considered in order to address the interaction between individual relevance systems and motivation (RQ 4). The adapted model of the social regulatory mechanism considers the role of the three specific personal meanings (EXPERIENCE OF AUTONOMY, COMPETENCE, and SOCIAL RELATEDNESS) to describe how these individual relevance systems (patterns of personal meaning) arise and unfold in mathematics classrooms. Regarding the predicted connection between social constructivism and personal meaning. I assumed that the application of this adapted cycle would provide a meaningful tool with which to interpret the empirical profiles in light of the social regulations experienced by learners in mathematics lessons. Theoretically, the learner constructs a complex pattern of personal meanings regarding mathematics as a school subject (Vollstedt, 2011b; Vorhölter, 2009) that is personally relevant for them. Accordingly, I predicted that learners rarely construct all the factors of a dimension of personal meaning (cf. Figure 5) at the same time. Moreover, I expected to find indicators for the predicted inner feedback mechanism (cf. Figure 4), that is, theoretical connections between the three specific meanings and the basic psychological needs. Accordingly, I hypothesized that the effects of the social regulatory mechanism (learners' individual relevance systems) would be reflected in the respective biological regulatory mechanism (motivational outcome) of both countries in terms of theoretical coherence.

In the case of RQ 5, at first glance, the profiles from Germany and Finland have similar relevance systems. Yet, empirical results (Mann–Whitney U tests) indicated significant differences between them. Since learners' social regulations are also determined by culture, I expected to find cultural differences that became visible through theoretical conclusions (Kleemann et al., 2013), that is, the work of interpreting the numerical profiles through the predicted conceptual framework.

Regarding this study's basic assumption (RQ 6), I expected to find evidence for the predicted theoretical connections between the boundary object of personal meaning and the two theoretical perspectives in order to explain the genesis of motivation in mathematics classroom.

For the coordination analysis procedure, I followed the steps described in section 4.2.2. Each profile's results subsection thus effectively consists of three steps – step 1: capture the learner's reaction to the outside social world in mathematics class in terms of the social regulatory mechanism; step 2: construct a hypothesis on the complementary motivational behaviour (biological regulation), and step 3: analyse the learner's inner world with reference to biological regulatory mechanism.

The *publication* stage (the starting point of the coordination analysis in step 1) of the social regulatory mechanism represents the learner's individual relevance system. To understand the learner's profile output, I initially summarized the constructed meanings with respect to the strong indicators represented by the constructed personal meanings ($M \ge 1.8$ cut-off point) and nonconstructed personal meanings ($M \le 1.2$ cut-off point). In some profiles, I included personal meanings with thresholds that are close to the cut-off values and with a tendency towards (non-) construction (e.g., M = 1.71 tendency towards construction and M = 1.29 tendency towards nonconstruction). Where possible, I described the profile summary by means of the eight dimensions of personal meaning. This was for the purpose of examining whether the predicted dimensions (dis)appear in the same way (conjoint or divergent for instance as single factors) and to see whether certain dimensions are characteristic of German and/or Finnish profiles' learning orientations. Based on this publication stage (profile summary), I reconstructed the learner's individually owned orientation pattern with respect to the processes of transformation and appropriation of the social regulatory mechanism. Through the reconstruction of these processes of social regulations, I aimed to get closer to student's social experiences and in particular their perception of the three specific personal meanings (EXPERIENCE OF AUTONOMY, COMPETENCE, AND SOCIAL RELATEDNESS). Again, this theoretical reconstruction of social experiences only focuses on learners' individual responses (ticked boxes on the Likert scale), and thus these insights only apply to the individual's perception of classroom interaction underlying their produced individual relevance system (and not an impartial perspective on the mathematics-classroom situation in terms of objective hermeneutics; Kleemann et al., 2013).

Furthermore, based on their profile's orientation pattern, I subsequently hypothesized the direction of the learner's motivation (step 2). To provide the biological nature of personal meaning, in step 3 I described the learner's motivation in terms of the (non-)existing regulatory styles (same cut-off points as for personal meanings) and the SDI (motivational tendency in the mathematics classroom).

In the following sections, I report in more detail on the theory-driven analyses found in the German (7.1) and Finnish data (7.2) according to the coordination analysis method.

7.1 Mathematics-Learning Orientations in Germany

In this section, the four profiles found in Germany are described using the coordination analysis method. At the end of this section, the learning orientations and motivation of the four identified profiles are compared as regards their similarities and differences.

7.1.1 Profile 1: Emotional-Social Integration with a Focus on Seeking Acceptance from the Teacher

The following figure, Figure 18, provides an overview of Profile 1, found in Germany, with respect to the strong indicators represented by the constructed ($M \ge 1.8$ cut-off point) and non-constructed ($M \le 1.2$ cut-off point) dimensions/kinds of personal meaning. This profile consisted of 34% of the sample (N = 264), of which 52% of the learners were female.

Figure 18





Note. Summarized overview of profile's pattern of personal meaning and the corresponding motivational behaviour. Profile 1 in Germany is depicted through the constructed and non-constructed kinds of personal meaning embedded within the corresponding dimension of personal meaning. The colour code (light) green is used for (a tendency towards) constructed personal meanings and (light) red is used for (a tendency towards) non-constructed personal meanings. Each dimension is coloured either green or red when all factors of the dimension behaved in a similar manner. Average values are greyed out. The colour code of the profile's frame reflects the profile's SDI, with orange for self-determined motivation, purple for controlled motivation, and grey when the SDI is balanced, i.e., neither positive nor negative.

Step 1. The publication stage (as the starting point of the coordination analysis) shows the learner's individual relevance system as having a mathematics-learning orientation that is focused on emotional-social integration in class. Referring to the learner's responses, I identified dimensions of personal meaning and single personal meanings that appeared as strong indicators with regard to construction: EXPERIENCE OF COMPETENCE and marks, together with two dimensions of personal meaning: emotional well-being in class and duty. I additionally considered reference to reality as part of the learner's relevance system, as this pointed to a tendency for construction

(M = 1.74, SD = .60). The dimension purism of mathematics appeared as a strong indicator for non-construction. The dimension of self-development arose partly as a strong indicator with regard to non-construction (self-perfection (M = 1.23, SD = .37), and cognitive challenge (M = 1.21, SD = .39) showed only a tendency, but the active practice of mathematics emerged as a strong indicator for non-construction). According to the results gathered about the publication stage, two out of the five multifaceted dimensions of personal meaning behaved in a similar manner. However, to understand the orientation pattern of this learner, I further examined the items behind the (non-)constructed personal meanings.

Based on strong indicators that manifested at the publication stage, it would appear that the learner studies mathematics for social reasons. Mathematics represents a compulsory subject (duty) in which they work for the teacher (dimension: emotional well-being in class). The teacher offers this emotional-social environment and allows the learner to experience competence (EXPERIENCE OF COMPETENCE) and make an effort in order to be rewarded with good grades (marks). These situations strengthen the learner's self-confidence. Further, the learner appreciates working on mathematics content that relates to everyday life (reference to reality). As a nonconstructed personal meaning, the dimension purism of mathematics and self-development showed that the relevance of mathematics as a subject matter (German: Sachgebiet) and the prospect of personal development through mathematics did not play a role for the particular learner. Here, a profile occurred where mathematics as a socially mediated scope becomes important. Based on this relevance system and their social experiences in class, the individual constructed two out of the three specific meanings, namely EXPERIENCE OF COMPETENCE and EXPERIENCE OF SOCIAL RELATEDNESS. Thus, they did not construct EXPERIENCE OF AUTONOMY in social interaction. To conclude, the reconstruction of transformation and appropriation of social regulation showed that the learner *identifies with* social processes that focus on a teacher who lets them feel competent, whereas they do not identify with individual processes in mathematics. In this profile, the learner's biographical orientation thus moves between these two opposing poles, which mutually represent opposing scopes for one another. However, in this orientation pattern they did not perceive EXPERIENCE OF AUTONOMY when learning mathematics. Based on the learner's publication, I reconstructed the levels of transformation and appropriation. Then, I considered this orientation pattern as an effect of social interaction in class, to investigate the biological regulation that could correspond to the social experiences of such a person.

Step 2. On the basis of this specific perception of social interaction in class, the learner's sense of success in terms of action in the classroom seemed to depend on their teacher. In particular, they did not perceive EXPERIENCE OF AUTONOMY in mathematics; that is, the learner did not experience themselves as the centre of action. Therefore, I formulated the following hypothesis in view of their motivational behaviour in mathematics class: If the learner publishes a relevance system whereby they socially experience their mathematics class through social processes, then their motivation tends to be controlled.

Step 3. The motivational regulation (as *enacting potential*) of this learner showed a high level of identified regulation. The other regulatory styles were low (intrinsic regulation, introjected

regulation, external regulation, and amotivation). Surprisingly, the SDI (0.09) tends towards selfdetermined motivation, so that there is a slight tendency towards self-determined motivation. However, this result is essentially balanced, that is, neither controlled nor self-determined. Here, my hypothesis (step 2) did not fit the data.

Summary of Profile 1. This relevance system represents the highest percentage of students (34%) out of the four profiles identified in Germany. Here, the difference in gender is balanced³⁸ $(\bigcirc: 52\%, \bigcirc: 48\%)$. This profile contains mostly average students (60 students had either grade 3 or 4), many high achievers (27 students had either grade 1 or 2), and fewer low achievers (which grade it was, 5 or 6). As a result of social interaction in class, the learner constructs a relevance system of personal meaning that basically focuses on emotional-social integration. The teacher is specifically important for their learning processes and enables them to EXPERIENCE COMPETENCE through their social involvement, even though they perceive this as a compulsory subject. In this context, they experience social integration. Here, mathematics is perceived as a socially mediated scope in which they study the subject less for individual than for social reasons. To conclude, this mathematics-learning orientation focusing on emotional-social integration scored a high level of identified regulation and showed motivational behaviour that was balanced, SDI = 0.09 (biological regulation), with a slight tendency towards self-determined motivation. This SDI is neither positive nor negative (for comparison only, the other German profiles had SDI values between -0.89 and 3.22). Since the learner's quality of motivation is well balanced, even though the specific meaning EXPERIENCE OF AUTONOMY was not constructed, a clarification of this biological experience is important (see section 7.4). Based on the profile orientation and the characteristic dimension of personal meaning (emotional well-being in class) that also emerged in a similar Finnish profile (but with a slightly different specification; see Profile 1–FIN in section 7.2.1), I termed this profile "Emotional-social integration with a focus on seeking acceptance from the teacher."

7.1.2 Profile 2: Enjoyment in Mathematics Learning

The following figure, Figure 19, provides an overview of Profile 2 found in Germany with respect to the strong indicators represented by the constructed ($M \ge 1.8$ cut-off point) and non-constructed ($M \le 1.2$ cut-off point) dimensions/kinds of personal meaning. This profile consisted of 27% of the sample (N = 264), of which 41% of the learners were female.

³⁸ Gender difference was not statistically tested.

Figure 19

Profile 2 in Germany: Enjoyment in Mathematics Learning.



Note. Summarized overview of profile's pattern of personal meaning and the corresponding motivational behaviour. Profile 2 in Germany is depicted through the constructed and non-constructed kinds of personal meaning embedded within the corresponding dimension of personal meaning. The colour code (light) green is used for (a tendency towards) constructed personal meanings and (light) red is used for (a tendency towards) non-constructed personal meanings. Each dimension is coloured either green or red when all factors of the dimension behaved in a similar manner. Average values are greyed out. The colour code of the profile's frame reflects the profile's SDI, with orange for self-determined motivation, purple for controlled motivation, and grey when the SDI is balanced, i.e., neither positive nor negative.

Step 1. The learner produced an individual relevance system with a mathematics-learning orientation that focused on enjoyment. Regarding the profile summary (*publication*), I identified seven out of the eight dimensions of personal meaning that appeared conjoint as strong indicators with regard to construction: purism of mathematics, freedom of action, self-development, professional qualification, emotional well-being in class, and positive image. The dimension of application in life also appeared conjoint, but reference to reality showed a tendency for construction (M = 1.75, SD = .69). Only duty was not constructed and showed a tendency for non-

construction (M = 1.24, SD = .94). On the basis of this profile summary, I sought to understand the learner's orientation framework. To capture their perceptions through social interaction, I reconstructed the transformation and appropriation processes of social regulations in mathematics class.

Based on strong evidence in the publication stage, the learner seemed to study mathematics for individual as well as for social reasons. The learner experienced mathematics as a socially mediated scope (dimensions: professional qualification, application in life, emotional well-being in class) in which individual development is possible (dimensions: self-development). The learner coped with learning processes independently and could use their skills without outside support (dimension: freedom of action). They worked with the teacher, and thereby mathematics was perceived as a school subject and at the same time as a scientific field (purism of mathematics). It was also important to them to get appreciation for their achievements in mathematics (positive image). Duty was a non-constructed personal meaning, which showed that the learner did not perceive mathematics class, the learner perceived all three specific meanings, namely EXPERIENCE OF AUTONOMY, COMPETENCE, and SOCIAL RELATEDNESS. With respect to the processes of *transformation* and *appropriation*, relevance as well as fun were *identified with* mathematics class, whereas duty was *not*. From these social experiences, the learner's biographical orientation pattern thus moves between these two opposing poles.

Step 2. Following the reconstructed social experiences in Step 1, the learner seemed to perceive themselves as self-directed and as the centre of action in mathematics lessons but did not connect classroom interaction with external pressure. Since they experienced all three specific meanings, I hypothesized this corresponding motivational behaviour: If the learner publishes a relevance system whereby they socially experienced their mathematics class as fun and as relevant, then their motivation tends to be self-determined.

Step 3. When considering the biological regulatory mechanism as enacting potential, the learner's motivation resulted in a high level of identified, intrinsic, and introjected regulation. The other types of regulation scored low mean values (external regulation and amotivation). As assumed, the SDI was positive, SDI = 2.45; that indicated self-determined motivation. Here the hypothesis (step 2) fits the data.

Summary of Profile 2. This relevance system is the second highest percentage of students out of the four profiles identified in Germany (27%). Here, the number of females is somewhat lower than the amount of males (\bigcirc : 41%, \bigcirc : 59%). This profile attracted mostly high achievers (41 students had either grade 1 or 2), many average students (29 students had either grade 3 or 4), and fewer low achievers (which grade it was, 5 or 6). Based on the social interactions in mathematics class, the learner constructed a relevance system of personal meaning that focused on individual as well as on social processes. The support of the teacher was not perceived as an external pressure, but as a multifaceted support. The students constructed EXPERIENCE OF AUTONOMY, COMPETENCE, and SOCIAL RELATEDNESS through their social regulations. Here, mathematics is experienced as a socially mediated scope that offers a learning environment in

RESULTS ACCORDING TO THE COORDINATION ANALYSIS METHOD

which to deal with individual processes. This mathematics-learning orientation associated with a self-determined motivational behaviour, SDI = 2.45 (the other profiles in Germany had SDI values between -0.89 and 3.22) that scored a high level of identified, intrinsic, and introjected regulation. On the basis of the learner's orientation pattern and the characteristic dimensions of personal meaning (application in life, emotional well-being in class, freedom of action, positive image, purism of mathematics, professional qualification, self-development) that emerged in the same manner in the Finnish profile (see Profile 2–FIN in section 7.2.2), I termed this profile "Enjoyment in mathematics learning".

7.1.3 Profile 3: Self-Improvement with a Focus on Mastery

The following figure, Figure 20, provides an overview of Profile 3 found in Germany with respect to the strong indicators represented by the constructed ($M \ge 1.8$ cut-off point) and non-constructed ($M \le 1.2$ cut-off point) dimensions/kinds of personal meaning. This profile consisted of 24% of the sample (N = 264), of which 40% of the learners were female.

Figure 20



Profile 3 in Germany: Self-Improvement With a Focus on Mastery.

Note. Summarized overview of profile's pattern of personal meaning and the corresponding motivational behaviour. Profile 3 in Germany is depicted through the constructed and non-constructed kinds of personal meaning embedded within the corresponding dimension of personal meaning. The colour code (light) green is used for (a tendency towards) constructed personal meanings and (light) red is used for (a tendency towards) non-constructed personal meanings. Each dimension is coloured either green or red when all factors of the dimension behaved in a similar manner. Average values are greyed out. The colour code of the profile's frame reflects the profile's SDI, with orange for self-determined motivation, purple for controlled motivation, and grey when the SDI is balanced, i.e., neither positive nor negative.

Step 1. The publication stage showed the learner's individual relevance system as having a mathematics-learning orientation focused on self-improvement. Regarding the learner's responses, I identified only one dimension of personal meaning (freedom of action) that appeared conjointly, but EXPERIENCE OF AUTONOMY showed a tendency for construction (M = 1.75, SD = .54). In addition, general education as a single personal meaning resulted as a strong indicator with regard to construction. Further, cognitive challenge (M = 1.79, SD = .64), classroom management (M = 1.75, SD = .64), and vocational preconditions (M = 1.73, SD = .61) showed a tendency for

construction. The dimensions of duty and positive image appeared as strong indicators with respect to non-construction. Hence, one out of the five multifaceted dimensions of personal meaning arose conjointly, even though one factor of this dimension was not a strong indicator. Next, to understand the orientation framework of this learner, I evaluated the items behind the (non-)constructed personal meanings.

Based on the personal meanings constructed at the publication stage, the student learns mathematics for individual reasons. The learner experiences mathematics as a scope in which to develop their own cognitive skills (general education) and seek cognitive challenges (cognitive challenge). This is necessary to fulfil their own competence requirements (EXPERIENCE OF COMPETENCE) and to have better chances in their professional life (vocational preconditions). Further, they cope with learning processes independently and can use their skills with no support from outside (dimension: freedom of action). The teacher ensures a good working environment is provided (classroom management). In terms of duty and positive image as non-constructed personal meanings, it was shown that the learner did not perceive the subject of mathematics as an obligation, nor did they study mathematics to please others. It was not important that the learner did what others asked them to do or received social appreciation for what they did in order to receive feedback from significant others. This was not relevant to them. Based on this specific relevance orientation and their social experiences in class, the learner experienced two specific meanings, namely EXPERIENCE OF AUTONOMY and COMPETENCE. Regarding the regulations of transformation and appropriation, the learner in this profile identified with mathematics class to fulfil their own competency requirements and to seek cognitive challenges-that is, they identified with the statement "Mathematics fulfils my competence requirements", whereas they did not identify with "I try my best in learning mathematics to please others". From these social experiences, the learner's biographical orientation pattern thus moves between these two contrary scopes, which mutually represent opposing poles for one another. In the next step, I made the following assumption, which may correspond to this orientation pattern in terms of the learner's motivational regulation.

Step 2. Following the reconstructed social experiences in Step 1, the learner seemed to perceive mathematics lessons as a means of self-improvement in order to fulfil their competence requirements and to feel cognitively challenged. Here, they experienced two specific meanings (dimension: freedom of action). Accordingly, the individual did not act for external reasons (like social appraisal) but valued experiencing themselves as a centre of action. Therefore, I hypothesized this corresponding motivational behaviour: If the learner publishes a relevance system where they socially experience fulfilment of their competency requirements, then their motivation tends to be self-determined.

Step 3. When considering the biological regulatory mechanism as *enacting potential*, the learner's motivation resulted in a high level of identified and intrinsic regulation. The other regulatory styles scored low mean values (introjected regulation, external regulation, and amotivation). As assumed, the SDI was positive, SDI = 3.22, which indicated self-determined motivation. Here the hypothesis (step 2) fits the data.

RESULTS ACCORDING TO THE COORDINATION ANALYSIS METHOD

Summary of Profile 3. This relevance system is the third highest percentage of students out of the four identified relevance systems in Germany (24%). Here, the percentage of females is lower than that of males (\bigcirc : 40%, \bigcirc : 59%). This profile attracted mostly high achievers (35 students had either grade 1 or 2) and many average students (27 students had either grade 3 or 4). Based on the social interactions in mathematics class, the learner constructed a relevance system that focused on individual processes. There was no dependency on the teacher, nor was a feeling of communality in class relevant. The teacher did not play such a big role in this profile. The teacher's support was important with respect to classroom management. The teacher gave the learners freedom to act and learn on their own. In this context, the learner constructed EXPERIENCES OF COMPETENCE and AUTONOMY and fulfilled their own competence requirements. Mathematics also had relevance for their future profession and allowed them to experience cognitive challenges. Mathematics was not perceived as irrelevant subject, nor was it studied to please others. This orientation (as a result of the social regulations) resulted in the highest self-determined motivational behaviour (SDI = 3.22; the other profiles in Germany had SDI values between -0.89 and 2.45) within Germany. Based on this specific relevance orientation in mathematics and the characteristic dimension of personal meaning (freedom of action) that also emerged in the similar Finnish profile (but with a different specification, see Profile 4–FIN in section 7.2.4), I termed this profile "Self-improvement with a focus on mastery".

7.1.4 Profile 4: External Pressure Focused on Adaption to the Teacher's Performance Requirements

The following figure, Figure 21, provides an overview of Profile 4, found in Germany, with respect to the strong indicators represented by the constructed ($M \ge 1.8$ cut-off point) and non-constructed ($M \le 1.2$ cut-off point) dimensions/kinds of personal meaning. This profile consisted of 15% of the sample (N = 264), of which 54% of the learners were female.

Figure 21

Profile 4 in Germany: External Pressure Focused on Adaption to the Teacher's Performance Requirements.



Note. Summarized overview of profile's pattern of personal meaning and the corresponding motivational behaviour. Profile 4 in Germany is depicted through the constructed and non-constructed kinds of personal meaning embedded within the corresponding dimension of personal meaning. The colour code (light) green is used for (a tendency towards) constructed personal meanings and (light) red is used for (a tendency towards) non-constructed personal meanings. Each dimension is coloured either green or red when all factors of the dimension behaved in a similar manner. Average values are greyed out. The colour code of the profile's frame reflects the profile's SDI, with orange for self-determined motivation, purple for controlled motivation, and grey when the SDI is balanced, i.e., neither positive nor negative.

Step 1. The publication stage of the social regulatory mechanism showed the learner's individual relevance system to have a mathematics-learning orientation that focused on external pressure. In this relevance system, the dimension of duty and the single meaning EXPERIENCE OF COMPETENCE were the strong indicators with regard to construction. The dimension of emotional well-being in class partly occurred, whereas the personal meaning of support by the teacher was constructed as a strong indicator, and the emotional-affective relation to the teacher was a threshold for construction (M=1.74, SD=.71). The dimensions of purism of mathematics, self-development,

application in life, positive image, and the single meaning EXPERIENCE OF AUTONOMY arose as strong indicators with regard to non-construction. The dimension of professional qualification partly emerged, whereas examination (M = 1.22, SD = .82) showed a tendency towards non-construction. At the publication stage, two out of the five multifaceted dimensions of personal meaning showed similar characteristics. Based on this profile summary, I approached the learner's orientation framework to reconstruct their social experience in mathematics class, that is, their transformation and appropriation level of social regulation.

Based on the constructed personal meanings that the individual published, the learner experienced mathematics through social processes. They experienced mathematics as an exclusively compulsory subject (dimension: duty) and the individual adapted to competence requirements (EXPERIENCE OF COMPETENCE) from the teacher (support by the teacher and emotional-affective relation to the teacher). In this learning environment, the individual was not involved in any individual processes supporting their individual development (non-construction of the dimension: self-development). They did not experience themselves as a centre of action (nonconstruction of EXPERIENCE OF AUTONOMY) or appreciate making a positive impression on others because of their achievement in mathematics (non-construction of positive image) while learning. Mathematics was neither perceived as an important school subject (non-construction of the dimension application in life, and the personal meanings examination and vocational precondition), nor as a specialist field (non-construction of purism of mathematics). Based on this specific perception of social interactions in class, the learner only experienced one specific meaning out of three; namely, EXPERIENCE OF COMPETENCE. In terms of the regulations of transformation and appropriation, the learner identified with mathematics as an adaption to the teacher's performance requirements, whereas they did not identify with mathematics as being interesting as an academic subject and as a specialist field. On the basis of these social experiences, the learner's biographical orientation framework thus exists between these two opposing poles. Next, I formulated the following assumption that may correspond with this orientation pattern in terms of the learner's motivational regulation in mathematics class.

Step 2. Based on the socially experienced relevance system in Step 1, the learner seemed to be dependent on the teacher and experienced mathematics lessons as an adaption to educational performance requirements. Here, they perceived only EXPERIENCE OF COMPETENCE as a specific meaning and did not construct EXPERIENCE OF AUTONOMY. Hence, I made the following assumption with respect to the learner's motivational behaviour: If the learner publishes a relevance system whereby they socially experience their mathematics class as an adaption to competency requirements from the teacher, then their motivation is tends to be controlled.

Step 3. When considering the biological regulatory mechanism as *enacting potential*, the learner's motivation did not result in any high mean values for regulatory styles. Identified regulation and amotivation scored moderate mean values and the others showed low mean values (intrinsic regulation, introjected regulation, and external regulation). However, the SDI was negative, SDI= -0.89, indicating a less controlled motivation. Here the hypothesis (step 2) fits the data.

Summary of Profile 4. This relevance system accounts for the smallest percentage of students out of the four identified profiles in Germany (15%). Here, the percentage of females is higher than that of males (\bigcirc : 54%, \bigcirc : 44%). This profile contained mostly average students (28 students had either grade 3 or 4), several high achievers (eight students had either grade 1 or 2), and fewer low achievers (four students had grade 5 or 6). Based on the social interactions in the mathematics class, the learner constructed a relevance system of personal meaning that focused on duty and on academic performance requirements. This adaption and the EXPERIENCE OF COMPETENCE was mediated by the teacher. The learner did not EXPERIENCE AUTONOMY and SOCIAL RELATEDNESS through their social interactions. Here, mathematics was experienced as a socially mediated compulsory subject that was purely an obligation to them. This mathematics-learning orientation (as a result of social regulations) resulted in less controlled behaviour (SDI = -0.89; the other German profiles had SDI values between 0.09 and 3.22) with moderate levels of identified regulation and amotivation. Based on this specific relevance orientation pattern in mathematics and the characteristic dimension of personal meaning (duty), which also emerged in the similar profile from Finland (but with a different specification, see Profile 3-FIN in section 7.2.3), I termed this profile "External pressure focusing on adaption to the teacher's performance requirements".

7.1.5 Comparison of German Learners' Mathematics-Learning Orientations

The identified profiles in Germany showed differences with respect to their mathematics-learning orientations. In all four German profiles, learners constructed the personal meaning EXPERIENCE OF COMPETENCE.

Two out of the four profiles constructed relevance systems that referred to social processes in mathematics class (Profile 1 and Profile 4). In Profile 1 and in Profile 4, from the individual's point of view the teacher had an essential and dominant role. Teacher-mediated situations in class were where learners EXPERIENCED COMPETENCE and emotional-social support, but at the same time they perceived mathematics learning as an obligation. The EXPERIENCE OF COMPETENCE appeared in both profiles, but EXPERIENCE OF SOCIAL RELATEDNESS only arose in Profile 1. In both relevance systems, the learner did not publish EXPERIENCE OF AUTONOMY. Here, only Profile 1 scored a high value in identified regulation, while Profile 4 showed moderate values for amotivation and identified regulation. Here, the motivational regulations resulted in a balanced (Profile 1, SDI = 0.09) and a less controlled motivation (Profile 4, SDI = -0.89).

On the contrary, Profile 2 perceived mathematics lessons through social and individual processes, and Profile 3 studied mathematics for individual reasons. In both profiles, the teacher took care of classroom management, whereas in Profile 2 the teacher additionally seemed to mediate multifaceted roles. Both individual relevance systems showed high values in intrinsic and identified regulation, and in Profile 2 introjected regulation was also represented. Profile 2 (SDI = 2.45) experienced all three specific meanings, whereas Profile 3 (SDI = 3.22) perceived EXPERIENCE OF AUTONOMY and EXPERIENCE OF COMPETENCE. These two types of orientation pattern showed self-determined motivation.

RESULTS ACCORDING TO THE COORDINATION ANALYSIS METHOD

Altogether, the inferred step, interpreting the numerical data into meaningful profiles, could be supported through the coordination analysis assisted by the three adjusted terminologies. In a nutshell, for Profile 1 the adapted model social regulatory mechanism of individual relevance system could not thoroughly explain the predicted motivational outcome. Although the learner perceived duty and did not construct EXPERIENCE OF AUTONOMY, this Profile 1 showed a tendency for self-determined regulated behaviour. It seems that these results highlighted certain crucial factors that shaped all profiles in view of their nature; these were all facets of the German educational culture. The (non-) construction of the specific meanings and their respective individual relevance patterns seem to stay in a specific relation to their corresponding motivational outcomes. To clarify these issues, as a last step, an interpretation of these results with regard to the educational system and the predicted conceptual framework would be fruitful, in particular to explore the interesting effects that emerged in Profile 1 (see section 7.4).

7.2 Mathematics-Learning Orientations in Finland

In this section, the four profiles found in Finland are described by using the coordination analysis method. At the end of this section, the four identified profiles' learning orientations and motivation are compared in terms of their similarities and differences.

7.2.1 Profile 1: Emotional-Social Integration Focused on Own Professional Qualification and Seeking Acceptance from the Teacher

The following figure 22 provides an overview of Profile 1 found in Finland with respect to the strong indicators represented by the constructed ($M \ge 1.8$ cut-off point) and non-constructed ($M \le 1.2$ cut-off point) dimensions/kinds of personal meaning. This profile consisted of 36% of the sample (N = 243), of which 45% of the learners were female.

Figure 22



Profile 1 in Finland: Emotional-Social Integration Focused on Own Professional Qualification and Seeking Acceptance From the Teacher.

Note. Summarized overview of profile's pattern of personal meaning and the corresponding motivational behaviour. Profile 1 in Finland is depicted through the constructed and non-constructed kinds of personal meaning embedded within the corresponding dimension of personal meaning. The colour code (light) green is used for (a tendency towards) constructed personal meanings and (light) red is used for (a tendency towards) non-constructed personal meanings. Each dimension is coloured either green or red when all factors of the dimension behaved in a similar manner. Average values are greyed out. The colour code of the profile's frame reflects the profile's SDI, with orange for self-determined motivation, purple for controlled motivation, and grey when the SDI is balanced, i.e., neither positive nor negative.

Step 1. The publication stage (as the starting point of the coordination analysis) shows the learner's individual relevance system as having a mathematics-learning orientation that is focused on emotional-social integration. Referring to the learner's responses, I identified two dimensions of personal meaning (emotional well-being in class and professional qualification) and single personal meanings (EXPERIENCE OF COMPETENCE, and general education) that appeared as strong indicators with regard to construction. I additionally considered positive image in a learner's

relevance system, as this showed a tendency for construction (M = 1.71, SD = .07). Purism of mathematics appeared as a strong indicator with regard to non-construction. According to the results gathered about the publication stage, two out of the five multifaceted dimensions of personal meaning appeared conjoint. However, to understand the orientation pattern of this learner, I examined the items behind the (non-) constructed personal meanings.

Based on the constructed personal meanings in the publication stage, the learner seemed to study mathematics for highly social reasons. Mathematics as a school subject was important to them with regard to their own future education (dimension: professional qualification and general education). Thereby the teacher was central to their learning process (dimension: emotional wellbeing in class). Further, it was important to them that they received appreciation and felt proud of their achievements in mathematics (EXPERIENCE OF COMPETENCE and positive image). As a nonconstructed personal meaning, the dimension of purism of mathematics showed that the personal relevance of mathematics as a specialist field did not play a role for them. Based on this relevance system and their social experiences in class, the individual constructed two out of the three specific meanings, namely EXPERIENCE OF COMPETENCE and SOCIAL RELATEDNESS. Thus, they did not construct EXPERIENCE OF AUTONOMY in the classroom. To conclude, the reconstruction of transformation and appropriation of social regulation showed that the learner *identified with* social processes that focused on their further education and the teacher. However, they did not identify with mathematics as a specialist field. In this profile, the learner's biographical orientation thus moved between these two opposing poles, which mutually represent opposing scopes for one another. Based on the learner's publication, I reconstructed the levels of transformation and appropriation. Further, I considered this orientation pattern as a result of social interaction in class, in order to make the following assumption about the biological regulation that corresponded to these social experiences in mathematics class.

Step 2. On the basis of this specific perception of social interaction in class, the learner seemed to experience the teacher's support in terms of emotional-social support and vocational preparation. In particular, they did not perceive EXPERIENCE OF AUTONOMY in the mathematics classroom and therefore did not perceive themselves as a centre of action. Here, their sense of the success of their actions depends on their teacher. Therefore, I formulated the following hypothesis with respect to motivational regulation: If the learner published a relevance system whereby they socially experienced their mathematics class as emotionally-socially supportive for their occupational qualification, then their motivation was somewhat controlled.

Step 3. In terms of *enacting potential*, the motivational regulation of this learner showed a high level of identified regulation. The other regulatory styles were low (intrinsic regulation, introjected regulation, external regulation, and amotivation). Surprisingly, the SDI was clearly positive, SDI = 1.30, which indicated self-determined motivation. Here my hypothesis (step 2, similar to Profile 1 in Germany) did not fit the data.

Summary of Profile 1. This relevance system represents the highest percentage of students (36%) out of the four identified profiles in Finland. Here, the difference³⁹ in gender is balanced

³⁹ Gender difference was not statistically tested.

(9:45%, 3:49%). The students in this profile were high achievers (64 students had grades 10 or 8) and average students (18 students had grades 7 or 6). As a result of social interactions in class, the learner constructed a relevance system of personal meaning that basically focused on emotional well-being in class and on their own professional qualification. Specifically, the teacher was crucial to their learning process. They EXPERIENCED COMPETENCE through their social involvement and a supportive social environment. In this context, they experienced social integration. Here, mathematics was experienced as a socially mediated scope and as an important school subject (not as a compulsory subject) for the learner's occupational career. The social processes were more important than the individual processes, and therefore the learner adapted to the social ones. In a nutshell, this mathematics-learning orientation (as a result of the social regulation) focusing on emotional-social relations resulted in a self-determined motivational behaviour where SDI = 1.30(biological regulation; the other Finnish profiles had SDI values between -0.19 and 3.89) and that scored high level of identified regulation. Hence, the learner's quality of motivation was autonomous even though the specific meaning EXPERIENCE OF AUTONOMY was not constructed. This surprising result needs to be clarified with respect to the classroom setting in Finland. Based on this specific relevance orientation pattern in mathematics and the characteristic dimension of personal meaning (emotional well-being in class) that also emerged in the similar profile from Germany (but with a different specification, see Profile 1–GER in section 7.1.1), I termed this profile "Emotional-social integration focused on own professional qualification and seeking acceptance from the teacher".

7.2.2 Profile 2: Enjoyment in Mathematics Learning

The following figure, Figure 23, provides an overview of Profile 2 found in Finland with respect to the strong indicators represented by the constructed ($M \ge 1.8$ cut-off point) and non-constructed ($M \le 1.2$ cut-off point) dimensions/kinds of personal meaning. This profile consisted of 26% of the sample (N = 243), of which 63% of the learners were female.

Figure 23

Profile 2 in Finland: Enjoyment in Mathematics Learning.



Note. Summarized overview of profile's pattern of personal meaning and the corresponding motivational behaviour. Profile 2 in Finland is depicted through the constructed and non-constructed kinds of personal meaning embedded within the corresponding dimension of personal meaning. The colour code (light) green is used for (a tendency towards) constructed personal meanings and (light) red is used for (a tendency towards) non-constructed personal meaning. Each dimension is coloured either green or red when all the factors of the dimension behaved in a similar manner. Average values are greyed out. The colour code of the profile's frame reflects the profile's SDI, with orange for self-determined motivation, purple for controlled motivation, and grey when the SDI is balanced, i.e., neither positive nor negative.

Step 1. The learner produced (*publication*) an individual relevance system showing a mathematics-learning orientation that focused on enjoyment. To understand the learner's profile output, I initially summarized the appeared meanings with respect to construction and non-construction. In this profile, I identified seven out of the eight dimensions of personal meaning that emerged as strong indicators with regard to construction: purism of mathematics, freedom of action, self-development, professional qualification, application in life, emotional well-being in class, and positive image. Only duty occurred as a strong indicator with regard to non-construction.

Next, on the basis of this profile summary I intended to understand the learner's orientation framework. To capture their perception of social interaction, I reconstructed the transformation and appropriation levels of social regulation.

Based on the constructed personal meanings in the publication stage, the learner seemed to study mathematics for individual as well as for social reasons. The learner experienced mathematics as a socially mediated (dimensions: professional qualification, application in life, emotional well-being in class) scope in which individual development was possible (dimensions: self-development). The learner coped with the learning process independently and could use their skills without outside support (dimension: freedom of action). They worked with the teacher, and thereby mathematics was perceived as a school subject and at the same time as a specialist field (purism of mathematics). They also enjoyed feeling pride in their achievements in mathematics (positive image). Duty, as a non-constructed personal meaning, showed that the learner did not perceive mathematics-learning situations as an obligation. Based on this specific relevance orientation and their social experiences in class, the learner experienced all three specific meanings, namely EXPERIENCE OF AUTONOMY, COMPETENCE, and SOCIAL RELATEDNESS. In terms of the processes of *transformation* and *appropriation*, the learner *identified with* the association of the mathematics class with fun and relevance, whereas they *did not identify* with the statement that mathematics lessons are a duty. Responding to these social experiences, the learner's orientation pattern thus moves between these two opposing poles. Further, I made the following assumption, which may correspond to this orientation pattern in terms of the learner's motivational regulation.

Step 2. Following the reconstructed social experiences in Step 1, the learner seemed to perceive mathematics lessons as enjoyment and not as pressure. Here, they experienced all three specific meanings and perceived the mathematics lessons as self-directed and themselves as a centre of action without external pressure. Therefore, I hypothesized the following corresponding motivational behaviour: If the learner publishes a relevance system whereby they socially experience their mathematics class as fun and relevant, then their motivation is tends to be self-determined.

Step 3. When considering the biological regulatory mechanism as *enacting potential*, the learner's motivation resulted in high levels of identified regulation and intrinsic regulation. The other types of regulation scored moderate (introjected regulation) to low mean values (external regulation and amotivation). As assumed, the SDI was positive, SDI = 3.58, which indicated self-determined motivation. Here the hypothesis (step 2) fit the data.

Summary of Profile 2. This relevance system has the second highest percentage out of the four identified profiles in Finland (26%). Here, the percentage of females is nearly twice that of males (\bigcirc : 63%, \bigcirc : 33%). This profile attracted mostly high achievers (61 students had either grade 10 or 8) and fewer average students (which grade it was, 7 or 6). Based on the social interaction in mathematics class, the learner constructed a relevance system of personal meaning that focused on individual and social processes. The teacher's support was not perceived as a control, but as a multifaceted support. The learner constructed EXPERIENCE OF AUTONOMY, COMPETENCE, and SOCIAL RELATEDNESS through their social interactions. Here, mathematics was experienced as a

RESULTS ACCORDING TO THE COORDINATION ANALYSIS METHOD

socially mediated scope in parallel to the experience of the environment as supportive in terms of dealing with individual processes in mathematics. This mathematics-learning orientation (as a result of the social regulation) is one of enjoyment for the students and resulted in a self-determined motivational behaviour (SDI = 3.58; the other Finnish profiles had SDI values between -0.19 and 3.89) that scored highly in identified regulation and intrinsic regulation. On the basis of the learner's orientation pattern and the characteristic dimensions of personal meaning (application in life, emotional well-being in class, freedom of action, positive image, purism of mathematics, professional qualification, self-development) that also emerged similarly in the profile from Germany (see Profile 2–GER in section 7.1.2), I termed this profile "Enjoyment in mathematics learning".

7.2.3 Profile 3: External Pressure Focusing on Adaption to the Educational System

The following figure, Figure 24, provides an overview of Profile 3 found in Finland with respect to the strong indicators represented by the constructed ($M \ge 1.8$ cut-off point) and non-constructed ($M \le 1.2$ cut-off point) dimensions/kinds of personal meaning. This profile consisted of 25% of the sample (N = 243), of which 43% of the learners were female.

RESULTS ACCORDING TO THE COORDINATION ANALYSIS METHOD

Figure 24



Profile 3 in Finland: External Pressure Focused on Adaption to the Educational System.

Note. Summarized overview of profile's pattern of personal meaning and the corresponding motivational behaviour. Profile 3 in Finland is depicted through the constructed and non-constructed kinds of personal meaning embedded within the corresponding dimension of personal meaning. The colour code (light) green is used for (a tendency towards) constructed personal meanings and (light) red is used for (a tendency towards) non-constructed personal meaning. Each dimension is coloured either green or red when all the factors of the dimension behaved in a similar manner. Average values are greyed out. The colour code of the profile's frame reflects the profile's SDI, with orange for self-determined motivation, purple for controlled motivation, and grey when the SDI is balanced, i.e., neither positive nor negative.

Step 1. The publication stage of the social regulatory mechanism showed the learner's individual relevance system as having a mathematics-learning orientation that was focused on external pressure. In this profile output, the personal meanings that emerged with respect to construction were fewer in number than meanings relating to non-construction. The dimension duty was the only strong indicator with regard to construction. The dimensions of purism of mathematics, self-development, application in life, positive image, and the single meanings EXPERIENCE OF AUTONOMY and examination appeared as strong indicators with regard to non-

construction. Based on this profile summary, I approached the learner's orientation framework in order to reconstruct their social experiences in the mathematics classroom, that is, their *transformation* and *appropriation* level of social regulatory mechanism.

Based on the constructed personal meanings that the individual published, the learner dealt with mathematics in class through social processes. They studied this subject out of obligation and the fear of sanctions. They experienced mathematics as an exclusively compulsory subject (dimension: duty) in which individual development was not possible (non-construction of dimension: self-development). They did not experience themselves as a centre of action (nonconstruction of EXPERIENCE OF AUTONOMY) or feel proud of their achievements in mathematics (non-construction of positive image) while learning. Mathematics was neither perceived as an important school subject (non-construction of the dimension application in life and examination) nor as a specialist field (non-construction of the dimension purism of mathematics). Surprisingly, there was no clear attitude towards the role of the teacher here, but the learner rejected learning mathematics. Based on this specific relevance orientation and their social experiences in class, the learner experienced none of the three specific meanings, namely EXPERIENCE OF AUTONOMY, COMPETENCE, and SOCIAL RELATEDNESS. Here, the learner *identified with* mathematics as a social adaption to the educational system, whereas they did not identify with mathematics lessons as interesting as an academic subject and as a specialist field. From these social experiences, the learner's orientation framework thus exists between these two opposing poles that mutually represent opposing scopes for one another. Further, I made the following assumption that may correspond to this orientation pattern in terms of the learner's motivational regulation.

Step 2. Based on the socially experienced relevance system in Step 1, the learner seemed to perceive mathematics lessons in the context of the pressure provided by the Finnish educational system. They experienced none of the three specific meanings and thus did not experience themselves as a centre of action or as involved in any individual or social processes (apart from duty). Therefore, I made the following assumption with respect to the learner's motivational behaviour: If the learner published a relevance system whereby they socially experienced their mathematics class as an adaption to the educational system, then their motivation tends to be controlled.

Step 3. When considering the biological regulatory mechanism as *enacting potential*, the learner's motivation showed a high level of identified regulation. The other regulatory styles scored moderate (amotivation) to low mean values (intrinsic regulation, introjected regulation, external regulation). Interestingly, the self-determination index (SDI) showed only a negative tendency, SDI = -0.19 of controlled motivation, but the value was not strong. Here my hypothesis (step 2) yielded an ambivalent result that needs to be interpreted with respect to the Finnish educational culture.

Summary of Profile 3. This relevance system has the third highest percentage of students out of the four identified profiles in Finland (25%). Here, the percentage of females is lower than that of males (\bigcirc : 43%, \circlearrowleft : 54%). This profile attracted mostly high achievers (39 students had either grade 10 or 8), several average students (16 students had either grade 7 or 6), and fewer low

RESULTS ACCORDING TO THE COORDINATION ANALYSIS METHOD

achievers (which grade it was, 5 or 4). Based on the social interaction in the mathematics class, the learner constructed a relevance system of personal meaning with a focus on obligation. The role of the teacher was neither perceived as support nor as control. The learner did not experience any specific personal meaning through their social interactions. Here, mathematics was experienced as a socially mediated scope that was purely an obligation to them. This mathematics-learning orientation (as a result of the social regulation) resulted in balanced motivational behaviour with a tendency towards controlled motivational behaviour; SDI = -0.19. This is neither positive nor negative (for comparison only, the other Finnish profiles had SDI values between 1.30 and 3.89, which scored a high level of identified regulation. This surprising result needs to be clarified in the context of the mathematics curricula in Finland. Based on this specific relevance orientation pattern in mathematics and the characteristic dimension of personal meaning (duty) that also emerged in the similar profile from Germany (but with a different specification, see Profile 4–GER in section 7.1.4), I termed this profile "External pressure focusing on adaption to the educational system".

7.2.4 Profile 4: Self-Improvement with a Focus on the Individual

The following figure, Figure 25, provides an overview of Profile 4 found in Finland with respect to the strong indicators represented by the constructed ($M \ge 1.8$ cut-off point) and non-constructed ($M \le 1.2$ cut-off point) dimensions/kinds of personal meaning. This profile consisted of 13% of the sample (N = 243), of which 39% of learners were female.
Figure 25



Profile 4 in Finland: Self-Improvement With a Focus on the Individual.

Note. Summarized overview of profile's pattern of personal meaning and the corresponding motivational behaviour. Profile 4 in Finland is depicted through the constructed and non-constructed kinds of personal meaning embedded within the corresponding dimension of personal meaning. The colour code (light) green is used for (a tendency towards) constructed personal meanings and (light) red is used for (a tendency towards) non-constructed personal meaning. Each dimension is coloured either green or red when all the factors of the dimension behaved in a similar manner. Average values are greyed out. The colour code of the profile's frame reflects the profile's SDI, with orange for self-determined motivation, purple for controlled motivation, and grey when the SDI is balanced, i.e., neither positive nor negative.

Step 1. The publication level of social regulation showed the learner's individual relevance system to have a mathematics-learning orientation focused on self-improvement. In this profile output, I identified the dimension purism of mathematics and the multifaceted ones (self-development, freedom of action) that appeared conjoint, partial (professional qualification), and single personal meanings (classroom management and general education) that resulted as strong indicators with regard to construction. The dimension of duty appeared as a strong indicator for non-construction. Hence, two out of the five multifaceted dimensions of personal meaning

appeared conjoint. Next, to reconstruct the learner's perceptions in class, I evaluated the items behind the (non-) constructed personal meanings.

Based on the constructed personal meanings at the publication stage, the learner studied mathematics for individual reasons. The learner experienced mathematics as a scope in which to develop their personal cognitive skills (dimension: self-development and general education) and seek self-improving challenges in mathematics. It was also seen as important to give them better opportunities in their professional life (vocational precondition and marks). Further, they coped with learning processes independently and could use their skills without support from the outside (dimension: freedom of action). The teacher ensured a good working environment (classroom management) to support their individual work, which also supported their perception of mathematics as a specialist field (purism of mathematics). Duty was a non-constructed dimension of personal meaning, which showed that the learner did not perceive mathematics-learning situations as an obligation. Based on this specific relevance orientation and their social experiences in class, the learner experienced two specific meanings, namely EXPERIENCE OF AUTONOMY and COMPETENCE. Regarding the regulations of transformation and appropriation, the learner identified with the statement that mathematics class was a "self-improving challenge", whereas they did not identify with the statement that mathematics was a "non-relevant discipline". From this social experience, the learner's biographical orientation pattern thus moves between these two opposing poles. In the next step, I made the following assumption that may correspond to this orientation pattern in terms of the learner's motivational regulation.

Step 2. Following the reconstructed social experiences in Step 1, the learner seemed to perceive mathematics lessons as supporting their own personal self-improvement and did not perceive the subject as a non-relevant discipline. Here, they experienced two specific meanings that showed that their success of action was independent of the teacher and that by focusing on individual processes, they perceived themselves as a centre of action. Therefore, I hypothesized this corresponding motivational behaviour: If the learner publishes a relevance system whereby they socially experience individual self-improvement in class, then their motivation tends to be self-determined.

Step 3. When considering the biological regulatory mechanism as *enacting potential*, the learner's motivation showed a high level of identified and intrinsic regulation. The other regulatory styles scored low mean values (introjected regulation, external regulation and amotivation). As assumed, the SDI was positive, SDI = 3.89, which indicated self-determined motivation. Here the hypothesis (step 2) fits the data.

Summary of Profile 4. This relevance system has the lowest percentage of students out of the four identified profiles in Finland (13%). Here, the percentage of females is somewhat lower than that of males (\bigcirc : 12%, \bigcirc : 17%). This profile attracted mostly high achievers (29 students had either grade 10 or 8) and fewer average students (one student had either grade 7 or 6). Based on the social interactions in mathematics class, the learner constructed a relevance system of personal meaning that focused on individual processes. There was neither a dependency on the teacher, nor a sense of community feeling in class. The teacher did not play a central role, but their support was

RESULTS ACCORDING TO THE COORDINATION ANALYSIS METHOD

important with respect to classroom management. The teacher left the learner free to act and they learnt for themselves. In this context, they constructed EXPERIENCE OF COMPETENCE and EXPERIENCE OF AUTONOMY. They sought self-improving challenges and experienced them in mathematics lessons. Therefore, mathematics was not perceived as a non-relevant subject. This mathematics-learning orientation (as a result of the social regulation) resulted in a self-determined motivational behaviour (SDI = 3.89; the Finnish profiles had SDI values between -0.19 and 3.58) that scored highly in identified regulation and intrinsic regulation. Based on this specific relevance orientation in mathematics and the characteristic dimension of personal meaning (freedom of action) that also emerged in the similar profile from Germany (but with a different specification, see Profile 3–GER in section 7.1.3), I termed this profile "Self-improvement with a focus on the individual".

7.2.5 Comparison of Finnish Learners' Mathematics-Learning Orientations

The identified profiles in Finland showed differences with respect to their mathematics-learning orientations. Two out of the four profiles were relevance systems that referred more to social processes in the mathematics class (Profile 1 and Profile 3). From the individual's point of view, the teacher and one's own professional qualification had an essential role in Profile 1. Teacher-mediated situations in class whereby the learner EXPERIENCED COMPETENCE and emotional-social integration were important. In Profile 3, the learner only participated in mathematics class to fulfil an obligation. The construction of EXPERIENCE OF COMPETENCE and OF EXPERIENCE OF SOCIAL RELATEDNESS resulted in Profile 1, whereas in Profile 3 none of the three specific meanings were perceived by the student. In both relevance systems, learners did not publish EXPERIENCE OF AUTONOMY. Here, Profiles 1 and 3 scored highly in identified regulation. Both profile 1 was self-determined, and the quality of motivation in Profile 3 showed only a tendency towards controlled motivation.

On the contrary, Profile 2 perceived mathematics lessons through social and individual processes, and Profile 4 was geared towards mathematics learning for individual reasons. In Profile 2 the teacher combined multifaceted roles, and in Profile 3 the teacher helped the learner to realize mathematics as a specialist field. Profile 2 experienced all three specific meanings, whereas Profile 4 perceived EXPERIENCE OF AUTONOMY and EXPERIENCE OF COMPETENCE. These two types of orientation pattern showed self-determined motivation. Both individual relevance systems showed high values in intrinsic and identified regulation.

In all four profiles, the students did not show clear controlled behaviour; only Profile 3 scored a tendency towards controlled motivation. In a nutshell, for two (Profile 1 and Profile 3) out of four cases the adapted model social regulatory mechanism of the individual relevance system could not thoroughly explain the hypothesized motivational outcome. Even though the learner did not construct EXPERIENCE OF AUTONOMY, Profile 1 showed self-determined regulated behaviour. Surprisingly, Profile 3, which only constructed duty, but none of the three specific meanings, resulted only in a tendency towards controlled behaviour in class (I expected the

RESULTS ACCORDING TO THE COORDINATION ANALYSIS METHOD

controlled behaviour to be more in evidence). The non-construction of the specific personal meanings and the respective individual relevance system showed no convincing connection to their corresponding motivational outcome in Profile 3. It seems that next to the three specific meanings, the mathematics curricula with the respective cultural settings that underpin them play an essential role in the social regulatory mechanism. Based on these results, the adapted model of social regulatory mechanism needs to include additional factors to clarify the social regulatory mechanism holistically. In particular, Profiles 1 and 3 seem to encompass interesting references that relate to a specifically Finnish classroom environment.

The results according to the coordination analysis of German and Finnish students made clear that the profile patterns from Germany and Finland that emerged had similar mathematicslearning orientations, but with some cultural variations. It seems that the subtle differences between the social regulatory mechanisms in classrooms not only play an essential role, but also result in significant differences in learners' motivational behaviour.

7.3 Summary of the Results According to Coordination Analysis

Based on the indicators found in the German and Finnish empirical data, personal meaning as a boundary object seems to have the characteristics to coordinate SDT and social constructivism. From the perspective of social constructivism, all the identified empirical profiles across countries could be interpreted as meaningful and valid empirical results by means of the adapted model, the social regulatory mechanism of the individual relevance system. In both countries, four different individual relevance systems could be identified that explain German and Finnish students' learning orientations towards emotional-social integration (characterized by the dimension: emotional well-being in class); enjoyment in mathematics learning (representative dimensions: all seven dimensions apart from duty); self-improvement (typical dimension: freedom of action); and external pressure (characterized by the dimension: duty) in mathematics classroom. The identified profiles showed similar patterns with differences in nuance, whereas the motivational outcomes varied across countries.

According to the findings, in both countries the profiles with emotional-social integration were most valued where learners focused more on social processes accompanied by the construction of EXPERIENCE OF COMPETENCE and SOCIAL RELATEDNESS. Whereas the German profile showed a more balanced motivation with a tendency towards self-determination, the Finnish relevance system scored a clearly positive SDI. The profiles of enjoyment in learning mathematics in which learners constructed all three specific meanings existed independently of culture, and, in both countries, showed strong self-determined motivation. In both groups, the learning orientations towards self-improvement with a strong focus on individual processes resulted in self-determined motivation when the learner constructed EXPERIENCE OF AUTONOMY and EXPERIENCE OF COMPETENCE. The mathematics-learning orientations towards external pressure reflected a tendency towards controlled motivation (Profile 3) in Finland and towards a less controlled motivation (Profile 4) in Germany. It became apparent that in particular the construction of EXPERIENCE OF AUTONOMY plays a central role within the three specific meanings.

The results from Finland, especially Profile 3 (external pressure focusing on adaption to the educational system), came up with surprising results. The socially experienced relevance system was not entirely reflected in the motivational outcome.

Although German and Finnish learners showed similar learning orientations in mathematics, their profile patterns seem to be specific to each country. It seems that next to the three specific meanings, the mathematics curricula with the respective cultural settings that underpin them play an essential role in the social regulatory mechanism. For this reason, additional mediators of social regulations exist that seem to affect the motivational outcome. These mediators should be added to the adapted model (social regulatory mechanism of the individual relevance system) in order to strengthen and balance the conceptual framework when coordinating SDT and social constructivism through the boundary object of personal meaning.

7.4 Theorizing the Results of the Coordination

The research activities considered in this dissertation are sequential-dependent (Schoonenboom & Johnson, 2017). For this reason, the coordination procedure, preceded by the quantitative procedure, should appropriately build on the previous empirical results. Based on the empirical results discussed in Chapter 6, I reported the results according to the coordination analysis (cf. 7.1 and 7.2). In the following sections, I interpret these results with respect to the main issue with which this dissertation is concerned; namely, the genesis of learners' motivation in the mathematics classroom.

To theorize the coordination of the results, I targeted three (further) RQs: RQ 4–How do the individual relevance systems and motivation interact in Germany and Finland, when viewed from a perspective coordinating SDT and social constructivism? RQ 5–How does the interaction found in German and Finnish data differ in view of learners' mathematics curricula? RQ 6–How does the conceptual framework built by coordinating SDT and social constructivism by means of the boundary object personal meaning explain the development of motivation in mathematics classroom?

By focusing the coordinating procedure on these questions, I reflect on the conceptual framework built by coordinating SDT and social constructivism in which personal meaning is conceptualized as situated at the boundary of both theoretical lenses (see section 7.4.1). This step intends to examine how far both theoretical lenses reflect one another. The theory-driven analyses emphasized that each country's relevance system has cultural tags that refer to their respective cultural setting. Therefore, I also contrasted the results from Germany (GER) and Finland (FIN) by referring to learners' cultural settings in the classroom. Through this step, I refined the interplay between theories and clarified surprising results whereby the adopted theoretical lenses were limited to explaining the coordination biological and social regulatory mechanism (see section 7.4.2). To support the interpretation procedure in view of comparing results from Germany and Finland, certain results (cf. 5.1–5.3) from quantitative statistical analyses are combined in section 7.4.2. Finally, I present the evidence from this study's coordination procedure in the shape of a theoretical model explaining the development of motivation by means of the linkage between

biological and social regulatory mechanisms, which seem to converge within the boundary object of personal meaning (see section 7.4.3).

7.4.1 Interpreting the Interaction Between Biological and Social Regulatory Mechanisms

The theory-driven analyses based on German and on Finnish empirical data helped to describe how learners' individual relevance systems (the content as *what*, described by the perspective of social constructivism) arise and unfold in mathematics classroom as a counterpart of the empirical evidence produced by the individuals and their motivational behaviour (information processing as *why*, explained through the perspective of SDT). Coordinating theories provided the potential to examine the empirical phenomenon—profiles of personal meaning—from two different theoretical perspectives, as well as to consider two complementary epistemic views as well-fitting elements (Prediger et al., 2008, p. 172; Sabena et al., 2014). This was done as a means of improving our understanding of how learners' quality of motivation develops in the mathematics classroom. The use of two different lenses also made it easier to mutually validate the conceptual framework and to interpret data from two different cultural settings, through theoretical coherence and theoretical contradiction in terms of "what and why".

In both countries, the empirical profiles of personal meaning could be separated into four kinds of mathematics-learning orientation, that is: emotional-social integration (Profile 1–GER and Profile 1–FIN); enjoyment in mathematics learning (Profile 2–GER and Profile 2–FIN); self-improvement (Profile 3–GER and Profile 4–GER); and external pressure (Profile 4–GER and Profile 3–FIN). In the following, I theorize the interaction between the biological and social regulatory mechanisms by means of both theoretical lenses, SDT and social constructivism.

Emotional-Social Integration. Through the repeated social experiences in their social environment, learners were facilitated in the classroom to construct the specific personal meanings EXPERIENCE OF COMPETENCE and EXPERIENCE OF SOCIAL RELATEDNESS. Based on these social experiences they produced learning orientations across countries that focused on emotional-social integration (Profile 1-GER and Profile 1-FIN). As EXPERIENCE OF AUTONOMY was missing, I expected both profiles to indicate relatively controlled behaviour. In both countries, these profiles have personal importance (identified regulation) as a relevant regulatory process (Ryan & Deci, 2017, p. 193) that resulted in a balanced motivation showing a tendency for self-determination (GER) and self-determined motivation (FIN) in mathematics class. In this connection, learners' specific profile patterns across countries implied the fulfilment of certain basic psychological needs: From the perspective of SDT, the fulfilment of the basic psychological needs for competence and relatedness takes place when learners start internalizing their motivations for performing a compulsory, externally imposed task (Vansteenkiste et al., 2018). In this development process, they move in the direction of autonomy fulfilment and therefore they need social integration in order to complete the process of internalization and to become autonomous (Niemiec & Ryan, 2009; Vansteenkiste et al., 2020). Therefore, learners particularly need a strong bond with their teacher. This bond provides them with the confidence to perform in learning situations, since they have no complete experience of psychological freedom, i.e., fulfilment of autonomy. In return for the teacher's support, they do not want to disappoint the teacher who has let them feel effective. Accordingly, they show their loyalty and try to please the teacher in order to get further support (Bartholomew et al., 2018; Haerens et al., 2015; Vansteenkiste et al., 2018) in mathematics lessons. This view from SDT can be reflected in both patterns of personal meaning that the learner experienced in their social setting (orientation towards emotional-social integration).

Enjoyment in Mathematics Learning. In both countries the social regulations in mathematics class supported the construction of all three specific personal meanings, EXPERIENCE OF COMPETENCE, AUTONOMY, and EXPERIENCE OF SOCIAL RELATEDNESS, which produce learning orientations towards enjoyment in mathematics learning (Profile 2-GER and Profile 2-FIN). These relevance systems go along with self-determined motivation, in which the respective regulatory processes (Ryan & Deci, 2017, p. 193) are enjoyment (intrinsic regulation) and personal importance (identified regulation), but also ego-involvement (introjected regulation in GER). In learning situations, the satisfaction of all basic psychological needs provides the nutriments of an increasing internalization process, integration, and well-being (Deci & Ryan, 2000; Ryan & Deci, 2017). Next to the energizing factors of intrinsic motivation (basic need for autonomy and competence: Deci & Rvan, 2000), the need for relatedness shows that the learner feels a strong connection to their significant others (Deci & Ryan, 2000; Krapp, 2005), but especially to the teacher (Vansteenkiste et al., 2018), who guides the processes of conventionalization in the mathematics classroom. Students are able to take ownership of their learning if they are provided with an accommodating social situation in which their basic psychological needs are met (Vansteenkiste et al., 2018). In Germany, ego-involvement is a notable factor, as studies from the perspective of SDT postulate that it is a typical effect of internally controlled feelings (Ryan & Deci, 2020) in which learners' self-esteem depends on their mathematics-learning outcomes (Frank, 1941; Ryan, 1982). Also here, the view from SDT can be reflected in the patterns of personal meaning that learners experienced in their social context (orientation towards enjoyment in mathematics learning).

Self-Improvement. For Profile 3–GER and Profile 4–FIN, the social regulations in mathematics class supported the construction of the specific personal meanings EXPERIENCE OF AUTONOMY and EXPERIENCE OF COMPETENCE that produce mathematics-learning orientations that focus on self-improvement. From the perspective of SDT, these profiles have *enjoyment* (intrinsic regulation) and *personal importance* (identified regulation) as relevant causes (Ryan & Deci, 2017, p. 193) for their high self-determined motivation. In regard to the learner's personal growth, the basic psychological needs for autonomy and competence are deeply interwoven (Deci & Ryan, 2000; Krapp, 2005). In general, the learner's autonomy refers to their need to experience themselves as a centre of action, whereas competence consists of the need to be capable of acting (Krapp, 2005, p. 635) and to feel optimally challenged (Deci & Ryan, 2000). However, an individual wishes for liberty of action if they feel self-assured about their abilities (Tafarodi et al., 1999; Krapp, 2005). Thus, in order to experience a feeling of competence, it is necessary for

the individual to desire autonomy (Krapp, 2005, p. 635). In these specific profile orientations, the two psychological needs, autonomy and competence, that are responsible for maintaining intrinsic motivation are addressed. Their satisfaction explains the individual's striving for self-improvement and to bring their competence to perfection (Krapp, 2005). However, relatedness, the less central need for the maintenance of intrinsic motivation (Deci & Ryan, 2000; Ryan & Deci, 2017), does not occur. Notably, the specific personal meaning EXPERIENCE OF SOCIAL RELATEDNESS was not constructed here. Also here, the view from SDT can be reflected in the patterns of personal meaning that learners experienced in their social context (orientation towards self-improvement).

External Pressure. In contrast, the social environment that supports the sole construction of EXPERIENCE OF COMPETENCE, Profile 4–GER, produced a learning orientation focusing on external pressure. As a result of this focus in the learner's relevance system, they show a motivational behaviour that directs them towards controlled behaviour. None of the regulatory styles show a clear regulation, but here the SDI indicates less controlled motivation. Analogously to Profile 1–GER, the teacher ensures the learner's basic need for competence is satisfied. The learner's performance is not completely internalized and as a result, their actions in class is neither autonomous nor fully committed (Vansteenkiste et al., 2018). Instead, their attention is centred on social integration, especially on gaining their teacher's approval in the classroom. The interpretation through SDT's lens finds theoretical coherence in the learner's socially regulated profile of personal meaning. Accordingly, the learner constructs personal meanings that reflect their adaption to external pressure.

Through the long-term mathematics lessons, Profile 3–FIN constructed none of the three specific personal meanings, and the social regulations in class supported a mathematics-learning orientation that focused on external pressure. Quite contrary to my hypothesis, this profile only resulted in balanced motivational behaviour and a (mild) tendency for controlled motivation to have personal importance (identified regulation) as a relevant regulatory process (Ryan & Deci, 2017, p. 193). From the perspective of SDT, the experience of basic psychological needs being thwarted can lead to psychological reactance (Brehm, 1966), malfunctioning, and ill-being (Vansteenkiste & Ryan, 2013). However, in this profile, the quality of motivation was not strongly controlled. Furthermore, the learner did not construct personal meanings that reflected their bond with the teacher. Accordingly, the Finnish learner did not connect external pressure with their teacher or need the teacher's help in order to feel effective or supported in terms of their internalization in mathematics class.

Certain mediators seem to exist within the Finnish social environment that guarantee the experience of self-determination in the mathematics classroom. In particular, for Profile 3–FIN, the hypothesized adapted model of social regulatory mechanism (cf. Figure 6) thus did not provide a holistic explanation of this profile's social experiences when these converge with the individual experiences explained through the perspective of SDT. Next to this notable result (Profile 3–FIN), each country's relevance systems, in particular their cross-cultural comparisons, provided strong evidence for the existence of these latent mediators affecting the learner's social experiences in the classroom. To entirely understand the individual's social regulations in mathematics class,

these additional mediating factors need to be included in the adapted model. In the following, using a comparison of the profiles from Germany and Finland, I point out why the mediating role of the teacher and the learner's cultural setting are essential mediating factors that complement the adapted model of the social regulatory mechanism of the individual relevance system.

7.4.2 Comparing the Individual Relevance Systems in Germany and Finland

Conspicuous features of German and Finnish profiles emerged, highlighting similarities, but also differences across countries. The contrasting of results led to an enrichment of the predicted adapted model, that is, the social regulatory mechanism of the individual relevance system.

Certain dimensions of personal meaning that found empirical support across countries (cf. 5.1) emerged consistently in analogous profile orientations in Germany and Finland. For that reason, they were used to specify profiles' learning orientations in general. As indicated earlier, German and Finnish learners identify with four similar mathematics-learning orientations: emotional-social integration (characterized by the dimension: emotional well-being in class); enjoyment in mathematics learning (characteristic dimensions: all except duty); self-improvement (characteristic dimension: freedom of action), and external pressure (representative dimension: duty). Although the students are from different cultural settings, they share certain commonalities. German and Finnish ninth graders, approximately aged 14 to 16 (OECD, 2020a, 2020b), belong to the age group of middle adolescence (ages 14 to 17). Accordingly, their similar ages and commonalities between the cultures of these Western European countries might clarify why they constructed similar relevance systems when learning mathematics.

In both countries, the profiles of emotional-social integration were the most commonly occurring relevance systems among the students. According to their orientations, these students focus more on social rather than on individual processes. Through the non-construction of EXPERIENCE OF AUTONOMY, I expected them to experience controlled motivation. However, when interpreting the profiles from SDT's lens, I found the opposite to be the case. In line with SDT, the student is undergoing a process of becoming a self-determined individual, and for this reason it is necessary for them to have an emotionally warm relationship with their teacher (Wentzel, 2012) who lets them feel effective in mathematics class. During middle adolescence, learners are in the period of emotional, physical, and sexual development associated with puberty, but are also going through neurological changes (Byrnes, 2001; Keating, 2004) with consequences that can extend well into adulthood (Wigfield et al., 2006). In this spirit, to develop an internalized motivation they need a strong bond with their teacher (Haerens et al., 2015; Niemiec & Ryan, 2009; Vansteenkiste et al., 2018). Although the basic psychological need of relatedness is less vital for intrinsic motivation (Deci & Ryan, 2000), this particular need assists the individual to move in the direction of autonomy satisfaction (Ryan & Deci, 2017). As indicated earlier (cf. 2.1.2), Assor et al. (2002) stressed that the importance of the learner's striving for autonomy exists regardless of their development stage. The existence of these specific profiles underlines how critical the innate need to become a self-determined human being is (Ryan & Deci, 2017), a need that parallels the other critical individual changes associated with the development that takes place in middle

adolescence (Steinberg, 2005; Wigfield et al., 2006). As is known, adolescence is affected by culture and thus varies over different cultural contexts (Wigfield et al., 2006). However, both German and Finnish learners value this emotional support greatly, and this emotional-social bond to the teacher exists in both cultures.

However, the Finnish profile showed self-determined motivational behaviour, whereas the profile in Germany scored a more balanced SDI. Even though both profiles are developing their internalization, for German students this process associates with duty in the sense of striving to perform well in order to please the teacher (focused on seeking acceptance from the teacher). The Finnish student connects the emotional-warmth relation with an additional focus on their professional qualification (focused on their own professional qualifications and on seeking acceptance from the teacher). As indicated earlier (cf. 6.2) due to the timing of the transition from lower to upper secondary education (academic or vocational track; OECD, 2020a), for Finnish ninth graders, good marks matter. In this connection, the Mann–Whitney U test pointed out that Finnish students perceived a higher relevance of vocational precondition and showed greater identified regulation than German learners (cf. 5.3). It becomes apparent that here, besides working for one's own qualifications, *specific support from the teacher* seems to result in a higher identified regulation, and this in turn seems to preponderate in Finnish students' self-determined SDI.

The second largest relevance system across countries is orientated towards enjoyment in mathematics learning; in this system, the social regulatory mechanism in the mathematics class appears to stimulate and fulfil all basic psychological needs through the realization of all three specific personal meanings. The complementary insights (through SDT and social constructivism) show that the learner socially experiences individual as well as social processes that result in a high degree of self-determination. Surprisingly, this specific learning orientation, enjoyment in mathematics learning, exists across cultures, without a culture-specific focus. Accordingly, the period of middle adolescence as a cultural phenomenon (Wigfield et al., 2006) shows almost exactly the same features, when all basic psychological needs are satisfied. Even though both groups construct almost identical orientation pattern in profiles, Finnish students scored higher relevance in the active practice of mathematics, self-perfection, marks, and vocational precondition (cf. 5.3), which associated with self-determined types of motivation within the correlation analyses (cf. 5.1). This could be why Finnish learners scored (again) a greater selfdetermination and seemed to be more satisfied in the context of their classroom setting than the German students, who additionally experienced internal pressure (introjected regulation). As indicated earlier, when learners are introjected, they may engage in this conduct because they are experiencing anxiety (Ryan & Deci, 2000a) guilt, or shame (Bartholomew et al., 2018), or because they are seeking self-worth to avoid a feeling of failure (Ryan & Deci, 2017). Ryan and Deci (2017, p. 186) suggested that this behavioural pattern is particularly common in highly pressured disciplines where a high level of competition and comparison between peers exists. The given differences in nuance between countries become more pronounced when contrasting the German and Finnish profiles that are oriented towards self-improvement and external pressure.

RESULTS ACCORDING TO THE COORDINATION ANALYSIS METHOD

For German learners, studying mathematics for self-improvement means satisfying their own competence requirements (focus on mastery), but for Finnish students it means strengthening their personal development (focus on the individual). Across countries, these profiles reflected the highest degree of self-determination. Even though both relevance systems perceive EXPERIENCE OF AUTONOMY (in Germany there is only a tendency for construction) and COMPETENCE, Finnish students constructed personal meanings that relate to intrinsic types of motivation (purism of mathematics, active practice of mathematics, and self-perfection) or perceived more relevance in them (cognitive challenge) than German learners (cf. 5.1). When compared, the Finnish profile again scored a greater SDI than the German relevance system. However, in Germany, the orientation towards self-improvement is the third largest profile, while in Finland it is the smallest.

The most surprising difference over groups concerned the mathematics-learning orientations towards external pressure. Whereas the German learner connects their feeling of obligation and the performance requirements with their teacher (focusing on adaption to teacher's performance requirements), the Finnish student does not personalize the subject mathematics as connected with their teacher (focusing on adaption to the educational system). Ouite the contrary; rather, they see the teacher as supporting them and probably relate mathematics to their Finnish educational system. Based on the learners' transactions with the social environment, personal meanings that reflect a collective relevance in mathematics, and the non-construction of EXPERIENCE OF AUTONOMY, I predicted these profiles would direct towards controlled motivational behaviour. Whereas in Germany, learners scored a higher relevance in EXPERIENCE OF COMPETENCE (cf. 5.3), which led to a less controlled quality of motivation, in Finland students only showed a tendency towards controlled motivation without constructing any specific personal meanings. As indicated above, interpretation of the German profile from a social constructivist perspective is in theoretical consensus with the lens of SDT, but the adapted model could not entirely explain SDT's perspective for Finnish Profile 3. Although Finnish learners constructed none of the specific personal meanings, the exclusive realization of duty in their learning orientation resulted in behaviour in mathematics class that was not explicitly controlled. In this example, it explicitly turned out that the adapted model (cf. Figure 6) could not provide convincing explanations regarding the learner's motivational outcome. In a nutshell, the adapted cycle fails to explain the support for self-determination in the Finnish classroom.

To conclude, contrasting the results from Germany and Finland by referring to learners' cultural setting in the classroom implied that the role of teacher is interpreted differently in German and Finnish schools. German students' relevance systems basically focus on performance-related learning orientations, whereas Finnish relevance systems in mathematics emphasize learning orientations that are inclined towards the individual learner. The results further show that in the German educational system, the teacher is seen as the personification of the subject mathematics and tends to emphasize the importance of academic performance—possibly mediated by the so-called *performance standards* within the German mathematics curriculum (German expression: *Leistungsstandards*; Blum, 2006, p. 15; Heymann, 2005). In particular, the consistent appearance of EXPERIENCE OF COMPETENCE in all profiles in Germany highlights the importance of academic

performance in the mathematics classroom. The Finnish students do not associate the external pressure that is placed on them with the teacher; rather, they see the teacher as being on their side. In Finland, external pressure in mathematics seems to relate more to the educational system.

Interestingly, these theory-driven analyses (investigating learners' social experiences) not only underpin and strengthen the empirical results that merged from the quantitative analyses, which pointed out that the classroom settings in mathematics vary greatly across countries; also, the results according to the coordination analysis indicated that analogously to the theoretical considerations (cf. 2.4), different foci within the mathematics curricula exist, for example academic performance and mathematical performance standards (GER; Blum, 2006, p. 15; Heymann, 2005; KMK, 2005) versus the equal support of academic and personal development (Finnish National Board of Education, 2016, p. 402). Like the empirical results (cf. 5.2), these findings also confirm the PISA results in which Finnish learners showed a lower level of anxiety about their mathematics performance in comparison to German learners (OECD, 2015, p. 2).

The mediating role of the teacher and the cultural context of the mathematics curricula seem to clarify why Finnish students show consistently higher levels of self-determined regulated behaviour in the mathematics classroom than is seen in German profiles. Apparently, the teacher mediates in the circular processes of the social regulatory mechanism and thus the teacher assists the process during all stages of social regulation, that is, appropriation, transformation, publication, and processes of conventionalization. To provide a holistic explanation of the learner's social regulatory mechanism, these mediators—the teacher's mediation activities and cultural amplifiers that reshape the learner's social regulation—should be included within the adapted model (see Figure 26).

Figure 26

Revised Overview: Learner's Social Regulatory Mechanism of the Individual Relevance System—Affected by the Mediating Role of the Teacher and Cultural Amplifiers.



Note. The empirical results of this study support the existence of latent mediators affecting learners' social regulations in the classroom. To provide a holistic explanation, teacher's mediation activities and cultural amplifiers are included. Cycle adapted from Ernest (2010).

7.4.3 Conclusions on the Effects Found in German and Finnish Mathematics-Learning Orientations

The coordination analysis came up with interesting insights that support the consideration of personal meaning as located at the boundary of SDT and social constructivism. From the perspective of social constructivism, the hypothesized adapted model, social regulatory mechanism of the individual relevance system, was not in conflict with the empirical profiles but did not provide an entire view of learners' motivational propensity. By means of this adapted model, the learner's whole relevance system can be understood as their experience of conventionalization processes, which thus emerges from the regulations of appropriation and transformation mediated by the teacher and is reshaped by cultural amplifiers.

Based on the empirical profiles across countries, it seems that the construction of the three specific personal meanings (EXPERIENCE OF AUTONOMY, EXPERIENCE OF COMPETENCE, and EXPERIENCE OF SOCIAL RELATEDNESS) in the social processes of appropriation and transformation increases the personal relevance of mathematics and regulates the publication of the learner's individual relevance system. In these regulations of appropriation and transformation, the three specific personal meanings play an essential role, as they seem to address the corresponding basic psychological needs.

The empirical evidence suggests that basic psychological needs can be stimulated by the construction (or fail to be stimulated in the case of non-construction) of the three specific personal meanings EXPERIENCE OF AUTONOMY, COMPETENCE, and SOCIAL RELATEDNESS through the learner's reaction to their social environment in the mathematics classroom. This in turn satisfies the respective basic psychological needs and regulates the learner's psychological well-being towards fulfilment (Deci & Ryan, 2000). The basic psychological needs, thus, wander into the regulations of appropriation and transformation and promote or restrict the process in view of their fulfilment. As a result of this (non-) satisfaction and in connection with significant others, in the processes of conventionalization, profiles are individually differentiated through strengths and weaknesses and emerge as patterns of personal meaning that are in line with the individual. Thus, the satisfaction of the respective basic needs seems to retroact to the learner's social space in class. Among the three specific personal meanings, the EXPERIENCE OF AUTONOMY has a crucial role as it seems to refer to the basic psychological need of autonomy, which is essential to experience deep-rooted personal importance and to put one's heart into learning situations in mathematics. In this way, the particular realization of EXPERIENCE OF AUTONOMY strengthens the relevance of both individual and mathematics-related personal meanings. If the learner's EXPERIENCE OF AUTONOMY is unrealized, they construct substitute personal meanings that weaken the relevance of both individual and mathematics-related personal meanings. On the basis of this mutual relevance reaction to the context of mathematics learning, the profiles of personal meaning can be explained not only by means of the adapted model (through the lens of social constructivism), but also through the theoretical lens of SDT that describes profiles' quality of motivation. In these profiles the three specific personal meanings show the same reaction, which are a result of the satisfaction of the respective basic psychological need, or the basic psychological needs mechanisms. As

predicted, these three meanings, EXPERIENCE OF AUTONOMY, COMPETENCE, and SOCIAL RELATEDNESS, with their specific natures, seem to be more closely related to the basic psychological needs than to other personal meanings. The effects found in both independent data sets, explained by two different epistemic views, strongly support this assumption.

Figure 27

Inner Feedback Mechanism Between Basic Psychological Needs and Specific Personal Meanings.



On a *micro level*, these effects indicate an inner feedback (see Figure 27) between the three specific personal meanings and the basic psychological needs that increases the learner's response regarding the personal relevance of learning mathematics. Not only does this particular inner feedback have an effect on the kind of relevance system the learner constructs through social interaction in the classroom setting, but it is also an essential component of the student's motivational behaviour - as the fulfilment of the innate basic psychological needs not only supports the individual's well-being, but also facilitates self-determined types of motivation (Deci & Ryan, 1993; Ryan & Deci, 2020). The published profiles in Germany and Finland highlight through both theoretical lenses the essential coherence between the three specific personal meanings and the basic psychological needs. Accordingly, a coordination between SDT and social constructivism provides a meaningful clarification of the content (or "what") of "why" the learner's motivational expression in mathematics lessons comes about.

7.4.4 Model of the Feedback Mechanism: The Genesis of Motivation in the Mathematics Classroom

The present research aimed to understand how the predicted biological regulatory mechanism (SDT) are coordinated with social regulatory mechanism (social constructivism) within the boundary object of personal meaning in light of the development of motivation in German and Finnish mathematics classrooms. As a response to this study's basic assumption, the effects found in the empirical evidence can be condensed into a hypothetical model that describes the development of the learner's quality of motivation in the mathematics classroom on two levels; namely, on a micro and on a macro level (see Figure 28).

RESULTS ACCORDING TO THE COORDINATION ANALYSIS METHOD

Figure 28

Hypothetical Model of the Feedback Mechanism Between Social and Biological Regulatory Mechanisms to Explain the Genesis of Motivation in the Mathematics Classroom.



RESULTS ACCORDING TO THE COORDINATION ANALYSIS METHOD

According to this coordination, from an epistemological view (*micro level*), the boundary object personal meaning mediates between the learner's outer and inner understanding. The link between SDT and personal meaning is the *inner feedback mechanism* (satisfaction of basic psychological needs) and the connection between social constructivism and personal meaning is the formation of the learner's *individual relevance system* (social experiences in the mathematics classroom). On the one hand, the learner's social regulatory mechanism regulates the construction of the individual relevance system. On the other hand, the biological regulatory mechanism regulates the trigger their biological regulatory mechanism. This in turn seems to feedforward, on the basis of the inner feedback mechanism, a feedback (reaction) to their social regulatory mechanism. So, they both have a natural effect, and these two effects meet within the boundary object of personal meaning.

Social regulations act on the individual and the individual reacts with biological regulations to what is offered within the classroom's social environment. In this process, the profile of personal meaning arises through repetitive processes that take place in mathematics class, and which intensify and produce a reinforcing profile pattern; that is, an individual relevance system is produced by the increasing number of regulations between the inner and outer worlds.

One can say that the dominant role is that of the biological regulatory mechanism. However, analogous to the butterfly effect (Ghys, 2015; Lorenz, 1995, pp. 181–184), the subtle differences in how the profiles develop seem to be socially regulated (affected by the mediating role of the teacher and cultural amplifiers). To conclude, the social direction seems to *trigger* and the biological seems to give *feedback*, and this takes place through the processes of personal meaning. Accordingly, personal meaning can be interpreted twice ontologically and twice epistemologically, and therefore it is of central importance in light of the learner's quality of motivation in mathematics class.

From an ontological viewpoint (*macro view*): On the left-hand side of Figure 28 there is the social regulatory mechanism, which focuses on the individual and the individual-social environment ("conversation", Ernest, 1998). Accordingly, there are the individual and social levels accompanied by the role of culture and the teacher's mediating activities in the mathematics classroom. The biological regulation (right-hand side of Figure 28) includes the basic psychological needs. In line with SDT, if certain basic psychological needs are not satisfied, this will be visible in the well-being and health of the person. This aspect links the psychological level and the physiological level. Accordingly, there are the individual and biological levels. Through this linkage of the social and biological regulatory mechanisms, the social, psychological, and biological levels are addressed. So actually, all three levels, social, psychological, and biological levels, are interwoven when looking at the genesis of quality of motivation in mathematics class. Through this interdependence, we cannot see the social and biological regulatory mechanisms independently of each other.

The proposed feedback mechanism model condenses the overall effects found in this investigation's coordination analysis to explain the genesis of quality of motivation in the mathematics classroom. The described inner mechanisms within the hypothetical model were not

directly observable, but their effects were visible in different individual relevance systems from Germany and Finland and could be explained by means of different theoretical lenses. These effects are not in conflict with the hypothesized model, but the model is not able to empirically illustrate the model's directionality, that is, the intensity of trigger or feedback effects.

8. Discussion

In this dissertation, I have examined the conceptual relationship between learners' individually constructed relevance systems (patterns of personal meaning) and the quality of their motivation in German and Finnish mathematics classrooms. Grounded on learners' empirical individual relevance systems, the specific factors within learners' inner biological regulations were examined through the lens of self-determination theory (SDT) to understand the corresponding motivational behaviour of each profile. By means of the lens of social constructivism, specific outer social regulations within the classroom context could be theoretically reflected within learners' inner biological regulations. Based on learners' individual experiences, the complementary perspective on their social experiences should help to increase our knowledge of teaching mathematics in the classroom in terms of possible external endeavour that addresses specific individual relevance systems which associate with autonomous motivation. At the same time, implications should be derived which aim to avoid the individual relevance systems that associate with controlled motivation.

In this chapter, I first present the main findings (see section 8.1) of this dissertation by summarizing the conceptual interconnections between learners' individual relevance systems (social regulatory mechanisms) and motivational behaviour (biological regulatory mechanisms), which theoretically explain the development of learners' motivation in the mathematics classroom. Secondly, to emphasize their generalizability and scope, I classify the main findings of this dissertation within the relevant theories of this work by relating them to existing knowledge; that is, within the landscape of mathematics-related affect research (see section 8.2) and SDTs research landscape (see section 8.3). Thirdly, I discuss the applied theoretical and the methodological approaches from a retrospective viewpoint (see section 8.4) to point out open-ended questions and suggest, from a prospective viewpoint, implications for future research, teaching mathematics in the classroom, and educational policy (see section 8.5). The final section of this chapter highlights the concluding remarks of this dissertation (see section 8.6).

8.1 Main Findings

The main findings within the empirical evidence of this study indicate that between the individual relevance system and motivation, a unique conceptual relationship exists that has an important role in students' mathematics learning in the classroom. In the following, I summarize the main findings of this dissertation: Learners' individual relevance systems found in Germany (GER) and Finland (FIN) can be differentiated into four kinds of mathematics-learning orientations in view of

- 1. Emotional-social integration—learners' starting internalization (either balanced (GER) or autonomous (FIN) motivation);
- 2. Enjoyment in mathematics learning—learners experience well-being and integration (autonomous motivation);
- 3. Self-improvement-learners feel freedom of action (autonomous motivation); and

- 4. External pressure—learners adapt to obligatory requirements (either balanced (FIN) or less controlled (GER) motivation).
- From the perspective of SDT, personal meaning is an effect of a basic psychological needs mechanisms, namely a biological regulatory mechanism.
- From the perspective of social constructivism, personal meaning results from interpersonal experiences in the classroom, namely a social regulatory mechanism.
- The biological and social regulatory mechanisms are coordinated within personal meaning, and the motivational construct of personal meaning links both regulations and can be interpreted as a boundary object through both theoretical lenses.
- Indicators within the empirical evidence support the finding that basic psychological needs can be stimulated by the construction (or fail to be stimulated in the event of non-construction) of the three specific personal meanings EXPERIENCE OF AUTONOMY, COMPETENCE and SOCIAL RELATEDNESS through the learner's reaction to their social environment in the mathematics classroom, termed as an inner feedback mechanism.
 - 1. Through this inner feedback mechanism, the biological regulatory mechanism regulates the learner's self-determination and well-being in the classroom (for autonomous to controlled motivation).
 - 2. Through this inner feedback mechanism, in the classroom setting the social regulatory mechanism regulates the direction of the learner's kind of individual relevance that is produced in line with the biological regulation (geared towards or away from the relevance in individual and mathematics-related personal meanings).
 - 3. The mediating role of the teacher and cultural amplifiers reshape the learner's social regulatory mechanism and this in turn explains why similar individual relevance systems result in different biological regulations across countries.
- Accordingly, basic psychological needs are the factors within personal meaning that have a strengthening or weakening effect on biological and social regulatory mechanisms.
- The effects (not directly observed, but not in conflict with the hypothesized model) found within the coordination of biological and social regulatory mechanisms can be modelled as a hypothetical feedback mechanism to explain the genesis of motivation in the mathematics classroom.
 - 1. The social regulatory mechanism triggers the biological regulatory mechanism and that in turn seems to feedforward to the social regulatory mechanism.
 - 2. Through this conceptual interrelation, one cannot regard the biological and social regulatory mechanisms as independent of each other.
 - 3. Within this conceptual interrelation, the dominant role is determined by the biological regulatory mechanism, but how the individual relevance system develops in terms of its subtle differences from other similar systems is socially regulated.

• To conclude, the boundary object personal meaning can be interpreted twice, ontologically (biological and social nature) as well as epistemologically (motivational behaviour or social interaction), and as a result of its multimodal ontological nature it consequently has major significance for the genesis of motivation in the mathematics classroom.

8.2 Complementing References to the Field of Mathematics-Related Affect

8.2.1 Networking Theories to Counter the Terminological Ambiguity

As indicated earlier (cf. 4.1.1), in qualitative empirical case studies the tradition of linking theoretical lenses is particularly widespread, but it is rarely realized in quantitative empirical investigations (Prediger et al., 2008). Connections between theories can take place at the level of principles, methodologies, and of questions, or by combining these elements (Radford, 2008a).

In this study, which belongs to the quantitative field, I considered the connections between theories on the level of principles (Radford, 2008b). The networking of theories approach sharpened the view of both theories' alignment (SDT and social constructivism) by means of coordinating the complementary analyses, that is, of the development of the individual relevance system and of the way this learning orientation is processed as motivational behaviour in the mathematics classroom. The applied theoretical knowledge (conceptualized through the principles of the theories) helped to view motivation from different angles by dissecting focus-relevant questions with respect to the role of cultural imprints. In particular, the consideration of the German and Finnish data helped to contrast and refine the relational understanding between learners' individual relevance systems and quality of motivation. This decision challenged the principles of both theories, SDT and social constructivism, in terms of their descriptive and explanatory power (Schoenfeld, 2002). This challenge led to working out (on a theoretical level) the mediating factors affecting the social regulations of the learners.

Although the results have certain limitations (see section 8.4), the above-mentioned decisions about the way this study was conducted, given the challenges associated with the need to take into account two independent data sets from different cultures and to target the selection of both theories, led to findings that supported the explanatory power (Schoenfeld, 2002) of the hypothetical model to a certain degree: The joint results (R as the fourth element; Radford, 2012) of SDT and social constructivism, obtained from this dissertation's networking practice, influenced and complemented each other at the same time. The results show that a one-sided view, either from the perspective of SDT or of social constructivism, of learners' motivation would have led to limited knowledge. Generated on the basis of a sequential-dependent methodology (M), both theoretical lenses (P) gave results that shed light on complementary insights answering common questions (Q) in view of learners' mathematics-related meanings and teachers' mediation activities reshaped by cultural amplifiers. This in turn strengthens the learners' individual and mathematics-related learning orientation (*what*),

and that explains *why* a self-determined motivational expression in mathematics lessons comes about. Bringing both lenses together by means of coordination, the thus-built conceptual framework creates a locally integrated conceptual tool (R; Prediger et al., 2008, p. 166) that works together on the same research problem by jointly complementing the resulting picture, thereby improving their new concept's (hypothetical model of a feedback mechanism, Figure 28) *explanatory power* as well as both theories' *descriptive power* (Schoenfeld, 2002). Accordingly, one can see that SDT and social constructivism are nicely aligned when used to investigate the role of personal meaning in learners' motivation.

This networking-of-theories approach also seems to be a helpful strategy with which to tackle the problem of terminological ambiguity in the definitions of motivational constructs (Hannula, 2012) in a sustainable way. Even though the coordination analyses, which follow the tradition of linking theoretical lenses, lead to a feedback mechanism that is hypothetical in nature, this contribution, creating a dialogue between constructs, establishes a fruitful research practice to counter the terminological ambiguity in mathematics-related affect research (Hannula, 2012; Suriakumaran et al., 2020) by elaborating the connections between the affective constructs. In fact, the present research approach helped to increase our knowledge as regards the theoretical conceptualization of personal meaning and to enrich our understanding concerning the interrelation between personal meaning and quality of motivation. However, although we have found empirical evidence for learners' patterns of personal meaning in two different cultural contexts (Vollstedt, 2011b; Vorhölter, 2009), all kinds of personal meaning could not be assessed within one measurement model (cf. 6.1). The conducted correlation analyses support an empirical link between the socially experienced personal meanings and the biologically perceived quality of motivation. Building on the insights gained so far, we could focus within future research on specific social conditions in the classrooms (e.g., the teacher's mediating role and the foci within mathematics curricula; see 8.5 for more), which seem to affect our learners' biological regulatory mechanisms. However, future research needs to find a solution that can methodologically estimate all factors (personal meanings) within one comprehensive model in order to move to the next level; for example, this might involve investigating the causal effects between personal meaning and motivation.

In summary, this research approach has not only contributed to addressing the problem of terminological ambiguity; also, the present investigation's application of networking theories intends to encourage the practice of "networking" with other fields from mathematics education to broaden scholars' horizons with regard to fruitful research strategies that could advance our research within the field of mathematics-related affect.

8.2.2 References to Meta-Theoretical Knowledge

Within the frame of this quantitative study, relevant indicators referring to the meta-theoretical knowledge (cf. 3.2.3) of this present study's research interest emerged in the evidence; this meta-theoretical knowledge is discussed in the following.

Relation to Learners' Interest in Mathematics. As indicated earlier (cf. 2.2.2), the construction of personal meanings is particularly influenced by various factors from learners' personal characteristics. Some of the factors may also come to the fore as consequences of a

socially experienced individual relevance system (Vollstedt, 2011b; Vorhölter, 2009). Although various concepts were considered as influencing factors during the coding process (see Vollstedt, 2011b), the three basic psychological needs, beliefs, and interest became relevant concepts (in the sense of Grounded Theory) for the construction of the learner's personal meaning (Vollstedt, 2011b; Vorhölter, 2009).

In this connection, the identified profiles of personal meaning in Germany and Finland indicate connections to the affective domain-related concept, namely *interest as a person-object relationship* (Krapp, 1992, 2002). Learners who are oriented towards "enjoyment" (Profile 2–GER and Profile 2–FIN) and "self-improvement with a focus on the individual" (Profile 4–FIN) seem to be interested in mathematics as a scientific discipline. In all these three profiles, learners specifically perceived the two specific personal meanings EXPERIENCE OF AUTONOMY and EXPERIENCE OF COMPETENCE and constructed the personal meaning "purism of mathematics". Referring to the inner feedback mechanism (cf. Figure 27), the learning situations' conditions thus seem to satisfy the basic psychological needs for autonomy and competence that in turn nurture the development and maintenance of intrinsic motivation. In these three cases (Profile 2–GER, Profile 2–FIN as well as Profile 4–FIN), in line with Bikner-Ahsbahs (2005, 2014), these results underline that the perception of *meaningful* mathematics lessons and active *involvement* (Mitchell, 1993) in activities (here through the EXPERIENCE OF AUTONOMY) can encourage their (situational) interest.

By contrast, in Profiles 1 in Germany and in Finland, learners are involved in the subject specifically with regard to social integration with the teacher and work *for* the teacher; they seem not to be interested in the subject of mathematics as a scientific discipline. However, as these students are on their way to becoming self-determined (starting internalization), the learning orientation towards emotional-social integration has a *fertile soil* in which learners' situational interest can be supported by offering them opportunities over time to benefit from EXPERIENCE OF AUTONOMY. Accordingly, the mathematics-learning orientations detected offer pointers for how lessons should be designed (see 8.5.2 for more) in order to promote interest and intrinsic motivation, conveyed through personal meaning.

Learners' Gender Performance Characteristics. Next to the references to related affective concepts, implicit knowledge about further characteristics of personal meaning emerged, namely gender and performance. Whereas the Profiles 1 (GER and FIN) of "emotional-social integration" did not display a noticeable difference in gender, they showed differences with respect to learners' performance. In Finland this profile is mostly preferred by high-achieving students, and in Germany most average students belong to this profile. Moreover, in both countries the learning orientation of "enjoyment" was mostly perceived by high-achieving students (in Germany it was also preferred by many average students). However, in Germany the proportion of male students is higher in this profile, while in Finland it was the other way around; that is, more female students engaged in learning through this orientation. Similar results were exhibited in profiles of "self-improvement"; in both countries more male students engaged in mathematics for self-improvement than female learners. As a profile this shows connection to the affective concept of interest; these results are in line with findings from the literature that state that male students are more interested in mathematics than girls are (Eccles et al., 1983; Watt, 2004). In both countries, this learning orientation was perceived mostly in high-achieving students, whereas in Germany this profile is also

experienced by average students. And finally, many high achievers and some average learners (in the case of Germany, some high achievers and many average students) engage in mathematics due to "external pressure". Whereas in Finland this orientation is mostly found in boys, in Germany there was no noticeable difference in gender.

Altogether, there is no strong trend regarding gender performance properties and profile orientations (which were also not tested in terms of their statistical significance), but it also becomes clear that in this context, cultural differences exist that could be clarified by having a closer look at the different classroom settings. If this is done, I recommend that these characteristics be discussed through a "fresh perspective" in future with languages that respects the individual's gender identity (American Psychological Association, 2020) and referring to all individuals as human beings first, i.e., avoiding the more common, dichotomous way of talking about gender.

State- or Trait-Like Characteristics. The discussion on the *state-* or *trait-*like properties of an affective concept has been an ongoing discourse within mathematics-related affect research (Hannula, 2012). Briefly, traits refer to relatively stable personality traits, but states of affectivity can vary over time or situationally (Kelava & Schermelleh-Engel, 2012). Referring to the state- or trait-like characteristics of personal meaning, published individual relevance systems represent an outcome after a series of experiences from the adapted cycle (cf. Figure 26). Accordingly, the realization of the three specific personal meanings that strengthen or weaken the individual- and mathematics-related personal meanings is based on multiple cycles in which the learner experiences their social interaction in the classroom. So far, there has been no clarification of the trait- or state-like properties of personal meaning. However, the conceptual relation between personal meaning and motivation implies some interesting clues with respect to this issue.

Although individual relevance systems are based on a series of experiences, the different kinds of learning orientations (emotional-social integration, enjoyment in mathematics learning, self-improvement, and external pressure) and the encompassed personal meanings can be tentatively related to state or trait. As indicated earlier (cf. 2.1.2), autonomous regulatory styles such as identified, integrated, and intrinsic regulation associate with persistence, whereas controlled regulatory styles do not. Furthermore, externally regulated students comply with expectations as long as the contingency lasts (Ryan & Deci, 2017). That is, in the absence of the external expectations in the classroom that constrain the conduct of students, those students would not exhibit consistently "good behaviour" in the long term and would show little engagement in their learning tasks, eventually ceasing to participate in them altogether (Ryan & Deci, 2017). Following this background knowledge, the thus-constructed kinds/dimensions of personal meaning and the profiles seem to have a more trait-like character (see Hannula, 2012) when they refer more to the subject of mathematics, the individual, and autonomous motivation (cf. 5.1 and 5.3), rather than to kinds/dimensions of personal meaning and the individual relevance systems that are associated with controlled motivation. The latter might have more state-like characteristics. Anyway, these relations provide limited cues; proper methodological procedures are required in order to dig deeper into this issue and to find solid answers.

8.2.3 The Body as an Essential Element of Mathematical Thinking

This dissertation's main findings can be condensed into a model which theoretically explains the genesis of motivation in the mathematics classroom as a function of learners' individual relevance systems. As indicated earlier (cf. 7.4.4), from an ontological viewpoint, this conceptual interrelation between personal meaning and quality of motivation can be theorized through the embodied, psychological, and social perspectives. Thus, all three perspectives seem to be interwoven when theorizing the genesis of motivation in the mathematics classroom. Besides the conceptual links between both constructs' central elements (cf. 2.3), the potential to theorize the hypothetical feedback mechanism from three different levels might be a further reason why a strong conceptual interrelation between personal meaning and motivation exists.

Hence, the knowledge in this model is not only underpinned by Hannula's (2012) metacube, which proposes to theorize affective concepts like quality of motivation or personal meaning through physiological, psychological, and social lenses. In line with Vygotsky (1962, 1987), this feedback mechanism also highlights why the emphasis on the monist perspective on mind and body, in which it is stated that affect "is never external to intellect" is important (Roth & Walshaw, 2019, p. 112; see also Hannula, 2012). Because, in accordance with Hickey (1997), this study's findings emphasize that motivation as an affective construct is not separable from cognitive activities when learning.

These important connections point out that the "corpus" of mathematics learning can neither be separated from body-related functions nor restricted to cognitive processes as an "unemotional activity" (Drodge & Reid, 2000, p. 249). In fact, this specific discussion on the study's findings goes beyond frontiers where mathematics learning or mathematical activities are considered as *embodied*, and as such I adhere to the research that considers and conceptualizes mathematical thinking as based on an interdependence of embodied, psychological, and social phenomena (de Freitas & Sinclair, 2014; Ferrari & Ferrara, 2020; Hannula, 2012; Roth & Walshaw, 2019).

8.3 Complementing References to SDT's Research Landscape

At the beginning of this thesis (cf. 1), it was stated that students can also be involved positively with mathematical activities through a learning environment that supports basic psychological needs and that highlights the personal relevance of mathematics for the student (more individually valued than socially, Ryan & Deci, 2017, 2020; Vansteenkiste et al., 2018). The knowledge gathered from the present study research can now clarify on a conceptual level why learners' personal relevance relates in particular to autonomous motivation when it refers more to the individual. It also shows why the consideration of all kinds of personal relevance (regardless of their relation to more or less internalized forms of motivation) was important to helping us holistically understand the conceptual interrelations between personal relevance and motivation in the mathematics classroom.

As indicated earlier, the learner's personal meaning is an effect of a basic psychological needs mechanisms; namely, a biological regulatory mechanism. Furthermore, the basic psychological needs are the factors within personal meaning which have a strengthening or weakening effect on biological and social regulatory mechanisms. Accordingly, personal

meanings with respect to the relevance in individual and mathematics-related personal meanings are in line with the learner's satisfaction of basic psychological needs, which in turn associate with autonomous forms of motivation. This aspect clarifies why the kinds of personal relevance that are close to the learner as an individual are associated with autonomous (extrinsic) regulatory styles (Vansteenkiste et al., 2018).

Moreover, Vansteenkiste et al. (2018) stated that the notion of personal relevance can only associate with autonomous forms of motivation. From the present knowledge, a differentiated consideration would take into account that learners' personal relevance can associate either with autonomous or controlled forms in line with the fulfilment of basic needs, and this depends on whether personal meanings are aligned with or diverge from the individual's personal relevance of mathematics (in an academic context). Based on these insights, the consideration of personal meanings which are less close to the individual or to mathematics was important to studying the complex associations of personal meaning with the various behavioural outcomes of self-determined and controlled types of motivation. In particular, the use of the composite score SDI (Levesque et al., 2004; Ryan & Deci, 2017, p. 195, 2020) was helpful to reflect learners' perceived relative autonomy in addition to the examination of the regulatory styles.

Altogether, these results from Germany and Finland underpin that basic psychological needs are not only universal (Vansteenkiste et al., 2020), but also significant factors in learners' *need for meaning* (Vinner, 2007) and thus are an essential constituent of their mathematics education. Moreover, Ryan and Deci (2017) discussed the notion of *meaning* on a more general level (e.g., meaning in life), and considered meaning to be a resulting effect of the satisfaction of basic psychological needs. Although both concepts, meaning and the learner's personal relevance in terms of studying mathematics, are not identical, the results of this dissertation indicate that Ryan and Deci's suggestion can be applied to the learner's mathematics-related personal meaning.

As indicated earlier, in comparison to other forms of autonomous motivation (intrinsic or integrated regulation), teachers' external endeavours can address identified regulation by encouraging students to understand (e.g.,) the personal relevance of mathematics learning for them. For this reason, it was a matter of particular interest with which individual relevance systems learners associate identified regulation. In Finland all profiles associated with particularly identified regulation, whereas in Germany all profiles apart from Profile 4, "external pressure," displayed identified regulation. Although Profile 3 in Finland was also oriented towards "external pressure", they still identified with the importance (Deci & Ryan, 2000) of mathematics learning. As mentioned earlier, Haerens et al. (2016) advocated that teachers should refrain from using classroom tactics of exerting pressure and control when students' conduct does not conform to their expectations, as this can cause frustration of their basic psychological needs. It seems that the degree to which the learner felt obligation towards the teacher and adapted themselves to the teacher's performance requirements (such as in Profile 4 in GER) influences the learner's (less) controlled behaviour.

Referring to the dimensions of personal meaning, in both cultural contexts the dimension of professional qualification and application in life associated highly positively with identified regulation. Accordingly, particularly when learners are not interested in the subject, highlighting the personal relevance learners attach to participating in the social environment and valuing mathematics knowledge as part of their identity, as well as understanding why mathematics will be important for their future profession—or further education—helps learners to actively internalize extrinsic content as part of their identities. To conclude, a focus on the students' relevance, or rationales that are student-centred and support the experience of autonomy, as well as autonomy-supportive teaching styles that foster a positive teacher–student relationship, contribute to students' positive learning processes and well-being.

While studies within SDT have identified motivational profiles in order to explore configurations of behavioural regulations in mathematics education (e.g., Wang et al., 2017), in this dissertation I have focused both on learners' individual mental compasses, which provide them with their orientations towards studying mathematics, and on the respective motivational behaviour they exhibit in the classroom. Considering learners' inner and outer worlds helped to move the analysis to the next level and focus on the interconnection between the learners' individual relevance systems and their behavioural regulations.

8.4 Retrospection

With regard to the general research interest in the genesis of motivation in the mathematics classroom, the given discussion has emphasized that several factors support the evidence gained about the hypothetical model of a feedback mechanism in the social and biological regulatory mechanisms: the effects can be understood through the joint lens that emerged by means of coordinating SDT and social constructivism, and the evidence behind the hypothetical model could be replicated, validated, and refined on the basis of two independent data sets, namely those from Germany and Finland. Furthermore, we were able to increase our knowledge regarding personal meaning's conceptualization and in this relation the power and range of SDT's basic psychological needs. The evidence of this dissertation fits into the renewed debate about the role of the body in learning and meta-theoretical links could be found between the identified empirical profiles and existing knowledge. Despite this support for the evidence of this present research, there exist several limitations that lead to certain open-ended questions.

8.4.1 Limitations of the Theoretical Approach

Reflection on Mathematics Curricula. With regard to the two mathematics curricula, there is no uniform curriculum in Germany. Since the German data set comes from various federal states (from west, north, and south), I looked at the overall cultural setting of many different school systems that are shaped by the KMK⁴⁰ and their educational standards (KMK, 2005) or, as Blum (2006) and Heymann (2005) refer to them, their "performance standards."

In contrast to the federal system in Germany, there is an overall mathematics curriculum in Finland. Regarding the Finnish curriculum, in which self-determination is explicitly promoted (e.g., in particular Profile 4, "self-improvement with a focus on the individual"), one can see that this mathematics curriculum has different effects compared to that of Germany, where such systematic promotion across all federal states simply does not exist—except for the standardization of the educational standards focusing on mathematics-related

⁴⁰ German: Kultusministerkonferenz-KMK

competencies. In Germany, these competence-related standards have to be implemented by everyone; that is, in all the respective federal states with their different educational systems and traditions. It is interesting that a uniform mathematics curriculum has the effect of fostering internalization, for example, students' readiness to work independently. In contrast, in the German federal system, which has a non-uniform mathematics curriculum which takes diversity of ability into account, this effect is relatively small.

What these differences effectively mean is that if the German context does not have these cultural *amplifiers or mediators*, we basically cannot expect to find similar profiles to those that are produced in Finland, because that would not work in a different cultural context. In terms of the mediators, the empirical evidence implied that learners' social regulations are affected by the cultural setting and mediating role of the teacher. Although classroom observations were not part of this research, based on learners' individual and interpersonal experiences it is thought that certain types of teacher behaviour in the classroom might support the construction of learners' individual relevance systems.

Reflection on the Teacher's Mediating Role. In particular, by discussing the teacher– learner interpersonal relation, the "model of sign appropriation and use" by Ernest (2010) describes how the zone of proximal development (ZPD, Vygotsky, 1978) may embrace the overall cycle that explains how an individual learns concept development though signs; that is, the ways in which learning and knowledge come to be realised (Ernest, 2010, p. 45). Ernest further stated that signs or texts are a kind of cultural representation (although there are others), and they appear in/are relevant to all four quadrants of the model. Since culture encompasses signs and texts, culture affects learning (Ernest, 2010).

Although the process of conventionalization and learners' social experiences in general in the various classrooms were not observed, application of the adapted model implied, in line with Ernest, that the cultural context affects learners' social experiences. Moreover, Ernest stated that the theoretical approach of social constructivism places emphasis (among other factors) on learner-teacher interactions (Ernest, 2010). The mediating role of the teacher also becomes a central component within the adapted model as it seems that the teacher triggers the profiles through specific behaviours. That is, within the different individual relevance systems across countries, the learners' points of view imply that they experience the various mediation activities of the teacher within the context of social interaction. Again, I have not observed the behaviour of teachers here, but using the data of specific profiles, we could predict which teacher behaviour would then support that profile, giving a spectrum of possible ways in which a teacher can act, and thus assumptions could be made about the ways in which a teacher supports different profiles in their actions and behaviour.

In this spirit, I briefly review teachers' mediation activities in terms of their effect on the process of internalization and benefits for mathematics learning. Generally, we can say that the existence of the eight profiles across countries implies that a teacher has more than one mediation activity and that mediations are not carried out by teachers in exactly the same way. They have to mediate very differently when students develop different profiles (e.g., within an inclusive mathematics classroom). The empirical profiles show implicit knowledge regarding teachers' mediation activities, which may take the form of an "emotional-social presence," caring for emotional-social integration in the class (Profile 1–GER, Profile 2–GER, Profile 1–FIN, and Profile 2–FIN); a "potential developer" with a versatile presence in class (Profiles 2–

GER and Profiles 2–FIN); a "self-improvement supporter", who, as the name suggests, supports self-improvement (Profiles 2–GER, Profile 3–GER, Profiles 2–FIN, and Profile 4–FIN); and finally, a "pressure supporter," who brings pressure to bear on students (Profile 4–GER). Almost all mediating roles were found across countries, apart from "pressure supporter", which only emerged in Profile 4–GER.

Taking into account the profiles' sample sizes, in both countries the profiles of emotionalsocial integration (Profile 1–GER and Profile 1–FIN) and enjoyment (Profile 2–GER and Profile 2–FIN) are preferred most. In light of middle adolescence's preference for an emotionally warm teacher–student relationship, the teacher supports learners' internalization in the mathematics class through an emotional-social presence. Since these profiles are the most preferred ones in all the groups, this specific mediation activity as an emotional-social presence supports learners during middle adolescence on the way to becoming self-determined human beings and thus has importance for their mathematics-learning processes as well as their personal development.

A prototype for acting in many ways was implied by profiles that were marked by "enjoyment in mathematics learning," these profiles emphasize how teachers serve many different relevance constructions in individual as well as social processes. Again, these profiles exist independently from cultural settings in both countries (having no culture-specific focus). Moreover, this learning orientation experienced all three specific personal meanings, and it is assumed that the social setting of these profiles fulfils all three basic psychological needs; this is because these learners identified with fun as well as relevance, and this directed them towards self-determined behaviour in the mathematics class. Accordingly, this profile, above all the others, seems excellent for teaching, and the activities of the teacher as a potential developer have a powerful effect in supporting learners' learning processes and autonomous motivation. Accordingly, this means that the teacher has to organize mediation in a variety of ways, as not all students are the same.

In conclusion, a teacher as a potential developer can support, from the learner's point of view, self-determined learning in mathematics if they act as a potential developer serving the roles of emotional-social presence and self-improvement supporter, but, importantly, *not* the role of pressure supporter. Of course, further research is necessary to underpin these limited insights. To summarize, the insights into learners' classroom experiences were not empirical and thus somewhat incomplete. Therefore, no profound knowledge can be gained about learners' experiences from their mathematics curricula and the teachers' mediating activities. Although implicit knowledge in this connection exists, we would have to observe these conventionalization processes, which affect the learners' development processes, from the perspective of the specific profiles in question.

8.4.2 Limitations of the Methodological Approach

To investigate the given ontological characteristics of the empirical phenomenon personal meaning, I carried out research activities following a sequential-dependent (Schoonenboom-Johnson, 2017) research design. Not only did the inclusion of these practices enriched the investigation by adding different methodological perspectives, but their integration also allowed a more detailed insight into the research topic and an increase in knowledge in terms

of the *problematique*, work on the conceptual understanding between personal meaning and motivation in mathematics.

The mono-methodological quantitative research approach as well as the coordinating research approach carried out in this dissertation produced evidence supporting the linkage between the social and the individual (SDT and social constructivism) as well as the existence of cultural differences, potentially affected by the mediating role of the teacher and the educational system in both cultural settings. Based on the quantitative approach, support of the dimensionality of personal meaning initially highlighted an implicit theoretical link between socially experienced personal meaning and the regulatory styles (SDT). The empirical stability of the predicted dimensions of personal meaning not only supported studying the psychometric properties in a meaningful way; their emergence within the empirical profiles also helped to characterize students' mathematics-learning orientations and to work out the spectrum of possible forms of action, based on profile levels, of how a teacher can act to support different profiles in their relevance construction and behaviour.

The correlation analyses across countries not only explored the predicted connection between these meta-factors-orientation towards the regulatory styles and to the relatedness of mathematics-that is, a positive linear connection between (kinds/dimensions of) personal meaning and quality of motivation; they also showed that Finnish students associate their personal meanings more strongly with self-determined types of motivation than German learners do. Analogous theoretical consistency emerged by contrasting the latent mean structure in which the student's constructed personal relevance was in line with the country's motivational peculiarity. Furthermore, these valid group comparisons undergirded and somewhat explained the correlational structures detected across groups: The correlation pattern in Germany showed most associations with controlled regulatory styles and resulted similarly in a latent mean structure in which German students identified less with individual and mathematics-related personal relevance than Finnish learners did. The sequential-dependent methodology also added to the value of certain empirical results and provided an enhanced insight into their role through the strategy of coordinating. That is, regarding the role of the three specific personal meanings within learners' social interactions, there was no clear evidence within the empirical work that clarified the roles of these three specific meanings. The coordination analysis helped to point out their roles for learners' individual relevance systems and to understand the differences in learners' biological regulations by theoretically describing their social experiences in the classroom. However, the quantitative statistical analyses provided the empirical basis for understanding these results in more depth within the model of feedback mechanism built by coordinating the theories.

Accordingly, the additional value of integrating these research activities, which are motivated by ontological assumptions (Buchholtz, 2021), can be underlined through the findings within the hypothetical model of a feedback mechanism. Although the networking approach compensated for the weaknesses of the mono-methodological quantitative approach, the strategy of coordinating presented the author with its own challenges when it came to conduct a theory-driven interpretation procedure of numerical data. In this spirit, the adjusted terminologies "identify with," "not identify with," and "enacting potential" (originating within qualitative research paradigm; Bohnsack, 1989) not only helped to validate the computed profiles in terms of theoretical consistency, that is, judged by the lens of social constructivism;

they also provided a secure framework in which to carefully conduct the coordinating analysis of standardized data (qualifying the quantitative data for each evaluation step that was carried out), and to benefit from the knowledge gained in light of this dissertation's specific networking activity and of motivation-relevant articulation of both theories (Prediger et al., 2008).

Limitations of Investigation. In terms of the methodological issues that led to the theoretical limitations of this dissertation, despite the usefulness of the two independent data sets, it is important to note that this study could not consider each country equally. Whereas the reconstructed personal meanings concern the kinds of personal relevance German (and Hong Kongese) students assign to mathematics learning, Finnish learners' personal meanings that specifically address the Finnish cultural context are not included. Because of this limitation it was not possible to decode the "what" (certain personal meanings) that support Finnish learners' autonomy experiences in Profile 3-FIN "External pressure focusing on adaptation to the educational system." Although the students did not construct EXPERIENCE OF AUTONOMY, COMPETENCE, and SOCIAL RELATEDNESS within their social context, the motivational orientation only showed tendency for controlled motivation to be associated with having identified regulation. Accordingly, the students seem to construct personal meaning(s) that reflect a personal importance (identified regulation) in mathematics learning. So, which personal relevance associates with identified motivation in this profile? In view of supporting autonomously motivated learning, it would be interesting to explore what kind of relevance Finnish culture offers to foster students' basic needs satisfaction which is not included within the German classroom (e.g., Profile 4–GER) considered here.

Further, this study's profiles cannot easily claim validity for the whole country. For this reason, the results and evidence gained here cannot be generalized to the whole country; that is, the generalizability and representativeness (a relatively small sample size is given) are very limited. Moreover, the hypothetical model of a feedback mechanism (cf. 7.4.4) is only based on theoretical knowledge, albeit evidence-based. To understand to what extent the social and the biological regulatory mechanisms interact in respect of the genesis of learners' motivation, further investigation is necessary. Accordingly, no causal effects can be derived.

Concerning the evidence found in Germany and Finland, the results did not include any classroom observations about the processes of conventionalization. For that reason, the social cultural context was only reconstructed on the basis of the individual patterns of personal meaning that learners constructed. Additional empirical observations could have helped explore interpersonal conversations between teachers and students and teachers' mediation activities (in connection with teachers' need experiences) in depth and also to investigate the question of what specific learning situations support students' personal experiences when constructing the specific personal meanings of EXPERIENCE OF AUTONOMY, COMPETENCE, AND SOCIAL RELATEDNESS.

In summary, in this study's context, connecting theories in quantitative studies is helpful not only to solve practical problems in research analysis but also to broaden one's mind with respect to implicit knowledge behind applied theoretical approaches and pointing out the methodical benefits and technical obstacles for future research.

8.5 Prospects

8.5.1 Implications for Future Research

In order to increase the understanding of autonomy-supportive teaching in connection with students' individual relevance systems, the focus in subsequent research should be on integrating qualitative approaches. As already pointed out by Ryan and Deci (2020), more qualitative studies are necessary to obtain a detailed and full picture of learners' everyday classroom experiences.

Supporting this demand, also with regard to the findings of the present dissertation, evidence from qualitative research would help to increase and develop our understanding of the interplay between learners' individual relevance systems and quality of motivation in multiple ways: based on the linkage which was emphasized by the hypothetical feedback mechanism, future research should adopt theoretical perspectives which theorize the learner's experiences on three levels: social, psychological, and embodied (Hannula, 2012). In this connection, empirical observations within the classroom can shed light on learners' everyday experiences, specifically in order to understand how the identified four profiles' social experiences-grounded on teacher-student and student-student interpersonal activitiesaffect their construction of the specific personal meanings (EXPERIENCE OF AUTONOMY, COMPETENCE, and SOCIAL RELATEDNESS). These insights might additionally highlight in which situations the experiences fluctuate on an intra-individual level and what decisive factors direct the profile formation. Alongside these observations, to get a full picture, teachers' mediation activities and especially their basic-psychological-needs experiences should also be observed as a counterpart to learners' experiences within educational reality. Ryan and Deci (2020) stated that the satisfaction of teachers' basic psychological needs has an impact on their students' learning processes. These insights can be used to explore to what extent teachers' needs fulfilment/frustration affects their mediation activities and in turn learners' construction of their individual relevance systems. The knowledge gained could subsequently help to design interventions (as suggested by Vansteenkiste et al., 2018), not only in order for learners to experience autonomy support in the mathematics classroom but also to allow teachers to satisfy their own basic psychological needs within their teaching activities.

Furthermore, on the basis of this qualitative research evidence, the intensity between social ("intensity of trigger") and biological regulatory mechanisms ("intensity of feedback") could subsequently be quantitatively assessed, whereby the networking approach could be applied to a qualitative and quantitative database in order to refine the coordinated conceptual framework that explains the development of motivation in the mathematics classroom. Hence, future research needs not only to care about students' autonomous motivations, but also teachers' well-being in mathematics education in order to sustainably support learners' autonomous motivation in the mathematics classroom.

8.5.2 Implications for Teaching and Educational Policy

Autonomously Oriented Mathematics Learning in the Classroom. What do these profiles mean for teaching mathematics? Based on the evidence gained, certain perspectives can be used to support learners' autonomously motivated learning in the classroom. Of the 17 personal

meanings, the three specific personal meanings (EXPERIENCE OF AUTONOMY, COMPETENCE, and SOCIAL RELATEDNESS) proved their power consistently in all profiles regardless of culture. Accordingly, teachers could create learning situations and impulses that support the experience of those three particular personal meanings. Thereby, even though it cannot be guaranteed that each learner (as an individual snowflake) can become interested in mathematics (Bikner-Ahsbahs, 2005; Ryan & Deci, 2017), there exists good potential for them to experience autonomous motivation and to personally value their mathematics learning.

As pointed out, Profile 2, "Enjoyment in mathematics learning", exists regardless of culture. Here, the teaching style was heterogeneous: as potential developers, teachers acted in a versatile manner: they supported emotional-social integration and self-improvement, importantly, free from pressure. Significantly, as emphasized by several scholars within SDT (Haerens et al., 2016; Reeve, 2016; Ryan & Deci, 2017), the exclusion of teachers' controlling behaviour also had here a positive impact on learners' experiences in the classroom that led them to construct a mental compass in terms of enjoyment and psychological well-being in that classroom, independent of the cultural context. That is, the teacher should perform and arrange various mediation activities in a number of ways in order to suit the individual profiles, which are different due to the biological development process considered here. Because all profiles across the countries displayed identified motivation apart from Profile 4 in Germany (where the teacher is connected with external pressure), the mediating role of "Pressure supporter" should be avoided to support learners' experiences of self-determination.

Other interesting evidence referred to the most valued profile in both countries, namely Profiles 1 in Germany and Finland, "Emotional–social integration". Irrespective of social cultural contexts and the foci within mathematics curricula, students wished for a warm bond with a mathematics teacher who mediated "emotional-social presence". SDT stated that feedback on learners' performance (Ryan & Deci, 2020) can have a meaning of that is of functional importance to students. However, these profiles showed that teachers' emotional-social presence is not limited to providing students with meaning and providing them with learning orientation in mathematics. This specific teacher–student interrelation can also be used to accompany the progressive intensification of students' internalization processes (EXPERIENCE OF COMPETENCE AND SOCIAL RELATEDNESS), particularly in the early stages of those processes.

Foci within Mathematics Curricula. Besides the differences between German and Finnish mathematics curricula, in both countries the identified overall learning orientations were similar, just having differences in nuance. Nonetheless, the social cultural influences were visible in respect of autonomy support in the mathematics classroom. An "excessive emphasis on grades, performance goals, (...)" (Ryan & Deci, 2020, p. 10) and also controlling teaching behaviour on the part of teachers (e.g., the mediating role of the teacher as "pressure supporter") were prominent within the profiles from Germany, but not within the Finnish mathematics-learning orientations. All profiles in Finland display identified motivation, whereas in Germany, Profile 4, oriented towards "external pressure," did not display identified motivation.

The comparison between the two countries did not intend to judge whether one is better than the other. Each country has to face individual issues within its educational system. Furthermore, heterogeneity in Germany (population ca. 83.1 million in 2020; Statista, 2021) is much greater than within Finland (population ca. 5.5 million in 2020; Statista, 2021), and accordingly the strategies and instructions within the Finnish educational system cannot be simply applied to German classrooms by ignoring the individual differences between them.

Each learner is connected to their cultural setting and therefore they should be treated as such. If the German context does not have these cultural amplifiers or mediators, we basically cannot expect that profiles will arise like those that are produced in Finland, because that does not work in a different cultural setting.

Instead of "copying" the Finnish instructions, we could learn from each other and "refresh" our view. The comparison intended to improve our perspective by questioning our own picture of the culture of mathematics education. If one investigates the school system where one has grown up, experience of something foreign helps one to question procedures that were hitherto accepted as natural. In this spirit, the reinterpretation of the German results made by contrasting them with the Finnish profiles showed that a differentiated perspective on the German mathematics curricula is necessary to support autonomous mathematics learning. Similar to the individual educational school systems of the 16 federal states, consideration of each learner's individual affective factors during their development is just as important. Although German policies have added further objectives (e.g., digitalization; KMK, 2019, p. 307) over recent years, there is no explicit objective concerning learners' personal development (like the transversal competencies in Finnish basic education; Finnish National Board of Education, 2016). Mentioning this topic explicitly within the mathematics curriculum may induce educators to take learners' personal development just as seriously as they do the "performance standards" (Blum, 2006; Heymann, 2005) for mathematics lessons.

8.6 Conclusion

The findings of this dissertation can be condensed into two central concluding remarks: Firstly, the evidence of the present investigation points out that besides the support of learners' basic psychological needs, the interpersonal relationship between teachers and students is very important. Like snowflakes, each learner is unique and thus they bring their individual motivations into the classroom to study mathematics. What is more, all learners are in the process of becoming self-determined individuals, and therefore they need an emotionally warm relationship with a teacher (Wentzel, 2009, 2012), who enables them to feel effective in the mathematics classroom. Because of this, experiencing an emotionally warm bond with the teacher or when the teacher mediates very differently in the classroom can support students to engage in mathematics learning for their own sake. This in turn simultaneously promotes learners' autonomous motivation.

In this regard, although teachers face difficulties in their everyday classroom routine, if they deliberately refrain from controlling teaching techniques (Reeve, 2009) this can help encourage learners to build a positive relationship with their teacher, which kicks off the internalization process and eventually leads to their attaining self-determination. Eschewing controlling educational styles does not entail a reduction in the structure and guidance provided, however. Indeed, many students find structure useful as a means of countering anxiety and uncertainty (Haerens et al., 2016; Mouratidis et al., 2013).

As these findings show, the interpersonal relationship between the teacher and the student is a decisive factor in students' learning processes as well as in their personal development. Although this aspect seems to be a helpful strategy in supporting the learner's autonomy experience as well as their mathematics-learning process, it creates a critical concern when access to mathematics education in school is limited (e.g., due to the COVID-19 pandemic crisis of 2020/2021) and it is not yet clear how educators can counterbalance the negative impacts of this on learners' cognitive development (Zacharopoulos et al., 2021).

This is because, according to the second concluding remark, mathematics learning is an interdependence of embodied, psychological, and social phenomena (de Freitas & Sinclair, 2014; Ferrari & Ferrara, 2020; Hannula, 2012; Roth & Walshaw, 2019). Therefore, the affective quality of the teacher–student relationship has a huge impact on learners' autonomously oriented motivation, emotional well-being (Wentzel, 2009), and the thus-resulting cognitive development (Hannula, 2012; Vygotsky, 1987). The present investigation's results emphasize (further) that besides a positive bond between teacher and student, we have to consider the individual's body and mind—or affect and intellect—through a monist perspective instead of a dualist approach (Hannula, 2012; Roth & Walshaw, 2019; Vygotsky, 1987). An intertwined view while teaching (and within our research) may holistically promote students' mathematics-learning processes, in order to support them on their individual journeys towards actualizing themselves as fully self-determined individuals.
References

- Abramson, L. Y., Seligman, M. E., & Teasdale, J. D. (1978). Learned helplessness in humans: Critique and reformulation. *Journal of Abnormal Psychology*, 87(1), 49–74. https://doi.org/10.1037/0021-843X.87.1.49
- Akkerman, S. F., & Bakker, A. (2011). Boundary crossing and boundary objects. *Review of Educational Research*, 81(2), 132–169. https://doi.org/10.3102/0034654311404435
- American Psychological Association. (2020). *Publication manual of the American Psychological Association* (7th ed.) https://doi.org/10.1037/0000165-000
- Asparouhov, T., Muthén, B. (2010, September 29). *Multiple imputation with Mplus: Technical implementation*. Retrieved September 5, 2020, from http://statmodel2.com/download/Imputations7.pdf
- Assor, A., Kaplan, H., & Roth, G. (2002). Choice is good, but relevance is excellent: Autonomy-enhancing and suppressing teacher behaviours predicting students' engagement in schoolwork. *The British Journal of Educational Psychology*, 72(2), 261–278. https://doi.org/10.1348/000709902158883
- Assor, A., Vansteenkiste, M., & Kaplan, A. (2009). Identified versus introjected approach and introjected avoidance motivations in school and in sports: The limited benefits of selfworth strivings. *Journal of Educational Psychology*, 101(2), 482–497. https://doi.org/10.1037/a0014236
- Assude, T., Boero, P., Herbst, P., Lerman, S., & Radford, L. (2008). *The Notions and Roles of Theory in Mathematics Education Research*. Paper presented by the Survey Team at ICME 11. Mexiko 2008. Retrieved June 2, 2021, from https://www.mathunion.org/fileadmin/ICMI/files/About_ICMI/Publications_about_ICMI/ ICME_11/Assude.pdf
- Autio, T. (2021). From Knowledge and Bildung Toward Competences and Skills in Finnish Curriculum Policy?: Some Theoretical, Historical, and Current Observations Related to Finland. In W. Zhao & D. Tröhler (Eds.), Euro-Asian Encounters on 21st-Century Competency-Based Curriculum Reforms (pp. 41-56). Springer. https://doi.org/10.1007/978-981-16-3009-5
- Backhaus, K., Erichson, B., & Weiber, R. (2015). *Fortgeschrittene Multivariate Analysemethoden*. Springer. https://doi.org/10.1007/978-3-662-46087-0
- Bakker, A., Smit, J., & Wegerif, R. (2015). Scaffolding and dialogic teaching in mathematics education: Introduction and review. *ZDM*, *47*(7), 1047–1065. https://doi.org/10.1007/s11858-015-0738-8
- Barkoukis, V., Taylor, I., Chanal, J., & Ntoumanis, N. (2014). The relation between student motivation and student grades in physical education: A 3-year investigation. *Scandinavian Journal of Medicine & Science in Sports*, 24(5), e406-14. https://doi.org/10.1111/sms.12174

REFERENCES

- Bartholomew, K. J., Ntoumanis, N., Mouratidis, A., Katartzi, E., Thøgersen-Ntoumani, C., & Vlachopoulos, S. (2018). Beware of your teaching style: A school-year long investigation of controlling teaching and student motivational experiences. *Learning and Instruction*, 53, 50–63. https://doi.org/10.1016/j.learninstruc.2017.07.006
- Bentler, P. M. (2009). Alpha, dimension-free, and model-based internal consistency reliability. *Psychometrika*, 74(1), 137–143. https://doi.org/10.1007/S11336-008-9100-1
- Benware, C. A., & Deci, E. L. (1984). Quality of learning with an active versus passive motivational set. *American Educational Research Journal*, *21*(4), 755–765. https://doi.org/10.3102/00028312021004755
- Bernard, H. R., & Ryan, G. W. (2010). *Analyzing qualitative data: Systematic approaches*. Sage.
- Biehler, R., & Hartung, R. (2006). Die Leitidee Daten und Zufall. In W. Blum, C. Drüke-Now, R. Hartung, & O. Köller (Eds.), *Bildungsstandards Mathematik konkret. Sekundarstufe I: Aufgabenbeispiele, Unterrichtsanregungen, Fortbildungsideen* (4th ed., pp. 51–80). Cornelsen.
- Bikner-Ahsbahs, A. (2005). Mathematikinteresse zwischen Subjekt und Situation: Theorie interessendichter Situationen - Baustein für eine mathematikdidaktische Interessentheorie.
 Zugl.: Flensburg, Univ., Habil.-Schr., 2003. Texte zur mathematischen Forschung und Lehre: Vol. 43. Franzbecker.
- Bikner-Ahsbahs, A. (2014). Turning Disinterest into Interest in Class: An Intervention Study. In C. Nicol, P. Liljedahl, S. Oesterle, & D. Allan (Eds.) *Proceedings of the Joint Meeting* 2 – 145 of PME 38 and PME-NA 36, Vol. 2, pp. 145-152. Vancouver, Canada: PME.
- Bikner-Ahsbahs, A. (2015). Empirically grounded building of ideal types. A methodical principle of constructing theory in the interpretative research in mathematics education. In A. Bikner-Ahsbahs, C. Knipping, & N. Presmeg (Eds.), *Advances in Mathematics Education. Approaches to Qualitative Research in Mathematics Education* (pp. 105–135). Springer Netherlands.
- Bikner-Ahsbahs, A., & Prediger, S. (2014a). Networking as research practices:
 Methodological lessons learnt from the case studies. In A. Bikner-Ahsbahs & S. Prediger (Eds.), *Networking of theories as a research practice in mathematics education* (pp. 235–247). Springer International Publishing.
- Bikner-Ahsbahs, A., & Prediger, S. (Eds.). (2014b). Networking of Theories as a Research Practice in Mathematics Education. Springer International Publishing. https://doi.org/10.1007/978-3-319-05389-9
- Biller, K. (1991). Habe Sinn und wisse Sinn zu wecken! Sinntheoretische Grundlagen der Pädagogik. Schneider.
- Blum, W. (2006). Einführung. In W. Blum, C. Drüke-Now, R. Hartung, & O. Köller (Eds.), Bildungsstandards Mathematik: konkret. Sekundarstufe I: Aufgabenbeispiele, Unterrichtsanregungen, Fortbildungsideen (4th ed., pp. 14–32). Cornelsen.

Blumenberg, H. (1981). Die Lesbarkeit der Welt. Suhrkamp.

Blumer, H. (1969). Symbolic interactionism: Perspective and method. Prentice-Hall.

- Boekaerts, M. (1999). Self-regulated learning: Where we are today. *International Journal of Educational Research*, *31*(6), 445–457. https://doi.org/10.1016/S0883-0355(99)00014-2
- Bofah, E. A.-t., & Hannula, M. S. (2015). Studying the factorial structure of Ghanaian twelfth-grade students' views on mathematics. In B. Pepin & B. Roesken-Winter (Eds.), Advances in Mathematics Education. From beliefs to dynamic affect systems in mathematics education (pp. 355–381). Springer International Publishing. https://doi.org/10.1007/978-3-319-06808-4 18
- Bohnsack, R. (1989). Generation, Milieu und Geschlecht: Ergebnisse aus Gruppendiskussionen mit Jugendlichen. Zugl.: Erlangen-Nürnberg, Univ., Habil.-Schr. Biographie und Gesellschaft: Vol. 8. Leske + Budrich.
- Bohnsack, R. (1999). *Rekonstruktive Sozialforschung*. *Einführung in Methodologie und Praxis qualitativer Forschung*. Opladen Leske + Budrich.
- Bohnsack, R. (2014). *Rekonstruktive Sozialforschung: Einführung in qualitative Methoden* (9th ed). *UTB Erziehungswissenschaft, Sozialwissenschaft: Vol. 8242*. Barbara Budrich. https://d-nb.info/1049403444/04
- Bollen, K. A. (1989). *Structural equations with latent variables*. John Wiley & Sons. https://doi.org/10.1002/9781118619179
- Borkenau, P. (2006). Selbstbericht. In F. Petermann, & M. Eid, (Eds.), *Handbuch der Psychologischen Diagnostik* (pp. 135–142). Göttingen Hogrefe.
- Brehm, J. W. (1966). *A theory of psychological reactance. Social psychology*. Academic Press.
- Brown, T. A. (2006). *Confirmatory factor analysis for applied research. Methodology in the social sciences.* The Guilford Press.
- Bruner, J. S. (2002). Acts of meaning. The Jerusalem-Harvard lectures. Harvard Univ. Press.
- Buchholtz, N. (2021). Voraussetzungen und Qualitätskriterien von Mixed-Methods-Studien in der mathematikdidaktischen forschung. *Journal für Mathematik-Didaktik*, 42(1), 219– 242. https://doi.org/10.1007/s13138-020-00173-0
- Bühner, M. (2011). *Einführung in die Test- und Fragebogenkonstruktion* (3rd ed). *Psychologie*. Pearson Studium.
- Büssing, J. (2016). Zur Dimensionalität der Sinnkonstruktionstypen "Erfüllung gesellschaftlich geprägter Anforderungen" und "Aktive Auseinandersetzung mit Mathematik": Eine quantitative Fragebogenstudie: [On the dimensionality of the types of personal meaning "interaction with mathematical contents" and "emotionally effected evolvement": A quantitative study using a survey]. (Unpublished master's thesis).
- Byrne, B. M. (2000). Measuring self-concept across culture: Issues, caveats, and practice. In R. G. Craven & H. W. Marsh (Eds.), Self-concept theory, research & practice: advances for the new millennium: Proceedings of the inaugural international conference, university of western Sydney, Australia, 5-6 October 2000 (pp. 30–41). University of Western Sydney, SELF Research Centre.

- Byrne, B. M., Shavelson, R. J., & Muthén, B. (1989). Testing for the equivalence of factor covariance and mean structures: The issue of partial measurement invariance. *Psychological Bulletin*, 105(3), 456–466. https://doi.org/10.1037/0033-2909.105.3.456
- Byrnes, J. P. (2001). *Minds, brains, and learning: Understanding the psychological and educational relevance of neuroscientific research.* Guilford Press.
- Chen, B., Vansteenkiste, M., Beyers, W., Boone, L., Deci, E. L., van der Kaap-Deeder, J., Duriez, B., Lens, W., Matos, L., Mouratidis, A., Ryan, R. M., Sheldon, K. M., Soenens, B., van Petegem, S., & Verstuyf, J. (2015). Basic psychological need satisfaction, need frustration, and need strength across four cultures. *Motivation and Emotion*, 39(2), 216– 236. https://doi.org/10.1007/s11031-014-9450-1
- Chen, F. F. (2007). Sensitivity of goodness of fit indexes to lack of measurement invariance. *Structural Equation Modeling: A Multidisciplinary Journal*, *14*(3), 464–504. https://doi.org/10.1080/10705510701301834
- Chen, F. F. (2008). What happens if we compare chopsticks with forks? The impact of making inappropriate comparisons in cross-cultural research. *Journal of Personality and Social Psychology*, 95(5), 1005–1018. https://doi.org/10.1037/a0013193
- Cheung, G. W., & Rensvold, R. B. (1998). Cross-cultural comparisons using non-invariant measurement items. *Applied Behavioral Science Review*, 6(1), 93-110.
- Cheung, G. W., & Rensvold, R. B. (2002). Evaluating goodness-of-fit indexes for testing measurement invariance. *Structural Equation Modeling: A Multidisciplinary Journal*, 9(2), 233–255. https://doi.org/10.1207/S15328007SEM0902_5
- Chirkov, V., Ryan, R. M., Kim, Y., & Kaplan, U. (2003). Differentiating autonomy from individualism and independence: A self-determination theory perspective on internalization of cultural orientations and well-being. *Journal of Personality and Social Psychology*, 84(1), 97–110. https://doi.org/10.1037/0022-3514.84.1.97
- Cobb, P., Yackel, E., & Wood, T. (1992). A constructivist alternative to the representational view of mind in mathematics education. *Journal for Research in Mathematics Education*, 23(1), 2–33. https://doi.org/10.5951/jresematheduc.23.1.0002
- Cohen, J. (1988). *Statistical Power Analysis for the Behavioral Sciences* (2nd ed.). Erlbaum. http://gbv.eblib.com/patron/FullRecord.aspx?p=1192162
- Combe, A. (2014). Dialog und Verstehen im Unterricht. Lernen im Raum von Phantasie und Erfahrung. In U. Gebhard (Ed.), Sinn im Dialog: Zur Möglichkeit sinnkonstituierender Lernprozesse im Fachunterricht (1st ed., pp. 51–66). Springer VS.
- Covington, M. (1998). Self-worth and the fear of failure. In M. V. Covington (Ed.), *The will to learn: A guide for motivating young people* (pp. 77–103). Cambridge University Press.
- Covington, M. (2009). Self-worth theory: retrospection and prospects. In K. R. Wenzel & A. Wigfield (Eds.), *Handbook of motivation at school* (pp. 141–169). Routledge/Taylor & Francis Group.
- Csikszentmihalyi, M. (1975). *Beyond boredom and anxiety. The Jossey-Bass behavioral science series.* Jossey-Bass. http://www.loc.gov/catdir/enhancements/fy0607/76007234-b.html

de Ayala, R. J. (2009). The theory and practice of item response theory. Guilford Press.

- de Charms, R. (1968). *Personal causation: The internal affective determinants of behavior*. Academic.
- de Freitas, E., & Sinclair, N. (2014). *Mathematics and the Body: Material Entanglements in the Classroom*. Cambridge University Press https://doi.org/10.1017/CBO9781139600378
- Deci, E. L. (1971). Effects of externally mediated rewards on intrinsic motivation. *Journal of Personality and Social Psychology*, 18(1), 105–115. https://doi.org/10.1037/h0030644
- Deci, E. L. (1975). Intrinsic motivation. Perspectives in social psychology. Plenum Press.
- Deci, E. L. (1992). The relation of interest to the motivation of behavior: A selfdetermination theory perspective. In K. A. Renninger, S. Hidi, & A. Krapp (Ed.), *The role* of interest in learning and development (pp. 43–70). Lawrence Erlbaum Associates, Inc.
- Deci, E. L., & Ryan, R. M. (1985). *Intrinsic motivation and self-determination in human behavior*. *Perspectives in social psychology*. Plenum.
- Deci, E. L., & Ryan, R. M. (1991). A motivational approach to self: Integration in personality. In R. A. Dienstbier (Ed.), *Nebraska symposium on motivation: Vol. 38. Perspectives on motivation* (pp. 237–288). Lincoln, NE: University of Nebraska Press.
- Deci, E. L., & Ryan, R. M. (1993). Die Selbstbestimmungstheorie der Motivation und ihre Bedeutung f
 ür die P
 ädagogik. Zeitschrift f
 ür P
 ädagogik, 39(2), 223–238. https://doi.org/10.25656/01:11173
- Deci, E. L., & Ryan, R. M. (1995). Human autonomy. In M. H. Kernis (Ed.), *Efficacy*, agency, and self-esteem (pp. 31–49). Springer US. https://doi.org/10.1007/978-1-4899-1280-0 3
- Deci, E. L., & Ryan, R. M. (2000). The "what" and "why" of goal pursuits: Human needs and the self-determination of behavior. *Psychological Inquiry*, 11(4), 227–268. https://doi.org/10.1207/S15327965PLI1104_01
- Deci, E. L., & Ryan, R. M. (Eds.). (2002). *Handbook of self-determination research*. University of Rochester Press.
- Deci, E. L., & Ryan, R. M. (2016). Optimizing students' motivation in the era of testing and pressure: A self-determination theory perspective. In W. C. Liu, J. C. K. Wang, & R. M. Ryan (Eds.), *Building autonomous learners* (pp. 9–29). Springer Singapore.
- Deci, E. L., Vallerand, R. J., Pelletier, L. G., & Ryan, R. M. (1991). Motivation and education: The self-determination perspective. *Educational Psychologist*, 26(3-4), 325– 346. https://doi.org/10.1080/00461520.1991.9653137
- Denzin, N. K. (1970). The Research Act. Chicago Aldine.
- Denzin, N. K. (1989). *The research act: A theoretical introduction to sociological methods* (3rd ed.). Prentice-Hall.
- Döring, N., & Bortz, J. (2016). Forschungsmethoden und Evaluation in den Sozial- und Humanwissenschaften. Springer https://doi.org/10.1007/978-3-642-41089-5

- Drodge, E. N., & Reid, D. A. (2000). Embodied cognition and the mathematical emotional orientation. *Mathematical thinking and learning*, 2(4), 249-267. Lawrence Erlbaum Associates, Inc.
- Eccles, J. S., Adler, T. F., Futterman, R., Goff, S. B., Kaczala, C. M., Meece, J. L., &
 Midgley, C. (1983). Expectancies, values and academic behaviors. In J. T. Spence (Ed.), *Achievement and achievement motives* (pp. 74-146). San Francisco, CA: W. H. Freeman.
- Enders, C. K. (2010). *Applied missing data analysis*. *Methodology in the social sciences*. Guilford Press.
- Ernest, P. (1998). Social Constructivism as a Philosophy of Mathematics. SUNY Press.
- Ernest, P. (2010). Reflections on theories of learning. In B. Sriraman & L. English (Eds.), *Theories of mathematics education* (pp. 39–47). Springer Berlin Heidelberg. https://doi.org/10.1007/978-3-642-00742-2 4
- Farmer, E., Papadopoulos, N., Emonson, C., Fuelscher, I., Pesce, C., McGillivray, J., Hyde, C., Olive, L., & Rinehart, N. (2020). A preliminary investigation of the relationship between motivation for physical activity and emotional and behavioural difficulties in children aged 8-12 years: The role of autonomous motivation. *International Journal of Environmental Research and Public Health*, 17(15). https://doi.org/10.3390/ijerph17155584
- Ferrari, G., & Ferrara, F. (2020). Affective Bonds and Mathematical Concepts: Speaking of Affect Through Sympathy. In C. Andrà, D. Brunetto, & F. Martignone (Eds.), *Theorizing* and Measuring Affect in Mathematics Teaching and Learning (pp. 11-20). Springer International Publishing. https://doi.org/10.1007/978-3-030-50526-4_2
- Fiedler, K., Kutzner, F., & Krueger, J. I. (2012). The long way from α-error control to validity proper: Problems with a short-sighted false-positive debate. *Perspectives on Psychological Science: A Journal of the Association for Psychological Science*, 7(6), 661– 669. https://doi.org/10.1177/1745691612462587
- Fielding, N., & Fielding, J. (1986). *Linking Data*. SAGE Publications, Inc. https://doi.org/10.4135/9781412984775
- Finnish National Agency for Education. (2017). Finnish education in a nutshell: Education in Finland. Retrieved January 21, 2021, from https://www.oph.fi/sites/default/files/documents/finnish education in a nutshell.pdf
- Finnish National Board of Education. (2016). *National Core Curriculum for Basic Education* 2014.
- Fischer, R. & Malle, G. (1985). *Mensch und Mathematik. Eine Einführung in didaktisches Denken und Handeln.* Wissenschaftsverlag.
- Flick, U. (2006). *An introduction to qualitative research* (3rd ed.). Sage. http://www.loc.gov/catdir/enhancements/fy0658/2005927681-d.html
- Flick, U. (2008). *Triangulation: Eine Einführung* (2nd ed.). *Qualitative Sozialforschung: Bd.* 12. VS Verlag für Sozialwissenschaften.
- Frank, J. D. (1941). Recent studies of the level of aspiration. *Psychological Bulletin*, 38(4), 218–226. https://doi.org/10.1037/h0059344

- Frankl, V. E. (1978). *The unheard cry for meaning*. Simon and Schuster. http://search.ebscohost.com/login.aspx?direct=true&scope=site&db=nlebk&AN=1973165
- French, B. F., & Finch, W. H. (2016). Detecting differential item functioning. In K. Schweizer & C. DiStefano (Eds.), *Principles and methods of test construction: Standards* and recent advances (pp. 197–217). Hogrefe Publishing.
- Frenzel, A. C., Goetz, T., Pekrun, R., & Watt, H. M. G. (2010). Development of mathematics interest in adolescence: Influences of gender, family, and school context. *Journal of Research on Adolescence*, 20(2), 507–537. https://doi.org/10.1111/j.1532-7795.2010.00645.x
- Frenzel, A. C., Pekrun, R., Dicke, A.-L., & Goetz, T. (2012). Beyond quantitative decline: Conceptual shifts in adolescents' development of interest in mathematics. *Developmental Psychology*, 48(4), 1069–1082. https://doi.org/10.1037/a0026895
- Gass, S. M., & Mackey, A. (2000). Stimulated recall methodology in second language research. Second language acquisition research Monographs on research methodology. Erlbaum. Retrieved July 22, 2021, from http://www.loc.gov/catdir/enhancements/fy0634/99036248-d.html
- Gebhard, U. (2014). Einleitung: Sinn im Dialog: Zur Möglichkeit sinnkonstituierender Lernprozesse in der Schule. In U. Gebhard (Ed.), *Sinn im Dialog: Zur Möglichkeit sinnkonstituierender Lernprozesse im Fachunterricht* (1st ed., pp. 7–8). Springer VS.
- Geiser, C. (2011). *Datenanalyse mit Mplus: eine anwendungsorientierte Einführung* (2nd ed.). VS Verlag für Sozialwissenschaften. https://doi.org/10.1007/978-3-531-92042-9
- Geldhof, G. J., Preacher, K. J., & Zyphur, M. J. (2014). Reliability estimation in a multilevel confirmatory factor analysis framework. *Psychological Methods*, 19(1), 72–91. https://doi.org/10.1037/a0032138
- Ghys, É. (2015). The butterfly effect. In S. J. Cho (Ed.), *The proceedings of the 12th international congress on mathematical education* (pp. 19–39). Springer International Publishing. https://doi.org/10.1007/978-3-319-12688-3_6
- Goldin, G. A. (2014). Perspectives on emotion in mathematical engagement, learning, and problem solving. In R. Pekrun, & L. Linnenbrink-Garcia (Eds.), *International handbook of emotions in education* (pp. 391–414). Routledge/Taylor & Francis Group.
- Haerens, L., Aelterman, N., Vansteenkiste, M., Soenens, B., & van Petegem, S. (2015). Do perceived autonomy-supportive and controlling teaching relate to physical education students' motivational experiences through unique pathways? Distinguishing between the bright and dark side of motivation. *Psychology of Sport and Exercise*, 16, 26–36. https://doi.org/10.1016/j.psychsport.2014.08.013
- Haerens, L., Vansteenkiste, M., Aelterman, N., & van den Berghe, L. (2016). Toward a systematic study of the dark side of student motivation: Antecedents and consequences of teachers' controlling behaviors. In W. C. Liu, J. C. K. Wang, & R. M. Ryan (Eds.), *Building autonomous learners* (pp. 59–81). Springer Singapore.

REFERENCES

- Halinen, I. (2018). The new educational curriculum in Finland. In M. Matthes, L. Pulkkinen,
 C. Clouder, B. Heys (Eds.) *Improving the Quality of Childhood in Europe* Volume 7 (pp. 75-89). Alliance for Childhood European Network Foundation.
- Hambleton, R. K., Swaminathan, H., & Rogers, H. J. (1991). Fundamentals of item response theory. Measurement methods for the social sciences series: Vol. 2. Sage Publ. http://www.loc.gov/catdir/enhancements/fy0655/91022005-d.html
- Hancock, G. R. (2001). Effect size, power, and sample size determination for structured means modeling and mimic approaches to between-groups hypothesis testing of means on a single latent construct. *Psychometrika*, 66(3), 373–388. https://doi.org/10.1007/BF02294440
- Hannula, M. S. (2004). 'Regulating motivation in mathematics': A paper presented at the Topic Study Group 24 of ICME-10 conference.
- Hannula, M. S. (2006). Motivation in mathematics: Goals reflected in emotions. *Educational Studies in Mathematics*, 63(2), 165–178. https://doi.org/10.1007/s10649-005-9019-8
- Hannula, M. S. (2012). Exploring new dimensions of mathematics-related affect: Embodied and social theories. *Research in Mathematics Education*, 14(2), 137–161. https://doi.org/10.1080/14794802.2012.694281
- Harter, S. (1981). A new self-report scale of intrinsic versus extrinsic orientation in the classroom: Motivational and informational components. *Developmental Psychology*, *17*(3), 300–312. https://doi.org/10.1037/0012-1649.17.3.300
- Havighurst, R. J. (1972). *Developmental tasks and education* (3rd edition). David McKay Company Inc. New York.
- Heymann, H. W. (2005). Garantieren "Standards" einen besseren Mathematikunterricht? *Praxis der Mathematik*, 47(1), 40–41.
- Hickey, D. T. (1997). Motivation and contemporary socio-constructivist instructional perspectives. *Educational Psychologist*, 32(3), 175–193. https://doi.org/10.1207/s15326985ep3203_3
- Hoyle, R. H., & Gottfredson, N. C. (2014). Sample size considerations in prevention research applications of multilevel modeling and structural equation modeling. *Prevention Science: The Official Journal of the Society for Prevention Research*, 16(7), 988–997. https://doi.org/10.1007/s11121-014-0489-8
- Hu, L.-t., & Bentler, P. M. (1999). Cutoff criteria for fit indexes in covariance structure analysis: Conventional criteria versus new alternatives. *Structural Equation Modeling: A Multidisciplinary Journal*, 6(1), 1–55. https://doi.org/10.1080/10705519909540118
- Hui, C. H., & Triandis, H. C. (1985). Measurement in cross-cultural psychology. *Journal of Cross-Cultural Psychology*, 16(2), 131–152. https://doi.org/10.1177/0022002185016002001
- Hull, C. L. (1943). *Principles of behavior: An introduction to behavior theory. The Century psychology series.* Appleton-Century-Crofts.

- Hulleman, C. S., Godes, O., Hendricks, B. L., & Harackiewicz, J. M. (2010). Enhancing interest and performance with a utility value intervention. *Journal of Educational Psychology*, 102(4), 880–895. https://doi.org/10.1037/a0019506
- IBM Corp. (2020). *IBM SPSS Statistics for Windows, Version 27.0* [Computer software]. Armonk, NY: IBM Corp.
- Jang, H. (2008). Supporting students' motivation, engagement, and learning during an uninteresting activity. *Journal of Educational Psychology*, 100(4), 798–811. https://doi.org/10.1037/a0012841
- Janßen, T. (2016). Ausbildung algebraischen Struktursinns im Klassenunterricht: Lernbezogene Neudeutung eines mathematikdidaktischen Begriffs [Dissertation]. Bremen Staats- u. Universitätsbibliothek. Retrieved May 17, 2021, from http://nbnresolving.de/urn:nbn:de:gbv:46-00105386-17
- Keating, D. P. (2004). Cognitive and brain development. In R. M. Lerner & L. Steinberg (Eds.), *Handbook of adolescent psychology* (pp. 45–84). John Wiley & Sons Inc.
- Kelava, A., & Schermelleh-Engel, K. (2012). Latent-State-Trait-Theorie (LST-Theorie). InH. Moosbrugger & A. Kelava (Eds.) Testtheorie und Fragebogenkonstruktion (pp. 363-381). Springer.
- Kelle, U., & Buchholtz, N. (2015). The combination of qualitative and quantitative research methods in mathematics education: A "mixed methods" study on the development of the professional knowledge of teachers. In A. Bikner-Ahsbahs, C. Knipping, & N. Presmeg (Eds.), Advances in Mathematics Education. Approaches to Qualitative Research in Mathematics Education (pp. 321–361). Springer Netherlands. https://doi.org/10.1007/978-94-017-9181-6 12
- Kelle, U., & Kluge, S. (2010). Vom Einzelfall zum Typus: Fallvergleich und Fallkontrastierung in der qualitativen Sozialforschung (2nd ed.). Qualitative Sozialforschung: Vol. 15. VS Verl. für Sozialwiss.
- Kelle, U., & Lüdemann, C. (2019). Bridge assumptions in rational choice theory: Methodological problems and possible solutions. In H.-p. Blossfeld, G. Prein & G. Prein (Eds.), *Rational choice theory and large-scale data analysis* (pp. 112–125). Routledge. https://doi.org/10.4324/9780429303753-7
- Kilpatrick J., Hoyles C., Skovsmose O., Valero P. (Eds.). (2005). *Mathematics Education Library,: vol 37. Meaning in Mathematics Education*. Springer.
- Kleemann, F., Krähnke, U., & Matuschek, I. (2013). *Interpretative Sozialforschung*. Springer Fachmedien Wiesbaden. https://doi.org/10.1007/978-3-531-93448-8
- Kleinke, K., Schlüter, E., & Christ, O. (2017). *Strukturgleichungsmodelle mit Mplus: Eine praktische Einführung*. De Gruyter Studium.
- KMK. (2005). Beschlüsse der Kultusministerkonferenz. Bildungsstandards im Fach Mathematik für den Hauptschulabschluss (Jahrgangsstufe 9). München: Luchterhand. Retrieved April 13, 2021, from

https://www.kmk.org/fileadmin/Dateien/veroeffentlichungen_beschluesse/2004/2004_10_ 15-Bildungsstandards-Mathe-Haupt.pdf

- KMK. (2016). The Education System in the Federal Republic of Germany 2014/2015: A description of the responsibilities, structures and developments in education policy for the exchange of information in Europe.
- KMK. (2019). The Education System in the Federal Republic of Germany 2017/2018: A description of the responsibilities, structures and developments in education policy for the exchange of information in Europe.
- Koestner, R., Otis, N., Powers, T. A., Pelletier, L., & Gagnon, H. (2008). Autonomous motivation, controlled motivation, and goal progress. *Journal of Personality*, *76*(5), 1201–1230. https://doi.org/10.1111/j.1467-6494.2008.00519.x
- Koller, H.-C. (2008a). Einleitung. In H.-C. Koller (Ed.), Sinnkonstruktion und Bildungsgang.
 Zur Bedeutung individueller Sinnzuschreibungen im Kontext schulischer Lehr-Lern-Prozesse. (pp. 7–12). Verlag Barbara Budrich.
- Koller, H.-C. (2008b). Sinnkonstruktion und Bildungsgang. Zur Bedeutung individueller Sinnzuschreibungen im Kontext schulischer Lehr-Lern-Prozesse. Verlag Barbara Budrich. https://doi.org/10.2307/j.ctvdf06q4
- Krapp, A. (1992). Das Interessenkonstrukt Bestimmungsmerkmale der Interessenhandlung und des individuellen Interesses aus der Sicht einer Person- Gegenstands-Konzeption. In A. Krapp & M. Prenzel (Eds.), Arbeiten zur sozialwissenschaftlichen Psychologie: Vol. 26. Interesse, Lernen, Leistung: Neuere Ansätze der pädagogisch-psychologischen Interessenforschung (297-329). Aschendorff.
- Krapp, A. (2000). Interest and human development during adolescence: An educationalpsychological approach. In J. Heckhausen (Ed.), Advances in Psychology. Motivational Psychology of Human Development: Developing Motivation and Motivating Development (Vol. 131, pp. 109–128). Elsevier. https://doi.org/10.1016/S0166-4115(00)80008-4
- Krapp, A. (2002). An educational-psychological theory of interest and its relation to sdt. In E.L. Deci & R. M. Ryan (Eds.), *Handbook of self-determination research* (pp. 405–427).University of Rochester Press.
- Krapp, A. (2005). Das Konzept der grundlegenden psychologischen Bedürfnisse. Ein Erklärungsansatz für die positiven Effekte von Wohlbefinden und intrinsischer Motivation im Lehr-Lerngeschehen. Zeitschrift für Pädagogik, 51(5), 626–641. Retrieved September 9, 2020, from

https://www.pedocs.de/volltexte/2011/4772/pdf/ZfPaed_2005_5_Krapp_Konzept_der_grundlegenden_psychologischen_Beduerfnisse_D_A.pdf

- Lamnek, S., & Krell, C. (2016). *Qualitative Sozialforschung: Mit Online-Materialien* (6th ed.). Beltz.
- Lawson, A. E. (2015). Hypothetico-deductive method. In R. Gunstone (Ed.), *Encyclopedia of science education* (pp. 471–472). Springer Netherlands. https://doi.org/10.1007/978-94-007-2150-0_260
- Lazarsfeld, P. F., & Henry, N. W. (1968). Latent Structure Analysis. Houghton Mifflin.
- León, J., Núñez, J. L., & Liew, J. (2015). Self-determination and stem education: Effects of autonomy, motivation, and self-regulated learning on high school math achievement.

Learning and Individual Differences, *43*, 156–163. https://doi.org/10.1016/j.lindif.2015.08.017

- Lepper, M. R. (1983). Social-control processes and the internalization of social values: An attributional perspective. In E.T. Higgins, D. N. Ruble, & W. W. Hartup (Ed.), *Social cognition and social development* (pp. 294–330). Cambridge University Press.
- Levesque, C., Zuehlke, A. N., Stanek, L. R., & Ryan, R. M. (2004). Autonomy and competence in German and American university students: A comparative study based on self-determination theory. *Journal of Educational Psychology*, 96(1), 68–84. https://doi.org/10.1037/0022-0663.96.1.68
- Lewin, K. (1951). *Field Theory in Social Science: Selected Theoretical Papers by Kurt Lewin*. Harper & Brothers Publishers New York.
- Little, T. D., Cunningham, W. A., Shahar, G., & Widaman, K. F. (2002). To parcel or not to parcel: Exploring the question, weighing the merits. *Structural Equation Modeling: A Multidisciplinary Journal*, 9(2), 151–173. https://doi.org/10.1207/S15328007SEM0902 1
- Liu, W. C., Wang, J. C. K., & Ryan, R. M. (2016). Understanding motivation in education: Theoretical and practical considerations. In W. C. Liu, J. C. K. Wang, & R. M. Ryan (Eds.), *Building autonomous learners* (pp. 1–7). Springer Singapore.
- Loewenthal, K. M. (2001). An Introduction to Psychological Tests and Scales. Psychology Press. https://doi.org/10.4324/9781315782980
- Lorenz, E. N. (1995). *The essence of chaos* (1st paperback ed.). *The Jessie and John Danz lectures*. University of Washington Press.
- Macho, S. (2015). *Moderne Testtheorie Manuskript zur Vorlesung*. Retrieved July 22, 2020, from https://docplayer.org/45002815-Moderne-testtheorie-manuskript-zur-vorlesung-autor-siegfried-macho.html
- MacIntosh, R., & Hashim, S. (2003). Variance estimation for converting mimic model parameters to irt parameters in dif analysis. *Applied Psychological Measurement*, 27(5), 372–379. https://doi.org/10.1177/0146621603256021
- Marsh, H. W., & Köller, O. (2004). Unification of theoretical models of academic selfconcept/achievement relations: Reunification of east and west german school systems after the fall of the berlin wall. *Contemporary Educational Psychology*, 29(3), 264–282. https://doi.org/10.1016/S0361-476X(03)00034-1
- Marsh, H. W., Lüdtke, O., Trautwein, U., & Morin, A. J. S. (2009). Classical latent profile analysis of academic self-concept dimensions: Synergy of person- and variable-centered approaches to theoretical models of self-concept. *Structural Equation Modeling: A Multidisciplinary Journal*, 16(2), 191–225. https://doi.org/10.1080/10705510902751010
- Martela, F., Bradshaw, E. L., & Ryan, R. M. (2019). Expanding the map of intrinsic and extrinsic aspirations using network analysis and multidimensional scaling: Examining four new aspirations. *Frontiers in Psychology*, 10, 2174. https://doi.org/10.3389/fpsyg.2019.02174
- Mason, J., & .Johnston-Wilder, S. (2004). *Fundamental constructs in mathematics education*. Routledge.

- Mason, J., & Waywood, A. (1996). The role of theory in mathematics education and research. In Bishop A.J., Clements K., Keitel C., Kilpatrick J., Laborde C. (Ed.), *International handbook of mathematics education* (pp. 1055–1089). Springer, Dordrecht.
- Maydeu-Olivares, Alberto. (2019). Re: The SRMR is 0.141 in CFA but other indices are ok. What is the implication?. Retrieved June 21, 2020, from: https://www.researchgate.net/post/The-SRMR-is-0141-in-CFA-but-other-indices-are-ok-What-is-the-implication/5dc75d07979fdca9423864d1/citation/download.
- Mayer, I. (2017). Thai secondary students' personal meaning with respect to mathematics. (Unpublished master's thesis).
- Meade, A. W., & Lautenschlager, G. J. (2004). A comparison of item response theory and confirmatory factor analytic methodologies for establishing measurement equivalence/invariance. *Organizational Research Methods*, 7(4), 361–388. https://doi.org/10.1177/1094428104268027
- Meredith, W. (1993). Measurement invariance, factor analysis and factorial invariance. *Psychometrika*, *58*(4), 525–543. https://doi.org/10.1007/BF02294825
- Meyer, M. A. (2008). Unterrichtsplanung aus der Perspektive der Bildungsgangforschung. In M. A. Meyer, M. Prenzel, & S. Hellekamps (Eds.), *Perspektiven der Didaktik: Zeitschrift für Erziehungswissenschaft. Sonderheft 9* (pp. 117–137). VS Verlag für Sozialwissenschaften.
- Middleton, J. A., & Spanias, P. A. (1999). Motivation for achievement in mathematics: Findings, generalizations, and criticisms of the research. *Journal for Research in Mathematics Education*, 30(1), 65. https://doi.org/10.2307/749630
- Millsap, R. E., & Olivera-Aguilar, M. (2012). Investigating measurement invariance using confirmatory factor analysis. In R. H. Hoyle (Ed.), *Handbook of structural equation modeling* (pp. 380–392). The Guilford Press.
- Mitchell, M. (1993). Situational interest: Its multifaceted structure in the secondary school mathematics classroom. *Journal of Educational Psychology*, *85*(3), 424–436. https://doi.org/10.1037/0022-0663.85.3.424
- Moosbrugger, H., & Höfling, V. (2006). Teststandards. In F. Petermann, & M. Eid, (Eds.), *Handbuch der Psychologischen Diagnostik* (pp. 407–419). Göttingen Hogrefe.
- Moosbrugger, H., & Schermelleh-Engel, K. (2006). Faktorenanalyse. In F. Petermann & M. Eid, (Eds.), *Handbuch der Psychologischen Diagnostik* (pp. 304–3017). Göttingen Hogrefe.
- Mouratidis, A., Vansteenkiste, M., Michou, A., & Lens, W. (2013). Perceived structure and achievement goals as predictors of students' self-regulated learning and affect and the mediating role of competence need satisfaction. *Learning and Individual Differences*, 23, 179–186. https://doi.org/10.1016/j.lindif.2012.09.001
- Müller, F. H., & Palekčić, M. (2005). Bedingungen und Auswirkungen selbstbestimmt motivierten Lernens bei kroatischen Hochschulstudenten. *Empirische P\u00e4dagogik*, 19(2), 134–165.

- Muthen, B. O., Kao, C.-F., & Burstein, L. (1991). Instructionally sensitive psychometrics: Application of a new irt-based detection technique to mathematics achievement test items. *Journal of Educational Measurement*, 28(1), 1–22. https://doi.org/10.1111/j.1745-3984.1991.tb00340.x
- Muthén, L. K., & Muthén, B. O. (1998-2017). *Mplus User's Guide: Statistical analysis with latent variables* (8th ed.). Muthén.
- Muthen, B. O., & Satorra, A. (1995). Complex sample data in structural equation modeling. *Sociological Methodology*, 25, 267. https://doi.org/10.2307/271070
- Neuman, W. L. (2011). *Social research methods: Qualitative and quantitative approaches* (7th ed.). *Relevance of social research*. Allyn and Bacon.
- Niemiec, C. P., & Ryan, R. M. (2009). Autonomy, competence, and relatedness in the classroom. *Theory and Research in Education*, 7(2), 133–144. https://doi.org/10.1177/1477878509104318
- Niss, M. (1994). Mathematics in society. In R. Biehler, R. Scholz, R. Strässer, & B. Winkelmann (Eds.), *Mathematics education library: Vol. 13. Didactics of mathematics as a scientific discipline* (pp. 367–378). Kluwer.
- Nuttin, J. (1984). *Motivation, planning, and action: A relational theory of behavior dynamics. Louvain psychology series. Studia psychologica.* Leuven University Press; Erlbaum.
- Nylund, K. L., Asparouhov, T., & Muthén, B. O. (2007). Deciding on the number of classes in latent class analysis and growth mixture modeling: A monte carlo simulation study. *Structural Equation Modeling: A Multidisciplinary Journal*, 14(4), 535–569. https://doi.org/10.1080/10705510701575396
- OECD. (2013). PISA 2012 Results: Ready to Learn: Students' Engagement, Drive and Self-Beliefs (Volume III): PISA. OECD Publishing. https://doi.org/10.1787/9789264201170-en
- OECD. (2014a). *Pisa 2012 results in focus: What 15-year-olds know and what they can do with what they know*. Retrieved February 20, 2020, from https://www.oecd.org/pisa/keyfindings/pisa-2012-results-overview.pdf
- OECD. (2014b). *PISA 2012 Technical report*. Retrieved February 20, 2020, from https://www.oecd.org/pisa/pisaproducts/PISA-2012-technical-report-final.pdf
- OECD (2015), "Does Math Make You Anxious?", *PISA in Focus*, No. 48, OECD Publishing, Paris, https://doi.org/10.1787/5js6b2579tnx-en.
- OECD. (2016). *PISA 2015 Results in Focus*. https://www.oecdilibrary.org/docserver/aa9237e6en.pdf?expires=1625851887&id=id&accname=guest&checksum=ED9AB3D1D4BFF4E5 AE9877770DA3254B https://doi.org/10.1787/aa9237e6-en
- OECD. (2019). *Pisa 2018 results (volume I): what students know and can do* [PISA]. https://doi.org/10.1787/5f07c754-en
- OECD. (2020a). *Education policy outlook: Finland* © *oecd 2020*. Retrieved April 22, 2021, from
 - https://gpseducation.oecd.org/Content/MapOfEducationSystem/FIN/FIN_2011_EN.pdf

OECD. (2020b). *Education policy outlook: Germany* © *oecd 2020*. Retrieved April 22, 2021, from

https://gpseducation.oecd.org/Content/MapOfEducationSystem/DEU/DEU_2011_EN.pdf

Pelletier, L. G., Fortier, M. S., Vallerand, R. J., & Brière, N. M. (2001). Associations among perceived autonomy support, forms of self-regulation, and persistence: A prospective study. *Motivation and Emotion*, 25(4), 279–306. https://doi.org/10.1023/A:1014805132406

Piaget, J. (1971). Biology and knowledge: An essay on the relations between organic regulations and cognitive processes. A phoenix book. Univ. of Chicago Press.

- Popper, K. (1968). *The Logic of Scientific Discovery* (2nd ed.). *Routledge Classics*. Taylor and Francis.
- Popper, K. R. (1989). *Logik der Forschung* (9th ed.). Die Einheit der Gesellschaftswissenschaften: Vol. 4. Mohr.
- Popper, K. R. (2000). Vermutungen und Widerlegungen: Das Wachstum der wissenschaftlichen Erkenntnis. Mohr Siebeck.
- Pospeschill, M. (2010). Testtheorie, Testkonstruktion, Testevaluation. utb.de-Bachelor-Bibliothek: Vol. 3431. Ernst Reinhardt Verlag.
- Postman, N. (1996). *The end of education: Redefining the value of school* (1st Vintage Books ed.). Vintage Books.
- Prediger, S., & Bikner-Ahsbahs, A. (2014). Introduction to networking: Networking strategies and their background. In A. Bikner-Ahsbahs & S. Prediger (Eds.), *Networking of theories as a research practice in mathematics education* (pp. 117–125). Springer International Publishing.
- Prediger, S., Bikner-Ahsbahs, A., & Arzarello, F. (2008). Networking strategies and methods for connecting theoretical approaches: First steps towards a conceptual framework. *ZDM*, 40(2), 165–178. https://doi.org/10.1007/s11858-008-0086-z
- Przyborski, A., & Wohlrab-Sahr, M. (2014). *Qualitative Sozialforschung: Ein Arbeitsbuch* (4th ed.). Lehr- und Handbücher der Soziologie. Oldenbourg Verlag.
- Radford, L. (2008a). Connecting theories in mathematics education: Challenges and possibilities. *ZDM*, 40(2), 317–327. https://doi.org/10.1007/s11858-008-0090-3
- Radford, L. (2008b). Theories in mathematics education: A brief inquiry into their conceptual differences. *Working paper. Prepared for the ICMI Survey Team 7. The notion and role of theory in mathematics education research*, 1-17.
- Radford, L. (2012). On the growth and transformation of mathematics education: Paper presented at the International Colloquium The Didactics of Mathematics: Approaches and Issues. A Homage to Michèle Artigue. Université de Paris VII. May 31 to June 1, 2012. Working Paper. Retrieved May 17, 2021, from https://www.researchgate.net/publication/319089825_On_the_growth_and_transformation _of_mathematics_education_theories
- Radford, L. (2014). Towards an embodied, cultural, and material conception of mathematics cognition. *ZDM*, *46*(3), 349–361. https://doi.org/10.1007/s11858-014-0591-1

REFERENCES

- Raykov, T. (2009). Evaluation of scale reliability for unidimensional measures using latent variable modeling. *Measurement and Evaluation in Counseling and Development*, 42(3), 223–232. https://doi.org/10.1177/0748175609344096
- Raykov, T. (2012). Scale construction and development using structural equation modeling.In R. H. Hoyle (Ed.), *Handbook of structural equation modeling* (pp. 472–492). The Guilford Press.
- Reeve, J. (2016). Autonomy-supportive teaching: What it is, how to do it. In W. C. Liu, J. C. K. Wang, & R. M. Ryan (Eds.), *Building autonomous learners* (pp. 129–152). Springer Singapore.
- Reeve, J., & Cheon, S. H. (2021). Autonomy-supportive teaching: Its malleability, benefits, and potential to improve educational practice. *Educational Psychologist*, *56*(1), 54–77. https://doi.org/10.1080/00461520.2020.1862657
- Rensvold, R. B., & Cheung, G. W. (1998). Testing measurement models for factorial invariance: A systematic approach. *Educational and Psychological Measurement*, 58(6), 1017–1034. https://doi.org/10.1177/0013164498058006010
- Rost, J. (2004). *Lehrbuch Testtheorie Testkonstruktion* (2nd ed.). Psychologie Lehrbuch. Huber.
- Rost, J. (2006). Latent-Class-Analyse. In F. Petermann & M. Eid (Eds.), Handbuch der Psychologischen Diagnostik (pp. 275-287). Göttingen Hogrefe.
- Roth, W.-M., & Walshaw, M. (2019). Affect and emotions in mathematics education: Toward a holistic psychology of mathematics education. *Educational Studies in Mathematics*, 102(1), 111–125. https://doi.org/10.1007/s10649-019-09899-2
- Ryan, R. M. (1982). Control and information in the intrapersonal sphere: An extension of cognitive evaluation theory. *Journal of Personality and Social Psychology*, 43(3), 450– 461. https://doi.org/10.1037/0022-3514.43.3.450
- Ryan, R. M., & Connell, J. P. (1989). Perceived locus of causality and internalization: Examining reasons for acting in two domains. *Journal of Personality and Social Psychology*, 57(5), 749–761. https://doi.org/10.1037/0022-3514.57.5.749
- Ryan, R. M., & Deci, E. L. (2000a). Intrinsic and extrinsic motivations: Classic definitions and new directions. *Contemporary Educational Psychology*, 25(1), 54–67. https://doi.org/10.1006/ceps.1999.1020
- Ryan, R. M., & Deci, E. L. (2000b). Self-determination theory and the facilitation of intrinsic motivation, social development, and well-being. *The American Psychologist*, 55(1), 68– 78. https://doi.org/10.1037/0003-066X.55.1.68
- Ryan, R. M., & Deci, E. L. (2002). Overview of self-determination theory: An organismicdialectical perspective. In E. L. Deci & R. M. Ryan (Eds.), *Handbook of selfdetermination research* (pp. 3–33). Univ. of Rochester Press.
- Ryan, R. M., & Deci, E. L. (2009). Promoting self-determined school engagement. In K. R.
 Wenzel & A. Wigfield (Eds.), *Handbook of motivation at school* (pp. 171–196).
 Routledge/Taylor & Francis Group.

- Ryan, R. M., & Deci, E. L. (2017). Self-determination theory: Basic psychological needs in motivation, development, and wellness. Guilford Press.
- Ryan, R. M., & Deci, E. L. (2020). Intrinsic and extrinsic motivation from a selfdetermination theory perspective: Definitions, theory, practices, and future directions. *Contemporary Educational Psychology*, *61*, Article 101860. https://doi.org/10.1016/j.cedpsych.2020.101860
- Ryan, R. M., Sheldon, K. M., Kasser, T., & & Deci, E. L. (1996). All goals are not created equal: An organismic perspective on the nature of goals and their regulation. In Gollwitzer, P. M. & Bargh, J. A. (Eds.), *The psychology of action: Linking cognition and motivation to behavior* (pp. 7–26). The Guilford Press.
- Sabena, C., Arzarello, F., Bikner-Ahsbahs, A., & Schäfer, I. (2014). The epistemological gap: A case study on networking of APC and IDS. In A. Bikner-Ahsbahs & S. Prediger (Eds.), *Networking of theories as a research practice in mathematics education* (pp. 179–200). Springer International Publishing.
- Satorra, A., & Bentler, P. M. (2001). A scaled difference chi-square test statistic for moment structure analysis. *Psychometrika*, 66(4), 507–514. https://doi.org/10.1007/BF02296192
- Savalei, V. (2012). The relationship between root mean square error of approximation and model misspecification in confirmatory factor analysis models. *Educational and Psychological Measurement*, 72(6), 910–932. https://doi.org/10.1177/0013164412452564
- Schermelleh-Engel, K., Kelava, A., & Moosbrugger, H. (2006). Gütekriterien. In F. Petermann, & M. Eid, (Eds.), *Handbuch der Psychologischen Diagnostik* (pp. 420–433). Göttingen Hogrefe.
- Schoenfeld, A. H. (2002). Research methods in (mathematics) education. In L. D. English (Ed.), *Handbook of international research in mathematics education*. (pp. 435–487). Lawrence Erlbaum Associates.
- Schoonenboom, J., & Johnson, R. B. (2017). How to construct a mixed methods research design. Kölner Zeitschrift Fur Soziologie Und Sozialpsychologie, 69(Suppl 2), 107–131. https://doi.org/10.1007/s11577-017-0454-1
- Schröder, M. (2016). Zur Dimensionalität der Sinnkonstruktionstypen "Kognitive Selbstentwicklung "und "Anwendungsrelevanz": Eine quantitative Fragebogenstudie): [On the dimensionality of the types of personal meaning "cognitive self-development" and "relevance of application": A quantitative study using a survey]. (Unpublished master's thesis).
- Sheldon, K. M., & Kasser, T. (2001). Getting older, getting better? Personal strivings and psychological maturity across the life span. *Developmental Psychology*, 37(4), 491–501. https://doi.org/10.1037//0012-1649.37.4.491
- Shi, D., Lee, T., & Maydeu-Olivares, A. (2019). Understanding the model size effect on SEM fit indices. *Educational and Psychological Measurement*, 79(2), 310–334. https://doi.org/10.1177/0013164418783530

- Shi, D., Maydeu-Olivares, A., & Rosseel, Y. (2020). Assessing fit in ordinal factor analysis models: SRMR vs. RMSEA. *Structural Equation Modeling: A Multidisciplinary Journal*, 27(1), 1–15. https://doi.org/10.1080/10705511.2019.1611434
- Sijtsma, K. (2009). On the use, the misuse, and the very limited usefulness of cronbach's alpha. *Psychometrika*, 74(1), 107–120. https://doi.org/10.1007/s11336-008-9101-0
- Skinner, E., Saxton, E., Currie, C., & Shusterman, G. (2017). A motivational account of the undergraduate experience in science: Brief measures of students' self-system appraisals, engagement in coursework, and identity as a scientist. *International Journal of Science Education*, 39(17), 2433–2459. https://doi.org/10.1080/09500693.2017.1387946
- Slevitch, L. (2011). Qualitative and quantitative methodologies compared: Ontological and epistemological perspectives. *Journal of Quality Assurance in Hospitality & Tourism*, 12(1), 73–81. https://doi.org/10.1080/1528008X.2011.541810
- Star, S. L., & Griesemer, J. R. (1989). Institutional ecology, 'translations' and boundary objects: Amateurs and professionals in berkeley's museum of vertebrate zoology, 1907-39. *Social Studies of Science*, 19(3), 387–420. https://doi.org/10.1177/030631289019003001
- Statista 2021a. (2021, June 8). *Einwohnerzahl Deutschland*. https://de.statista.com/themen/20/einwohnerzahl/
- Statista 2021b. (2020, January 29). *Finnland: Gesamtbevölkerung 2020*. https://de.statista.com/themen/1616/finnland/
- Steenkamp, J.-B. E. M., & Baumgartner, H. (1998). Assessing measurement invariance in cross-national consumer research. *Journal of Consumer Research*, 25(1), 78–107. https://doi.org/10.1086/209528
- Steinberg, L. (2005). Cognitive and affective development in adolescence. *Trends in Cognitive Sciences*, 9(2), 69–74. https://doi.org/10.1016/j.tics.2004.12.005
- Steingut, R. R., Patall, E. A., & Trimble, S. S. (2017). The effect of rationale provision on motivation and performance outcomes: A meta-analysis. *Motivation Science*, 3(1), 19–50. https://doi.org/10.1037/mot0000039
- Stenling, A., Ivarsson, A., Hassmén, P., & Lindwall, M. (2017). Longitudinal associations between athletes' controlled motivation, ill-being, and perceptions of controlling coach behaviors: A bayesian latent growth curve approach. *Psychology of Sport and Exercise*, 30, 205–214. https://doi.org/10.1016/j.psychsport.2017.03.002
- Steyer, R., & Eid, M. (2001). *Messen und Testen*. Springer Berlin Heidelberg. https://doi.org/10.1007/978-3-642-56924-1
- Strauss, A. L., & Corbin, J. M. (1996). *Grounded theory: Grundlagen qualitativer* Sozialforschung. Weinhelm Beltz.
- Suriakumaran, N., Hannula, M. S., & Vollstedt, M. (2019). Investigation of Finnish and German 9th grade students' personal meaning with relation to mathematics. *Lumat: International Journal of Math, Science and Technology Education*, 7(2). https://doi.org/10.31129/LUMAT.7.2.411
- Suriakumaran, N., Hannula, M. S., & Vollstedt, M. (2020). Comparing and contrasting personal meaning and value. In C. Andrà, D. Brunetto, & F. Martignone (Eds.), *Theorizing*

and measuring affect in mathematics teaching and learning (pp. 67–76). Springer International Publishing. https://doi.org/10.1007/978-3-030-50526-4 7

- Tafarodi, R. W., Milne, A. B., & Smith, A. J. (1999). The confidence of choice: Evidence for an augmentation effect on self-perceived performance. *Personality and Social Psychology Bulletin*, 25(11), 1405–1416. https://doi.org/10.1177/0146167299259006
- Taylor, G., Jungert, T., Mageau, G. A., Schattke, K., Dedic, H., Rosenfield, S., & Koestner, R. (2014). A self-determination theory approach to predicting school achievement over time: The unique role of intrinsic motivation. *Contemporary Educational Psychology*, 39(4), 342–358. https://doi.org/10.1016/j.cedpsych.2014.08.002
- Thomas, A. E., & Müller, F. H. (2011). Skalen zur motivationalen Regulation beim Lernen von Schülerinnen und Schülern. Skalen zur akademischen Selbstregulation von Schüler/innen SRQ-A: Wissenschaftliche Beiträge aus dem Institut für Unterrichts- und Schulentwicklung Nr. 5. Institut für Unterrichts- und Schulentwicklung.
- Törner, G. (2002). Epistemologische Grundüberzeugungen verborgene Variablen beim Lehren und Lernen von Mathematik. *Der Mathematikunterricht*, 48(4/5), 103–128.
- Vallerand, R. J., & Blssonnette, R. (1992). Intrinsic, extrinsic, and amotivational styles as predictors of behavior: A prospective study. *Journal of Personality*, *60*(3), 599–620. https://doi.org/10.1111/J.1467-6494.1992.TB00922.X
- Vallerand, R. J., Pelletier, L. G., Blais, M. R., Briere, N. M., Senecal, C., & Vallieres, E. F. (1992). The academic motivation scale: A measure of intrinsic, extrinsic, and amotivation in education. *Educational and Psychological Measurement*, 52(4), 1003–1017. https://doi.org/10.1177/0013164492052004025
- Valverde, G.A., Bianchi, L. J., Wolfe, R. G., Schmidt, W. H., & Houang, R. T. (2002). According to the book: using TIMSS to investigate the translation policy into practice through the world of textbooks. Dordrecht: Kluwer.
- Vandenberg, R. J., & Lance, C. E. (2000). A review and synthesis of the measurement invariance literature: Suggestions, practices, and recommendations for organizational research. *Organizational Research Methods*, 3(1), 4–70. https://doi.org/10.1177/109442810031002
- Vansteenkiste, M., Aelterman, N., Muynck, G.-J. de, Haerens, L., Patall, E., & Reeve, J. (2018). Fostering personal meaning and self-relevance: A self-determination theory perspective on internalization. *The Journal of Experimental Education*, 86(1), 30–49. https://doi.org/10.1080/00220973.2017.1381067
- Vansteenkiste, M., Niemiec, C. P., & Soenens, B. (2010). The development of the five minitheories of self-determination theory: An historical overview, emerging trends, and future directions. In T. C. Urdan & S. A. Karabenick (Eds.), Advances in Motivation and Achievement. The Decade Ahead: Theoretical Perspectives on Motivation and Achievement (Vol. 16, pp. 105–165). Emerald Group Publishing Limited. https://doi.org/10.1108/S0749-7423(2010)000016A007
- Vansteenkiste, M., & Ryan, R. M. (2013). On psychological growth and vulnerability: Basic psychological need satisfaction and need frustration as a unifying principle. *Journal of Psychotherapy Integration*, 23(3), 263–280. https://doi.org/10.1037/a0032359

- Vansteenkiste, M., Ryan, R. M., & Soenens, B. (2020). Basic psychological need theory: Advancements, critical themes, and future directions. *Motivation and Emotion*, 44(1), 1– 31. https://doi.org/10.1007/s11031-019-09818-1
- Vansteenkiste, M., Simons, J., Lens, W., Soenens, B., & Matos, L. (2005). Examining the motivational impact of intrinsic versus extrinsic goal framing and autonomy-supportive versus internally controlling communication style on early adolescents' academic achievement. *Child Development*, 76(2), 483–501. https://doi.org/10.1111/J.1467-8624.2005.00858.X
- Verbund FDB. (2020). Übersicht über die länderspezifischen Besonderheiten für Befragungen an Schulen. Retrieved June 5, 2020, from http://www.forschungsdatenbildung.de/genehmigungen?la=de
- Vinner, S. (2007). Mathematics education: Procedures, rituals and man's search for meaning. *The Journal of Mathematical Behavior*, 26(1), 1–10. https://doi.org/10.1016/j.jmathb.2007.03.004
- Vollstedt, M. (2011a). On the classification of personal meaning: Theory-governed typology vs. Empiricism-based clusters. In B. Ubuz (Chair), *PME.*, Ankara, Turkey.
- Vollstedt, M. (2011b). Sinnkonstruktion und Mathematiklernen in Deutschland und Hongkong: Eine rekonstruktiv-empirische Studie. Vieweg+Teubner Verlag / Springer Fachmedien Wiesbaden GmbH Wiesbaden. https://doi.org/10.1007/978-3-8348-9915-6
- Vollstedt, M., & Duchhardt, C. (2019). Assessment and structure of secondary students' personal meaning related to mathematics. In M. S. Hannula, G. C. Leder, F. Morselli, M. Vollstedt, & Q. Zhang (Eds.), *ICME-13 Monographs. Affect and Mathematics Education* (pp. 137–164). Springer International Publishing. https://doi.org/10.1007/978-3-030-13761-8_7
- Vollstedt, M., & Vorhölter, K. (2008). Zum Konzept der Sinnkonstruktion am Beispiel von Mathematiklernen. In H.-C. Koller (Ed.), Sinnkonstruktion und Bildungsgang. Zur Bedeutung individueller Sinnzuschreibungen im Kontext schulischer Lehr-Lern-Prozesse. (pp. 25–46). Verlag Barbara Budrich.
- Vorhölter, K. (2009). Sinn im Mathematikunterricht: Zur Rolle von Modellierungsaufgaben bei der Sinnkonstruktion von Schülerinnen und Schülern. Studien zur Bildungsgangforschung: Vol. 27. Barbara Budrich.
- Vygotski, L. S. (1962). Thought and language (Rev. and expanded ed.). MIT Press.
- Vygotsky, L. S. (1978). Mind in society: The development of higher psychological processes.M. Cole, V. John-Steiner, S. Scribner, & E. Souberman (Eds.), Harvard University Press.
- Vygotsky, L. S. (1987). The collected works of L. S. Vygotsky, vol. 1: Problems of general psychology. Springer.
- Wang, C. K. J., Liu, W. C., Nie, Y., Chye, Y. L. S., Lim, B. S. C., Liem, G. A., Tay, E. G., Hong, Y.-Y., & Chiu, C.-Y. (2017). Latent profile analysis of students' motivation and outcomes in mathematics: An organismic integration theory perspective. *Heliyon*, 3(5), e00308. https://doi.org/10.1016/j.heliyon.2017.e00308

- Wang, J.C.K., Ng, B.L.L., Liu, W. C., & Ryan, R. M. (2016). Can being autonomy-supportive in teaching improve students' self-regulation and performance? In W. C. Liu, J. C. K. Wang, & R. M. Ryan (Eds.), *Building autonomous learners* (pp. 227–243). Springer Singapore.
- Warburton, V. E., Wang, J. C. K., Bartholomew, K. J., Tuff, R. L., & Bishop, K. C. M. (2020). Need satisfaction and need frustration as distinct and potentially co-occurring constructs: Need profiles examined in physical education and sport. *Motivation and Emotion*, 44(1), 54–66. https://doi.org/10.1007/s11031-019-09798-2
- Watt, H. M. G. (2004). Development of adolescents' self-perceptions, values, and task perceptions according to gender and domain in 7th- through 11th-grade Australian students. *Child Development*, 75(5), 1556–1574. https://doi.org/10.1111/j.1467-8624.2004.00757.x
- Weber, M. (1972). Wirtschaft und Gesellschaft: Grundriß der verstehenden Soziologie (Studienausg., 5th rev. ed.). Mohr Siebeck.
- Wentzel, K. R. (2009). Students' relationships with teachers as motivational contexts. In K. R. Wenzel & A. Wigfield (Eds.), *Handbook of motivation at school* (pp. 301–322). Routledge/Taylor & Francis Group.
- Wentzel, K. R. (2012). Teacher-student relationships and adolescent competence at school. In T. Wubbels, P. d. Brok, J. van Tartwijk, & J. Levy (Eds.), *Interpersonal relationships in education* (pp. 19–35). Sense Publishers. https://doi.org/10.1007/978-94-6091-939-8_2
- White, R. W. (1959). Motivation reconsidered: The concept of competence. *Psychological Review*, *66*(5), 297–333. https://doi.org/10.1037/h0040934
- Wieferich, A. (2016). Zur Dimensionalität der Sinnkonstruktionstypen "Wohlbefinden durch eigene Leistung" und "Emotional-affektiv geprägte Entfaltung": Eine quantitative Fragebogenstudie: [On the dimensionality of the types of personal meaning "well-being due to own achievement" and "emotional-affective development": A quantitative study using a survey]. (Unpublished master's thesis).
- Wigfield, A., Byrnes, J. P., & Eccles, J. S. (2006). Development during early and middle adolescence. In P. A. Alexander & P. H. Winne (Eds.), *Handbook of educational psychology* (pp. 87–113). Lawrence Erlbaum Associates Publishers.
- Wigfield, A., & Eccles, J. S. (2000). Expectancy–value theory of achievement motivation. *Contemporary educational psychology*, 25(1), 68–81.
- Zacharopoulos, G., Sella, F., & Cohen Kadosh, R. (2021). The impact of a lack of mathematical education on brain development and future attainment. *Proceedings of the National Academy of Sciences of the United States of America*, 118(24). https://doi.org/10.1073/pnas.2013155118
- Zucker, S., Sassman, C., & Case, B. J. (2004). *Cognitive labs. Technical Report.* Pearson Inc. Retrieved May 9, 2017, from

http://images.pearsonassessments.com/images/tmrs/tmrs_rg/cognitivelabs.pdf

Dear 9th Graders,

Thank you for agreeing to take part in this important survey⁴¹. Within the scope of our project, we would like to find out what you think about mathematics. It is about your attitude towards mathematics and mathematics class respectively. Please fill out the following questionnaire.

There are no right or wrong answers and you do not have to be an expert in mathematics. It is important for us that you complete the questionnaire carefully and honestly. It would be a pity if your thoughts get lost.

The questionnaire is completely anonymous. Therefore, it is not apparent who has completed which questionnaire. Hence, do not write your name on the questionnaire.

<u>General</u>

What is your gender?

Female	
Male	
Other	
Don't want to answer	

What was your math grade in the last academic year?

Notes:

- Please answer the following questions quick and in sequence.
- Read the individual questions carefully and tick <u>that answer which describes you most</u> <u>applicable</u>.

Please tick only <u>one</u> box belonging to each statement.		X	
If you want to <i>correct</i> your answer than <u>colour</u> <u>the wrong box completely</u> and tick the right one!			X

Let's start! ©

⁴¹ Please contact the author for the German or the Finnish version of the survey.

<u>Part: 1</u>

► Ma	athematics at school	strongly	rather	rather	strongly
	Statement	disagree	disagree	agree	agree
Emo3	If I do not like my math teacher, I also don't like the math lessons.				
Aus1	It is important to me that there are recovery phases in mathematics lesson in which I can lean back.				
Kom3	I am proud of myself, when I realize what I have learned in the last years in math lessons.				
Emo2i	I don't care who teaches us, as contents matter and not the teacher.				
Pos4	I intensively deal with the mathematical contents so that I can help others in lessons.				
Emo4	I pay more attention in math lessons if I like the teacher.				
Kom7	I am proud of myself, when I can present my solution in front of my class.				
Klas1.2/ St81Q01	It annoys me if no one listens to what the teacher says.				
Soz4	It is important to me that there is a positive classroom climate during math lessons.				
Amo1	In math lessons, I am just physically present but I don't think.				
Aut7	It is important to me to explore new mathematical contents independently.				
Klas2.3	It is important to me that the central theme in mathematics lesson is recognizable clearly.				
Amo2	If the teacher doesn't recognize I deal with other things.				
Lzk7	I work carefully on my math homework in order to pass quizzes and tests.				

> Mathematics at school

	Statement	strongly disagree	rather disagree	rather agree	strongly agree
Aus5	I enjoy if e.g. historical knowledge about mathematics is mentioned in class.				
Klas2.2/ St81Q02	It bothers me if the math lessons is noisy and chaotic.				
Lzk6	I learn mathematical contents in order to be able to show my calculations at the blackboard.				
Soz1	It is important to me that a good sense of community does exist in math lessons.				
Klas3.2/ St81Q03	I don't like if my teacher has to wait a long time until it becomes quiet in the lesson.				
Lzk2	Without any exams or tests in mathematics, I would be less engaged in subject matter.				
Aus4	I like it when we can finally relax after a strenuous working phase in mathematics.				
Klas4.2/ St81Q04	It is important to me that we can work well in math lessons.				
Akt4i	As soon as I am aktively challenged, I lose interest/pleasure in math lessons.				
Emo5	I prefer to go to the math lessons when I realize that the teacher is interested in me.				
Klas5.2/ St81Q05	It annoys if the students start working only a long time after the lesson has already begun.				
Lzk3	The examination are an important reason for me to deal with mathematics.				
Klas1.3/ Steve1_1	It is important to me that the math lessons follow a logical order.				

Mathematics at school

	Statement	strongly disagree	rather disagree	rather agree	strongly agree
Lzk1	I deal with mathematics in order to pass exams and tests.				
Aus3	It is important to me to sometimes play games in math lessons.				
Klas3.3	It is important to me that the contents in the math lesson build on each other stringently.				
Akt13	When I am aktively challenged in lessons, I have the feeling to understand the math contents easily.				
Amo3	I do not care about the math lessons.				
Lzk8	I prepare intensely before a math examination in order to do well.				
Klas4.3	It is important to me that also things are discussed, which are a bit off topic.				
Her5	I am unmotivated when the tasks in the math lessons are too simple.				
Akt11	I understand more in math lessons due to phases of active practice.				
EfU1	It is important to me that my teacher supports me in learning mathematics.				
Klas5.3	It is important to me that the math lesson is not interrupted for too long by interposed questions.				

Learr	ning and understanding mathematics	strongly	rather	rather	strongly
	Statement	disagree	disagree	agree	agree
Pur6	Due to the logical structure of mathematics, it is easy for me to understand the contents.				
Akt10	Working actively on tasks helps me understand the contents.				
Sel4	It is particularly important for me that I really understand the ideas between the mathematical relationships.				
Lzk4	I calculate exceptionally many tasks in order to have a good feeling before exams.				
EfU2	Without any support from my teacher, it is more difficult for me to understand mathematics.				
Soz8	It is important to me that we work on new contents together.				
Sel3	It is important to me to perceive my learning progress.				
Lzk5	Before exams I work on more tasks than those given to us so that I know that I can do the topic.				
EfU3	It is important to me that my teacher explains the contents well to me.				
Akt9	When I deal actively with the contents It is easier for me learn mathematics.				
Sel1	I work on many mathematical tasks in order to become better and faster.				
EfU4	My teacher's support is important for my learning process in mathematics.				

Mathematics and I

	Statement	strongly disagree	rather disagree	rather agree	strongly agree
Kom1	I have a positive feeling when I succeed in mathematics.				
Aut2	I like to search for own approaches to solve mathematical tasks.				
Pur1	In its purity, mathematics is uniquely beautiful to me.				
Zen1	I deal with mathematics in order to be proud of my marks.				
Sel2	I deal with mathematics in order to improve my logical thinking.				
Akt1	I enjoy calculating.				
Her7	I get bored when the mathematical contents are too simple.				
Kom2	I deal with mathematics because my learning success makes me feel good.				
Pur2	The structure of mathematics fascinates me.				
Sel5	I deal with mathematics in order to increase my mathematical self-confidence.				
Zen3	I make an effort in mathematics in order to be pleased with good marks.				
Pfl2	I deal with mathematics mainly because it is a compulsory subject.				
Pur3	The beauty of mathematics appears to me in simple results for difficult problems.				
Akt2	I am happy when I can do mathematical tasks.				

≻ Ma	thematics and I	strongly	rathor	rathor	strongly
	Statement	disagree	disagree	agree	agree
Pfl5	Dealing with mathematics is nothing more than an obligation.				
Pur4	I am overjoyed by the uniqueness of the mathematical technical language.				
Sel8	It is important to me to become more confident in mathematics (e.g. by practicing).				
Pur7	I am excited about mathematics because mathematical statements are either right or wrong.				
Zen4	I deal with mathematics as good marks strenghten my self-confidence.				
Akt12	I look for additional tasks myself and work on them, in order to better keep up in math lessons.				
Sel6	Mathematics is important to me because I can develop myself in a number of ways.				
Pfl3	I only deal intensively with mathematics if I get a reward (e.g. money,).				
Akt3	I like to deal actively with mathematics.				
Her2	If I cannot solve tricky tasks immediately, I think about it until I got it.				
Kom6	I deal with mathematics because then I realize what I am capable of.				
Her1	I work on additional mathematical tasks in order to challenge myself.				
Aut8	When dealing with mathematics it is important to me to recognize that I can manage tasks without any help.				

Mathematics and I

		strongly disagree	rather disagree	rather agree	strongly agree
	Statement				
Kom4	I am proud of myself when I've solved a mathematical task which was difficult for me.				
Her3	It is important to me that I feel challenged from mathematics.				
All6	Mathematics is important to me because thinking mathematically has many advantages.				
Akt5	I like to deal with varied tasks.				
Her4	I have great pleasure with mathematics if I succeed in solving complex problems.				
Pur5	Mathematics is beautiful to me as it is unique in its formalism.				
Zen5	Bad marks in mathematics frustrate me.				
Aut1	It is important to me to independently come to the solution for new types of tasks.				
Aut5	It is important to me to organize the time for working on mathematical tasks on my own.				
Brf5i	I will not need school-mathematics for my professional life.				
Aut3	Working autonomously is important to me to when it comes to mathematics.				
Pfl1	I mainly deal with mathematics because I have to.				
Aut6	I like when I may choose math exercises on my own.				
Sel7	Mathematical competitions motivate me to deal more with contents.				
Pfl6	To deal with mathematics is an obligation to me.				

> Mathematics in everyday life

	Statement	strongly disagree	rather disagree	rather agree	strongly agree
All3	I deal with mathematics so that I do not lack important knowledge later on.				
Rea3	I prefer mathematical tasks which take up situations from real life.				
Brf4	I learn mathematical contents because I have to apply them during my studies / vocational training.				
All5	Learning mathematics is important to me because in doing so you learn for life.				
Rea2	I understand mathematics easier if tasks refer to examples from my life.				
Brf2	I deal with mathematics because I will have a larger selection of professions later on.				
All7	I deal with mathematics because logical thinking pertains to general education.				
Brf1	I deal with mathematics as I need it for my desired profession.				
Rea1	It is important to me that the contents from math lessons refer to everyday life.				
All4	I think that learning mathematics is important because it is of major importance to society.				
Rea4	Tasks referring to everyday life make me occupy myself more deeply with the mathematical content.				
All2	Learning mathematics is meaningful/sensible/reasonable because responsible citizens should have mathematical skills.				

> Mathematics in everyday life

	Statement	strongly disagree	rather disagree	rather agree	strongly agree
Rea5	Learning mathematics is important to me because I can apply it to everyday life.				
All8	I learn mathematics as otherwise a participation in life is only possible to a limited extent.				
Rea6	I think that learning mathematics is important because it is of great importance for other sciences.				
All1	I deal with mathematics because it pertains to general education.				
Rea8	Learning mathematics is important to me in order to understand processes found in everyday life.				
Brf3	I try to be good at mathematics because thereby I will have better chances on a profession later on.				
Rea7	Doing mathematics is important to me because it occurs everywhere in life.				

Other people and I

	Statement	strongly disagree	rather disagree	rather agree	strongly agree
Pos1	It is important to me that I am given credit for my achievement in mathematics from people who are important for me.				
EfU5	It is important to me that my teacher communicates my progress in mathematics to me.				
Kom5	I feel good when I can help others in math lessons.				
Aus2	I think It is good to have group or partner work in math lessons because then I can also talk about other things than mathematics.				

Other people and I strongly rather rather st					strongly
	Statement	disagree	disagree	agree	agree
Her6	In difficult tasks I try to be faster than the others.				
Zen2	It is embarrassing when I have worse marks in mathematics than the others.				
Pos2	It is important to me that others get a positive impression of me through my mathematics achievement.				
Soz2	It is important to me that I can collaborate with others in math lessons.				
Pfl4	I deal with mathematics because it is expected from me.				
Pos3	I deal with mathematics in order to be a role model for others.				
Soz5	If I can, I am happy to help the others with their problems in mathematics.				
Pos5	If my teacher has a poor impression of me in mathematics I make special effort in order to change that.				
Emo1	It is important to me that I have a good relationship to my teacher in math lessons.				
Soz3	It is important to me that preferably everybody should understand the contents in math lessons.				
Aut4	In math lessons it is important to me to choose with whom to collaborate with.				
Emo6	It is important to me that I feel valued by my teacher.				
Pfl7	I learn mathematics because my family wants me to.				
Pos7	I deal with mathematics in order to get appreciation for my achievement.				

Other people and I		strogly	rather	rather	strongly
	Statement	disagree	diagree	agree	agree
Soz6	I prefer to do mathematics when I collaborate with others in a group.				
Pos8	It is important to me that my family can be proud of my mathematics achievement.				
Soz7	I like group work because one can help each other.				
EfU6	It is important to me that my teacher is interested in my progress in math lessons.				
Pos6i	I do not care what others think about my mathematics performance.				

<u>Part: 2</u>

> Learning mathematics from my point of view

	l work and learn in the subject mathematics	strongly disagree	rather disagree	rather agree	strongly agree
Olntrinsisch1	because it is fun.				
Oldentifiziert1	because thereby I will have more opportunities to choose profession.				
OIntrojiziert1	because I want the teacher to think I am a good student.				
OExternal1	because otherwise I will get into trouble at home.				
OIntrinsisch2	because I like to deal with mathematics.				
Oldentifiziert2	because the things I learn here will be useful later on.				
Olntrojiziert2	because I like to be better than my classmates.				
Appendix A Original Survey Instrument – English Version

> Learning mathematics from my point of

Learning n	I work and learn in the subject mathematics	strongly disagree	rather disagre	ra e ag	ther gree	strongly agree
OExternal2	because otherwise I will get in trouble with my teacher.					
OIntrinsisch3	because I like to solve mathematical tasks.					
Oldentifiziert3	in order to be able to do a further education (e.g. school, vocational training or studies).					
OIntrojiziert3	because I would feel guilty.					
OExternal3	because my parents request that from me.					
OIntrinsisch4	because I like to think about the things in the subject mathematics.					
Oldentifiziert4	because thereby I can get a better profession.					
OIntrojiziert4	because I want the other students to think I am smart.					
OIntrinsisch5	because I enjoy to deal with mathematics.					
OIntrojiziert5	because I like to receive praise.					
OExternal4	because I have to.					

Thank you very much for your participation! ©

APPENDIX B

Appendix B.

Table—Confirmatory Factor Analysis of 17–Factor Model of Personal Meaning in German and Finnish Students.

Model	χ^2	df	CFI	TLI	RMSEA	SRMR	
17 Factors (GER)	3140.704	2074	.848	.834	.043	.061	
17 Factors (FIN)	3243.324	2074	.857	.843	.047	.062	

Note. CFI =comparative fit index; TLI = Tucker–Lewis index; RMSEA = root mean square error of approximation; SRMR = standardized root mean square residual.

Appendix C1.

Tabl	e—Intercorrel	lations of	17.	Personal	Meanings	in .	Finish	Students.

Pe	ersonal meaning	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16
1	Active practice	-															
2	Classroom	.50															
	management																
3	Cognitive challenge	.87	.39														
4	Duty	70	27	59													
5	Emotional-affective	.28	.52	.31	(04)												
	relation to the teacher																
6	Examination	.48	.43	.36	25	.46											
7	Experience of	.81	.47	.82	49	.33	.41										
	Autonomy																
8	Experience of	.70	.67	.72	34	.70	.64	.72									
	competence																
9	Experience of	.37	.73	.42	(09)	.78	.43	.37	.79								
	relatedness																
1() General education	.76	.43	.80	57	.42	.49	.68	.68	.49							
11	l Marks	.47	.45	.44	22	.56	.56	.47	.73	.61	.51						
12	2 Positive image	.46	.32	.48	19	.55	.48	.54	.69	.58	.54	.70					
13	3 Purism of	.90	.36	.87	62	.29	.42	.70	.56	.33	.78	.40	.45				
	mathematics																
14	A Reference to reality	.14	.35	.21	(01)	.40	.29	(.10)	.38	.50	.30	.16	.31	.24			
15	5 Self-perfection	.89	.48	.80	61	.44	.64	.77	.74	.45	.92	.56	.60	.88	.37		
16	6 Support by the	(.16)	.48	.18	(.00)	.91	.47	.21	.63	.66	.33	.50	.59	.20*	.44	.39	
	teacher																
17	7 Vocational	.58	.34	.54	50	.40	.45	.50	.60	.36	.80	.49	.58	.51	.25	.68	.35
	precondition																

Notes. Correlations in parentheses were not statistically significant, correlations printed in italics were significant at p < .01, all other correlations reported vere significant at p < .001

Appendix C2.

Table—Intercorrelations of	f 17 Personal Meanings in Germa	n Students.
	, , , , , , , , , , , , , , , , , , , ,	

Pe	ersonal meaning	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16
1	Active practice																
2	Classroom																
	management	.34															
3	Cognitive challenge	.84	.32														
4	Duty	80	25	59													
5	Emotional-affective																
	relation to the teacher	(.05)	.33	(.14)	(.17)												
6	Examination	(.21)	.25	(.13)	(05)	(.19)											
7	Experience of																
	Autonomy	.73	.49	.88	51	.25	.34										
8	Experience of																
	competence	.30	.41	.25	(13)	.60	.51	.55									
9	Experience of				, í												
	relatedness	.24*	.57	(.20)	(14)	.52	.34	.48	.74								
1() General education	.71	.37	.73	51	.28	.31	.73	.40	.37							
1	l Marks	.20*	.37	(.14)	(03)	.38	.58	.39	.58	.42	.40						
12	2 Positive image	.26	.41	(.18)	(07)	.42	.55	.42	.70	.40	.30	.75					
13	3 Purism of				, í												
	mathematics	.85	.30	.80	66	.21	.29	.75	.37	.28	.72	.26	.35				
14	A Reference to reality	(.14)	.30	.26	(.09)	.41	(.14)	.28	.34	.41	.27	(.09)	.25	.20			
1.	5 Self-perfection	.77	.43	.70	(52)	.26	.53	.78	.48	.35	.83	.53	.53	.83	(.19)		
10	5 Support by the				, í												
	teacher	(02)	.38	(10)	.29	.77	.35	(.22)	.68	.50	(.17)	.54	.59	(.00)	.32	.22	
17	7 Vocational			```				× /			. /			. /			
	precondition	.40	.23*	.49	29	.16*	.21	.45	.28	.27	.73	.40	.29	.43	.23	.53	(.13)

Notes. Correlations in parentheses were not statistically significant, correlations printed in italics were significant at p < .01, all other correlations reported vere significant at p < .001

Appendix D1.

					3												
Item	Act	Cla	Cog	Dut	Emo	Exa	Au	Com	Soc	Gen	Mark	Pos	Puri	Rea	Self	Supp	Voc
Akt1	.791																
Akt2	.664																
Akt3	.715																
Klas1		.776															
Klas2		.680															
Klas3		.676															
Klas4		.600															
Klas5		.589															
Her2			.580														
Her3			.828														
Her5			.469														
Pfl1				.892													
Pfl2				.863													
Pfl5				.764													
Pfl6				.620													
Emo1					.733												
Emo4					.433												
Emo5					.532												
Emo6					.850												
Lzk4						.859											
Lzk5						.797											
Lzk8						.673											
Aut1							.678										
Aut3							.709										
Aut7							.547										
Aut8							.585										
Kom1								.635									

Table—Factor Loadings of Personal Meaning Items in Finish Students.

Kom4	.717							
Kom5	.589							
Kom7	.608							
Soz1		.692						
Soz3		.504						
Soz4		.601						
Soz5		.624						
A112		-	.659					
All4			.685					
All6			.856					
A117			.727					
Zen1				.853				
Zen3				.756				
Zen4				.772				
Posl					.645			
Pos2					.690			
Pos7					.781			
Pos8					.641			
Pur1						.788		
Pur2						.839		
Pur3						.739		
Pur4						.796		
Pur5						.828		
Pur6						.575		
Pur7						.693		
Rea1							.659	
Rea2							.721	
Rea3							.606	
Rea4							.834	
Sel1								.625
Sel2								.768

Sel5	.737
Sel6	.745
EfU4	.636
EfU5	.809
EfU6	.746
Brf1	.708
Brf2	.767
Brf3	.767
Brf4	.580
Brf5u	.492

Notes. Act=Active practice; Cla=Classroom management; Cog=Cognitive challenge; Dut=Duty; Emo=Emotional-affective relation to the teacher; Exa=Examination; Au= Experience of autonomy; Co= Experience of competence; So= Experience of social relatedness; Gen=General education; Mark= Marks; Pos=Positive image; Puri=Purism of mathematics; Rea= Reference to reality; Sel=Self-perfection; Supp=Support by the teacher; Voc= Vocational precondition; Am = Amotivation; Ext = External; Intro = Introjected; Iden = Identified; Intr = Intrinsic. Item abbreviations and wording are displayed in Appendix A.

Appendix D2.

		0	- oj = e: ~ e		8												
Item	Act	Cla	Cog	Dut	Emo	Exa	Au	Com	Soc	Gen	Mark	Pos	Puri	Rea	Self	Supp	Voc
Akt1	.740																
Akt2	.820																
Akt3	.775																
Klas1		.447															
Klas2		.748															
Klas3		.805															
Klas4		.551															
Klas5		.642															
Her2			.615														
Her3			.782														
Her5			.501														
Pfl1				.794													
Pfl2				.731													
Pfl5				.801													
Pfl6				.487													
Emo1					.618												
Emo4					.493												
Emo5					.627												
Emo6					.692												
Lzk4						.881											
Lzk5						.665											
Lzk8						.662											
Aut1							.713										
Aut3							.578										
Aut7							.547										
Aut8							.599										
Kom1								.576									

Table—Factor Loadings of Personal Meaning Items in German Students.

Kom4	563							
Kom5	.505 677							
Kom7	.600							
Sozl	.000	688						
Soz3		389						
Soz4		592						
Soz5		515						
A112			600					
A 114			.000					
A116			.0 1 0 703					
A117			681					
Zen1			.001	824				
Zeni				.024 800				
				.009				
Zen4				./02	676			
Post					.020			
P082					.030			
					.625			
Pos					.627	7(0		
Purl						.768		
Pur2						.7/4		
Pur3						.647		
Pur4						.728		
Pur5						.734		
Pur6						.574		
Pur7						.644		
Real							.730	
Rea2							.739	
Rea3							.736	
Rea4							.705	
Sel1								.483
Sel2								.603

Sel5	.637
Sel6	.742
EfU4	.565
EfU5	.723
EfU6	.705
Brf1	.746
Brf2	.672
Brf3	.685
Brf4	.722
Brf5u	.549

Notes. Act=Active practice; Cla=Classroom management; Cog=Cognitive challenge; Dut=Duty; Emo=Emotional-affective relation to the teacher; Exa=Examination; Au= Experience of autonomy; Co= Experience of competence; So= Experience of social relatedness; Gen=General education; Mark= Marks; Pos=Positive image; Puri=Purism of mathematics; Rea= Reference to reality; Sel=Self-perfection; Supp=Support by the teacher; Voc= Vocational precondition; Am = Amotivation; Ext = External; Intro = Introjected; Iden = Identified; Intr = Intrinsic. Item abbreviations and wording are displayed in Appendix A.

Appendix D3.

		Regulation				
Item	Amotivation	External	Introjected	Identified	Intrinsic	
Amo1	.79					
Amo2	.59					
Amo3	.54					
Ext1		.81				
Ext2		.47				
Ext3		.62				
Intro2			.73			
Intro4			.93			
Intro5			.70			
Ident1				.85		
Ident3				.78		
Ident4				.87		
Intrin1					.88	
Intrin2					.91	
Intrin3					.91	
Intrin4					.81	
Intrin4					.87	

Table—Factor Loadings of Regulatory Style Items in Finnish Students.

Notes. Item abbreviations and wording are displayed in Appendix A.

Appendix D4.

		Regulation				
Item	Amotivation	External	Introjected	Identified	Intrinsic	
Amo1	.81					
Amo2	.68					
Amo3	.70					
Ext1		.79				
Ext2		.39				
Ext3		.91				
Intro2			.78			
Intro4			.91			
Intro5			.59			
Ident1				.90		
Ident3				.76		
Ident4				.87		
Intrin1					.87	
Intrin2					.91	
Intrin3					.93	
Intrin4					.77	
Intrin4					.87	

Table—Factor loadings of Regulatory Style Items in German Students.

Notes. Item abbreviations and wording are displayed in Appendix A.